Electroweak Theory*)

-----Framework of On-Shell Renormalization and Study of Higher-Order Effects-----

Ken-ichi AOKI,**) Zenrō HIOKI, Rokuo KAWABE,* Michiji KONUMA** and Taizo MUTA***

Department of Physics, Kyoto University, Kyoto 606 *Department of Physics, Osaka Medical College, Takatsuki 569 **Reseach Institute for Fundamental Physics, Kyoto University, Kyoto 606 ***Department of Physics, Hiroshima University, Hiroshima 730

(Received August 9, 1982)

The electroweak theory (the Weinberg-Salam theory) is reviewed emphasizing its aspect of a renormalizable gauge field theory with spontaneous symmetry breaking. The on-shell renormalization procedure is developed where all the renormalization constants are fixed on the mass shell of gauge bosons, fermions and Higgs bosons. It is applied to the calculation of radiative corrections to leptonic processes $\nu_{\mu}e \rightarrow \nu_{\pi}e$, $\overline{\nu}_{\mu}e \rightarrow \overline{\nu}_{\mu}e$, $\mu_{\mu}e \rightarrow \mu_{\nu}e$ and $\mu \rightarrow e\overline{\nu}_{e}\nu_{\mu}$. The experimental significance of the radiative corrections and the effect of the corrections to the values of physical masses of W^{\pm} and Z are discussed. The relation among different renormalization procedures is clarified.

Contents

- 1. Introduction
 - 1.1 Gauge unification of particle interactions
 - 1.2 Development of the Weinberg-Salam theory
 - 1.3 Field-theoretical aspect of the Weinberg-Salam theory
 - 1.4 Plan of the paper
- 2. Structure of Electroweak Theory
 - 2.1 Gauge principle
 - 2.2 Quantization
 - 2.3 Higgs mechanism
 - 2.4 Ward-Takahashi identities
 - 2.5 Renormalizability

^{*)} This work is supported in part by the Grant-in-Aid for scientific research from the Ministry of Education, Science and Culture.

^{**)} Present address: Research Institute for Fundamental Physics, Kyoto University, Kyoto 606.

- 3. On-Shell Renormalization in Electroweak Theory. I
 - 3.1 Renormalization schemes
 - 3.2 On-shell renormalization condition
 - 3.3 Two-point functions in the Weinberg-Salam theory
 - 3.4 Charge universality
- 4. On-Shell Renormalization in Electroweak Theory. II
 - 4.1 Structure of the full Lagrangian
 - 4.2 Feynman rules
 - 4.3 Renormalization conditions and counter terms
- 5. Electroweak Radiative Corrections to Leptonic Processes
 - 5.1 Leptonic processes and the Weinberg-Salam theory
 - 5.2 Scenario of the calculation
 - 5.3 Neutral current processes: $\nu_{\mu}e^{-} \rightarrow \nu_{\mu}e^{-}, \ \bar{\nu}_{\mu}e^{-} \rightarrow \bar{\nu}_{\mu}e^{-}$
 - 5.4 Charged current processes: $\nu_{\mu}e^{-} \rightarrow \mu^{-}\nu_{e}, \ \mu^{-} \rightarrow e^{-}\bar{\nu}_{e}\nu_{\mu}$
- 6. Order α Corrections to Physical Quantities
 - 6.1 Higher order effects and W^{\pm} , Z boson masses
 - 6.2 Order α corrections to cross sections and decay width
- 7. Brief Survey of Studies on Electroweak Radiative Corrections
 - 7.1 Conventional perturbation
 - 7.2 Leading-log calculation
- 8. Conclusion and Outlook

Acknowledgements

Appendices

- A Dimensional regularization
- B Formulas for one-loop Feynman amplitudes
- C 1/M expansion of some Feynman amplitudes
- D F_n and G_n functions
- E Renormalization constants
- F Spence function
- G Formulas for the generating functional and the effective action

References

2

Referral Guide of This Paper:

For details on the on-shell renormalization formalism, refer to



For some practical calculations in the Weinberg-Salam theory, refer to



For formulas for calculations in the Weinberg-Salam theory, refer to



For the results of electroweak radiative corrections of $O(\alpha)$ to leptonic processes, refer to



Chapter 1

Introduction

§1.1 Gauge unification of particle interactions

Classical electrodynamics was brought to completion in the form of the Maxwell equation where the vector potential appears as a fundamental field. As is well-known, the vector potential is regarded as a gauge field according to the local gauge invariance of the theory. The covariant quantization of the theory was performed consistently by introducing the indefinite metric and the theory turned out to be quantum electrodynamics (QED). It is the renormalizable gauge field theory where higher order effects in perturbative expansion for given electrodynamic processes can be systematically calculated. [Tati and Tomonaga 48, Fukuda et al. 49, Schwinger 48, 49a, b, Feynman 48, 49]. Those predictions based on higher order calculations have been extensively tested in various accurate experiments.

The theory of weak interactions originated from Fermi's theory [Fermi 34] on β decays^{*}) and was settled in the effective 4-fermion interactions of the V-A type [Feynman and Gell-Mann 58, Sudarshan and Marshak 58]. Although it was quite successful in the phenomenological sense, the 4-fermion theory was not satisfactory because of its violation of unitarity and of its nonrenormalizability which prevents us from dealing with weak higher order effects in a convincing way. Theories of weak interactions mediated by massive intermediate bose fields instead of the 4-fermion interactions had been proposed to attain universality [Lee et al. 49]. In these theories unitarity is recovered. This type of the theory is, however, still nonrenormalizable and thus there is no improvement of the situation. The improvement could only be possible if the weak vector fields would be regarded as non-Abelian gauge fields [Yang and Mills 54] just like the electromagnetic field as a gauge field in quantum electrodynamics. While the weak vector fields must be massive in order to reproduce the 4-fermion interaction effectively, gauge fields are required to be massless for the gauge invariance.

An attempt to solve this discrepancy and to unify the electromagnetic and weak interactions was proposed by Glashow in the use of $SU(2) \times U(1)$ gauge symmetry [Glashow 61]. A possible mechanism of giving a mass to the gauge field was found by Higgs in the case that the gauge symmetry is

^{*)} It is interesting to note that Yukawa, as early as Fermi, applied his idea of intermediate bosons to β decays [Yukawa 35]. This is the first attempt to unify strong and weak interactions.

spontaneously broken [Higgs 64, 66]. This is known as the Higgs mechanism. A realistic model of weak interactions was proposed by Weinberg and Salam where the Higgs mechanism was applied to a non-Abelian gauge field theory [Weinberg 67, Salam 68]. It should be noted here that to construct the realistic model of weak interactions it was necessary to unify the theories of weak and electromagnetic interactions in the non-Abelian gauge field theory. According to the proof by 't Hooft such a non-Abelian gauge field theory with spontaneous breakdown of gauge symmetries is renormalizable ['t Hooft 71b]. Experimental evidence for the model proposed by Weinberg and Salam has been accumulated and the model is widely believed to be the real theory of weak and electromagnetic interactions. The theory of this type is often called the electroweak theory.

The strong interaction of hadrons is supposed to originate from the interaction of their constituents, quarks, which is described by the non-Abelian gauge field theory based on the local color gauge symmetry of quarks [Fritzsch et al. 73]. This gauge theory, called quantum chromodynamics (QCD) has many desirable aspects in nonperturbative regime (as a review, see, e.g. [Marciano and Pagels 78]) and has also received strong phenomenological support particularly in short distance processes (for a recent review, see, e.g. [Buras 81]).

Just as QED and the weak interaction theory were unified to be the electroweak theory, the electroweak theory and QCD are expected to be unified as a gauge theory which is called the grand unified theory (GUT) (one of the models is due to [Georgi and Glashow 74]. For recent reviews, see, e.g. [Langacker 81] and [Konuma and Maskawa 81]). Experimental tests of GUT are currently under preparation and information on the validity of GUT will be obtained in the near future. It is, however, also important to continue more phenomenological tests on the electroweak theory and QCD in order to put on a solid basis for these theories which are the basic building blocks of GUT.

In the present review article we try to elucidate the structure of the electroweak theory as a gauge field theory with the intention of testing experimentally its field theoretical aspect.

§1.2 Development of the Weinberg-Salam theory

There could be a variety of models for the electroweak theory concerning its detailed structure. The difference of models is typically reflected by the neutral current structure of the theory. According to the experimental analyses of the neutral current structure in recent years, the original model by Weinberg and Salam was finally chosen as a unique model of the electroweak theory.

The Weinberg-Salam model was originally formulated for leptons only.

At that time the charm quark had not yet been discovered while two kinds of neutrinos were known, and the lepton-quark correspondence was not complete.^{*)} The introduction of a constituent with charm quantum number resolved this tantalizing situation [Maki 64a, b, Hara 64, Bjorken and Glashow 64]. In order to incorporate quarks with the Weinberg-Salam model, it is also necessary to fulfill the lepton-quark correspondence. The incorporation of the charm quark was crucial to explain the absence of the strangeness-changing neutral current owing to the GIM mechanism [Glashow et al. 70]. It is now possible to accommodate in the Weinberg-Salam model any number of generations of leptons and quarks in the sequential form. We assign the left-handed (righthanded) quarks and leptons to the doublets (singlets) of the group SU(2) $\times U(1)$. In the present paper we confine ourselves to the case of 3 generations.

$\left(\begin{matrix} \nu_e \\ e \end{matrix} \right)_{\rm L}$	$\begin{pmatrix} \nu_{\mu} \\ \mu \end{pmatrix}_{L}$	$\left(\begin{matrix} \boldsymbol{\nu}_{\tau} \\ \boldsymbol{\tau} \end{matrix} ight)_{\mathbf{L}}$
VeR	$\nu_{\mu \mathrm{R}}$	ν_{rR}
e_{R}	μ_{R}	$ au_{ m R}$
$\begin{pmatrix} u \\ d' \end{pmatrix}_{\mathrm{L}}$	$\left(\begin{array}{c} c\\ s' \end{array} ight)_{\rm L}$	$\left(\begin{array}{c} t\\ b \end{array} \right)_{\rm L}$
$u_{\rm R}$	c _R	t_R
$d_{\mathtt{R}}$	\$ _R	b_{R}

Here suffixes R and L refer to the right- and left-handed projections $(1 \pm \gamma_5)/2$ respectively and d' and s' represent the Cabibbo-mixed states [Cabibbo 63]. We neglect the Kobayashi-Maskawa mixing [Kobayashi and Maskawa 73] throughout the present paper. Since, in general, neutrinos may acquire masses, we have included ν_{eR} , $\nu_{\mu R}$ and ν_{rR} in the above tabulation. Though *t*-quark has not yet been observed, it has also been included.

The group $SU(2) \times U(1)$ reflects the symmetry of electroweak interactions and may be regarded as a local gauge group. The requirement of the $SU(2) \times U(1)$ local gauge invariance leads to the existence of the gauge fields which turn out to be the photon A and the weak bosons W^{\pm} and Z. To make the weak bosons massive we utilize the Higgs mechanism in connection with the spontaneous breakdown of the $SU(2) \times U(1)$ symmetry. The most economical way of doing this is to introduce a doublet of the Higgs scalar field, $\boldsymbol{0}$, the vacuum expectation value of which is given by

$$\langle \mathbf{\Phi} \rangle = \begin{pmatrix} 0 \\ v \end{pmatrix}.$$

^{*)} The lepton-quark correspondence originated from the lepton-hadron correspondence [Gamba et al. 59, Katayama et al. 62, Maki et al. 62].

By this choice of the Higgs field, the $SU(2) \times U(1)$ symmetry is broken down to the U(1) symmetry leaving only the electric charge conserved.

The model described above is, in a sense, minimal as a model of electroweak interactions. One may, of course, consider a model with more parameters. As mentioned before, it is possible to choose a proper model by examining the neutral current structure of the models. By careful analyses of neutrino neutral-current processes (for a review of experimental data, see, e.g. [Hioki 78, Baltay 79]), it was concluded that the Weinberg-Salam model is the unique solution in the quark sector as far as the first generation of quarks is concerned [Sehgal 77, Abbott and Barnett 78a, 78b, Komatsu 78, Hung and Sakurai 79]. The experiment on the parity violation in polarized electrondeuteron scattering [Prescott et al. 78] gave important information on the interference of the weak neutral current and electromagnetic current. By the analysis of this data the Weinberg-Salam model was uniquely chosen in the electron sector [Konuma and Oka 78, Abbott and Barnett 78b]. In Fig. 1.1 we present experimentally allowed regions for effective coupling constants g_{y} and g_A of the electron neutral current, where g_V and g_A are defined through the effective Hamiltonian for neutral current interactions:

$$H_{\rm eff} = (G_{\rm F}/\sqrt{2}) \bar{\nu} \gamma_{\mu} (1-\gamma_5) \nu \bar{e} \gamma^{\mu} (g_{\nu} - g_A \gamma_5) e ,$$



Fig. 1. 1. Experimentally allowed strengths of the neutral current couplings of the electron.

- (1) Area between two ellipses is allowed from data of $\nu_{\mu}e \rightarrow \nu_{\mu}e$ scattering.
- (2) Allowed area from data of $\bar{\nu}_{\mu}e \rightarrow \bar{\nu}_{\mu}e$.
- (3) Area surrounded by three dashed curves is consistent with data of reactor neutrino scattering $\overline{\nu}_{e}e \rightarrow \overline{\nu}_{e}e$.
- (4) Allowed region from polarized electron deuteron scattering $\vec{e}d \rightarrow eX$.
- (5) Two small regions are consistent with data of $e^+e^- \rightarrow e^+e^-$, $\mu^+\mu^-$, $\tau^+\tau^-$ at PETRA.

The black region is the solution consistent with all the above experimental data and includes the values in the Weinberg-Salam theory with $\sin^2\theta_W \simeq 0.25$.

where ν and e represent the neutrino and electron fields respectively. For the Weinberg-Salam model

$$g_r = -\frac{1}{2} + 2\sin^2\theta_w; \ g_A = -\frac{1}{2}.$$

Experimental constraints set by $\nu_{\mu}e \rightarrow \nu_{\mu}e$, $\bar{\nu}_{\mu}e \rightarrow \bar{\nu}_{\mu}e$, $\bar{\nu}_{e}e \rightarrow \bar{\nu}_{e}e$ and polarized $\vec{e}d \rightarrow eX$ data single out the Weinberg-Salam model with $\sin^{2}\theta_{W}\simeq 0.25$. Recent experimental analyses at DESY in $e^{+}e^{-} \rightarrow e^{+}e^{-}$, $\mu^{+}\mu^{-}$, $\tau^{+}\tau^{-}$ and $e^{+}e^{-} \rightarrow$ hadrons have added new information to the validity of the Weinberg-Salam theory in the energy region which covers 3 generations of leptons [Davier 82].

These tests of the model, however, have been restricted to the tree level, i.e., the field-theoretical aspect of the model has never been tested. In order for the Weinberg-Salam model to be established as electroweak theory, there are still more hurdles to be cleared. The following is a list of some of further important experimental tests of the Weinberg-Salam theory.

- 1) Aspect of the model...e.g., Observation of W^{\pm} and Z.
- 2) Aspect of the renormalizable field theory...e.g., Experimental check of radiative corrections.
- 3) Aspect of the spontaneous breakdown of the $SU(2) \times U(1)$ gauge symmetry...e.g., Observation of the Higgs boson.

In this review article we concentrate our attention on the second aspect of the above list. For this purpose we develop a convenient renormalization procedure and apply it to some practical calculations of radiative corrections to leptonic processes.

§ 1.3 Field-theoretical aspect of the Weinberg-Salam theory

Electroweak theory is a non-Abelian gauge field theory based on the local $SU(2) \times U(1)$ gauge symmetry. The difficulty lies in the fact that the quantization and renormalization procedures in non-Abelian gauge field theories are not straightforward generalization of those in Abelian gauge field theories like QED.

It was found by Feynman in 1962 that, in the covariant quantization of non-Abelian gauge field theories, one needs ghost fields even at one-loop level by the requirement of unitarity and gauge invariance [Feynman 63]. This can be easily seen by calculating the one-loop self-energy part of gauge bosons corresponding to the Feynman diagrams illustrated in Fig. 1.2. In fact it does not satisfy the gauge invariance and its discontinuity does not fulfill the uni-



Fig. 1. 2. The Feynman diagram for the one-loop self-energy part of the gauge boson.

tarity requirement. This difficulty may be eliminated by introducing a contribution of a ghost field which obeys an anticommutation relation although it is a scalar field. Feynman's idea was generalized to multi-loops by de Witt, but the origin of the ghost was still mysterious [de Witt 67]. In 1967 Faddeev and Popov gave a beautiful formulation for the quantization of non-Abelian gauge field theories based on the path-integral method [Faddeev and Popov 67]. According to their formulation the appearance of the ghost (Faddeev-Popov ghost) in the covariant quantization of non-Abelian gauge field theories is now clearly understood and the covariant Feynman rules are systematically written down.

In order to study the structure of non-Abelian gauge field theories, however, it is more convenient and more powerful to use the canonical operator formalism than the path integral formalism. In 1978 Kugo and Ojima succeeded in completing the manifestly covariant quantization in the canonical operator formalism, which guarantees the unitary S-matrix, based on the Becchi-Rouet-Stora (BRS) invariance [Kugo and Ojima 78, 79, Becchi et al. 75, 76]. Their formulation may be regarded as a natural generalization of the Nakanishi-Lautrap formalism in QED [Nakanishi 66, Lautrap 67]. In the present review article we follow the Kugo-Ojima formulation to quantize electroweak theory.

Once the problem of quantization is settled, the perturbative calculation can be systematically performed. In calculating higher order effects, however, one encounters ultraviolet divergences in the loop integrations and the renormalization procedure is required to be developed. It was found by 't Hooft and Lee and Zin-Justin that non-Abelian gauge field theories are renormalizable even with the spontaneous symmetry breaking ['t Hooft 71b, Lee and Zin-Justin 72a, b, c]. Although the renormalizability of the electroweak theory is obvious according to the works of 't Hooft and Lee and Zin-Justin, it is still necessary to formulate the renormalization procedure in order to obtain the finite part in divergent amplitudes for a specific electroweak process. There have been proposed several renormalization procedures to deal with electroweak radiative corrections. None of them, however, are performed entirely on the mass shell. We develop, in the present paper, the on-shell renormalization scheme.*' Since there exists some confusion in understanding the processes of calculating electroweak radiative corrections on the basis of various renormalization schemes, we will discuss the comparison of the schemes proposed so far (Chapter 7).

One of the main sources of the confusion lies in the complication due to the existence of many independent parameters. The free parameters existing in the Weinberg-Salam theory are g and g' (SU(2) and U(1) gauge coupling

^{*)} The original idea of the on-shell renormalization scheme was given in [Aoki 79]. See also [Aoki and Hioki 79, Aoki et al. 81, Inoue et al. 80, Bardin et al. 82].

constants respectively), μ^2 and λ (parameters in the Higgs potential) and f_i (the Higgs-fermion Yukawa-coupling constant where *i* refers to the type of fermions). There is another important parameter v, the vacuum expectation value of the Higgs field. This parameter v is not an independent parameter and should be fixed at the minimum point of the Higgs potential and hence is given in terms of the above free parameters. Since this set of independent parameters sets which consist of physical parameters. Some examples of the set of physical parameters are

(1)
$$e, \sin \theta_W, g, m_i, m_{\phi},$$

(2) $e, \sin \theta_W, M_W, m_i, m_{\phi},$
(3) $e, M_Z, M_W, m_i, m_{\phi},$

where e is the electromagnetic coupling constant, θ_{W} the Weinberg angle, and M_{W} , M_{Z} , m_{t} and m_{ϕ} are masses of the W- and Z- boson, the fermion of type i and the Higgs boson ϕ respectively. These various choices of the set of independent parameters are equivalent and related to each other by simple relations at the tree level. However, their relations become very complicated once higher order corrections are taken into account. It is this stage where the proper choice of the set of independent parameters is important.

Several authors have adopted set (1) or (2). While M_W and M_Z are well-defined as pole positions of Green functions, the definition of the parameter $\sin \theta_W$ is rather obscure except at the tree level where $\tan \theta_W = g'/g$. When higher order corrections are taken into account, there is no more unique and a priori definition of $\sin \theta_W$. Hence one must introduce the definition of $\sin \theta_W$ by hand. Moreover, unlike physical masses, $\sin \theta_W$ is not directly measurable (see § 3.1).*' For this reason we do not adopt $\sin \theta_W$ as one of the independent parameters. In our renormalization scheme we choose set (3) as independent parameters where, except for e the exact value of which is known in QED experiments, all the parameters are masses which are physical observables. Since W, Z and Higgs bosons have not been observed, the values of M_W , M_Z and m_ϕ are still unknown. However, if all these mass parameters are fixed, our scheme will be considered to be superior. A detailed explanation of our scheme will be given in Chapter 3, where we will set up all the renormalization conditions on the mass shell.

§1.4 Plan of the paper

This paper reviews the field theoretical structure of electroweak theory,

^{*)} One may define $\sin \theta_W$ such that $\cos \theta_W = M_W/M_Z$ in terms of physical masses M_W and M_Z . The scheme with this definition, however, turns out to be equivalent to the on-shell scheme.

11

presents the on-shell renormalization scheme and deals with some applications of the scheme to higher-order calculations in leptonic processes.

Introduction

Chap. 1

Chapter 2 contains the review of the general framework in electroweak theory. Starting from the gauge principle, we derive the Lagrangian for electroweak theory (§ 2.1), to which the Faddeev-Popov ghost term is added in connection with the covariant quantization (§ 2.2) and the Higgs field term in connection with the spontaneous symmetry breakdown (§ 2.3). The generalized Ward-Takahashi identities are derived (§ 2.4) and the renormalizability of the theory governed by the above Lagrangian is then proved (§ 2.5) by the full use of these identities.

Chapter 3 consists of the description of the on-shell renormalization scheme which we believe is the most convenient and useful. After an introductory remark on renormalization schemes in general (§ 3.1), the on-shell renormalization scheme is introduced and on-shell renormalization conditions are settled (§ 3.2). A further detailed discussion on two-point functions is given in § 3.3. The universality of the electric charge, the on-shell coupling constant of the photon, is then proved carefully (§ 3.4).

Chapter 4 presents a collection of useful tools to perform higher order calculations. The full Lagrangian given in § 2.3 is rearranged explicitly in terms of physical fields (§ 4.1) and the Feynman rules derived from this Lagrangian are tabulated (§ 4.2). All the counter terms necessary to carry out the on-shell renormalization are presented in terms of the physical parameters (§ 4.3).

Chapter 5 is devoted to the detailed description of applications of our renormalization scheme to the calculation of radiative corrections to leptonic processes. We first make a short review on leptonic processes in the Weinberg-Salam theory (\S 5.1). After the brief explanation of the calculational procedure (\S 5.2), the neutral (\S 5.3) and charged (\S 5.4) current processes are taken up for the calculation of radiative corrections.

In Chapter 6 we discuss the numerical results. The values of the Wand Z-boson masses are discussed on the basis of the results in Chapter 5 (§ 6.1). The numerical estimates of the total cross sections with one-loop corrections are presented (§ 6.2).

Chapter 7 deals with discussions and comparison of calculational devices adopted by various groups. All the approaches are in principle equivalent, but have quite different appearances. Hence this chapter may help understanding relations among different approaches. The discussion is given in the conventional (§ 7.1) and renormalization-group (§ 7.2) approaches separately.

Chapter 8 is devoted to the concluding remarks.

Finally we would like to suggest that readers who are mainly interested in practical calculations in higher orders may skip Chapters 2 and 3 and start from Chapter 4. It is recommended that serious readers who wish to see the detail of the on-shell renormalization scheme read through Chapters 2 and 3. [See the Referral Guide of This Paper placed after contents.]

Notations

We follow the notations of the text book by Bjorken and Drell in most cases [Bjorken and Drell 65]. Thus the metric is $g^{\mu\nu} = (1, -1, -1, -1)$ and γ_5 is defined by $\gamma_5 = i\gamma^0\gamma^1\gamma^2\gamma^3$. The inner product of a γ matrix with vector p is denoted by Feynman's slash: $\gamma_{\mu}p^{\mu} \equiv p$.

See § 4.2 for a slight difference between our notations and those by Bjorken and Drell.

Downloaded from https://academic.oup.com/ptps/article-abstract/doi/10.1143/PTPS.73.1/1848359 by guest on 06 January 2019

Chapter 2

Structure of Electroweak Theory

In this chapter we shall briefly describe the formal structure of electroweak theory. Starting with the gauge principle, we shall derive the classical Lagrangian for electroweak theory and perform quantization in covariant gauge following the canonical operator formalism. We shall then introduce the Higgs mechanism to give masses to gauge bosons and fermions. The renormalizability of the theory will be shown by the use of generalized Ward-Takahashi identities.

§2.1 Gauge principle

It has been well-known that the weak interaction is described by the effective current-current Lagrangian of the following form,

$$\mathcal{L}_{\mathbf{W}} = \frac{G_{\mathbf{F}}}{\sqrt{2}} j_{\mu}^{\dagger} j^{\mu} , \qquad (2 \cdot 1)$$

where j_{μ} is the weak charged current. Here, for simplicity, we shall consider only one generation of leptons and quarks, and further restrict our argument to leptons only. A full generalization to include quarks and multi-generations is straightforward and will be given in § 2.3. In the present case, the current takes the form

$$j_{\mu} = \overline{\psi}_{e} \gamma_{\mu} (1 - \gamma_{5}) \psi_{\nu} \tag{2.2}$$

with ψ_e and ψ_{ν} the fields of the electron and the electron-neutrino respectively. On the other hand, the electromagnetic interaction of charged leptons is expressed by the Lagrangian

$$\mathcal{L}_{\rm EM} = e \; j_{\mu}^{\ \rm em} A^{\mu} \,, \tag{2.3}$$

where A^{μ} is the photon field and j_{μ}^{em} is the electromagnetic current with the form

$$j_{\mu}^{\text{em}} = \overline{\psi}_{e} \gamma_{\mu} \psi_{e} . \qquad (2 \cdot 4)$$

Thus the weak and electromagnetic interactions are describable in terms of the currents of leptons, j_{μ} and j_{μ}^{em} .

Rearranging the structure of j_{μ} and j_{μ}^{em} , one easily finds that they can be reexpressed in terms of currents with definite $SU(2) \times U(1)$ transformation properties. In fact, we have

K-I. Aoki, Z. Hioki, R. Kawabe, M. Konuma and T. Muta

$$j_{\mu} = 2 \left(J_{\mu}^{1} - i J_{\mu}^{2} \right), \qquad (2 \cdot 5)$$

$$j_{\mu}^{\ \rm em} = J_{\mu}^{\ 0} - J_{\mu}^{\ 3} , \qquad (2 \cdot 6)$$

where J_{μ}^{0} and J_{μ}^{n} (n=1, 2, 3) are given by

$$J_{\mu}^{\ 0} = \overline{\psi}_{e} \gamma_{\mu} \frac{1 + \gamma_{5}}{2} \psi_{e} + \frac{1}{2} \overline{\psi} \gamma_{\mu} \frac{1 - \gamma_{5}}{2} \psi,$$

$$J_{\mu}^{\ n} = \overline{\psi} \gamma_{\mu} \frac{1 - \gamma_{5}}{2} \frac{\tau^{n}}{2} \psi \qquad (2 \cdot 7)$$

with $\psi = (\psi_{\nu}, \psi_{e})$ and $\tau^{n}/2$ the SU(2) generator in the fundamental representation. The currents J_{μ}^{0} and J_{μ}^{n} may be identified with those corresponding to the U(1) and SU(2) symmetries respectively. Under the SU(2) symmetry the left-handed (right-handed) fermion $(1-\tau_{5})\psi/2$ ($(1+\tau_{5})\psi/2$) can be classified in a doublet (singlet) representation. This observation suggests that the weak and electromagnetic interactions maintain the $SU(2) \times U(1)$ symmetry. In the following we shall take the $SU(2) \times U(1)$ symmetry as a basis of our argument and shall determine the Lagrangian of electroweak theory by regarding the $SU(2) \times U(1)$ symmetry as a local gauge symmetry.

The algebra of the $SU(2) \times U(1)$ symmetry is generated by T^n and Y (generators of the $SU(2) \times U(1)$ group) with the following commutation relations,

$$[T^{l}, T^{m}] = i\varepsilon^{lmn}T^{n}, \qquad (2\cdot8)$$

$$[T^{n}, Y] = 0, \qquad (2 \cdot 9)$$

where ε^{lmn} is the antisymmetric tensor ($\varepsilon^{123} = 1$ and $\varepsilon^{213} = -1$, etc.) and the repeated index should be summed over. For later convenience we introduce indices a, b, c, \cdots which run over 0, 1, 2 and 3, and define

$$T^{\bullet} \equiv Y/2 . \tag{2.10}$$

Equations (2.8) and (2.9) now read

$$[T^a, T^b] = i f^{abc} T^c , \qquad (2.11)$$

where $f^{obc} = 0$, and $f^{abc} = \varepsilon^{abc}$ for $\{a, b, c\} = \{1, 2, 3\}$. It should be noted that the eigenvalue of Y in each representation of $SU(2) \times U(1)$ is determined so that the relation $Q = T^{3} + Y/2$ holds where Q is the charge operator.

The leptons e and ν_e constitute a singlet and doublet in $SU(2) \times U(1)$ as has already been mentioned in § 1.2. They form the following sets (the neutrino is assumed to be left-handed),

$$R = \phi_{eR}, \qquad L = \begin{pmatrix} \psi_{\nu L} \\ \psi_{eL} \end{pmatrix}, \qquad (2 \cdot 12)$$

where suffixes R and L refer to the right- and left-handed components of field ψ respectively:

$$\psi_{\mathrm{R},\mathrm{L}} = \frac{1 \pm \gamma_{\mathfrak{s}}}{2} \psi \,. \tag{2.13}$$

We now introduce the local gauge transformation based on the SU(2) $\times U(1)$ symmetry. The transformation property of the field ψ reads

$$\delta\psi(x) = \theta^a(x)\,\delta^a\psi(x),\qquad(2\cdot 14)$$

where the θ^{a} 's are gauge-transformation parameters and

$$\delta^a \psi(x) = i g^a T^a \psi(x)$$
 (no summation on a) (2.15)

with

$$g^{a} = \begin{cases} g & \text{for } a = 1, 2, 3, \\ g' & \text{for } a = 0. \end{cases}$$
(2.16)

Here g and g' are SU(2) and U(1) gauge coupling constants respectively. Applying the gauge principle we may derive the classical Lagrangian in a standard manner (see, e.g. [Abers and Lee 73]). The Lagrangian acquires the following expression,

$$\mathcal{L} = \mathcal{L}_{G} + \mathcal{L}_{F}, \qquad (2 \cdot 17)$$

$$\mathcal{L}_{G} = -\frac{1}{4} F^{a}_{\mu\nu} F^{\ \mu\nu}_{a}, \qquad (2 \cdot 18)$$

$$\mathcal{L}_{\mathrm{F}} = i \overline{L} \mathcal{D}_{\mathrm{L}} L + i \overline{R} \mathcal{D}_{\mathrm{R}} R , \qquad (2 \cdot 19)$$

where

$$F^{a}_{\mu\nu} = \partial_{\mu}W^{a}_{\nu} - \partial_{\nu}W^{a}_{\mu} + g^{a}f^{abc}W^{b}_{\mu}W^{c}_{\nu}, \qquad (2\cdot20)$$

(no summation on a)

$$D_{\mathrm{L}\mu} = \partial_{\mu} - ig^{a}T^{a}W_{\mu}^{a}, \qquad (2\cdot 21)$$

$$D_{\mathbf{R}\boldsymbol{\mu}} = \partial_{\boldsymbol{\mu}} - ig'T^{\mathbf{0}}W_{\boldsymbol{\mu}}^{\mathbf{0}} \tag{2.22}$$

with W_{μ}^{a} the $SU(2) \times U(1)$ gauge field. Note that $T^{0} = -1/2$ for L $(-1 \text{ for } \mathbb{R})$. The transformation rule of W_{μ}^{a} is given by

$$\delta W_{\mu}{}^{a} = \partial_{\mu} \theta^{a} + g^{a} f^{abc} \theta^{b} W_{\mu}{}^{c}. \quad (\text{no summation on } a) \qquad (2 \cdot 23)$$

Obviously W_{μ}^{a} for a=1, 2, 3 belongs to the adjoint representation of SU(2).

The gauge fields and matter fields described by the Lagrangian (2.17) are all massless while, in the real world, almost all fermions are massive and the gauge bosons mediating weak interaction are required to be very heavy.

15

Thus the Lagrangian (2.17) which has been obtained by the simple application of the gauge principle is unrealistic in the present form. Before going into this problem, we shall consider the problem of quantization of the theory in § 2.2.

§ 2.2 Quantization

We would like to perform the covariant quantization of the theory specified by the Lagrangian (2.17). For this purpose we define the variable canonically conjugate to the field W_{μ}^{a} ,

$$\Pi_{\mu}{}^{a} \equiv \frac{\partial \mathcal{L}}{\partial \dot{W}^{a\mu}} = -F^{a}_{0\mu} \,.$$

According to this definition, the 0-th component of $\Pi_{\mu}{}^{a}$ vanishes: $\Pi_{0}{}^{a}=0$. This means that the propagator of gauge fields is singular for the system governed by the local-gauge invariant Lagrangian. Under this circumstance it is impossible to make the covariant quantization consistently.

The singularity disappears if we do not respect the local gauge invariance by adding a gauge fixing term, and we can perform the covariant canonical quantization. At this stage we need to introduce the Faddeev-Popov ghost [Faddeev and Popov 67], in particular, in the covariant quantization of non-Abelian gauge field theory. In this scheme, however, some of the states corresponding to gauge fields become unphysical, i.e., acquire negative norm. Accordingly we have to restrict the states by subsidiary conditions.

Here we first explain the quantization based on the path integral formulation (see, e.g. [Abers and Lee 73, Faddeev and Slavnov 80]) to show how the Faddeev-Popov ghost comes into play. We then perform the covariant canonical quantization in the operator formulation [Kugo and Ojima 78].

To the Lagrangian (2.17) we add the covariant gauge fixing term,

$$\frac{1}{2\alpha} (\partial^{\mu} W_{\mu}^{\ a})^{2}, \qquad (2\cdot 24)$$

where α is the so-called gauge parameter. With this gauge fixing term the theory is now nonsingular and hence we can write down the Feynman rule for the theory. One may then calculate Feynman diagrams which involve gauge-field loops. For example, the proper one-loop bare self-energy part (Fig. 1.2) of the field W_{μ}^{α} can be easily calculated using the dimensional regularization to regulate the ultraviolet divergence. It reads in the Feynman gauge ($\alpha = 1$),

$$\begin{split} \Pi^{ab}_{\mu\nu}(q) &= (\delta_{ab} - \delta_{a0}\delta_{b0}) \frac{g^2}{16\pi^2} \bigg[\left\{ \frac{19}{6} \Big(\frac{1}{\varepsilon} - \gamma - \ln \frac{-q^2}{4\pi\mu^2} \Big) + \frac{58}{9} \right\} q^2 g_{\mu\nu} \\ &- \left\{ \frac{11}{3} \Big(\frac{1}{\varepsilon} - \gamma - \ln \frac{-q^2}{4\pi\mu^2} \Big) + \frac{67}{9} \right\} q_{\mu}q_{\nu} \bigg], \qquad (2 \cdot 25) \end{split}$$

where μ^2 is the regularization mass scale squared, $\varepsilon = (4-D)/2$ with D the space-time dimensions and γ is the Euler constant. In Eq.(2.25) only gauge-field loop has been taken into account. Apparently we see that Eq. (2.25) violates gauge invariance, i.e., $q^{\mu}\Pi^{ab}_{\mu\nu}(q) \neq 0$. This situation does not change even if we take account of the contribution of fermion loops, since it is by itself gauge invariant. One can also show that it does not satisfy unitarity in each order of perturbation theory. These inconsistencies may be traced back to the fact that polarizations of gauge fields are not necessarily physical in the covariant gauge. To recover gauge invariance and unitarity, Feynman in 1963 devised a procedure in which one adds a contribution to $\Pi^{ab}_{\mu\nu}$ from a hypothetical scalar particle which is anticommuting and belongs to the adjoint representation of SU(2) [Feynman 63]. In fact, if one takes into account the contribution of the Feynman diagram in Fig. 2.1 with a proper coupling type and an overall negative sign due to the assumed anticommutation property, one finds

$$\Pi^{ab}_{\mu\nu}(q) = \delta_{ab} \frac{g^2}{72\pi^2} \left[15 \left(\frac{1}{\varepsilon} - \gamma - \ln \frac{-q^2}{4\pi\mu^2} \right) + 31 \right] (q^2 g_{\mu\nu} - q_{\mu} q_{\nu}), \quad (2 \cdot 26)$$

which is obviously gauge-invariant. This procedure can also be applied to the Feynman amplitude with many external lines and may be extended to the case of multiloop amplitudes [de Witt 67].



A clear-cut formulation to justify this heuristic procedure was given by Faddeev and Popov using the Feynman path-integral method [Faddeev and Popov 67]. Let us consider a transition amplitude defined by the path integral,

$$Z = \int \mathcal{D}\phi \exp\left(i \int d^4 x \mathcal{L}\right), \quad (\mathcal{D}\phi = \prod_i \mathcal{D}\phi^i)$$
 (2.27)

where the ϕ^{i} 's represent all the involved fields and index *i* expresses Lorentz and group indices altogether. We set the gauge fixing condition,

$$F_i{}^a\phi^i = f^a , \qquad (2 \cdot 28)$$

where F_i^a is an operation on ϕ^i which may or may not be a differentiation and f^a is a given function. For example, if $F_i^a = \partial_\mu$ and $f^a = 0$, Eq. (2.28) reads $\partial^{\mu}W_{\mu}^{\ a} = 0$. With this gauge fixing the path integral (2.27) must be performed on the restricted paths which satisfy Eq. (2.28). This constraint may be explicitly taken into account by using the functional $\Delta[\phi^i]$ which is defined by

17

K-I. Aoki, Z. Hioki, R. Kawabe, M. Konuma and T. Muta

where $\mathcal{D}g$ is an invariant measure of group element g in local $SU(2) \times U(1)$ which has property $\mathcal{D}(gg') = \mathcal{D}(g)$ and ϕ_g^i is a transform of ϕ^i under the group element g. Apparently from Eq. (2.29) we see that $\mathcal{\Delta}[\phi^i]$ is invariant under the transformation g:

$$\Delta[\phi^i] = \Delta[\phi_g^i]. \tag{2.30}$$

Using Eqs. $(2 \cdot 29)$ and $(2 \cdot 30)$, Eq. $(2 \cdot 27)$ is rewritten as

$$Z = \int \mathcal{D}g \int \mathcal{D}\phi \mathcal{\Delta}[\phi^i] \delta \left(F_i^a \phi^i - f^a\right) \exp\left(i \int d^4 x \mathcal{L}\right), \qquad (2 \cdot 31)$$

where we have taken into account the gauge invariance of $\mathcal{D}\phi^i$ and \mathcal{L} . Since the integrand in Eq.(2.31) does not depend on g, one may factor out $\int \mathcal{D}g$ which is an infinite constant. By this factorization one obtains the welldefined transition amplitude Z. Equation (2.31) shows that the application of the gauge fixing condition enforces the measure of the path integral to be deformed resulting in an appearance of $\mathcal{\Delta}[\phi^i]$.

The presence of $\Delta[\phi^i]$ in Eq.(2.31) is an obstacle to derive ordinary Feynman rules. In order to circumvent this difficulty, we try to exponentiate $\Delta[\phi^i]$ and redefine the Lagrangian. For this purpose we derive an explicit expression of $\Delta[\phi^i]$ in Eq.(2.31). Noting that $\phi_g^i \simeq \phi^i + \theta^a \delta^a \phi^i$ and $\mathfrak{D}g \simeq \mathfrak{D}\theta$ for $g\simeq 1$, we find, under the condition $F_i^b \phi^i = f^b$,

$$\begin{split} \mathcal{\Delta}[\phi^{i}] &= \left[\int \mathcal{D}\theta \delta \left(\delta^{a} \left(F_{i}^{b} \phi^{i} \right) \theta^{a} - f^{b} \right) \right]^{-1} \\ &= \det \left(\delta^{a} \left(F_{i}^{b} \phi^{i} \right) \right). \end{split} \tag{2.32}$$

Hence in Eq. (2.31) we may effectively replace $\Delta[\phi^i]$ by the determinant in Eq. (2.32). We here introduce auxiliary fields c^a and \bar{c}^a which are scalar fields following anticommutation rules and belonging to the adjoint representation. The field possesses the property of the Grassmann number. With the help of these fields c and \bar{c} , we can rewrite det $(\delta^a(F_i^b\phi^i))$ as follows,

$$\det\left(\delta^{a}(F_{i}^{b}\phi^{i})\right) = \int \mathcal{D}\bar{c} \mathcal{D}c \exp\left[-i \int d^{4}x \,\bar{c}^{a}\delta^{a}(F_{i}^{b}\phi^{i})c^{b}\right]. \quad (2\cdot33)$$

Note here that the anticommutation property of the fields c^a and \bar{c}^a is necessary to obtain det $(\delta^a(F_i^b\phi^i))$ in Eq. (2.33) instead of $1/\det(\delta^a(F_i^b\phi^i))$. The fields c^a and \bar{c}^a are called Faddeev-Popov ghosts (FP ghosts) [Faddeev and Popov 67]. The delta function $\delta(F_i^a\phi^i - f^a)$ in Eq. (2.31) can also be exponentiated if we average it over the parameter f^a with the weight $\exp[-i(f^a)^2/2\alpha]$: Chap. 2 Structure of Electroweak Theory

$$\int \mathcal{D}f \,\delta\left(F_{i}{}^{a}\phi^{i}-f^{a}\right) \exp\left(-i \int d^{4}x (f^{a})^{2}/2\alpha\right)$$
$$= \int \mathcal{D}B \exp\left[i \int d^{4}x \left(B^{a}F_{i}{}^{a}\phi^{i}+\frac{\alpha}{2}B^{a}B_{a}\right)\right], \qquad (2.34)$$

where we have introduced an auxiliary field B^a to linearize the gauge term. We finally find

$$Z = \int \mathcal{D}g \int \mathcal{D}\phi \mathcal{D}\bar{c} \mathcal{D}c \mathcal{D}B \exp\left(i \int d^4x \mathcal{L}'\right), \qquad (2.35)$$

where the modified Lagrangian \mathcal{L}' reads

$$\mathcal{L}' = \mathcal{L} + \mathcal{L}_{GF} + \mathcal{L}_{FP}, \qquad (2 \cdot 36)$$

$$\mathcal{L}_{\rm GF} = B^a F_i^a \phi^i + \frac{\alpha}{2} (B^a)^2, \qquad (2 \cdot 36a)$$

$$\mathcal{L}_{\rm FP} = -\bar{c}^a \delta^a (F_i^b \phi^i) c^b \,. \tag{2.36b}$$

In the covariant global symmetric gauge,

$$F_i{}^a\phi^i = \partial^{\mu}W_{\mu}{}^a \quad \text{and} \quad \delta^a (F_i{}^b\phi^i) = \partial^{\mu}D_{\mu}{}^{ab}, \qquad (2\cdot37)$$

where

$$D_{\mu}{}^{ab} = \delta^{ab}\partial_{\mu} + g^a f^{abc} W_{\mu}{}^c . \qquad (2 \cdot 37a)$$

Performing the integration by parts, we obtain

$$\mathcal{L}_{\rm FP} = \left(\partial^{\mu} \bar{c}^{\,a}\right) D_{\mu}{}^{ab} c^{b} \,. \tag{2.38}$$

The Lagrangian \mathcal{L}' in Eq. (2.36) together with Eq. (2.38) is the basic Lagrangian in the covariant gauge from which the Feynman rules follow.

The above procedure of the quantization can be recapitulated by an operator formulation. As was observed before, in the course of quantization one was forced to introduce the gauge fixing term and then the local gauge invariance was violated. At the same time, however, the Faddeev-Popov ghost term was necessitated. The final Lagrangian thus obtained has a new global symmetry under the transformation including Faddeev-Popov ghost fields. This was found by Becchi, Rouet and Stora and is called the BRS transformation [Becchi et al. 75, 76].

We can now reverse the argument: We start with the BRS transformation and determine \mathcal{L}_{FP} so that the total Lagrangian is invariant under the BRS transformation. The BRS transformation is defined by the following set of infinitesimal transformations,

$$\delta^{\text{BRS}}\psi(x) = i\lambda g^a c^a(x) T^a \psi(x),$$

K-I, Aoki, Z. Hioki, R. Kawabe, M. Konuma and T. Muta

$$\begin{split} \delta^{\text{BRS}} W_{\mu}{}^{a}(x) &= \lambda D_{\mu}{}^{ab} c^{b}(x), \\ \delta^{\text{BRS}} \bar{c}^{a}(x) &= \lambda B^{a}(x), \\ \delta^{\text{BRS}} c^{a}(x) &= -\frac{\lambda}{2} g^{a} f^{abc} c^{b}(x) c^{c}(x), \quad \text{(no summation on } a) \\ \delta^{\text{BRS}} B^{a}(x) &= 0, \qquad (2 \cdot 39) \end{split}$$

where λ is a parameter independent of x which anticommutes with c(x), $\bar{c}(x)$ and $\psi(x)$. Note that the transformations of $\psi(x)$ and $W_{\mu}^{a}(x)$ are obtained from their local gauge transformations (2.14) and (2.23) by replacing $\theta^{a}(x)$ by $\lambda c^{a}(x)$. We require that the total Lagrangian $\mathcal{L}' = \mathcal{L} + \mathcal{L}_{GF} + \mathcal{L}_{FP}$ be invariant under the BRS transformation (2.39) leaving \mathcal{L}_{FP} unknown. Obviously $\mathcal{L}(=\mathcal{L}_{G} + \mathcal{L}_{F})$ is invariant under (2.39) as it has originally the local gauge invariance. On the other hand, \mathcal{L}_{GF} transforms under (2.39) such that

$$\delta^{\text{BRS}} \mathcal{L}_{\text{GF}} = \lambda B^a \delta^a (F_i^b \phi_i) c^b \,. \tag{2.40}$$

To maintain $\delta^{BRS} \mathcal{L}' = 0$, we must compensate Eq.(2.40) by $\delta^{BRS} \mathcal{L}_{FP}$. By taking the form (2.36b) as \mathcal{L}_{FP} , we find

$$\delta^{\text{BRS}} \mathcal{L}' = -\bar{c}^a \delta^{\text{BRS}} (\delta^b (F_i^a \phi_i) c^b), \qquad (2 \cdot 41)$$

which vanishes. In fact, by applying Jacobi's identity for operator O,

$$\frac{\delta(\delta^a O)}{\delta \phi_i} \delta^b \phi_i - \frac{\delta(\delta^b O)}{\delta \phi_i} \delta^a \phi_i = -g^a f^{abc} \delta^c O, \qquad (2 \cdot 42)$$

we can show that

$$\begin{split} \delta^{\mathrm{BRS}}(\delta^{b}(F_{i}{}^{a}\phi_{i})c^{b}) &= -\lambda \frac{g}{2}f^{abc}\delta^{c}(F_{i}{}^{a}\phi_{i})c^{d}c^{b} \\ &+ \delta^{b}(F_{i}{}^{a}\phi_{i})\left(-\lambda \frac{g}{2}\right)f^{bcd}c^{c}c^{d} = 0 \;. \end{split}$$

Thus the Lagrangian \mathcal{L}' with the form (2.36) is the one which satisfies our requirement.

Following the usual procedure through the Lagrangian \mathcal{L}' , we can define canonically conjugate momenta and set up equal-time canonical commutation relations. For the FP ghost fields anticommutation relations are assumed and the Fermi statistics is adopted. According to the Noether theorem, there exists a conserved current J_{μ}^{BRS} as \mathcal{L}' is invariant under the BRS transformation. The charge $Q_{\rm B}$ defined by

$$Q_{\rm B} = \int d^3x J_0^{\rm BRS}(x) \tag{2.43}$$

generates the BRS transformation $(2 \cdot 39)$. For the Lagrangian \mathcal{L}' in the covariant global symmetric gauge $(2 \cdot 37)$, Q_B takes the form

$$Q_{\rm B} = \int d^{s}x \Big(B^{a} \overline{\partial}_{0} c^{a} + g^{a} f^{abc} B^{a} A_{0}^{b} c^{c} - \frac{1}{2} g^{a} f^{abc} \partial_{0} \overline{c}^{a} c^{b} c^{c} \Big). \qquad (2 \cdot 44)$$

A remarkable property of the BRS charge $Q_{\rm B}$ is the nilpotency

$$Q_{\rm B}^{\ 2} = 0$$
. (2.45)

This property may be checked directly by using Eq. $(2 \cdot 44)$ and suggests that applying the BRS transformation twice, one gets the vanishing result:

$$\delta^{\text{BRS}}(\delta^{\text{BRS}}O) = 0. \qquad (2 \cdot 46)$$

The proof of $\delta^{BRS} \mathcal{L}' = 0$ in Eq. (2.41) is in fact an example of this property. The physical state |phys> in this formalism is the state satisfying

$$Q_{\rm B}|{\rm phys}\rangle = 0$$
. (2.47)

Using the nilpotency of $Q_{\rm B}$, one can show that the space spanned by |phys> has the positive semidefinite norm [Kugo and Ojima 78, 79]. According to this fact, the unitarity of the physical S-matrix, the S-matrix projected on the physical space, is guaranteed. Thus a consistent quantization of the theory is maintained in the covariant operator formalism.

It is quite important to note that the FP ghosts c^a and \bar{c}^a and the auxiliary field B^a (the scalar component of W_{μ}^{a}) are combined with the longitudinal component W_{L}^{a} of W_{μ}^{a} and only the combination of them with zero norm can appear in the physical space. Hence c^a , \bar{c}^a , B^a and W_L^a effectively decouple from the physical states. According to this formalism, we can also make a lucid explanation of the Higgs mechanism to be described in § 2.3. In this case, four fields c^a , \bar{c}^a , B^a and χ^a (Nambu-Goldstone mode) are combined to disappear from the physical space and three remaining components of the gauge field represent a massive spin-1 particle [Nambu and Jona-Lasinio 61, Goldstone 61]. The mechanism in which four unphysical components of fields may appear in the physical space only as a combination of zero-norm states is called the quartet mechanism. One may find all possible asymptotic states by constructing the representation of the algebra of $Q_{\rm B}$ and $Q_{\rm c}$ (conserved charge of the ghost number). In the gauge theory there appear only two kinds of particles: the physical particles (singlet) and the quartets which are unobservable [Nakanishi 79].

§ 2.3 Higgs mechanism

As stressed at the end of § 2.1, gauge fields cannot acquire masses in a gauge invariant (or, strictly speaking, BRS invariant) Lagrangian. Also fermion

21

mass terms are not included in the Lagrangian (2.36), because $\overline{\psi}_{L}\psi_{R}$ is a doublet representation of SU(2) and gauge non-invariant. To make the theory realistic we need important ideas in quantum field theory, i.e., the spontaneous symmetry breakdown and the Higgs mechanism.

The idea of spontaneous symmetry breakdown was introduced in elementary particle physics first by Nambu and Jona-Lasinio in connection with the chiral symmetry [Nambu and Jona-Lasinio 61]. Consider a continuous symmetry of Lagrangian \mathcal{L} . According to the usual Noether theorem, there exists a conserved charge Q which generates the symmetry transformation,

$$[Q,\phi] = i\delta\phi, \qquad (2\cdot48)$$

and commutes with the Hamiltonian:

$$[Q, H] = 0. (2.49)$$

In the case of quantum mechanics where the degrees of freedom are finite, the vacuum (the minimum energy state) is unique and is a simultaneous eigenstate of H and Q. On the contrary, in quantum field theory with infinite degrees of freedom, it is possible to have degenerate vacuum states since transitions between these vacuum states are possible only by an infinite number of operations. Here all the states are constructed on one vacuum which is no longer an eigenstate of the charge Q:*)

$$Q|0\rangle \neq 0$$
. (2.50)

In other words, the symmetry of the Lagrangian is realized as the degeneration of vacuums and it is not manifest in the real world constructed on one vacuum. This is the spontaneous symmetry breakdown. The remnant of the symmetry is, however, observed through the scalar mode called the Nambu-Goldstone (NG) particle which, roughly speaking, corresponds to the state (2.50). Taking account of the commutativity (2.49), we find that this mode is a massless excitation.

In order to describe spontaneous symmetry breakdown, it is most convenient to introduce scalar fields which transform as in Eq. (2.48). We assume that the vacuum expectation value of $\delta\phi$ does not vanish:

$$\langle 0|\delta\phi|0\rangle \neq 0$$
, (2.51)

which implies Eq. $(2 \cdot 50)$ and hence spontaneous symmetry breakdown. To have the non-vanishing vacuum expectation value, we may add to \mathcal{L} a potential term of $\delta\phi$ which assures the stability of this vacuum. We may regard this scalar as an elementary field or a tool of effective expression of the spontaneous

^{*)} Strictly speaking, the charge Q is ill-defined in this case, but, for simplicity, we use Q in the present argument.

symmetry breakdown. In the Hilbert space constructed on this vacuum, it is convenient to expand $\delta\phi$ around its vacuum expectation value and separate it into classical and quantum-fluctuation parts. Thus we obtain the effective Lagrangian on this vacuum.

In the case that the symmetry is a local symmetry, that is, gauge symmetry, the Nambu-Goldstone mode is absorbed into the gauge boson as its longitudinal component and consequently the gauge boson turns out to be massive. This is called the Higgs mechanism [Higgs 64, 66]. The above phenomenon is now explained in the following. The gauge invariant kinetic term of scalar ϕ which is called the Higgs scalar takes the form,

$$|D_{\mu}\phi|^{2} \equiv |(\partial_{\mu} - igT^{a}A_{\mu}^{a})\phi|^{2}, \qquad (2.52)$$

where T^a is the generator in the representation of ϕ . We suppose that the symmetry generated by charge Q^a is spontaneously broken:

$$Q^a|0\rangle \neq 0. \tag{2.53}$$

The above equation is represented in terms of scalar ϕ such that

$$\langle \delta^a \phi \rangle = i T^a \langle \phi \rangle \neq 0 . \tag{2.54}$$

The gauge boson A_{μ}^{a} corresponding to Q^{a} has, on this vacuum, the mass matrix generated from $(2 \cdot 52)$:

$$\mathcal{L}_{\text{mass}} = M_{ab} A_{\mu}{}^{a} A^{b\mu}, \qquad (2.55)$$

$$M_{ab} \equiv \langle \phi^* \rangle T^a T^b \langle \phi \rangle . \tag{2.56}$$

The mechanism of the absorption of the Nambu-Goldstone particle (corresponding to the field $\delta\phi$) is understood in a covariant gauge as the fact that the NG mode becomes a member of an unphysical quartet.

Fermions can also have masses through spontaneous symmetry breakdown. If scalar ϕ interacts with fermions in a gauge invariant way:

$$\mathcal{L}_{\text{int}} = \sum \bar{\psi} \psi \phi , \qquad (2.57)$$

the following mass terms emerge,

$$\mathcal{L}_{\text{mass}} = \sum \overline{\phi} \phi \langle \phi \rangle . \qquad (2.58)$$

Now let us discuss the Weinberg-Salam model. A common expectation based on the phenomenology is that the weak interaction is mediated by massive gauge bosons, while the electromagnetic interaction is described by the photon which is a massless gauge field. Thus the $SU(2) \times U(1)$ sym metry of the Lagrangian (2.36) must be broken spontaneously in the following manner,

$$SU(2) \times U(1) \rightarrow U(1)_{\text{EM}}$$
.

The symmetry breaking of this pattern is made possible by introducing an adequate Higgs scalar which develops a vacuum expectation value as follows:

$$T^{a}\langle\phi\rangle\neq0, \quad (a=0,1,2,3)$$

$$Q\langle\phi\rangle=0, \quad (2.59)$$

where Q is the electromagnetic charge generator defined by

$$Q = T^{\mathfrak{s}} + T^{\mathfrak{o}} \,. \tag{2.60}$$

On this vacuum only one gauge boson which couples to Q remains massless and others acquire masses.

In order to make fermions massive, it is necessary to introduce SU(2) doublet Higgs fields, because the $\overline{\psi}_{L}\psi_{R}$ term belongs to the doublet representation and Yukawa interactions $\overline{\psi}_{L}\psi_{R}\phi$ must be gauge invariant.

Taking account of the above argument, we introduce one SU(2) doublet Higgs field for which the eigenvalue of T^0 is 1/2. As potential terms of the Higgs field we take the form,

$$V(\phi) \equiv -\mu^2 \boldsymbol{\varPhi}^{\dagger} \boldsymbol{\varPhi} + \lambda \left(\boldsymbol{\varPhi}^{\dagger} \boldsymbol{\varPhi} \right)^2$$
$$= \lambda \left(\boldsymbol{\varPhi}^{\dagger} \boldsymbol{\varPhi} - \frac{\boldsymbol{\upsilon}^2}{2} \right)^2 - \frac{\mu^4}{4\lambda}, \qquad (2 \cdot 61)$$

where v is introduced to denote the minimum point of the potential $V(\phi)$. The vacuum expectation value of Φ satisfies

$$\langle \boldsymbol{\varPhi}^{\dagger} \rangle \langle \boldsymbol{\varPhi} \rangle = \frac{1}{2} v^2 \,.$$
 (2.62)

Taking a suitable SU(2) phase convention, we may write in general,

$$\langle \mathbf{0} \rangle = \begin{pmatrix} 0 \\ v/\sqrt{2} \end{pmatrix}$$
. (v: real) (2.63)

This convention corresponds to our representation of charge generator Q, (2.60), and it is easily verified that the symmetry generated by Q is not broken on this vacuum:

$$Q\langle \boldsymbol{\varPhi} \rangle = \begin{pmatrix} 1 & 0 \\ 0 & 0 \end{pmatrix} \begin{pmatrix} 0 \\ \boldsymbol{\upsilon}/\sqrt{2} \end{pmatrix} = 0 .$$
 (2.64)

It should be noted here that a single Higgs doublet field necessarily led to the breakdown of $SU(2) \times U(1)$ down to U(1). This remaining U(1)symmetry is interpreted as that of QED in any convention with an appropriate redefinition of the charge generator.

The expression of Higgs \mathcal{O} on this vacuum is given in terms of hermitian fields ϕ , χ_1 , χ_2 and χ_3 such that

$$\varPhi = \frac{1}{\sqrt{2}} \begin{pmatrix} i\chi_1 + \chi_2 \\ v + \phi - i\chi_3 \end{pmatrix}.$$
(2.65)

The Lagrangian for the Higgs field reads

$$\mathcal{L}_{\mathrm{H}} = (D_{\mathrm{L}}^{\mu} \boldsymbol{\varPhi})^{\dagger} (D_{\mathrm{L}\mu} \boldsymbol{\varPhi}) + \mu^{2} \boldsymbol{\varPhi}^{\dagger} \boldsymbol{\varPhi} - \lambda (\boldsymbol{\varPhi}^{\dagger} \boldsymbol{\varPhi})^{2} . \qquad (2 \cdot 66)$$

Through \mathcal{L}_{H} the gauge boson mass matrix is generated as follows:

$$\mathcal{L}_{\text{mass}} = M_{ab} W_{\mu}^{a} W^{b\mu},$$

$$M_{ab} = \frac{\begin{vmatrix} W_{1} & W_{2} & W_{3} & W_{6} \end{vmatrix}}{\begin{vmatrix} W_{1} & g^{2} & 0 & 0 & 0 \end{vmatrix}} \\ W_{2} & 0 & g^{2} & 0 & 0 \\ W_{3} & 0 & 0 & g^{2} & -gg' \\ W_{3} & 0 & 0 & -gg' & g'^{2} \end{vmatrix}$$

$$(2.67)$$

By diagonalizing the above mass matrix, we obtain

$$\mathcal{L}_{\text{mass}} = M_W^2 W_{\mu}^{+} W^{-\mu} + \frac{1}{2} M_Z^2 Z_{\mu} Z^{\mu} , \qquad (2.68)$$

$$M_W = \frac{1}{2} gv , \qquad M_Z = \frac{1}{2} \sqrt{g^2 + {g'}^2} v ,$$

$$W_{\mu}^{\pm} = (W_{\mu}^{-1} \mp i W_{\mu}^{-2}) / \sqrt{2} ,$$

$$Z_{\mu} = (g W_{\mu}^{-3} - g' W_{\mu}^{-0}) / \sqrt{g^2 + {g'}^2} ,$$

$$A_{\mu} = (g' W_{\mu}^{-3} + g W_{\mu}^{-0}) / \sqrt{g^2 + {g'}^2} , \qquad (2.69)$$

where W_{μ}^{\pm} , Z_{μ} and A_{μ} represent the charged weak bosons, the neutral weak boson and the photon respectively. There is no mass term corresponding to the photon because it couples to the charge of the remaining symmetry, Q.

The corresponding Nambu-Goldstone bosons of W_{μ}^{\pm} and Z_{μ} are represented by $\chi^{\pm} (= (\chi_1 \mp i \chi_2) / \sqrt{2})$ and χ_s , and they are all massless as easily seen in \mathcal{L}_{H} (2.66). These NG bosons form three "quartets" with *B* fields (scalar polarizations of gauge fields) and FP-ghosts and antighosts. These "quartets" are all unphysical and can appear only with zero norm in the physical Hilbert space. This is guaranteed by the physical state condition (2.47) (see §§ 2.2 and 3.3). There are also mixing terms of the form $\partial_{\mu}\chi \cdot W^{\mu}$ in \mathcal{L}_{H} . In the 't Hooft gauge which we adopt in the following, they are all cancelled away by the corresponding ones coming from the gauge fixing terms (see $\S 4.1$).

Thus three components of Higgs field $\boldsymbol{\Phi}$ are unphysical NG bosons and there remains the field $\boldsymbol{\phi}$ which is the physical Higgs boson. This particle has mass,

$$m_{\phi} = \sqrt{2}\,\mu\,,\tag{2.70}$$

and is observable though it has not yet been found.

The Yukawa interaction of fermions with Higgs field ϕ is given by

$$\mathcal{L}_{\mathrm{M}} = -f\bar{L}\phi R + \mathrm{h.c.}, \qquad (2.71)$$

which generates fermion mass terms,

$$\mathcal{L}_{\text{mass}} = -f(v/\sqrt{2})\cdot\overline{\psi}_e\psi_e$$
.

The Yukawa coupling f is a free parameter in our Lagrangian and hence the electron mass is an input parameter as in QED.

We finally obtain the full Lagrangian by adding \mathcal{L}_{H} (2.66) and \mathcal{L}_{M} (2.71) to \mathcal{L}' of Eq.(2.36),

$$\mathcal{L} = \mathcal{L}_{G} + \mathcal{L}_{F} + \mathcal{L}_{GF} + \mathcal{L}_{FP} + \mathcal{L}_{H} + \mathcal{L}_{M}. \qquad (2.72)$$

This is our basic Lagrangian with which we start our argument on renormalization and application to the calculation of higher order effects in leptonic processes.

Here we study the physical contents of the electroweak interactions described by our Lagrangian in the tree approximation. The term of our Lagrangian corresponding to the photon coupling is

$$\mathcal{L}_{\rm EM} = (gg'/\sqrt{g^2 + g'^2}) A^{\mu} \overline{\psi}_{e} \gamma_{\mu} \psi_{e} + \cdots, \qquad (2.73)$$

where only the photon-fermion coupling is explicitly shown. In order for this interaction to reproduce QED, we identify the electromagnetic coupling constant as follows,

$$e \equiv gg' / \sqrt{g^2 + {g'}^2} \,. \tag{2.74}$$

Next we consider effective four-fermion interactions in the low-energy limit mediated by massive gauge bosons. There are two kinds of these effective four-fermion interactions. One is the charged current interaction mediated by the W^{\pm} boson:

$$\mathcal{L}_{\text{charged}}^{\text{eff}} = (g^2 / 8M_W^2) j_{\mu}^{\dagger} j^{\mu}, \qquad (2.75)$$

where j_{μ} is defined in Eq. (2.5). Comparing it with Eq. (2.1), we obtain the following relation:

26

Chap. 2 Structure of Electroweak Theory

$$G_{\rm F}/\sqrt{2} = g^2/8M_{\rm W}^2 = 1/2v^2, \qquad (2.76)$$

which determines the vacuum expectation value, the mass scale of the symmetry breakdown.

The other is the neutral current interaction mediated by the Z boson, which is parametrized as follows:

$$\mathcal{L}_{\text{neutral}}^{\text{eff}} = \sqrt{2} G_{\text{F}} j_{\mu}^{\text{NC}} j^{\text{NC}\mu}, \qquad (2.77)$$

where

$$j_{\mu}^{NC} = \overline{\psi} \left\{ T^{\mathfrak{s}}(1 - \gamma_{\mathfrak{s}}) - 2 \sin^2 \theta_{W} Q \right\} \psi, \qquad (2 \cdot 78)$$

$$\sin \theta_{\rm w} = g' / \sqrt{g^2 + {g'}^2} \,. \tag{2.79}$$

The angle θ_W is the rotation angle between (W^3, W^0) and (Z, A) and is called the Weinberg angle. We can express it by the masses of W^{\pm} and Z:

$$\sin^2\!\theta_{\rm W} = 1 - M_{\rm W}^2 / M_Z^2 \,. \tag{2.80}$$

This neutral current interaction is very important to determine the electroweak theory among many gauge models as mentioned in § 1.2. We should notice that the neutral current is parametrized in a simple way with one unknown parameter $\sin^2\theta_w$, but this simple parametrization is possible only in the case of tree approximation (see Chapter 7).



Fig. 2.2. Relations among tree parameters g(SU(2) coupling), $g^{*}(U(1) \text{ coupling})$, θ_{W} (the Weinberg angle), M_{W} and M_{Z} (gauge boson masses) and the electromagnetic coupling e.

Since the beginning of § 2.1, we have restricted ourselves to the case of single generation of leptons. The final form of our Lagrangian (2.72)includes only the electron and electron-neutrino as a fermion family. We are now in a position to extend our scheme to the more realistic case of multigenerations of leptons and quarks.

As has already been mentioned in $\S1.2$, leptons and quarks constitute

27

singlets and doublets in $SU(2) \times U(1)$. We represent fermions in the following way:

$$R_{i} = (\psi_{i})_{\mathrm{R}}, \quad R_{I} = (\psi_{I})_{\mathrm{R}}, \quad L_{I} = \begin{pmatrix} \psi_{I} \\ \psi_{I}' \end{pmatrix}_{\mathrm{L}}, \quad (2 \cdot 81)$$

where *i* and I(=1, 2, 3) are indices representing 3 generations of leptons and quarks, ψ_i and ψ_I correspond to the usual up $(T^s = +1/2)$ and down $(T^s = -1/2)$ components of leptons and quarks and suffixes R and L represent the right-handed and left-handed components of field ψ :

$$\psi_{\mathbf{R},\mathbf{L}} = \frac{1 \pm \gamma_5}{2} \psi \,. \tag{2.82}$$

The primes of ψ_i and ψ_i express the mixing among leptons or quarks,

$$\psi_i' = U_{iI}\psi_I, \quad \psi_I' = U_{Ii}^{-1}\psi_i \tag{2.83}$$

with U the mixing matrix. If the neutrinos are massless, then U=1 for leptons (this corresponds to the case of no mixing in the minimum Higgs scheme). For later convenience we introduce a doublet

$$L_{i} = \begin{pmatrix} \psi_{i}' \\ \psi_{i} \end{pmatrix}_{\mathrm{L}} . \tag{2.84}$$

It should be noted that L_i and L_I is related by

$$L_i = U_{iI} L_I \,. \tag{2.85}$$

The full Lagrangian corresponding to Eq. (2.72) can be derived in the present general case just in the same way as before. The resulting Lagrangian is given by

$$\mathcal{L} = \mathcal{L}_{G} + \mathcal{L}_{F} + \mathcal{L}_{GF} + \mathcal{L}_{FP} + \mathcal{L}_{H} + \mathcal{L}_{M}, \qquad (2 \cdot 86)$$

where \mathcal{L}_{G} , \mathcal{L}_{GF} , \mathcal{L}_{FP} and \mathcal{L}_{H} are the same as the previous ones, i.e., (2.18), (2.36a), (2.36b) and (2.66) respectively. Only the parts of the Lagrangian which include fermions are subject to change,

$$\mathcal{L}_{\mathrm{F}} = i \sum_{I} \overline{L}_{I} \mathcal{D}_{\mathrm{L}} L_{I} + i \sum_{n=i,I} \overline{R}_{n} \mathcal{D}_{\mathrm{R}} R_{n}, \qquad (2.87)$$

$$\mathcal{L}_{\mathrm{M}} = -\sum_{i} f_{i} \overline{L}_{i} \varPhi R_{i} - \sum_{I} f_{I} \overline{L}_{I} (i\tau_{2} \varPhi^{*}) R_{I} + \text{h.c.}, \qquad (2 \cdot 88)$$

where f_i and f_I are the Yukawa coupling constants. Owing to the nonvanishing vacuum expectation value of the Higgs field $\boldsymbol{\emptyset}$, fermions acquire masses through the Lagrangian (2.88). The masses of fermions are given in the tree approximation by Chap. 2 Structure of Electroweak Theory

$$m_{i,I} = f_{i,I} v / \sqrt{2}$$
 (2.89)

If we use Eq. (2.76) with the value of G_F determined by the experimental data on the muon decay width in the tree approximation, i.e.,

$$G_{\rm F} = 1.166 \times 10^{-5} {\rm GeV}^{-2}, \qquad (2.90)$$

we obtain

$$v/\sqrt{2} = 174 \,\mathrm{GeV}$$
 (2.91)

This is a typical mass scale at which the symmetry $SU(2) \times U(1)$ breaks down to $U(1)_{\text{EM}}$ spontaneously. Using Eqs.(2.69), (2.80) and (2.91), we find

$$M_{\rm w} = 37.3/|\sin \theta_{\rm w}|$$
, (GeV)
 $M_z = 74.6/|\sin 2\theta_{\rm w}|$. (GeV) (2.92)

Recent experimental data on $\nu_{\mu}e$ and $\nu_{\mu}N$ scattering and polarized eN scattering suggest (assuming the massless neutrino) that

$$\sin^2\theta_{\rm W}\simeq 0.235\,,\qquad(2\cdot93)$$

giving

$$M_w \simeq 77 \text{ GeV}$$
,
 $M_z \simeq 88 \text{ GeV}$. (2.94)

The above arguments on parameters M_W , M_Z , G_F and $\sin^2\theta_W$ are restricted to the tree level. If one takes into account higher order effects and performs the necessary renormalization, one finds that those relations like Eqs. (2.69) (2.70), (2.74), (2.76) and (2.89) may be modified. A detailed discussion of the higher order effects will be given in Chapter 7.

As we have seen, the experimental information on e, $G_{\rm F}$ and $\sin^2 \theta_{\rm W}$ determines all the parameters relevant to all effective four-fermion weak interactions (at the tree level). Thus the gauge theory of electroweak interactions proves itself to have higher predictive power.

§ 2.4 Ward-Takahashi identities

The Ward-Takahashi (WT) identities are the relations among Green functions given by symmetries of the theory. Originally the WT identities were discovered in QED [Ward 50, Takahashi 57]. In these days generalized ones are usually referred to. There are an infinite number of these WT identities. In this section we restrict ourselves to derivation of the WT identities based on the BRS symmetry which has been stated in the previous section.

29

The BRS symmetry plays an essential role to construct the operator formalism of gauge theories as a physically consistent quantum field theory in a covariant way. For example, consider the unitarity of physical S-matrix. It is guaranteed in terms of the quartet mechanism as explained in § 2. 2. This mechanism states that unphysical quartet members appear in the physical Hilbert space only as zero-norm states. This is equivalent to the existence of the appropriate WT identities of Green functions including unphysical particles in the final state although the following type of the WT identities is not explicitly used in the proof of the quartet mechanism [Kugo and Ojima 78].

We derive the WT identities for the effective action given by the BRS symmetry. These identities will be used to prove the renormalizability ($\S 2.5$), the consistency of the renormalization conditions ($\S 3.3$) and the charge universality ($\S 3.4$).

Let us define the generating functional W[J, K] of Green functions.

$$\exp iW[J,K] \equiv \langle 0|\mathrm{T} \exp iS[J,K]|0\rangle, \qquad (2.95)$$

$$S[J,K] = \int d^4x \left[J_i(x)\hat{\phi}_i(x) + K_i(x)\delta^{\text{BRS}}\hat{\phi}_i(x) \right], \qquad (2.96)$$

where $\hat{\phi}_i(x)$ stand for all field operators^{*}) in the theory and $J_i(x)$ and $K_i(x)$ are *c*-number source currents with the same commutativity as that of corresponding $\hat{\phi}_i(x)$ and $\delta^{\text{BRS}} \hat{\phi}_i(x)$. It should be mentioned that the differentiation with respect to anticommuting η is understood to be the so-called leftdifferentiation, which corresponds to the distribution formula:

$$D_{\eta}(AB) = (D_{\eta}A)B + (-1)_{\eta}^{A}A(D_{\eta}B),$$

where $(-1)_{\eta}^{A}$ is +1 (-1) when A is commutative (anticommutative) with η .

Differentiation of the above W[J, K] with respect to J_i leads to a connected Green function of the field $\hat{\phi}$:

$$\frac{1}{i^{n-1}} \frac{\delta^n W[J,K]}{\delta J_1 \delta J_2 \cdots \delta J_n} \bigg|_{J=K=0} = \langle 0 | \mathrm{T} \hat{\phi}_1 \cdots \hat{\phi}_n | 0 \rangle^c . \tag{2.97}$$

The proof of this equation is given in Appendix G. As a next step, following the usual procedure, we define classical fields ϕ_t in terms of W,

$$\phi_i = \frac{\delta}{\delta J_i} W[J, K]. \tag{2.98}$$

We define the effective action $\Gamma[\phi, K]$ as the Legendre transform of W[J, K] with respect to J:

$$\Gamma[\phi, K] \equiv W[J[\phi, K], K] - J_i[\phi, K]\phi_i. \qquad (2.99)$$

^{*)} In the previous sections, ϕ_i has been used for field operators. In this section, ϕ_i stand for classical fields and field operators are written as $\hat{\phi}_i$.

It should be noticed that the source J is a functional of ϕ and K. In fact, with the help of Γ the source J is expressed as

Chap. 2

$$J_i = -(+) \frac{\delta \Gamma}{\delta \phi_i}$$
 for commuting (anticommuting) field. (2.100)

By successive differentiations with respect to ϕ , we obtain one-particle irreducible Green functions:

$$\frac{1}{i} \frac{\delta^{2} \Gamma[\phi, K]}{\delta \phi_{i} \delta \phi_{j}} \Big|_{\phi = \langle \hat{\phi} \rangle, K = 0} \langle 0 | \mathrm{T} \hat{\phi}_{i} \hat{\phi}_{k} | 0 \rangle^{c} = \delta_{jk}, \qquad (2 \cdot 101)$$

$$i \frac{\delta^{n} \Gamma[\phi, K]}{\delta \phi'_{i_{1}} \cdots \delta \phi'_{i_{n}}} \Big|_{\phi = \langle \hat{\phi} \rangle, K = 0} \langle 0 | \mathrm{T} \hat{\phi}_{i_{1}} \hat{\phi}_{i_{1}'} | 0 \rangle^{c} \cdots \langle 0 | \mathrm{T} \hat{\phi}_{i_{n}} \hat{\phi}_{i_{n}'} | 0 \rangle^{c}$$

$$= \langle 0 | \mathrm{T} \hat{\phi}_{i_{1}} \cdots \hat{\phi}_{i_{n}} | 0 \rangle^{c} \text{ for } n \geq 3. \qquad (2 \cdot 102)$$

See Appendix G for the proof of these formulas. In these formulas the ordering of operators should be kept carefully so that the correct sign is obtained for anticommuting fields.

Let us derive the WT identity for the effective action Γ . We start with a trivial relation:

$$\langle 0| \left[i \lambda Q_{\rm B}, \, \mathrm{T} \exp i S[J, \, K] \right] | 0 \rangle = 0 , \qquad (2 \cdot 103)$$

the validity of which is clear if we remember the physical state condition $(2\cdot47)$,

$$Q_{\rm B}|0\rangle = 0$$

Noticing Eq. $(2 \cdot 46)$,

$$\delta^{\text{BRS}}(\delta^{\text{BRS}}\hat{\phi}_i) = 0$$
,

we obtain

$$\int d^4x J_i(x) \langle 0| \mathrm{T}\delta^{\mathrm{BRS}} \hat{\phi}_i(x) \exp iS[J,K] |0\rangle = 0, \qquad (2 \cdot 104)$$

which is expressed in terms of the effective action Γ as

$$\int d^4x \frac{\delta\Gamma}{\delta\phi_i(x)} \frac{\delta\Gamma}{\delta K_i(x)} \equiv \frac{\delta\Gamma}{\delta\phi_i} \cdot \frac{\delta\Gamma}{\delta K_i} = 0, \qquad (2.105)$$

where use has been made of Eq. $(2 \cdot 100)$ and

$$\frac{\delta W}{\delta K_i}\Big|_{J \text{ fixed}} = \frac{\delta \Gamma}{\delta K_i}\Big|_{\phi \text{ fixed}}.$$
(2.106)

Equation $(2 \cdot 105)$ is a goal of our derivation. We write all fields in Eq. $(2 \cdot 105)$ explicitly:

$$\frac{\delta\Gamma}{\delta W_{\mu}{}^{a}} \cdot \frac{\delta\Gamma}{\delta K_{a}{}^{\mu}} + \frac{\delta\Gamma}{\delta \psi_{i}} \cdot \frac{\delta\Gamma}{\delta K_{\psi_{i}}} + \frac{\delta\Gamma}{\delta \overline{\psi}_{i}} \cdot \frac{\delta\Gamma}{\delta K_{\overline{\psi}_{i}}} + \frac{\delta\Gamma}{\delta \varphi_{i}} \cdot \frac{\delta\Gamma}{\delta K_{\theta_{i}}} + \frac{\delta\Gamma}{\delta \overline{c}^{a}} \cdot \frac{\delta\Gamma}{\delta K_{e^{a}}} = 0, \qquad (2 \cdot 107)$$

where we have used the following relation:

$$\frac{\delta\Gamma}{\delta K_{\bar{c}}} = \frac{\delta W[J,K]}{\delta J_B} = B. \qquad (2.108)$$

The identity (2.107) can be changed to a more useful form. We start with the equations of motion for fields \hat{B}^a and \hat{c}^a ,

$$F_i{}^a\hat{\phi}_i + \alpha \hat{B}^a = 0, \qquad (2 \cdot 109)$$

$$\delta^b(F_i{}^a\hat{\phi}_i)\hat{c}^b=0, \qquad (2\cdot110)$$

where we have taken the same gauge fixing Lagrangian as that in §2.3:

$$\mathcal{L}_{\rm GF} = F_i{}^a \hat{\phi}_i \hat{B}^a + \frac{\alpha}{2} \hat{B}^a \hat{B}^a \,. \label{eq:GF}$$

We rewrite the trivial relations

$$\langle 0|T(F_i{}^a\hat{\phi}_i + \alpha \hat{B}^a) \exp iS[J, K]|0\rangle = 0, \qquad (2.111)$$

$$\langle 0|T\left(\delta^{b}(F_{i}^{a}\widehat{\phi}_{i})\widehat{c}^{b}\right)\exp\,iS[J,K]|0\rangle = 0\,. \tag{2.112}$$

The covariant combination $\partial^{\mu} \widehat{W}_{\mu}{}^{a}$ should appear in the terms $F_{i}{}^{a} \widehat{\phi}_{i}$ because the canonical conjugate momentum of $\widehat{W}_{0}{}^{a}$, $\widehat{\Pi}_{\widehat{W}_{0}a} \equiv \delta \widehat{\mathcal{L}} / \delta \widehat{W}_{0}{}^{a}$, can be obtained from the gauge fixing terms, $\widehat{B} \cdot F_{i}{}^{a} \widehat{\phi}_{i}$. We extract the differential operator out of the *T*-product taking into account the equal time commutator terms between $\widehat{W}_{0}{}^{a}$ and exp i S[J, K]. Noting that only *B* field, i.e., $\widehat{\Pi}_{\widehat{W}_{0}a}$ in S[*J*, *K*] does not commute with $\widehat{W}_{0}{}^{a}$ at equal time, we obtain from Eq.(2.111)

$$(F_{i}^{a}\phi_{i}+\alpha B^{a})\langle 0|T \exp iS[J,K]|0\rangle - \langle 0|T[\hat{W}_{0}^{a},\exp iS[J,K]]_{ETC}|0\rangle$$

$$= (F_{i}^{a}\phi_{i}+\alpha B^{a}+J_{B}^{a})\langle 0|T \exp iS[J,K]|0\rangle$$

$$= \left(F_{i}^{a}\phi_{i}+\alpha B^{a}-\frac{\delta\Gamma}{\delta B^{a}}\right)\langle 0|T \exp iS[J,K]|0\rangle$$

$$= 0. \qquad (2.113)$$

Similarly, in Eq. (2.112) we notice that in the terms $\delta^b(F_i^a \hat{\phi}_i) \hat{c}^b$ there are differential terms of the canonical conjugate momentum of \hat{c}^a , $\partial^{\mu} \hat{\Pi}_{\hat{c}^a}$, which con-

tribute as equal time commutator terms. Then we obtain

$$F_{i}^{a}\langle 0|T\delta^{BRS}\hat{\phi}_{i} \exp iS[J, K]|0\rangle - \langle 0|T[\hat{H}_{\hat{\epsilon}^{a}}, \exp iS[J, K]]_{ETC}|0\rangle$$

$$= \left(F_{i}^{a}\frac{\delta W}{\delta K_{i}} + J_{\bar{\epsilon}^{a}}\right)\langle 0|T \exp iS[J, K]|0\rangle$$

$$= \left(F_{i}^{a}\frac{\delta \Gamma}{\delta K_{i}} + \frac{\delta \Gamma}{\delta \bar{\epsilon}^{a}}\right)\langle 0|T \exp iS[J, K]|0\rangle$$

$$= 0. \qquad (2.114)$$

In this way Eqs. $(2 \cdot 111)$ and $(2 \cdot 112)$ are changed to

$$F_i^a \phi_i + \alpha B^a = \frac{\delta \Gamma}{\delta B^a}, \qquad (2.115)$$

$$F_i^a \frac{\delta \Gamma}{\delta K_i} = -\frac{\delta \Gamma}{\delta \bar{c}^a}.$$
 (2.116)

Although these relations are obtained by using the equations of motion, they should be regarded as kinematical relations based on the BRS invariance since they are related only with \mathcal{L}_{GF} and \mathcal{L}_{FP} .

It is more convenient to introduce the modified effective action $\widetilde{\varGamma}$ defined by

$$\widetilde{\Gamma} = \Gamma - \int \mathcal{L}_{\rm GF} d^4 x \,, \qquad (2.117)$$

where \mathcal{L}_{GF} is understood as described in terms of classical fields. $\widetilde{\Gamma}$ is independent of B:

$$\frac{\delta \widetilde{\Gamma}}{\delta B^a} = \frac{\delta \Gamma}{\delta B^a} - (F_i^a \phi_i + \alpha B^a) = 0, \qquad (2.118)$$

where use has been made of Eq.(2.115). Substituting $\tilde{\Gamma}$ into Eq.(2.116), we obtain

$$F_i^{a} \frac{\delta \tilde{\Gamma}}{\delta K^{i}} + \frac{\delta \tilde{\Gamma}}{\delta \bar{c}^{a}} = 0.$$
(2.119)

Let us rewrite Eq. (2.107) by using $\widetilde{\Gamma}$:

$$\begin{split} \frac{\delta\widetilde{\Gamma}}{\delta W_{\mu}{}^{a}} \cdot \frac{\delta\widetilde{\Gamma}}{\delta K_{a}{}^{\mu}} + \frac{\delta\widetilde{\Gamma}}{\delta \psi_{i}} \cdot \frac{\delta\widetilde{\Gamma}}{\delta K_{\phi_{i}}} + \frac{\delta\widetilde{\Gamma}}{\delta \overline{\psi}_{i}} \cdot \frac{\delta\widetilde{\Gamma}}{\delta K_{\phi_{i}}} + \frac{\delta\widetilde{\Gamma}}{\delta \phi_{i}} \cdot \frac{\delta\widetilde{\Gamma}}{\delta K_{\sigma^{a}}} + \frac{\delta\widetilde{\Gamma}}{\delta c^{a}} \cdot \frac{\delta\widetilde{\Gamma}}{\delta K_{\sigma^{a}}} \\ &= -\frac{\delta}{\delta \phi_{i}} \int \mathcal{L}_{\rm GF} d^{4}x \cdot \frac{\delta\widetilde{\Gamma}}{\delta K_{i}} - \frac{\delta\widetilde{\Gamma}}{\delta \overline{c}} \cdot B \,. \end{split}$$

The right-hand side terms vanish according to Eq. (2.119) as follows:

Downloaded from https://academic.oup.com/ptps/article-abstract/doi/10.1143/PTPS.73.1/1848359 by guest on 06 January 2019

K-I. Aoki, Z. Hioki, R. Kawabe, M. Konuma and T. Muta

$$\frac{\delta}{\delta\phi_i}\int \mathcal{L}_{\rm GF} d^4x \cdot \frac{\delta\widetilde{\Gamma}}{\delta K_i} + \frac{\delta\widetilde{\Gamma}}{\delta\overline{c}^a} \cdot B = B \cdot \left(F_i^a \frac{\delta\widetilde{\Gamma}}{\delta K_i} + \frac{\delta\widetilde{\Gamma}}{\delta\overline{c}^a}\right) = 0.$$

The final form of the WT identity is

$$\frac{\delta \widetilde{\Gamma}}{\delta W_{\mu}^{a}} \cdot \frac{\delta \widetilde{\Gamma}}{\delta K_{a}^{\mu}} + \frac{\delta \widetilde{\Gamma}}{\delta \psi_{i}} \cdot \frac{\delta \widetilde{\Gamma}}{\delta K_{\psi_{i}}} + \frac{\delta \widetilde{\Gamma}}{\delta \overline{\psi}_{i}} \cdot \frac{\delta \widetilde{\Gamma}}{\delta K_{\overline{\psi}_{i}}} + \frac{\delta \widetilde{\Gamma}}{\delta \mathcal{K}_{o_{i}}} + \frac{\delta \widetilde{\Gamma}}{\delta \mathcal{K}_{c^{a}}} \cdot \frac{\delta \widetilde{\Gamma}}{\delta K_{c^{a}}} = 0. \qquad (2.120)$$

We differentiate Eq. $(2 \cdot 120)$, for example, with respect to ϕ and set ϕ to their vacuum expectation values: $\phi = \langle \hat{\phi} \rangle$. Each obtained term is a product of two one-particle irreducible Green functions:



Fig. 2.3. Graphical representation of a Ward-Takahashi identity (2.121).

We express Eq.(2.121) graphically as in Fig. 2.3. In this graph the black dot at the center represents an operator product $\delta^{BRS}\hat{\phi}_j$ and the right half of the diagram stands for the Green function into which the operator product $\delta^{BRS}\hat{\phi}_j$ is inserted. The dotted line represents the ghost propagator. In the case that $\delta^{BRS}\hat{\phi}_j$ contain linear terms of fields, special attention is required. For gauge fields, for example, the right half of Eq.(2.121) is

$$\langle 0|T\delta^{\mathrm{BRS}}\hat{W}_{\mu}{}^{a}\cdots|0
angle_{\mathrm{proper}}{}^{c}=\langle 0|T(gf^{abc}\hat{W}_{\mu}{}^{b}\hat{c}{}^{c}+\partial_{\mu}\hat{c}{}^{a})\cdots|0
angle_{\mathrm{proper}}{}^{c}$$

As it stands, the contribution of the second term on the right-hand side is not included in Fig. 2.3. We understand, however, that the diagram in the figure represents the whole contribution including that of $\partial_{\mu}\hat{c}^{\alpha}$ for simplicity of graphical expression.

In the following sections, these WT identities, Eqs. $(2 \cdot 118) \sim (2 \cdot 120)$ play an important role in proving the renormalizability and charge universality and in analyzing consistency of the renormalization conditions.

§ 2.5 Renormalizability

It was proved first by 't Hooft that non-Abelian gauge theories are re-

normalizable ['t Hooft 71b, c]. Further contributions have clarified the structure of gauge theories [Lee and Zinn-Justin 72a, b, c, Slavnov 72, Taylor 71, t' Hooft and Veltman 72b, Fujikawa et al. 72, for review articles, see Abers and Lee 73, Lee 76]. In this section general theory of the renormalizability is briefly reviewed and the Ward-Takahashi (WT) identities are derived which constitute the essential part of the renormalizability of gauge theories.

Consider a theory described by a Lagrangian $\mathcal{L}(g_0, \phi_0)$ where g_0 and ϕ_0 generically represent bare coupling constants (including masses) and bare fields respectively. The effective action Γ of the theory defined in § 2.4 is calculated by the loop expansion. This Γ as a functional of g_0 and ϕ_0 includes divergences. The renormalizability of the theory dictates that the effective action is a finite functional Γ_R of g_R and ϕ_R ,

$$\Gamma_{\mathbf{R}}(g_{\mathbf{R}},\phi_{\mathbf{R}}) = \Gamma(g_{\mathbf{0}},\phi_{\mathbf{0}}), \qquad (2 \cdot 122)$$

where g_{R} and ϕ_{R} are defined by an appropriate transformation of g_{0} and ϕ_{0} . We introduce renormalization constants Z through the transformation,

$$g_0 = {}_g Z g_{\mathrm{R}} , \qquad (2 \cdot 123 \mathrm{a})$$

$$\phi_0 = Z_s^{1/2} \phi_R , \qquad (2 \cdot 123b)$$

and assume the loop expansion of the above renormalization constants,

¢

$$Z = 1 + Z^{(1)} + Z^{(2)} + \cdots .$$
 (2.124)

These $Z^{(n)}$'s generate counter terms for the effective action Γ_{R} in the loop expansion. The problem is whether all divergent integrals in Γ_{R} can be cancelled out by available counter terms loop by loop.

In each order of the loop expansion counter terms play a role to subtract polynomials in momentum variable from Feynman amplitudes. The resultant renormalized Feynman integrals are written in the closed form, where subtraction is expressed by Taylor operators such as

$$1 - \sum_{i} \tau_{i}, \quad \tau_{i} \equiv (\partial_{p})^{i}|_{p=\mu}.$$

By the use of this closed form for renormalized Feynman integrals, it was strictly proved that if sufficient counter terms are prepared to subtract all superficial divergences, all renormalized Feynman integrals are convergent and finite, that is, the theory is renormalizable [Bogoliubov and Parasiuk 57, Hepp 66, Zimmermann 69]. Here the superficial degree of divergence of a Feynman integral is the degree of divergence obtained from purely dimensional analysis.

Renormalizability requires first of all that the number of types of superficially divergent graphs is finite, otherwise we must prepare an infinite number of (bare) parameters to renormalize the theory. This requirement is satisfied only in the case where all coupling constants g_0 in \mathcal{L} have non-negative dimension in the momentum scale [Sakata et al. 51, 52]. Gauge theories satisfy the condition: Gauge coupling constants are dimensionless. If we introduce scalar fields, the Yukawa couplings with fermions and scalar self couplings up to quartic ones are allowed.

Secondly the consistency should be proved between the degrees of freedom of counter terms and independent divergences of amplitudes. In gauge theories (with simple Lie group) all gauge coupling constants, for example, are g for three-vertices and g^2 for four-vertices. This uniqueness of couling constants is nothing but the gauge symmetry which gives the theory a very strong predictive power. The uniqueness of various (bare) couplings indicates that the counter terms are strongly constrained. In order to absorb all divergences in amplitudes into constrained counter terms corresponding relations should exist among divergent amplitudes. These relations are given by the WT identities derived in § 2. 4.

In the Weinberg-Salam (WS) theory, for example, the electron mass is dynamically related to the Higgs coupling constant with the electron. This is one of the above-mentioned constraints. Especially in the gauge theory with spontaneous symmetry breakdown, the renormalizability of the theory is nontrivial if we are involved in the perturbation calculation on the symmetry breaking vacuum.

Let us review briefly the proof of the renormalizability of gauge theories by the use of the induction. We define the operation * by

$$F * G \equiv \frac{1}{2} \left(\frac{\delta F}{\delta \phi_i} \cdot \frac{\delta G}{\delta K_i} + \frac{\delta G}{\delta \phi_i} \cdot \frac{\delta F}{\delta K_i} \right), \qquad (2.125)$$

where ϕ_i and K_i are relevant fields and corresponding BRS sources respectively. The WT identity (2.120) is expressed as follows by this operation,

$$\widetilde{\Gamma} * \widetilde{\Gamma} = 0. \qquad (2.126)$$

We renormalize ϕ_i and K_i properly so that the WT identity (2.126) holds in the expression with renormalized fields and BRS sources.

In the loop expansion, we write

$$\widetilde{\Gamma}_{R} \equiv \sum_{i=0}^{\infty} \Gamma_{R}^{(i)} . \qquad (2 \cdot 127)$$

We also expand the action \widetilde{S} with source terms,

$$\widetilde{S} \equiv \sum_{i=0}^{\infty} \widetilde{S}^{(i)}, \qquad (2.128)$$

where $\tilde{S}^{(i)}$ $(i \ge 1)$ are counter terms. We suppose that $\tilde{\Gamma}_{R}$ is a finite functional to the *n*-loop order with counter terms to the same order:
Chap. 2 Structure of Electroweak Theory

$$\widetilde{\Gamma}_{\mathbf{R}}^{(j)} \left[\sum_{i=0}^{n} \widetilde{S}^{(i)}\right] = \text{finite for } j \leq n.$$
(2.129)

The (n+1)-loop part of the identity $(2 \cdot 126)$ is written as

$$\widetilde{\Gamma}_{R}^{(0)}[\widetilde{S}^{(0)}] * \widetilde{\Gamma}_{R}^{(n+1)}[\sum_{i=0}^{n+1} \widetilde{S}^{(i)}] + [\text{finite functionals}] = 0.$$
 (2.130)

Taking out the divergent part of Eq. (2.130), we have

$$\widetilde{S}^{(0)} * \widetilde{\Gamma}^{(n+1)}_{\text{div}} \left[\sum_{i=0}^{n+1} \widetilde{S}^{(i)} \right] = 0, \qquad (2 \cdot 131)$$

where we have used the fact that the effective action to the tree approximation is the tree action,

$$\widetilde{\Gamma}_{R}^{(0)} = \widetilde{S}^{(0)} . \qquad (2 \cdot 132)$$

The (n+1)-loop part $\widetilde{\Gamma}_{div}^{(n+1)} [\sum_{i=0}^{n+1} \widetilde{S}^{(i)}]$ consists of two kinds of terms,

$$\widetilde{\Gamma}_{\rm div}^{(n+1)} \Big[\sum_{i=0}^{n+1} \widetilde{S}^{(i)} \Big] = \widetilde{\Gamma}_{\rm div}^{(n+1)} \Big[\sum_{i=0}^{n} \widetilde{S}^{(i)} \Big] + \widetilde{S}^{(n+1)} .$$
(2.133)

The first term on the right-hand side of Eq. $(2 \cdot 133)$ is amplitudes calculated with counter terms up to *n*-loop order and its inner divergences have been eliminated. The second term, $S^{(n+1)}$, is the (n+1)-loop counter terms to cancel out superficial divergences. By substituting Eq. $(2 \cdot 133)$ into Eq. $(2 \cdot 131)$, we obtain

$$\widetilde{S}^{(0)} * \widetilde{\Gamma}^{(n+1)}_{\text{div}} \left[\sum_{i=0}^{n} \widetilde{S}^{(i)} \right] = 0 . \qquad (2 \cdot 134)$$

We note here that the relation

$$\widetilde{S}^{(0)} * \widetilde{S}^{(n+1)} = 0$$
, (2.135)

holds since $\widetilde{S}^{(n+1)}$ is a local gauge invariant functional. Equation (2.134) is called the renormalization equation. [Lee 74].

The general solution of this functional equation is obtained as follows [Jogleker and Lee 76],

$$\widetilde{\Gamma}_{\rm div}^{(n+1)} \left[\sum_{i=0}^{n} \widetilde{S}^{(i)} \right] = a_{j} G_{j} + \widetilde{S}^{(0)} * F , \qquad (2 \cdot 136)$$

where G_j is a local gauge invariant functional of gauge and matter fields and F is a functional of all fields and BRS sources. We should recall here that it is sufficient for the renormalizability that superficially divergent parts are cancelled out. The superficially divergent part of $\tilde{\Gamma}$ is the functional with dimension up to four. The solution (2.136) with dimension up to four is shown

37

to have the form of the action with a suitable redefinition of coupling constants, fields and sources. Therefore we can take the (n+1)-loop counter terms as

$$S^{(n+1)} \equiv -\Gamma_{\rm div}^{(n+1)} \left[\sum_{i=0}^{n} S^{(i)} \right]$$
 (2.137)

with appropriate renormalization constants. Thus the divergent part of $\Gamma^{(n+1)}[\sum_{i=0}^{n+1} S^{(n+1)}]$ vanishes. Combining with the fact that in the tree approximation (the zero-loop order) the theory is finite, we have completed the proof of renormalizability to all orders.

We explain below the relation between a symmetric theory and a symmetry broken theory. We show that the theory with spontaneous symmetry breakdown (symmetry broken theory) can be actually renormalized with the same counter terms as those for the symmetric theory. In principle the above proof for the renormalizability holds regardless of the structure of the vacuum, symmetric or symmetry broken, since we deal with the effective action which simultaneously describes systems with all possible vacua. However, the renormalization condition (2.137) actually requires symmetric calculation of the perturbation theory. In the symmetric calculation of a symmetry broken theory, the propagator of the corresponding Higgs field ϕ ($\langle \phi \rangle \neq 0$), that is, the propagator expanded from the point $\langle \phi \rangle = 0$, has negative mass squared. Therefore, it is non-trivial whether a consistent symmetric and mass-independent renormalization procedure exists.

Symmetric and mass-independent renormalization schemes are obtained by several authors. [Weinberg 73, 't Hooft 73, Kugo 77, see, also, Collins and Macfarlane 74]. The most convenient one is the minimal subtraction scheme by 't Hooft. Consider a gauge theory with a Higgs potential Lagrangian,

$$\mathcal{L}_{\rm V} = -\frac{1}{2} m_0^2 \phi_0^2 - \frac{1}{4!} \lambda_0 \phi_0^4 \,. \tag{2.138}$$

In this theory we define renormalization constants as usual. For m_0^2 we define

$$m_0^2 = Z_m m^2 \,. \tag{2.139}$$

In the minimal subtraction scheme we subtract only pole terms at D=4 in the *D*-dimensional regularization. In this scheme all dimensionless renormalization constants including Z_m are determined as follows:

$$Z_{i} = \sum_{n=1}^{\infty} C_{n} \frac{1}{(D-4)^{n}}, \qquad (2 \cdot 140)$$

where C_n is a function of renormalized coupling constants (dimensionless). By using this scheme, we renormalize symmetric $(m^2 \ge 0)$ and symmetry broken $(m^2 < 0)$ theory simultaneously with common counter terms. The difficulty which may be caused by the negative mass squared does not occur. In a usual mass-dependent renormalization scheme a negative mass squared of the propagator brings about the difficulty that renormalization constants which should be real are actually estimated to be complex.

By using the above symmetric and mass-independent renormalization scheme, we obtain the effective action for the symmetry broken theory $(m^2 < 0)$ as follows. First of all, we calculate the effective action in the symmetric theory with a positive m^2 and make it finite by renormalization. Then we set m^2 to be a negative value. We find the ϕ_{\min} which gives the absolute minimum of the effective potential. Finally, we reexpand the effective action with $\tilde{\phi} = \phi$ $-\phi_{\min}$ and obtain the effective action of the symmetry broken theory expressed by the true excitation mode of fields. If it is necessary, we may impose an additional finite renormalization on the obtained effective action. Of course, no difficulty takes place in the procedure of this additional renormalization.

The renormalization conditions for amplitudes in the minimal subtraction scheme is not clear. We present explicitly a set of renormalization conditions for a symmetric and mass-independent renormalization scheme [Kugo 77]. The point is to renormalize the effective action so that it becomes a finite functional of both fields and mass m^2 with common counter terms. The renormalization condition appropriate for this purpose is, for example,

$$\Gamma^{(2)}(p^2=0, \lambda, m^2=0)=0,$$
 (2.141a)

$$\frac{\partial \Gamma^{(2)}}{\partial m^2}(p^2=0, \lambda, m^2)|_{m^2=\mu^2}=1, \qquad (2.141b)$$

$$\frac{\partial \Gamma^{(2)}}{\partial p^2}(p^2, \lambda, m^2 = \mu^2)|_{p^2 = 0} = -1, \qquad (2.142)$$

$$\Gamma^{(4)}(p^2 = 0, \lambda, m^2 = \mu^2) = -\lambda, \qquad (2 \cdot 143)$$

where $\Gamma^{(n)}$ is the *n*-point function of Higgs ϕ in the momentum representation,

$$\Gamma^{(n)} = \frac{1}{n!} \mathrm{FT} \frac{\delta^n \Gamma}{\delta \phi(x_1) \cdots \delta \phi(x_n)},$$

and μ^2 is a renormalization point. We have not set μ^2 equal to zero on account of the infrared divergences. Here we should notice that a usual mass renormalization condition is split into two conditions $(2 \cdot 141)$. This is necessary to renormalize the effective action comprehensively as a functional of fields, coupling constants and mass m^2 with mass independent counter terms. In order to satisfy the above renormalization conditions, we must prepare two terms for the mass counter terms; one is proportional to m^2 and the other is a constant independent of m^2 . We introduce renormalization constants and separate $\mathcal{L}_{\mathbf{V}}$ into the tree part and the counter-term parts as follows: K-I. Aoki, Z. Hioki, R. Kawabe, M. Konuma and T. Muta

$$\phi_0 = Z_{\phi}^{1/2} \phi, \qquad (2 \cdot 144a)$$

$$\lambda_0 = Z_{\lambda} Z_{\lambda}^{-2} \lambda \qquad (2 \cdot 144b)$$

$$\lambda_0 - \mathcal{L}_\lambda \mathcal{L}_\phi \quad \lambda, \qquad (2.1440)$$

$$m_{0}^{2} = Z_{m} Z_{\phi}^{-1} m^{2} + \delta m^{2} Z_{\phi}^{-1}, \qquad (2 \cdot 144c)$$

$$\mathcal{L}_{\mathbf{v}} = \mathcal{L}_{\mathbf{v}}^{t} + \mathcal{L}_{\mathbf{v}}^{c}, \qquad (2 \cdot 145)$$

$$\mathcal{L}_{v}^{t} = -\frac{1}{2} m^{2} \phi^{2} - \frac{\lambda}{4!} \phi^{4}, \qquad (2 \cdot 146a)$$

$$\mathcal{L}_{v}^{c} = -\frac{1}{2}m^{2}(Z_{m}-1)\phi^{2} - \frac{1}{2}\delta m^{2}\phi^{2} - \frac{\lambda}{4!}(Z_{\lambda}-1)\phi^{4}. \qquad (2 \cdot 146b)$$

By introducing mass renormalization constants Z_m and δm^2 redundantly as in Eq. (2.144c), we have necessary mass counter terms for the renormalization conditions (2.141). All the renormalization constants $(Z_{\phi}, Z_{\lambda}, Z_m, \delta m^2)$ are determined mass- $(m^2$ -) independently. The counter term δm^2 can be identically zero especially in the minimal subtraction scheme.

The relation between the symmetric $(\langle \phi \rangle = 0)$ and symmetry broken $(\langle \phi \rangle \neq 0)$ theories which is explained above is easily understandable by graphical manner. Consider a fermion loop contribution to the self-energy of a gauge boson in a symmetry broken theory. The fermion is assumed to be massless in a symmetric theory and to acquire mass from a vacuum expectation value of Higgs scalar ϕ . We expand the self-energy graph such that

$$\mathbf{m} = \mathbf{m} + \mathbf{m} +$$

where the fermion line represents massive (massless) propagator on the left (right)-hand side. The ultraviolet divergences occur only in the first and second terms on the right-hand side of Eq. (2.147). These two divergences correspond to those in the symmetric theory; one corresponds to the two-point function and the other corresponds to the four-point function of gauge bosons and Higgs scalars. Thus counter terms for the symmetric theory also cancel out divergences in the symmetry broken theory. As will be shown below, some kinds of universalities between amplitudes hold in the symmetric theory. In the symmetry broken theory, for example, in Eq. (2.147) only the first and second terms on the right-hand side satisfy these universalities and the rest terms break them. Though the breaking terms are finite, these universalities do not hold in the symmetry broken theory. This point will also be mentioned later.

In the following we give explicit expressions of the WT identities in the form of Green functions which guarantee the renormalizability in a symmetric theory with a simple group. We deal with four examples: a) two-point function of gauge bosons, b) three-point functions of gauge bosons and ghosts, c) three-point functions of gauge bosons and fermions and d) four-point functions of gauge bosons. In a) we show that gauge bosons remain massless with higher order corrections. In b) \sim d) we clarify constraints among various vertices which are indispensable for the proof of the renormalizability because all relevant couplings in the Lagrangian are described by unique coupling constant (g and g²).

a) We begin with the WT identity $(2 \cdot 120)$:

Chap. 2

$$\frac{\delta\widetilde{\Gamma}}{\delta W_{\nu}^{\alpha}} \cdot \frac{\delta\widetilde{\Gamma}}{\delta K_{\alpha}^{\nu}} + \frac{\delta\widetilde{\Gamma}}{\delta\psi_{i}} \cdot \frac{\delta\widetilde{\Gamma}}{\delta K_{\phi_{i}}} + \frac{\delta\widetilde{\Gamma}}{\delta\overline{\psi}_{i}} \cdot \frac{\delta\widetilde{\Gamma}}{\delta K_{\phi_{i}}} + \frac{\delta\widetilde{\Gamma}}{\delta C_{\phi_{i}}} \cdot \frac{\delta\widetilde{\Gamma}}{\delta K_{c^{\alpha}}} + \frac{\delta\widetilde{\Gamma}}{\delta K_{c^{\alpha}}} \cdot \frac{\delta\widetilde{\Gamma}}{\delta K_{c^{\alpha}}} = 0, \quad (2.148)$$

where Greek letters represent group indices of the adjoint representation. We make differentiation of Eq. (2.148):

$$\frac{\delta^2}{\delta W_{\mu}^{\ \beta} \delta c^{\tau}} \mathrm{Eq.} (2.148) \bigg|_{0},$$

where the symbol $|_0$ represents setting all field variables to zero. Taking account of the ghost number conservation, we have only one term,

$$\frac{\delta^2 \widetilde{\Gamma}}{\delta W_{\mu}^{\ \beta} \delta W_{\nu}^{\ \alpha}} \cdot \frac{\delta^2 \widetilde{\Gamma}}{\delta K_{\alpha}^{\ \nu} \delta c^r} \bigg|_{0} = 0 . \qquad (2 \cdot 149)$$

In the momentum representation we define invariant amplitudes as follows:

$$\widetilde{\Gamma}^{\mu\nu}_{\alpha\beta}(p) \equiv \int d^4x \, e^{ip(x-y)} \frac{\delta^2 \widetilde{\Gamma}}{\delta W_{\mu}^{\,\alpha}(x) \, \delta W_{\nu}^{\,\beta}(y)} \\ \equiv \delta^{\alpha\beta} \Big\{ A\left(p^2\right) \left(g^{\mu\nu} - \frac{p^{\mu} p^{\nu}}{p^2}\right) + B\left(p^2\right) \frac{p^{\mu} p^{\nu}}{p^2} \Big\}, \qquad (2.150)$$
$$G^{\mu}_{\alpha\beta}(p) \equiv \int d^4x \, e^{ip(x-y)} \frac{\delta^2 \widetilde{\Gamma}}{\delta K_{\mu}^{\,\alpha}(x) \, \delta c^{\beta}(y)}$$

$$\equiv p^{\mu} \delta^{\alpha\beta} G(p^2). \tag{2.151}$$

We write them graphically as (see Appendix G),

$$\widetilde{\Gamma}^{\mu\nu}_{\alpha\beta}(p) \equiv \underbrace{\overset{\mu}{}}_{\alpha} \underbrace{\overset{\nu}{}}_{\beta} , \qquad (2.152)$$

$$G^{\mu}_{\alpha\beta}(p) \equiv {}^{\mu}_{\alpha} \int_{\beta} \int_{\beta} (2 \cdot 153)$$

41

where the dotted line represents the current of the conserved ghost number and the black dot stands for the local operator product of BRS transform $(\delta^{BRS}W_{\mu}^{a}$ in Eq.(2.153)). We note here that a derivative of the effective action Γ with respect to a BRS source K equals that of W defined in (2.95):

$$\frac{\delta \widetilde{\Gamma}}{\delta K} = \frac{\delta \Gamma}{\delta K} = \frac{\delta W}{\delta K} \,. \tag{2.154}$$

With the above graphical notations Eq. $(2 \cdot 149)$ is expressed as

$$\overset{\nu}{\underset{\beta}{\longrightarrow}} \overset{\nu}{\underset{\alpha}{\longrightarrow}} \overset{\nu}{\underset{\alpha}{\longrightarrow}} \overset{\rho}{\underset{\alpha}{\longrightarrow}} \overset{\rho}{\underset{\alpha}{\longrightarrow}} = c \quad . \tag{2.155}$$

In terms of invariant functions we have from Eq. (2.155)

$$B(p^2)G(p^2) = 0. (2.156)$$

As will be shown in Eq. $(2 \cdot 169)$, the function $p^2G(p^2)$ is equal to the inverse propagator of the ghost and it cannot be identically zero. Hence we have the identity,

$$B(p^2) \equiv 0. \tag{2.157}$$

Because of this identity the propagator of the gauge boson without gauge fixing term contribution is

$$\widetilde{D}^{\mu\nu}_{\alpha\beta}(p^2) = \delta^{\alpha\beta} A^{-1}(p^2) \left(g^{\mu\nu} - \frac{p^{\mu} p^{\nu}}{p^2} \right), \qquad (2.158)$$

that is, transverse. The pole position of the propagator remains zero because of Eq. (2.157) and another identity,

$$A(0) = B(0). (2.159)$$

This identity (2.159) is nothing but the requirement that the one-particle irreducible graph (the inverse propagator $\widetilde{\Gamma}^{\mu\nu}$) does not have any pole at least in the perturbation theory. Thus the gauge boson is massless in any order of the perturbation theory and the second order divergence does not appear in the gauge boson self-energy. We can make the self-energy finite only by adjusting the wave function renormalization constant of the gauge boson field.

b) We proceed to the three-point vertices of gauge bosons and ghosts. The differentiation of the WT identity (2.148) gives the following relations:

$$\frac{\delta^{3}}{\delta W_{\mu}^{\alpha} \delta W_{\nu}^{\beta} \delta c^{\gamma}} \operatorname{Eq.} (2.148) \Big|_{\mathfrak{g}} :$$

$$\stackrel{P \downarrow}{\longrightarrow} \overset{\mu}{\longrightarrow} \overset{\rho}{\longrightarrow} \overset{\rho}{\longrightarrow} \overset{\mu}{\longrightarrow} \overset{\rho}{\longrightarrow} \overset{\mu}{\longrightarrow} \overset{\mu}{\longrightarrow} \overset{\rho}{\longrightarrow} \overset{\mu}{\longrightarrow} \overset{\mu}{\longrightarrow} \overset{\rho}{\longrightarrow} \overset{\mu}{\longrightarrow} \overset{\rho}{\longrightarrow} \overset{\mu}{\longrightarrow} \overset{\rho}{\longrightarrow} \overset{\mu}{\longrightarrow} \overset{\rho}{\longrightarrow} \overset{\mu}{\longrightarrow} \overset{\rho}{\longrightarrow} \overset{\mu}{\longrightarrow} \overset{\rho}{\longrightarrow} \overset{\rho}{\longrightarrow} \overset{\mu}{\longrightarrow} \overset{\rho}{\longrightarrow} \overset{\rho}{\longrightarrow} \overset{\rho}{\longrightarrow} \overset{\mu}{\longrightarrow} \overset{\rho}{\longrightarrow} \overset{\rho}{\overset{\rho}{\longrightarrow} \overset{\rho}{\longrightarrow} \overset{\rho}{\overset{\rho}{\longrightarrow} \overset{\rho}{\overset{\rho}{\overset{\rho}{\overset{\rho}{\longrightarrow} \overset{$$

The invariant amplitude of the tree three-vertex of gauge bosons corresponds to $g^{\mu\nu}(p+q)^{\rho}f^{\alpha\beta\tau}$ term. The relevant part of Eq.(2.160) is written as

$$i\Gamma(P, k) g^{\mu\nu} P^{\rho} f^{\alpha\beta\delta} k_{\rho} \delta^{\delta\gamma} G(k) + \delta^{\alpha\delta} A\left(\frac{P-k}{2}\right) g^{\mu\rho} \cdot g^{\rho\nu} H(P, k) f^{\delta\beta\gamma} + \delta^{\beta\delta} A\left(\frac{P+k}{2}\right) g^{\nu\rho} \cdot g^{\rho\mu} H(-P, k) f^{\delta\alpha\gamma} = 0, \qquad (2.161)$$

where

$$P = q + p, \quad k = q - p, \qquad (2 \cdot 162)$$

and we define the invariant functions Γ and H as follows:

$$\begin{array}{c} \stackrel{P}{\longrightarrow} \stackrel{\downarrow}{\longrightarrow} \stackrel{\downarrow}{\longrightarrow} \stackrel{\rho}{\longrightarrow} \stackrel{\sigma}{\longrightarrow} \stackrel{\sigma}{\longrightarrow} ig^{\mu\nu}P^{\rho}f^{\alpha\beta\tau}\Gamma\left(P,k\right) + [\text{other invariant amplitudes}], \quad (2.163) \\ \stackrel{Q}{\longrightarrow} \stackrel{\downarrow}{\longrightarrow} \stackrel{\rho}{\longrightarrow} \stackrel{\sigma}{\longrightarrow} \frac{ig^{\mu\nu}P^{\rho}f^{\alpha\beta\tau}\Gamma\left(P,k\right)}{ig^{\mu\nu}P^{\rho}f^{\alpha\beta\tau}\Gamma\left(P,k\right)} = 0$$

$$\underset{\delta}{\overset{p}{\xrightarrow{}}}_{\beta} \overset{\rho}{\xrightarrow{}}_{\beta} = g^{\rho\nu} f^{\delta\beta\tau} H(P, k) + [\text{other invariant amplitudes}]. \quad (2.164)$$

With the operation of $\delta/\delta k_{\rho}|_{k_{\rho}=0, P^{2}=0}$ to Eq. (2.160) we have

$$iG(0)\Gamma(0,0) = A'(0)H(0),$$
 (2.165)

where for simplicity we neglect the effect of the infrared divergences contained in the above functions. No form factor other than $g^{\mu\nu}P^{\rho}f^{\alpha\beta\tau}$ term in (2.163) contributes to Eq. (2.165).

Next we use the second type of the WT identities. We adopt a gauge fixing Lagrangian,

$$\mathcal{L}_{\rm GF} = \partial_{\mu} W_{\beta}^{\mu} B^{\beta} + \frac{\alpha}{2} B_{\beta} B^{\beta} , \qquad (2 \cdot 166)$$

which preserves the global symmetry. In this gauge the WT identity $(2 \cdot 119)$ reads

$$\partial^{\mu} \frac{\delta \widetilde{\Gamma}}{\delta K_{\alpha}{}^{\mu}} + \frac{\delta \widetilde{\Gamma}}{\delta \overline{c}^{\alpha}} = 0. \qquad (2 \cdot 167)$$

By differentiating Eq. $(2 \cdot 167)$ with respect to c^{β} , we have

$$-ip^{\mu} \stackrel{\mu}{\alpha} \stackrel{\mu}{\beta} + \stackrel{\alpha}{\alpha} \stackrel{\alpha}{\beta} = 0 , \qquad (2.168)$$

43

that is,

$$p^{2}G(p^{2}) = i\gamma(p^{2}), \qquad (2 \cdot 169)$$

where we define the ghost inverse propagator as

$$\int d^4x e^{ip(x-y)} \frac{\delta^2 \Gamma}{\delta \bar{c}^{\,\alpha}(x) \,\delta c^{\,\beta}(y)} \equiv \overset{\alpha}{\cdots} \overset{\beta}{\longleftrightarrow} = -\delta^{\,\alpha\beta} \gamma \,(p^2). \tag{2.170}$$

We obtain a useful relation from Eq. $(2 \cdot 169)$,

$$G(0) = i\gamma'(0). (2.171)$$

Further differentiation $\delta^2/\delta W_{\nu}^{\beta}\delta c^r$ of Eq. (2.167) gives

that is,

$$H(P, k) = -\Gamma_c(P, k), \qquad (2.173)$$

where Γ_{c} is defined as

$$\sum_{\alpha}^{p} \underbrace{\bigcap_{\alpha}}_{\gamma}^{k} = -ip^{\nu}f^{\alpha\beta\gamma}\Gamma_{c}(P,k) + [\text{other invariant amplitudes}].$$
(2.174)

Consequently we obtain the identity,

$$\frac{\Gamma(0,0)}{A'(0)} = \frac{-iH(0,0)}{G(0)} = \frac{\Gamma_{\mathfrak{c}}(0,0)}{\gamma'(0)}, \qquad (2.175)$$

where we have used the relations (2.165), (2.171) and (2.173).

The corresponding renormalized functions are the coefficient functions of the renormalized field variables of the effective action:

$$\Gamma_{\mathbf{R}} \equiv Z_s^{3/2} \Gamma , \qquad (2 \cdot 176a)$$

$$A_{\mathbf{R}} = Z_{\mathbf{s}} A , \qquad (2 \cdot 176 \mathrm{b})$$

$$\Gamma_{cR} \equiv Z_s^{1/2} Z_c \Gamma_c , \qquad (2.176c)$$

 $\gamma_{R} \equiv Z_{c} \gamma$, (2.176d)

where Z factors are defined as follows:

 $W^{\mu} \equiv Z_{s}^{1/2} W_{R}^{\mu},$ (2.177a)

$$c = Z_c^{1/2} c_{\rm R},$$
 (2.177b)

Chap. 2 Structure of Electroweak Theory

$$\bar{c} = Z_c^{1/2} \bar{c}_R$$
. (2.177c)

In terms of these renormalized amplitudes the identity $(2 \cdot 175)$ is rewritten as

$$\frac{\Gamma_{\mathfrak{R}}(0,0)}{A_{\mathfrak{R}}'(0)} = \frac{\Gamma_{\mathfrak{c}\mathfrak{R}}(0,0)}{\gamma_{\mathfrak{R}}'(0)} \left(= \frac{-iH(0,0)}{G(0)} Z_{\mathfrak{s}}^{1/2} \right). \tag{2.178}$$

This relation represents that if we can make $\Gamma_{\rm R}$ finite, which is possible, of course, by adjusting the charge (g) renormalization constant, we obtain the finite $\Gamma_{\rm cR}$ automatically. In other words, the counter term to make $\Gamma_{\rm R}$ finite also makes $\Gamma_{\rm cR}$ finite. This property is an ingredient of the renormalizability of gauge theories.

Here it should be noted that we have not neglected any finite term in deriving the relation $(2 \cdot 178)$. Taking the on-shell renormalization conditions (see § 3.2),

$$A'(0) = \gamma'(0) = -1, \qquad (2.179)$$

the equality of the on-shell charge (the charge universality) between the cubic self coupling of gauge bosons and the gauge-ghost coupling is proved:

$$\Gamma_{\rm R}(0,0) = \Gamma_{e\rm R}(0,0) \left(= \frac{iH(0,0)}{G(0)} Z_{\mathfrak{z}}^{1/2} \right). \tag{2.180}$$

This is also called the Slavnov-Taylor identity [Slavnov 72 and Taylor 71]. This relation is not a direct result of the renormalizability of the theory, since it shows an equality including all finite contributions. In this sense the renormalization equation $(2 \cdot 134)$ does not assure the charge universality, which is no more than closed relations among infinite parts.

c) Next we investigate gauge couplings of fermions. By a similar differentiation of the WT identity (2.148), we obtain

We rewrite it as follows:

$$\Gamma_{F}^{\mu}(P,k) \cdot k_{\mu}G(k^{2}) = K\left(\frac{P+k}{2}\right)H_{F}(P,k) + \overline{H}_{F}(P,k)K\left(\frac{P-k}{2}\right), \quad (2.182)$$

where we have used invariant functions defined by

$$\frac{1}{i} \bigoplus_{j=1}^{p} \equiv \delta^{ij} K(p), \text{ (inverse propagator)}$$
(2.183)

45

K-I. Aoki, Z. Hioki, R. Kawabe, M. Konuma and T. Muta

$$\stackrel{\mathsf{p}}{\underset{\mathsf{q}}{\overset{\mathsf{\psi}}}} \stackrel{\mathsf{j}}{\underset{\mathsf{i}}{\overset{\mathsf{k}}{\overset{\mathsf{a}}}}} \stackrel{\alpha}{\underset{\mathsf{\mu}}{\overset{\alpha}{=}}} T^{\alpha}_{ij} \Gamma_{F}^{\mu}(P,k),$$
 (2.184)

$$p j \stackrel{i}{\underset{\alpha}{\overset{k}{\longrightarrow}}} = T^{\alpha}_{ij} \overline{H}_F(P,k).$$
 (2.185b)

Here T_{ij}^{α} is generator matrices of the representation of the fermion. In the above definitions we have supposed that different multiplets of fermions do not mix with each other for simplicity of the argument. It should be noted that K and H_F are matrices with spinor indices and are not commutable with each other. We define renormalized quantities as follows:

$$\psi = Z_F^{1/2} \psi_{\mathbf{R}}, \quad \overline{\psi} = Z_F^{1/2} \overline{\psi}_{\mathbf{R}}, \qquad (2 \cdot 186)$$

$$K_{\mathbf{R}} = Z_{\mathbf{F}} K , \qquad (2 \cdot 187)$$

$$\Gamma_{FR}^{\mu} = Z_{3}^{1/2} Z_{F} \Gamma_{F}^{\mu} . \qquad (2 \cdot 188)$$

By analysing the structure of K and $\overset{(\longrightarrow)}{H_F}$ and using the on-shell renormalization conditions for $K_{\mathbb{R}}$ (see § 3.2.2), we obtain the final result,

$$\Gamma_{FR}^{\mu}(P^{2}=4m_{F}^{2},k=0)=-\gamma^{\mu}\overline{H}_{F}(P^{2}=4m_{F}^{2},k=0)Z_{3}^{1/2}/G(0). \quad (2\cdot189)$$

We differentiate furthermore the WT identity (2.148) in order to deduce information on \overline{H}_{F} :

$$\frac{\delta^{4}}{\delta \overline{\psi}_{i} \delta \psi_{j} \delta c^{\alpha} \delta W_{\mu}^{\beta}} \operatorname{Eq.} (2.148) \Big|_{0} :$$

$$\stackrel{P \bullet}{=} \int_{q \bullet}^{j} \int_{1}^{q} \int_{k}^{r_{\mu}} \int_{k}^{\mu} \int_{k}^{\beta} \int_{k}^{\mu} \int_{k}^{k} \int_{$$

where momenta k and r are set equal to zero and all amplitudes are sandwitched by on-shell spinors, which are represented by the symbol \dashv . Taking corresponding form factors, we obtain

$$T_{ij}^{\gamma}f^{\gamma\beta\alpha}\gamma^{\nu}\cdot g^{\nu\mu}H + T_{ik}^{\beta}\gamma^{\mu}\cdot T_{kj}^{\alpha}H_{F} + T_{ik}^{\alpha}\overline{H}_{F}\cdot T_{kj}^{\beta}\gamma^{\mu} = 0. \qquad (2.191)$$

Equation (2.191) is reduced to

$$H(0,0) = i\overline{H}_F(P^2 = 4m_F^2, k=0), \qquad (2.192)$$

$$[T^{\alpha}, T^{\beta}] = i f^{\alpha \beta \tau} T^{\tau} . \qquad (2.193)$$

From Eqs. (2.178), (2.189) and (2.192), we have

$$\Gamma_{FR}^{\mu}(P^{2}=4m_{F}^{2}, k=0) = \gamma^{\mu} \frac{iH(0,0)}{G(0)} Z_{s}^{1/2} = \gamma^{\mu} \Gamma_{R}(0,0) = \gamma^{\mu} \Gamma_{cR}(0,0).$$
(2.194)

This identity assures the renormalizability for gauge couplings of fermions and the on-shell charge universality among gauge couplings of fermions, of ghosts and of gauge bosons themselves.

d) Finally, we study the four-point function of gauge bosons. We differentiate the WT identity (2.148):

$$\frac{\delta^{4}}{\delta W_{\mu}^{\alpha} \delta W_{\nu}^{\beta} \delta W_{\rho}^{\gamma} \delta c^{\delta}} \operatorname{Eq.} (2.148) \Big|_{\mathfrak{g}} :$$

$$\stackrel{q}{\underset{r \uparrow \rho}{\mathfrak{g}}} \stackrel{\varphi}{\underset{\lambda}{\mathfrak{g}}} \stackrel{$$

where all particles are on their mass shell. We define the invariant amplitudes of the four point function:

$$\stackrel{\mathbf{p} \downarrow \alpha}{\leftarrow} \stackrel{\mathbf{p} \downarrow \alpha}{\stackrel{\nu}{\rightarrow}} \stackrel{\mu}{\longrightarrow} \stackrel{\delta}{\longrightarrow} = g^{\mu\rho} g^{\nu\lambda} f^{\alpha\beta\omega} f^{\gamma\delta\omega} \Gamma_4(p,q,r) + [\text{other invariant amplitudes}].$$

$$(2.196)$$

By extracting $f^{\alpha\beta\omega}f^{\gamma\delta\omega}g^{\mu\rho}(p+q)^{\nu}$ term in Eq.(2.195), we obtain

$$\Gamma_4(0,0,0)G(0) = i\Gamma(0,0)H(0,0). \qquad (2.197)$$

With the aid of Eq. (2.165) and the definition of Γ_{4R} ,

$$\Gamma_{4R} = Z_3^2 \Gamma_4,$$
 (2.198)

we have the identity,

$$\Gamma_{4\mathbf{R}}(0,0,0) = - \{\Gamma_{\mathbf{R}}(0,0)\}^{2} / A_{\mathbf{R}}'(0). \qquad (2 \cdot 199)$$

Thus $\Gamma_{4R}(0, 0, 0)$ is renormalized to be g_R^2 under the on-shell renormalization conditions: $\Gamma_R(0, 0) = g_R$ and $A_R'(0) = -1$ (Eq. (2.179)).

We have shown how the identities assuring the renormalizability are obtained from the WT identities. Those identities express the on-shell charge universality as a by-product in a gauge theory without spontaneous symmetry breakdown. Two properties of the theory, the renormalizability and the charge universality, are essentially independent. Actually in a symmetry broken theory the charge universality is broken though the renormalizability holds. The difference between these properties is in the fact that the renormalizability is related to the infinite part of amplitudes, while the charge universality is the property of full amplitudes with all finite terms (see also § 3.4).

Chapter 3

On-Shell Renormalization in Electroweak Theory. I

We start with a general discussion on renormalization procedure and, at the same time, explain problems associated with choice of renormalization schemes. We then restrict ourselves to the on-shell renormalization scheme in the Weinberg-Salam theory. A detailed discussion will be given on the derivation of on-shell conditions. Special attention will be paid to two-point functions which exhibit complicated structure. Finally we give a proof of the universality of the electric charge by the full use of our on-shell scheme.

§ 3.1 Renormalization schemes

We would like to present a general procedure in renormalization calculations. In this process several problems arise: There are some ambiguities in defining renormalized parameters in the theory and also in giving numerical values to these parameters from the knowledge of experimental data.

Suppose that we calculate physical quantities, F, in a renormalizable field theory which includes a bare parameter g_0 (coupling constant) among others. Physical quantities, F, expressed in terms of g_0 are divergent in general and the renormalization is necessary.^{*)} The definition of the renormalized coupling constant g, however, is not unique. In order to give an explicit expression for F in terms of g, we must fix the scheme. The numerical values of F are predicted by determining the value of g. To give the value of g we choose some reliable experimental data.

If the theory can be solved exactly, this apparent ambiguity in defining g does not affect physical predictions. We have, however, only perturbative method at hand and we have to truncate the series at some order. In this case the physical predictions, in general, may be affected by the ambiguities in g (e.g., "scheme dependence problem" in the perturbative QCD). Moreover there appears another ambiguity in determining the numerical value of g from the experimental data. For example, suppose that we have two independent physical quantities, $\sigma(g)$ and $\Gamma(g)$, and experimental data on σ and Γ are given. The value of g determined perturbatively by using the data on σ may be different from that for Γ by the amount which is interpreted as higher order effect.

Let us now specialize to the Weinberg-Salam theory in which several

^{*)} We here mention only the ultraviolet (UV) divergences. Infrared divergences will be discussed in practical calculation (Chapter 5).

independent parameters appear. Different choices of set of independent parameters give rise to the apparent difference in expressions of the Lagrangian. For example, several authors use the set of parameters,^{*)} g (SU(2) coupling), sin $\theta_{\rm W}$ and $M_{\rm W}$. It is also possible to choose the set e, $M_{\rm W}$ and $M_{\rm Z}$, which is our choice in this paper. (Accordingly, the Feynman rules are expressed by e, $M_{\rm W}$ and $M_{\rm Z}$ as shown in § 4.2.)

With a set of parameters suitably chosen, one can calculate a relevant S-matrix element as a function of these parameters. To eliminate ultraviolet divergences which may appear in the course of the calculation, one applies an appropriate renormalization scheme. As stated before, the choice of the renormalization scheme is rather arbitrary. The so-called MS ['t Hooft 73], $\overline{\text{MS}}$ [Bardeen et al. 78] and MOM [Celmaster and Gonsalves 79] schemes in quantum chromodynamics are some of the typical examples. If we know an exact solution of the theory, different choices of the schemes do not cause any difference in the resulting S-matrix element although its expression as a function of the renormalized parameters may differ. In truncated perturbation theories, however, different choices of the schemes can lead to different physical predictions.

In the Weinberg-Salam theory it is often a standard manner to use, as in QED, the fine structure constant $\alpha = e^2/4\pi$ defined in the Thomson limit as an expansion parameter. This means that we choose a specific scheme to define α , i.e., the on-shell scheme. As we have more parameters other than α , i.e., M_W and M_Z , we must specify the prescription to split bare parameters M_{W0} and M_{Z0} into renormalized parts and counter terms, $M_{W0}^2 = M_W^2 + \delta M_W^2$ and $M_{Z0}^2 = M_Z^2 + \delta M_Z^2$. We adopt the scheme which is straightforward generalization of the one commonly used in QED: We determine, for example, counter terms δM_W^2 and Z_W (W field renormalization constant) so that the transverse part of the renormalized W boson self-energy, $A_R^W(q^2)$, behaves as

$$A_{\mathbb{R}}^{W}(q^2)$$
 and $\frac{d}{dq^2}A_{\mathbb{R}}^{W}(q^2) \rightarrow 0$ as $q^2 \rightarrow M_{W}^2$. (3.1)

With this condition, the renormalized mass M_W is identical to the pole position of the W propagator, i.e., the physical mass. For this reason our scheme is called the on-shell scheme.

The S-matrix element is now obtained perturbatively as a function of renormalized parameters e, M_W and M_z in a specific renormalization scheme (the on-shell scheme in our case). To obtain the physical prediction on the Smatrix element, we must determine the values of these parameters with the help of given experimental data. The values of e, M_W and M_z can be de-

^{*)} The WS theory with one Higgs doublet includes five kinds of independent parameters. As two parameters other than those shown here, m_f (fermion mass) and m_{ϕ} (Higgs mass) are usually used.

termined if we have three pieces of independent experimental information. (The value of e is actually determined in the Thomson limit without recourse to the perturbative argument. Thus the value of e is exact and this scheme is a preferred scheme in the W-S theory as well as in QED.) It is this stage where an ambiguity in predicting radiative corrections to the S-matrix element creeps in as far as we use the perturbation theory. Of course this ambiguity does not exist if we know an exact solution. As a set of experimental data, α^{\exp} , Γ^{\exp} and R^{\exp} are often utilized, where Γ is the total width for the decay $\mu \rightarrow e \bar{\nu} \nu$ and R is a ratio of neutral current cross section to charged current one or a ratio of cross sections of neutrino and antineutrino processes (e.g., $R = \sigma (\bar{\nu} e \rightarrow \bar{\nu} e) / \sigma (\nu e \rightarrow \nu e)$).*' Assume that we perform the calculation to *n*-loop order and determine the values of M_W and M_Z using the data Γ^{\exp} and R^{\exp} . We denote those values by $M_W^{(n)}$ and $M_Z^{(n)}$, and Γ and R calculated to *n*-loop order by $\Gamma^{(n)}$ and $R^{(n)}$ respectively. We have

$$\Gamma^{(n)}(M_{W}^{(n)}, M_{Z}^{(n)}) = \Gamma^{\exp}, \qquad (3 \cdot 2)$$

$$R^{(n)}(M_{W}^{(n)}, M_{Z}^{(n)}) = R^{\exp}.$$
(3.3)

If the masses of W and Z bosons will be measured directly in the near future, we may also choose the set of parameters M_W^{\exp} and M_Z^{\exp} as input data instead of Γ^{\exp} and R^{\exp} (another obvious parameter α is neglected), and substitute M_W^{\exp} and M_Z^{\exp} directly into the renormalized parameters M_W and M_Z . In our previous papers [Aoki et al. 80 and 81], we have taken the latter set^{**)} and given the numerical results. In this article we will consider both sets and give corresponding results. Another set we analyze in this article is the one e^{\exp} , Γ^{\exp} and M_Z^{\exp} . This set is interesting in the sense that the high accuracy of the mass determination of Z boson is expected in coming experiments.

We have given a brief summary concerning the renormalization schemes. We show it schematically in Table 3.1.

Finally we wish to stress that our scheme in which parameters α , M_w and M_z are used and the subtraction of UV divergence is made on mass shell, is a natural extension of the renormalization in QED, and is very convenient and physical since the parameters M_w^2 and M_z^2 directly correspond to the pole positions of gauge boson propagators, that is, the physical masses. (For example, if one chooses another scheme with the Weinberg angle, θ_w , as one of the parameters, there appears an ambiguity in defining the angle θ_w since it is not the physical quantity. This point will be discussed in § 7.1.)

^{*)} We have put the suffix "exp" to α , Γ and R to stress that they express experimental values.

^{**)} Since W and Z bosons are not yet observed, we have assumed suitable values for M_W^{exp} and M_Z^{exp}.

Table 3.1. Flow chart for renormalization procedure.



a) The full list of counter terms is given in §4.3.

b) $\Gamma \equiv \Gamma(\mu \rightarrow e \nu \overline{\nu}), R \equiv \sigma(\overline{\nu}_{\mu} e \rightarrow \overline{\nu}_{\mu} e) / \sigma(\nu_{\mu} e \rightarrow \nu_{\mu} e).$

§ 3.2 **On-shell renormalization condition**

In this section we present a general discussion about the on-shell renormalization conditions without special recourse to the Weinberg-Salam theory. We start with the case of scalar fields. We recapitulate the reduction formula for the S-matrix in order to clarify the relation between the S-matrix element and the corresponding Green function. The on-shell renormalization conditions are defined and their explicit forms are given in the case with particle mixing. We proceed to the case of Dirac fields where non-commutativity of propagators should be taken into account. Finally the case of vector bosons is studied,

where the mixing with scalars including Nambu-Goldstone bosons makes a little complication.

3.2.1 Scalar fields

First we would like to recapitulate the framework of dealing with the S-matrix for scalar fields in the case without mixing. The S-matrix element is given by, in the Heisenberg picture,

$$S = \langle q_1, \cdots, q_m : \text{out} | k_1, \cdots, k_n : \text{in} \rangle, \qquad (3 \cdot 4)$$

where $|in\rangle$ and $|out\rangle$ are in and out asymptotic states respectively which are eigenstates of the total Hamiltonian and of other conserved charges including total momentum. These asymptotic states are constructed by repeated applications of creation operators $a_{out}^{\dagger}(k_i)$ to the vacuum $|0\rangle$. These creation operators form corresponding asymptotic fields:

$$\phi_{\text{in}}(x) = \int d^3k \left[a_{\text{in}}(k) f_k(x) + a_{\text{in}}^{\dagger}(k) f_k^*(x) \right], \qquad (3.5)$$

where $f_k(x)$ are normalized solutions of the Klein-Gordon equation. The asymptotic field satisfies a free field equation:

$$(\Box + m^2) \phi_{\rm as}(x) = 0, \qquad (3 \cdot 6)$$

where ϕ_{as} represents ϕ_{in} or ϕ_{out} . The Feynman propagator for ϕ_{as} reads

$$\begin{aligned} \mathcal{A}_{as}(p^{2}) &\equiv i \int \langle 0 | \mathrm{T}\phi_{as}(x) \phi_{as}(y) | 0 \rangle d^{4}(x-y) e^{-ip(x-y)} \\ &\equiv \mathrm{FT} i \langle 0 | \mathrm{T}\phi_{as}(x) \phi_{as}(y) | 0 \rangle \\ &= \frac{1}{m^{2} - p^{2} - i\varepsilon}, \end{aligned}$$
(3.7)

where FT denotes the Fourier transformation defined above.

This asymptotic field is related to the Heisenberg field as

$$\phi(x) \to \widetilde{Z}^{1/2} \phi_{\text{in}}(x) , \quad t \to \mp \infty .$$
(3.8)

It should be noted that the limit in Eq. $(3\cdot 8)$ is the so-called weak limit. This relation is the well-known LSZ asymptotic condition [Lehmann et al. 55]. The factor $\tilde{Z}^{1/2}$ appearing in Eq. $(3\cdot 8)$ is interpreted as an amplitude that $\phi(x)$ creates the one-particle state out of the vacuum:

$$\langle p: \operatorname{as} | \phi(x) | 0 \rangle = \tilde{Z}^{1/2} \langle p: \operatorname{as} | \phi_{\operatorname{as}}(x) | 0 \rangle$$

= $\tilde{Z}^{1/2} f_{k}^{*}(x) .$ (3.9)

Using this relation, the spectral representation of ϕ propagator [Umezawa and Kamefuchi 51, Källén 52, Lehmann 54],

K-I. Aoki, Z. Hioki, R. Kawabe, M. Konuma and T. Muta

$$\Delta(p^2) = \int_0^\infty d\sigma \rho(\sigma) \frac{1}{\sigma^2 - p^2 - i\varepsilon}, \qquad (3 \cdot 10)$$

is separated into one-particle and continuous-spectrum contributions:

$$\Delta(p^{2}) = \frac{\widetilde{Z}}{m^{2} - p^{2} - i\varepsilon} + \int_{\text{th}}^{\infty} d\sigma \rho(\sigma) \frac{1}{\sigma^{2} - p^{2} - i\varepsilon} .$$
(3.11)

Note that the propagator $\Delta(p^2)$ is defined analogously by Eq. (3.7) if one replaces ϕ_{as} by ϕ . Equation (3.11) shows that the factor \tilde{Z} defined in the asymptotic condition (3.8) is calculated as a pole residue of the full propagator.

With the aid of the asymptotic condition the S-matrix element defined by Eq. $(3 \cdot 4)$ is related to the Green function:

$$\langle q_{1}\cdots q_{m}: \operatorname{out}|k_{1}\cdots k_{n}: \operatorname{in} \rangle$$

$$= \frac{i^{m+n}}{(\widetilde{Z}^{1/2})^{m+n}} \int \prod_{i=1}^{n} d^{4}x_{i} \prod_{j=1}^{m} d^{4}y_{j} \cdot f_{k_{i}}(x_{i}) (\overrightarrow{\square}_{x_{i}} + m^{2})$$

$$\times \langle 0|T\phi(y_{1})\cdots\phi(y_{m})\phi(x_{1})\cdots\phi(x_{n})|0\rangle (\overleftarrow{\square}_{y_{j}} + m^{2}) f_{q_{j}}^{*}(y_{j}) . \qquad (3.12)$$

This is called the reduction formula [Lehmann et al. 55]. Schematically the S-matrix element (3.12) is represented by

$$G : (m^{2} - k_{1}^{2}) \frac{1}{\widetilde{Z}^{1/2}} \\ (m^{2} - q_{m}^{2}) \frac{1}{\widetilde{Z}^{1/2}} \\ k_{1^{2}, \cdots, q_{m}^{2} = m^{2}}, \qquad (3.13)$$

where G is the corresponding Green function in the momentum representation.

Taking account of the expression of the propagator $(3\cdot11)$, we notice that each external-line factor $(m^2 - k_i^2) \tilde{Z}^{-1/2}$ in $(3\cdot13)$ picks out the pole part of the propagator leaving the factor $\tilde{Z}^{1/2}$. Consequently the reduction formula states that the S-matrix element is just the corresponding Green function with the external legs amputated where the external momenta are put on the mass shell and the factors $\tilde{Z}^{1/2}$ are multiplied on each leg:

$$S = \left(G_{amp} \right) \widetilde{Z}^{1/2}, \qquad (3.14)$$

where the cross mark stands generally for all amputated legs.

Now we discuss the renormalization conditions. The renormalization conditions are necessary to calculate Green functions, although they have no effect on the representation of the S-matrix itself which is expressed only by Heisenberg fields and well-defined $\tilde{Z}^{1/2}$ factor. In order to follow the renormalization procedure, we introduce a wave-function renormalization constant $Z^{1/2}$ which is determined by a suitable renormalization condition,

$$\phi(x) \equiv Z^{1/2} \phi_{\text{ren}}(x)$$
 . (3.15)

It should be noted that Eq. (3.15) is an operator equation and the factor $Z^{1/2}$ has nothing to do with $\tilde{Z}^{1/2}$ introduced before. One can, however, choose the renormalization condition

$$Z^{1/2} = \widetilde{Z}^{1/2}$$
. (3.16)

With this choice of renormalization scheme the $\tilde{Z}^{1/2}$ factors in Eq. (3.14) can be regarded as usual counter terms for renormalized Green functions. Such a renormalization condition that ensures the equality of $Z^{1/2}$ and $\tilde{Z}^{1/2}$ is called the on-shell renormalization condition. This scheme is simple and convenient noticing that we need $\tilde{Z}^{1/2}$ to obtain S-matrix elements.

Next we discuss the problem of mass renormalization. Although the mass renormalization has no direct effect on the relation between the S-matrix and Green function, the mass shift in Green functions should be taken into account in perturbative calculation. The full propagator takes the form

$$\Delta(p^{2}) = \frac{1}{m^{2} - p^{2} - \Pi(p^{2})},$$

where *m* is the renormalized mass which is taken to be equal to the physical mass. The proper self-energy part is denoted by $\Pi(p^2)$ in which mass counter term δm^2 is included. Referring to the reduction formula (3.12), we adjust δm^2 so that the pole position of the propagator coincides with the mass squared m^2 loop by loop,

$$\Delta^{-1}(m^2) = 0. (3.17)$$

By the on-shell condition we mean that, in addition to Eq. (3.16), the condition (3.17) is implicitly imposed.

We define the renormalized propagator by

$$\Delta_{\rm ren} \equiv {\rm FT} \, i \langle 0 | {\rm T} \phi_{\rm ren}(x) \phi_{\rm ren}(y) | 0 \rangle = \frac{1}{Z} \Delta \, .$$

The pole part of Δ_{ren} is given by

$$\mathcal{A}_{\text{ren}}[\text{pole}] = \frac{1}{Z} \mathcal{A}[\text{pole}] = \frac{\widetilde{Z}}{Z} \frac{1}{m^2 - p^2} = \frac{\widetilde{Z}}{Z} \mathcal{A}_{\text{as}}, \qquad (3.18)$$

where the mass renormalization has already been taken care of. By requiring Eq. (3.16), we get an equality between $\Delta_{\rm ren}$ [pole] with the on-shell condition and $\Delta_{\rm as}$. Now we write down the on-shell renormalization conditions explicitly in terms of the inverse propagator,

$$\Delta_{\rm ren}^{-1}(m^2) = 0$$
, $\Delta_{\rm ren}^{-1'}(m^2) = -1$, (3.19)

or in terms of the proper self-energy part,

K-I. Aoki, Z. Hioki, R. Kawabe, M. Konuma and T. Muta

$$\Pi_{\rm ren}(m^2) = 0$$
, $\Pi'_{\rm ren}(m^2) = 0$. (3.20)

We proceed to the case with particle mixing. In this case all particles other than the lightest one are unstable and can decay into the lightest particle with emission of photons. Rigorously speaking, such unstable particles have no corresponding asymptotic field, and their propagators have poles at the position $m_j^2 - i\Gamma_j$, where Γ_j is the total width. In the following, however, we adopt the approximation in which we neglect the finite width of particles and consider only the real part of the propagators.

Let us take the sector with N particles. Their asymptotic fields (in the sense of the above approximation) are related to Heisenberg fields with the matrix $\tilde{Z}^{1/2}$ factor:

$$\phi_i(x) \longrightarrow \widetilde{Z}_{ij}^{1/2} \phi_j^{as}(x) , \qquad (3 \cdot 21a)$$

$$\widetilde{Z}_{ij}^{-1/2} \phi_j(x) \longrightarrow \phi_i^{as}(x) . \qquad (3.21b)$$

The propagator of this sector,

$$\Delta_{ij} \equiv i \operatorname{FT} \langle 0 | \operatorname{T} \phi_i(x) \phi_j(y) | 0 \rangle, \qquad (3.22)$$

has N poles, positions of which are m_i^2 $(i=1\cdots N)$, the mass of ϕ_i^{as} . The pole part of the propagator is estimated by using Eq. (3.21):

$$\begin{aligned} \mathcal{\Delta}_{ij}[\text{pole}] &= i \; \mathrm{FT} \cdot \tilde{Z}_{im}^{1/2} \tilde{Z}_{jn}^{1/2} \langle 0 | \mathrm{T}\phi_m^{\,\mathrm{as}}(x) \, \phi_n^{\,\mathrm{as}}(y) \, | 0 \rangle \\ &= \tilde{Z}_{im}^{1/2} \tilde{Z}_{jn}^{1/2} \frac{\delta_{mn}}{m_n^2 - p^2} \,, \end{aligned}$$
(3.23)

which shows the relation between \tilde{Z}_{ij} and the pole structure of the propagator Δ_{ij} .

The S-matrix element with external line i, S_i , is related to the Green function G in such a way as

$$S_{i} = \bigoplus_{\text{G}_{amp}} \underbrace{j}_{(m_{i}^{2} - p^{2})} \widetilde{Z}_{ij}^{-1/2}, \qquad (3.24)$$

where by suffix amp we mean that the Green function G is amputated. Substituting Eq.(3.23) for the propagator into Eq.(3.24), we obtain

$$S_{i} = \bigoplus_{k \in \mathbb{Z}_{km}} \tilde{Z}_{km}^{1/2} \tilde{Z}_{jn}^{1/2} \frac{\delta_{mn}}{m_{n}^{2} - p^{2}} (m_{i}^{2} - p^{2}) \tilde{Z}_{ij}^{-1/2}$$

$$= \bigoplus_{k \in \mathbb{Z}_{ki}} \tilde{Z}_{ki}^{1/2}. \qquad (3.25)$$

The wave function renormalization constants are introduced in the matrix form:

$$\phi_i \equiv Z_{ij} \phi_j^{\text{ren}} \,. \tag{3.26}$$

The renormalized propagator is defined by

$$\varDelta_{ij}^{\text{ren}} \equiv i \operatorname{FT} \langle 0 | \operatorname{T} \phi_i^{\text{ren}}(x) \phi_j^{\text{ren}}(y) | 0 \rangle$$

and its pole part reads

$$\begin{aligned} \mathcal{A}_{ij}^{\text{ren}}[\text{pole}] &= Z_{im}^{1/2} Z_{jn}^{1/2} \mathcal{A}_{mn}[\text{pole}] \\ &= Z_{im}^{1/2} Z_{jn}^{1/2} \widetilde{Z}_{mk}^{1/2} \widetilde{Z}_{nl}^{1/2} \mathcal{A}_{kl}^{\text{as}} , \qquad (3 \cdot 27) \end{aligned}$$

where mass renormalization has already been taken into account. Renormalization conditions in which $Z_{ij}^{1/2} = \tilde{Z}_{ij}^{1/2}$ are given by

$$\mathcal{\Delta}_{ij}^{\text{ren}}[\text{pole}] = \frac{\delta_{ij}}{m_i^2 - p^2} \,. \tag{3.28}$$

These are the on-shell renormalization conditions with mixing.

We transform these conditions to those for the inverse propagator. In the region of p^2 very close to m_n^2 the renormalized propagator has a pole in (n, n) component and other components are regular (O(1)):

$$\Delta_{ij}^{\mathrm{ren}}\Big|_{p^{2} \to m_{n^{2}}} = n \begin{pmatrix} n \\ \vdots \\ \cdots \\ 1/\varepsilon \cdots \end{pmatrix}, \qquad (3 \cdot 29)$$

$$\varepsilon = m_n^2 - p^2 \,. \tag{3.30}$$

The inverse propagator is obtained by inverting the matrix of Eq. (3.29):

$$\Delta_{ij}^{\text{ren}-1} = (\det \Delta^{\text{ren}})^{-1} A_{ij}, \qquad (3.31)$$

where det Δ^{ren} is the determinant of the matrix Δ_{ij}^{ren} and A_{ij} is the (i, j) cofactor of Δ_{ij}^{ren} . The determinant of Δ^{ren} is given by

det
$$\Delta^{\text{ren}} = \frac{1}{\varepsilon} A_{nn} + O(1),$$
 (3.32)

and therefore,

$$(\det \Delta^{\operatorname{ren}})^{-1} = A_{nn}^{-1} \cdot \varepsilon + O(\varepsilon^2). \qquad (3 \cdot 33)$$

On the other hand, the cofactors are of O(1) for i=n or j=n while they are of $O(1/\varepsilon)$ for $i\neq n$ and $j\neq n$.

Hence we obtain

K-I. Aoki, Z. Hioki, R. Kawabe, M. Konuma and T. Muta

In other words,

$$\mathcal{\Delta}_{ij}^{\text{ren-1}}\Big|_{p^2 \to m_n^2} = \begin{cases} \varepsilon & \text{for } i = j = n ,\\ O(\varepsilon) & \text{for } i = n, \ j \neq n \text{ and } i \neq n, \ j = n , \ (3 \cdot 35) \\ O(1) & \text{otherwise} . \end{cases}$$

It is now obvious from Eq. (3.35) that our on-shell renormalization conditions for $\mathcal{I}_{ij}^{\text{ren}-1}$ read for all n $(1 \cdots N)$:

i) i=j=n,

$$\Delta_{nn}^{ren-1}(m_n^2) = 0, \qquad (3 \cdot 36a)$$

$$\frac{\partial}{\partial p^2} \mathcal{A}_{nn}^{\text{ren}-1}(m_n^2) = -1, \qquad (3 \cdot 36b)$$

ii) $i=n, j\neq n$,

$$\Delta_{nj}^{\text{ren-1}}(m_n^2) = 0, \qquad (3.37)$$

iii) $i \neq n, j = n$,

$$\Delta_{in}^{\text{ren-1}}(m_n^2) = 0. \qquad (3.38)$$

Note that we need not impose the condition on $\mathcal{I}_{ij}^{\text{ren}-1}$ for $i \neq n$ and $j \neq n$. These conditions ensure the equality $Z_{ij}^{1/2} = \tilde{Z}_{ij}^{1/2}$ and the external $\tilde{Z}_{ki}^{1/2}$ factor in Eq. (3.25) is interpreted as the counter term generated by the replacement of Eq. (3.26) in the Lagrangian.

In the following we argue that the number of renormalization conditions $(3\cdot36) \sim (3\cdot38)$ is equal to that of independent renormalization constants by which the renormalized propagator is determined. Not all renormalization conditions are independent of each other. In fact only the conditions on the (i, j) components $(i \geq j)$ are independent since $\mathcal{I}_{ij}^{\text{ren}}$ is symmetric by exchange of *i* and *j*. First of all, the condition $(3\cdot36a)$ is satisfied by adjusting the mass counter terms δm_i^2 . There remain N conditions $(3\cdot36b)$ and two conditions for all i > j: $(3\cdot37)$ and $(3\cdot38)$. The number of the conditions in total amounts to N^2 $(=N+2_NC_2)$. These N^2 conditions can be satisfied by the N^2 degrees of freedom of $Z_{ij}^{1/2}$. It should be noticed that the matrix $Z_{ij}^{1/2}$ is not orthogonal but a general real matrix, that is, all N^2 components of it are independent of each other.

Finally, we remark that the above property of $Z_{ij}^{1/2}$ means that the asymptotic fields or states are not related to Heisenberg fields through a general rotation in O(N) space. This fact is one of the most important points to understand and construct the on-shell renormalization scheme. This point is related to the ambiguity in defining $\sin^2\theta_W$ including higher order terms in the Weinberg-Salam theory. There have been many confusions about this point in discussing the renormalization scheme of the Weinberg-Salam theory, which will be discussed later.

3.2.2 Dirac fields

For particles with spin 1/2 the derivation of the on-shell renormalization conditions is essentially the same as that for scalar particles except the fact that the propagators have spin indices and do not commute with each other.

Consider the Weyl fields corresponding to irreducible representations of the Lorentz group,

$$\psi_{\mathrm{L},i}(x), \qquad \psi_{\mathrm{R},j}(x),$$

where we denote the chirality of the Weyl fields by the sufficies L and R. In general the mixing among all the Weyl fields should be taken into account in the renormalization procedure. The right-handed Weyl fields $\psi_{R,j}(x)$ may be changed to the fields with the left-handed chirality such that

$$\psi_{\mathbf{L},j}(x) = C \overline{\psi}_{\mathbf{R},j}(x), \qquad (3 \cdot 39)$$

where C is the charge conjugation matrix. Then all the Weyl fields possess the left-handed chirality and mix with each other.

In practical cases there are conservation laws which partly prohibit the mixings. The electric charge, for example, is assumed to be conserved. This assumption guarantees that only particles with the same charge can mix with each other. In the charged particle sector we adopt the Dirac field representation.*' In this case we define the relevant fields by

$$\psi_i = \psi_{\mathrm{L},i} + \psi_{\mathrm{R},i}, \qquad (3 \cdot 40)$$

$$\psi_{\mathrm{L},i} \equiv \frac{1}{2} (1 - \gamma_5) \psi_i,$$
 (3.41a)

$$\psi_{\mathbf{R},i} \equiv \frac{1}{2} \left(1 + \gamma_5 \right) \psi_i \,. \tag{3.41b}$$

We assume the following asymptotic conditions,

^{*)} It is possible to represent a charged particle by the two-component Weyl field. Such a particle is necessarily massless and is unrealistic. An extension of the formulation to include such a massless charged particle is straightforward (see the last part of this subsection).

$$\psi_{\mathrm{L},i}(x) \rightarrow \widetilde{Z}_{\mathrm{L},ij}^{1/2} \psi_{\mathrm{L},j}^{\mathrm{as}}(x) , \qquad (3 \cdot 42\mathrm{a})$$

$$\psi_{\mathbf{R},i}(x) \to \widetilde{Z}_{\mathbf{R},ij}^{1/2} \psi_{\mathbf{R},j}^{\mathrm{as}}(x) \ . \tag{3.42b}$$

We note here that the left- and right-chirality fields do not mix with each other because-of the fermion-number conservation which is directly related to the charge conservation.

For neutral particles, e.g., neutrinos, however, the charge conservation has nothing to do with the fermion number conservation. In fact neutrinos can have the Majorana type masses as well as the Dirac type masses. In the case where the fermion number is not a good quantum numbers, we must deal with a full mixing of all Weyl fields. We will comment on the on-shell renormalization conditions in this case at the end of this subsection.

We go back to the case of the Dirac fields described by Eqs. (3.40) and (3.42). The renormalized fields are defined by

$$\psi_{\mathrm{L},i}(x) \equiv Z_{\mathrm{L},ij}^{1/2} \psi_{\mathrm{L},j}^{\mathrm{ren}}(x), \qquad (3 \cdot 43a)$$

$$\psi_{\mathbf{R},i}(x) \equiv Z_{\mathbf{R},ij}^{1/2} \psi_{\mathbf{R},j}^{\mathrm{ren}}(x) \,. \tag{3.43b}$$

If the parity transformation which exchanges L and R chirality is a symmetry of the theory, we have the identity,

$$Z_{\mathbf{L},ij}^{1/2} = Z_{\mathbf{R},ij}^{1/2} \,. \tag{3.44}$$

Because of this relation we need not introduce independent renormalization constants for $\psi_{\rm L}$ and $\psi_{\rm R}$ in QED. In the Weinberg-Salam theory the relation (3.44) does not hold, and $Z_{{\rm L},ij}^{1/2}$ and $Z_{{\rm R},ij}^{1/2}$ are independent constants.

The on-shell renormalization conditions for the propagator

$$S_{ij} \equiv \text{FT} \ i \langle 0 | \mathsf{T} \psi_i(x) \bar{\psi}_j(y) | 0 \rangle, \qquad (3 \cdot 45)$$

are

$$S_{ij}^{\text{ren}}[\text{pole}] = S_{ij}^{\text{as}}, \qquad (3 \cdot 46)$$

where the propagator for asymptotic fields is given by

$$S_{ij}^{as}(p) = \frac{\delta_{ij}}{m_i - p} . \qquad (3.47)$$

The renormalized propagator has a pole only in the (n, n) channel at $p^2 = m_n^2$,

$$S_{ij}^{\text{ren}}(p) \xrightarrow{p^2 \to m_n^2} n \left(\begin{array}{c} & \ddots \\ & \vdots \\ & \ddots \\ & \frac{1}{m_n - p} \end{array} \right)$$
(3.48)

The inverse propagator

$$K_{ij}^{\text{ren}}(p) \equiv \{S^{\text{ren}}(p)\}_{ij}^{-1}, \qquad (3 \cdot 49)$$

behaves as

$$K_{ij}^{\text{ren}} \xrightarrow{p^{i} \to m_{n}^{2}} n \begin{pmatrix} & & & & \\ & & & & \\ & & & & \\ & & & \\ & & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & &$$

In the component form Eq. (3.50) is expressed as

$$(m_n-p)$$
 for $i=n, j=n$, $(3\cdot51a)$

$$M^{in}(m_n-p) \quad \text{for} \quad i\neq n, \quad j=n, \quad (3.51b)$$

$$K_{ij}^{\text{ren}} \xrightarrow{p^2 \to m_n^2} \begin{cases} 1 & (m_n - p) & \text{for } i = n, \ j \neq n, \end{cases} (0 \text{ or } j) \\ (m_n - p) M^{nj} \quad \text{for } i = n, \ j \neq n, \end{cases} (3 \cdot 51c)$$

(arbitrary for
$$i \neq n$$
, $j \neq n$. (3.51d)

Note that M^{ij} is generally uncommutable with p because of the existence of r^5 in it. The ordering in Eqs. (3.50) and (3.51) guarantees that the inverse of K reproduces Eq. (3.48) properly.

In order to write down the on-shell renormalization conditions (3.51) explicitly, it is convenient to introduce on-shell spinors u(m) and $\overline{u}(m)$ which are the solutions of the Dirac equations with mass m:

$$(p-m)u(m) = 0$$
, $(3.52a)$

$$\overline{u}(m)(p-m) = 0, \qquad (3.52b)$$

where momentum p in u(m) and $\overline{u}(m)$ is suppressed. Using these u(m) and $\overline{u}(m)$, we can represent the conditions (3.51) in the following form:

$$K_{ij}^{\mathrm{ren}}u(m_j) = 0, \qquad (3.53a)$$

$$\overline{u}(m_i)K_{ij}^{\text{ren}}=0, \qquad (3.53b)$$

$$\left\{\frac{1}{m_i - p} K_{ii}^{\text{ren}}\right\} u\left(m_i\right) = u\left(m_i\right) , \qquad (3.54a)$$

$$\overline{u}(m_i)\left\{K_{ii}^{\text{ren}}\frac{1}{\not p - m_i}\right\} = \overline{u}(m_i), \qquad (3.54b)$$

where the momentum squared p^2 is set on the mass shell $p^2 = m_f^2$ in Eq. (3.53a) and $p^2 = m_i^2$ in Eqs. (3.53b) and (3.54). The above expressions (3.54) correspond to the fact that external-line corrections vanish automatically in the on-shell renormalization scheme:

$$G_{amp} \underbrace{j}_{j} \underbrace{\upsilon}_{i} u(m_{i}) = 0,
 \qquad (3.55)$$

where Σ is the self-energy part.

We split the inverse propagator K(p) into four invariant functions:

$$K_{ij}^{\text{ren}}(p) = K_{1}^{ij}(p^2) \cdot \mathbf{l} + K_{5}^{ij}(p^2) \cdot \gamma^5 + K_{\gamma}^{ij}(p^2) \cdot p + K_{5\gamma}^{ij}(p^2) \cdot p\gamma^5, \quad (3.56)$$

where terms containing $\sigma^{\mu\nu}$ do not appear since only one momentum variable p_{μ} is available to contract $\sigma^{\mu\nu}$. Using Eq. (3.56), the renormalization conditions (3.53, 3.54) are

$$[i=j] K_1^{ii}(m_i^2) + m_i K_r^{ii}(m_i^2) = 0, (3.57a)$$

$$2m_{i}K_{1}^{ii\prime}(m_{i}^{2}) + K_{r}^{ii}(m_{i}^{2}) + 2m_{i}^{2}K_{r}^{ii\prime}(m_{i}^{2}) = 1, \qquad (3.57b)$$

$$K_5^{ii}(m_i^2) = 0$$
, (3.57c)

$$K_{5r}^{ii}(m_i^2) = 0$$
, (3.57d)

$$[i \neq j] K_1^{ij}(m_j^2) + m_j K_r^{ij}(m_j^2) = 0, (3.58a)$$

$$K_1^{ij}(m_i^2) + m_i K_r^{ij}(m_i^2) = 0, \qquad (3.58b)$$

$$K_{5}^{ij}(m_{j}^{2}) - m_{j}K_{57}^{ij}(m_{j}^{2}) = 0, \qquad (3.58c)$$

$$K_{5}^{ij}(m_{i}^{2}) + m_{i}K_{5r}^{ij}(m_{i}^{2}) = 0.$$
(3.58d)

The condition (3.57b) is concisely written as

$$\frac{\partial}{\partial p} \left\{ K_1(p^2) + p K_r(p^2) \right\}_{p=m} = 1.$$
(3.59)

We then discuss the consistency between the numbers of renormalization conditions and renormalization constants. Because of the hermiticity of the effective action Γ which is formally guaranteed by the hermiticity of the Lagrangian,

$$\Gamma^{\dagger} = \Gamma , \qquad (3 \cdot 60)$$

the inverse propagator has the following symmetry:

$$K_{ij}^{\text{ren}}(p) = \gamma^{0\dagger} K_{ij}^{\text{ren}\dagger}(p) \gamma^{0} . \qquad (3.61)$$

In deriving the above symmetry we have used the relation

$$K_{ij}^{\text{ren}}(p) = \int d^4(x-y) \, e^{ip(x-y)} \frac{\delta^2 \Gamma}{\delta \psi_i^{\text{ren}}(x) \, \delta \overline{\psi}_j^{\text{ren}}(y)} \,. \tag{3.62}$$

It is easy to check that the above hermiticity condition is consistent with the renormalization conditions. The renormalization condition (3.53a) for i>j,

$$K_{ij}^{\mathrm{ren}}u(m_j) = 0, \qquad (3.63)$$

implies

$$\overline{u}(m_j) K_{ji}^{\text{ren}} \gamma^0 = 0, \qquad (3 \cdot 64)$$

which is nothing but the condition (3.53b). Thus it is sufficient to impose the renormalization conditions only on K_{ij} for $i \ge j$. Those counter terms determined in the sector $i \ge j$ are sufficient to satisfy the renormalization conditions for the sector i < j. In the sector i = j we obtain, from the hermiticity condition (3.61),

$$K_1^{ii*} = K_1^{ii}, \qquad (3.65a)$$

$$K_{5}^{ii*} = -K_{5}^{ii}, \qquad (3.65b)$$

$$K_r^{ii*} = K_r^{ii} , \qquad (3.65c)$$

$$K_{br}^{ii*} = K_{br}^{ii} . \tag{3.65d}$$

Equations (3.65) indicate that all diagonal invariant functions are purely real or imaginary. According to these hermiticity conditions, the renormalization conditions (3.53a) and (3.53b) and also [(3.54a) and (3.54b) for the sector i=j] are not independent of each other.

Now we are ready for counting the number of independent conditions. In the sector i=j we have four conditions $(3\cdot57)$ since each equation is purely real or imaginary as seen in Eqs. $(3\cdot65)$. The conditions $(3\cdot58)$ for i>jare all complex equations, which correspond to eight real conditions. Hence the number of necessary conditions is 4N for i=j and 4N(N-1) for i>j. Thus it amounts to $4N^2$. Counter terms for the self-energy part are generated by the $Z_{\rm L}$ and $Z_{\rm R}$ factors and mass parameters as follows:

$$p(Z_{\rm L}^{1/2\dagger}Z_{\rm L}^{1/2})_{ij}\frac{1-\gamma_{\rm 5}}{2} + p(Z_{\rm R}^{1/2\dagger}Z_{\rm R}^{1/2})_{ij}\frac{1+\gamma_{\rm 5}}{2} - (Z_{\rm L}^{1/2\dagger}MZ_{\rm R}^{1/2})_{ij}\frac{1+\gamma_{\rm 5}}{2} - (Z_{\rm R}^{1/2\dagger}M^{\dagger}Z_{\rm L}^{1/2})_{ij}\frac{1-\gamma_{\rm 5}}{2}, \qquad (3\cdot66)$$

where M is the bare mass matrix. The general mass matrix M can be diagonalized with real positive eigenvalues by an appropriate biunitary transformation:

$$\{VMU\}_{ij} = \delta_{ij}m_i. \tag{3.67}$$

These unitary matrices U and V are absorbed into the definition of the $Z_{\rm L}^{1/2}$ and $Z_{\rm R}^{1/2}$ matrices, and so independent mass parameters are only the m_i 's $(i=1\sim N)$ in Eq. (3.67). Furthermore the common phase rotation of $Z_{\rm L}^{1/2}$ and $Z_{\rm R}^{1/2}$,

$$Z_{\mathrm{L},ij}^{1/2} \rightarrow e^{i\theta_i} \delta_{ik} Z_{\mathrm{L},kj}^{1/2} , \qquad (3.68a)$$

$$Z_{\mathbf{R},ij}^{1/2} \rightarrow e^{i\theta_i} \delta_{ik} Z_{\mathbf{R},kj}^{1/2}, \qquad (3.68b)$$

does not modify the counter terms (3.66). Consequently the number of degrees of freedom of counter terms is

$$4N^{2}[Z_{\rm L}^{1/2}, Z_{\rm R}^{1/2}] + N[m_{i}] - N[\theta_{i}] = 4N^{2}, \qquad (3.69)$$

which coincides with that of the renormalization conditions. We should notice here that the matrices $Z_{\rm L}^{1/2}$ and $Z_{\rm R}^{1/2}$ are not unitary. This implies that the on-shell states cannot be obtained by the unitary transformation from those states corresponding to the Heisenberg fields. The situation is similar to the case of scalar fields. The above counting is not sufficient for the proof of renormalizability, but one can easily show that all equations of renormalization conditions have a unique solution for counter terms.

Here we discuss the case where the fermion number is not necessarily conserved, for example, the case of neutrinos. In general all fermions are the Majorana particles in this case. We represent these fermions by the four-component Majorana fields which satisfies the self-charge-conjugate condition,

$$C\overline{\psi}_{\mathbf{M}} = \psi_{\mathbf{M}} \,. \tag{3.70}$$

This representation is more convenient than a two-component representation because a trace including both Dirac fields and Majorana fields has to be often calculated. The propagator of (asymptotic) Majorana fields is

$$S_{\phi\phi}^{as} \equiv FT \ i \langle 0 | T \phi_{M}^{as}(x) \overline{\phi}_{M}^{as}(y) | 0 \rangle$$
$$= \frac{1}{m - p}, \qquad (3.71a)$$

$$S_{\bar{\psi}\bar{\psi}}^{\mathrm{as}} \equiv \mathrm{FT} \; i\langle 0 | \mathrm{T}\bar{\psi}_{\mathrm{M}}^{\mathrm{as}}(x) \bar{\psi}_{\mathrm{M}}^{\mathrm{as}}(y) | 0 \rangle$$
$$= C^{-1} S_{\bar{\psi}\bar{x}}^{\mathrm{as}}, \qquad (3.71c)$$

where we have used the Majorana condition (3.70). The on-shell renormalization condition is

$$S_{\psi\bar{\psi}}^{\mathrm{ren}} = S_{\psi\bar{\psi}}^{\mathrm{as}}$$

We only deal with the Dirac-like part of the propagator of Majorana fields (3.71a). The Majorana condition (3.70) for renormalized fields guarantees the correct form for $S_{\psi\psi}^{\text{ren}}$ and $S_{\overline{\psi}\overline{\psi}}^{\text{ren}}$. Hence we obtain the same formulas of on-shell renormalization conditions as those for Dirac fields. The invariant amplitudes of the inverse propagator are, however, constrained by the Majorana condition (3.70) and the number of the independent renormalization conditions is reduced to $2N^2 + N$ for the sector of N Majorana particles. On the other hand, the Majorana condition (3.70) for the renormalized fields imposes the relation

$$Z_{\rm L} = Z_{\rm R}^*$$
,

and the degrees of freedom of the renormalization constants are also $2N^2 + N$.

Finally, we comment on the model which we will adopt for the estimation of the radiative corrections in Chapter 5. In the model, neutrinos are described by pure left-handed fields and are to be massless assuming the fermion number conservation. There is no mixing among different generations. This is a special case and the renormalization procedure becomes very simple. The inverse propagator of neutrinos is constrained so that

$$K_{\nu} = K_{\nu}(p^2) \not p(1-\gamma_5).$$

The renormalization constant $Z_i^{1/2}$ is introduced as

$$\nu_{\mathrm{L},i} \equiv Z_i^{1/2} \nu_{\mathrm{L},i}^{\mathrm{ren}}.$$

We adjust this $Z_i^{1/2}$ factor so that the renormalized inverse propagator K_{ν}^{ren} takes the form

$$K_{\nu}^{\rm ren} = -\frac{1}{2} \not p (1 - \gamma_5). \qquad (3.72)$$

This is the on-shell renormalization condition for the pure left-handed neutrinos.

3.2.3 Vector boson sector

In the vector boson sector mixings take place not only among vector fields but also among vector and scalar fields. Here the scalar fields are Nambu-Goldstone (NG) bosons or physical Higgs particles. In the case of the Weinberg-Salam (WS) theory the neutral gauge boson sector is most complicated where Z and A bosons, NG scalar χ_s and physical Higgs scalar ϕ take part. If CP is a good symmetry, the physical Higgs $\phi(CP=+)$ decouples from the neutral boson sector (CP=-1). On the other hand, as far as we adopt a manifestly renormalizable covariant gauge, NG bosons, which are absorbed into the longitudinal components of gauge bosons, must be taken into account in the calculation of Feynman graphs.

We consider mixing of vector bosons V_{i}^{μ} and scalars χ_i . The asymptotic fields of vector bosons are given by

$$\widetilde{Z}_{ij}^{-1/2} V_j^{\mu}(x) + \widetilde{Z}_{\chi ij}^{-1/2} \partial^{\mu} \chi_j(x) \to V_i^{\mu, \mathrm{as}}(x) . \qquad (3.73)$$

Here $\partial_{\mu}\chi_{j}$ term, i.e., the scalar mode, plays a role of guaranteeing that the asymptotic vector field satisfies the physical polarization condition:

$$\partial_{\mu} V_i^{\mu, \mathrm{as}}(x) = 0 . \qquad (3.74)$$

According to the usual reduction formula, the S-matrix element with a V_i^{μ} vector boson is obtained from the Green function as shown below,

$$(\mathbf{G}_{amp}) \stackrel{k}{\longrightarrow} (\widetilde{Z}_{ij}^{-1/2} V_j^{\mu} + \widetilde{Z}_{xij}^{-1/2} \partial^{\mu} \chi_j) \times (M_i^2 - p^2) \varepsilon_{\mu}, \qquad (3.75)$$

where M_i is the mass of the $V_i^{\mu,as}$ vector boson and ε_{μ} is a polarization vector of the emitted (absorbed) vector boson, which satisfies

$$p^{\mu}\varepsilon_{\mu}=0. \qquad (3.76)$$

By this physical polarization condition the $\partial^{\mu}\chi_{j}$ part of the Green function in Eq. (3.75) does not contribute to the S-matrix element. Similarly the Green function with χ_{j} on the k side in Eq. (3.75) does not contribute since the $V_{j}^{\mu}\chi_{k}$ transition propagator is proportional to the momentum p^{μ} , which is contracted with ε_{μ} . Thus the part of the k-j transition propagator which contributes to the S-matrix element (3.75) is the vector-vector transition part alone.

We decompose the vector-vector propagator into transversal and longitudinal parts:

where $T(p^2)$ and $L(p^2)$ are invariant functions. In Eq. (3.75) only T part survives under condition (3.76). The pole part of T is given by

$$T_{ij}[\text{pole}] = \tilde{Z}_{im}^{1/2} \tilde{Z}_{jn}^{1/2} \frac{\delta_{mn}}{M_n^2 - p^2}.$$
 (3.78)

The S-matrix element is expressed as follows:

$$\begin{cases} G_{amp}^{k} \tilde{Z}_{km}^{1/2} \tilde{Z}_{jn}^{1/2} \frac{\delta_{mn}}{M_{n}^{2} - p^{2}} \tilde{Z}_{ij}^{-1/2} (M_{i}^{2} - p^{2}) \varepsilon_{\mu} \\ = G_{amp}^{k} \tilde{Z}_{ki}^{1/2} \varepsilon^{\mu} . \end{cases}$$

$$(3.79)$$

We introduce the renormalization constant $Z_{ij}^{1/2}$ such that

$$V_i^{\mu}(x) \equiv Z_{ij}^{1/2} V_j^{\mu, \text{ren}}(x). \qquad (3.80)$$

The on-shell renormalization conditions in which $Z_{ij}^{1/2} = \widetilde{Z}_{ij}^{1/2}$ are

$$T_{ij}^{\text{ren}}[\text{pole}] = \frac{\delta_{ij}}{M_i^2 - p^2}.$$
(3.81)

The inverse propagator may be expressed as

$$\mathcal{A}_{ij}^{-1\mu\nu}(p) = A_{ij}(p^2) \left(g^{\mu\nu} - \frac{p^{\mu}p^{\nu}}{p^2} \right) + B_{ij}(p^2) \frac{p^{\mu}p^{\nu}}{p^2}, \qquad (3.82)$$

where

$$A_{ij}(p^2) = T_{ij}^{-1}(p^2), \quad B_{ij}(p^2) = L_{ij}^{-1}(p^2).$$

The on-shell renormalization conditions in terms of $A_{ij}(p^2)$ read

$$\begin{bmatrix} i = j \end{bmatrix}$$

$$A_{ii}(M_i^2) = 0, \quad A_{ii'}(M_i^2) = -1, \quad (3.83a)$$

$$\begin{bmatrix} i \neq j \end{bmatrix}$$

$$A_{ij}(M_i^2) = 0, \quad A_{ij}(M_j^2) = 0. \quad (3.83b)$$

Just as in the case of scalars, the wave function renormalization matrix $Z_{ij}^{1/2}$ is not orthogonal but is a general real matrix. An important point concerning vector-scalar mixing is that the renormalization constants corresponding to the $\tilde{Z}_{xij}^{1/2}$ factor in Eq. (3.73) are unnecessary for the purpose of obtaining the S-matrix element. Hence it is sufficient to introduce the renormalization constants only for vector-vector mixing part. Another point is that the on-shell renormalization conditions are imposed only on $A_{ij}(p^2)$. The counter terms for $B_{ij}(p^2)$ which are already fixed by the above conditions on $A_{ij}(p^2)$, actually cancel the infinity of $B_{ij}(p^2)$ and the full renormalized inverse propagator is made finite automatically if the theory is renormalizable.

Finally, we would like to make some comments. We can easily generalize the former results to the case of fields with arbitrary spin. First of all, we need not consider explicitly mixing of lower-spin fields such as scalar mixing in the vector sector. The contribution of the lower spin fields vanishes because of the on-shell polarization condition. Just to obtain the S-matrix element, we need not mix lower-spin fields.

Next, the on-shell renormalization conditions are imposed on the part of the propagator which survives by the application of on-shell polarization.

Schematically the on-shell renormalization conditions read for all i and j (no summation for repeated indices)

$$\langle i|K_{ij}=0, \qquad (3\cdot 84a)$$

$$K_{ij}|j\rangle = 0, \qquad (3.84b)$$

$$\langle i | \{ K_{ii} S_i^{\mathrm{T}} \} = 1, \qquad (3 \cdot 84c)$$

$$\{S_i^T K_{ii}\} |i\rangle = 1, \qquad (3 \cdot 84d)$$

where K_{ij} is the inverse propagator, S_i^{T} is the diagonalized tree propagator and $|i\rangle$ denotes the on-shell particle state (wave function). The momentum variables in Eq. (3.84) are understood to be on the mass shell of the corresponding state.

We can interpret the conditions of Eq. (3.84) from another point of view: Equation (3.84a) shows that the matrix $K_{ij}(p^2)$ has eigenvalue zero at $p^2 = m_i^2$ and the state vector $\langle i |$ is the left eigenstate for it. Equation (3.84c) represents normalization of this eigenvector. We will explain this interpretation of the on-shell renormalization conditions further in § 3.3.2.

There are many cases where we cannot adopt the on-shell renormalization scheme defined above. In a theory with some constraints independent mass parameters of all physical particles cannot be supplied by all independent parameters in the Lagrangian. In other words, there may be some relations in tree mass parameters and these relations may not hold with higher order corrections. The so-called pseudo-Goldstone particle, which is massless in the tree approximation and becomes massive through higher loop corrections, is an example of such a constrained system. In these cases we cannot make the tree masses physical ones. One may regard the difference between tree and physical masses as counter terms. This procedure, however, destroys the loop expansion of the effective action and necessarily breaks the renormalizability of the theory. We must adopt the scheme in which the particle masses (the pole positions of the propagator) get corrected loop by loop. In the standard Weinberg-Salam theory the number of the independent bare parameters in the Lagrangian is sufficient to supply the full degrees of freedom of physical particle masses, and hence we perform the renormalization procedure in the on-shell scheme without any problem.

§ 3.3 Two-point functions in the Weinberg-Salam theory

In the preceding section we have studied the on-shell renormalization scheme and proposed the renormalization conditions. In this section we investigate the structure of two-point functions in the Weinberg-Salam (WS) theory. First dealing with the charged gauge boson sector, we see the manifestation of the quartet mechanism which assures the unitarity of the physical *S*-matrix and we prove the renormalizability of this sector explicitly. Then we proceed to the neutral gauge boson sector, where we prove that the onshell renormalization conditions can be imposed consistently with the aid of the Ward-Takahashi (WT) identities [Aoki 79].

Although in this and the next section we make sure of the validity and the consistency of our renormalization scheme, readers who are interested only in explicit renormalization calculation may jump over these sections (§§ 3.3 and 3.4).

3.3.1 Charged gauge boson sector

We begin with the WT identities for the effective action, Eq. $(2 \cdot 120)$ derived in § 2.4,

$$\frac{\delta\widetilde{\Gamma}}{\delta W^a_{\mu}} \cdot \frac{\delta\widetilde{\Gamma}}{\delta K^a_{a}} + \frac{\delta\widetilde{\Gamma}}{\delta \varphi} \cdot \frac{\delta\widetilde{\Gamma}}{\delta K_{\phi}} + \frac{\delta\widetilde{\Gamma}}{\delta \psi} \cdot \frac{\delta\widetilde{\Gamma}}{\delta K_{\phi}} + \frac{\delta\widetilde{\Gamma}}{\delta \overline{\psi}} \cdot \frac{\delta\widetilde{\Gamma}}{\delta K_{\overline{\phi}}} + \frac{\delta\widetilde{\Gamma}}{\delta c^a} \cdot \frac{\delta\widetilde{\Gamma}}{\delta K_{c^a}} = 0, \quad (3.85)$$

where $\widetilde{\Gamma}$ is defined as before in Eq. (2.117):

$$\widetilde{\Gamma} = \Gamma - \int \mathcal{L}_{\rm GF} d^4 x \,, \qquad (3.86)$$

and W_{μ}^{a} , ϕ , ψ ($\overline{\psi}$) and c^{a} represent (bare) gauge fields, scalars, fermions and ghost fields respectively. The corresponding BRS source terms are K_{a}^{μ} , K_{ϕ} , K_{ψ} ($K_{\overline{\psi}}$) and $K_{c^{a}}$ respectively.

In the charged boson sector relevant fields are charged gauge bosons $(W_{\mu}^{+}, W_{\mu}^{-})$, charged Nambu-Goldstone (NG) bosons (χ^{+}, χ^{-}) and charged ghosts and antighosts $(c^{+}, c^{-}, \bar{c}^{+}, \bar{c}^{-})$. The ghosts decouple from the W- χ sector because of the ghost number conservation. We define two-point functions as follows:

$$\widetilde{\Gamma}_{\mu\nu} \equiv \operatorname{FT} \frac{\delta^{2} \Gamma}{\delta W^{+\mu}(x) \, \delta W^{-\nu}(y)} \bigg|_{0}$$

$$\equiv A\left(p^{2}\right) \left(g_{\mu\nu} - \frac{p_{\mu} p_{\nu}}{p^{2}}\right) + B\left(p^{2}\right) \frac{p_{\mu} p_{\nu}}{p^{2}}$$

$$\equiv A\left(p^{2}\right) T_{\mu\nu} + B\left(p^{2}\right) L_{\mu\nu}, \qquad (3.87)$$

K-I. Aoki, Z. Hioki, R. Kawabe, M. Konuma and T. Muta

$$\begin{split} \widetilde{\Gamma}_{W_{\mu}x}^{+-} &= \operatorname{FT} \frac{\delta^{2} \widetilde{\Gamma}}{\delta W^{+\mu}(x) \delta \chi^{-}(y)} \Big|_{0} \\ &\equiv i p_{\mu} C^{+-}(p^{2}), \qquad (3 \cdot 88) \\ \widetilde{\Gamma}_{xx} &= \operatorname{FT} \frac{\delta^{2} \widetilde{\Gamma}}{\delta \chi^{+}(x) \delta \chi^{-}(y)} \Big|_{0} \\ &\equiv p^{2} F(p^{2}), \qquad (3 \cdot 89) \\ \widetilde{\Gamma}_{cc}^{+-} &= \operatorname{FT} \frac{\delta^{2} \widetilde{\Gamma}}{\delta c^{+}(x) \delta \overline{c}^{-}(y)} \Big|_{0} \\ &\equiv \gamma^{+-}(p^{2}), \qquad (3 \cdot 90) \end{split}$$

where FT denotes Fourier transform defined in Eq. (3.7) and the notation $|_0$ represents that each field is set to its vacuum expectation value respectively. In Eq. (3.89) we have written the factor p^2 explicitly; this factorization will be justified in Eq. (3.99b). Taking account of the hermiticity of the effective action, we obtain the relation:

$$\{C^{-+}(p^2)\}^* = C^{+-}(p^2), \qquad (3.91)$$

$$\{\gamma^{-+}(p^2)\}^* = \gamma^{+-}(p^2), \qquad (3.92)$$

both sides of which are purely real quantities in CP invariant case.

In the tree approximation these invariant functions take the values as

$$A(p^{2}) = M_{W}^{2} - p^{2},$$

$$B(p^{2}) = M_{W}^{2},$$

$$C^{+-}(p^{2}) = C^{-+}(p^{2}) = M_{W},$$

$$F(p^{2}) = 1,$$

(3.93)

where M_W is the charged gauge boson mass in the tree level. In order to write the explicit form of $\gamma(p^2)$ we need to fix the gauge. For example, in the 't Hooft gauge which will be defined in Eq. (3.103), they are given by

$$\gamma^{+-}(p^2) = \gamma^{-+}(p^2) = \alpha M_W^2 - p^2. \qquad (3.94)$$

Two-point functions $(3.87) \sim (3.90)$ are expressed in the following matrix form:

		W_{ν}^{-}	x ⁻	<i>c</i> -	<u></u> <i>c</i> -
$\widetilde{\Gamma}^{(2)} = -$	W _# +	$A(p^2)T_{\mu\nu}+B(p^2)L_{\mu\nu}$	$ip_{\mu}C^{+-}(p^2)$	0	0
	χ+	$-ip_{\mu}C^{-+}(p^2)$	$p^2F(p^2)$	0	0
	c+	0	0	0	$\gamma^{+-}(p^2)$
	<i>c</i> +	0	0	$-\gamma^{-+}(p^2)$	0
					(3.95)

where the minus signs attached to C^{-+} and γ^{-+} appear on account of the change of order of differentiation.

In order to obtain the relations among invariant functions imposed by gauge invariance we differentiate the WT identity (3.85) and set all the fields to their vacuum expectation values which are zero in this case.

$$\frac{\delta^{2}}{\delta W_{\mu}^{+}\delta c^{-}} \operatorname{Eq.} (3.85) \Big|_{0} :$$

$$\widetilde{\Gamma}_{\mu\nu}^{+-} \cdot \operatorname{FT} \frac{\delta^{2}\widetilde{\Gamma}}{\delta K_{\nu}^{+}\delta c^{-}} \Big|_{0} + \widetilde{\Gamma}_{W\mu\nu}^{+-} \cdot \operatorname{FT} \frac{\delta^{2}\widetilde{\Gamma}}{\delta K_{\lambda}^{+}\delta c^{-}} \Big|_{0} = 0, \quad (3.96)$$

$$\frac{\delta^{2}}{\delta \chi^{+}\delta c^{-}} \operatorname{Eq.} (3.85) \Big|_{0} :$$

$$\widetilde{\Gamma}_{W\mu\nu}^{+} \cdot \operatorname{FT} \frac{\delta^{2}\widetilde{\Gamma}}{\delta K_{\nu}^{+}\delta c^{-}} \Big|_{0} + \widetilde{\Gamma}_{\lambda\lambda}^{+-} \cdot \operatorname{FT} \frac{\delta^{2}\widetilde{\Gamma}}{\delta K_{\lambda}^{+}\delta c^{-}} \Big|_{0} = 0. \quad (3.97)$$

Here use has been made of the ghost number conservation. With the definition of invariant functions:

$$p^{\nu}J(p^{2}) \equiv \operatorname{FT} \frac{\delta^{2}\widetilde{\Gamma}}{\delta K_{\nu}^{+}\delta c^{-}}\Big|_{0}, \qquad (3.98a)$$

$$I(p^2) = \operatorname{FT} \frac{\delta^2 \widetilde{\Gamma}}{\delta K_{\chi}^+ \delta c^-} \bigg|_{0}, \qquad (3.98b)$$

Eqs. (3.96) and (3.97) are rewritten as

$$p_{\mu}B(p^{2})J(p^{2})+ip_{\mu}C^{+-}(p^{2})I(p^{2})=0, \qquad (3.99a)$$

$$-ip^{2}C^{-+}(p^{2})J(p^{2})+p^{2}F(p^{2})I(p^{2})=0.$$
 (3.99b)

Equation (3.99b) assures the factorizability in Eq. (3.89). The p^2 factor comes from the fact that the χ field represents the massless NG mode. From the above WT identities an important relation of $\tilde{\Gamma}^{(2)}$ components is derived:

,

K-I. Aoki, Z. Hioki, R. Kawabe, M. Konuma and T. Muta

$$B(p^{2}) F(p^{2}) = |C^{+-}(p^{2})|^{2}. \qquad (3.100)$$

The inverse propagator $\Gamma^{(2)}$ is obtained by adding the contribution of a gauge fixing Lagrangian to $\tilde{\Gamma}^{(2)}$. For the gauge fixing term we take the bilinear form without using auxiliary *B* fields,

$$\mathcal{L}_{GF} = -\frac{1}{\alpha} \left(\partial^{\mu} W_{\mu}^{+} - \alpha X^{+} \cdot \chi^{+} \right) \left(\partial^{\mu} W_{\mu}^{-} - \alpha X^{-} \cdot \chi^{-} \right) + \cdots, \qquad (3.101)$$

where only the charged gauge boson part is explicitly written and $X^{\pm} \cdot \chi^{\pm}$ denotes the convolution integral in the coordinate space as follows:

$$X^{\pm} \cdot \chi^{\pm} \equiv \int d^4 y X^{\pm} (x - y) \chi^{\pm} (y), \qquad (3 \cdot 102)$$

which is a local product in the momentum space. The case where X^{\pm} function is taken to be

$$X^{\pm}(x) = \delta^{4}(x) M_{W}, \qquad (3.103)$$

is called the 't Hooft gauge.

The contribution of the gauge fixing part of the effective action,

$$\Gamma_{\rm GF} = \int d^4 x \mathcal{L}_{\rm GF} \,, \qquad (3 \cdot 104)$$

to the two-point functions is calculated as follows:

$$\Gamma_{GF\mu\nu}^{+-} \equiv FT \frac{\delta^2 \Gamma_{GF}}{\delta W^{+\mu}(x) \,\delta W^{-\nu}(y)}$$
$$= -\frac{1}{\alpha} p_{\mu} p_{\nu} , \qquad (3.105)$$

$$\Gamma_{\rm GFW_{\mu}\chi}^{+-} = i p_{\mu} \widetilde{X}^{-}(p^2), \qquad (3.106)$$

$$\Gamma_{\mathrm{GFW}_{\mu}\chi}^{-+} = i p_{\mu} \widetilde{X}^{+}(p^{2}), \qquad (3 \cdot 107)$$

$$\Gamma_{\rm GFxx}^{+-} = -\alpha \widetilde{X}^{-}(p^2) \widetilde{X}^{+}(p^2), \qquad (3.108)$$

where $\widetilde{X}^{\pm}(p^2)$ is the Fourier transform of $X^{\pm}(x)$. For simplicity of the argument we take $\widetilde{X}^{\pm}(p^2)$ so as to cancel the $W\chi$ mixing part of the inverse propagator:

$$\widetilde{X}^{\pm}(p^2) = -C^{\pm\pm}(p^2). \tag{3.109}$$

Finally the full inverse propagator in this gauge fixing condition (3.109) is
$$\Gamma^{(2)} = \frac{W_{\mu}^{-}}{\chi^{+}} \frac{W_{\nu}^{-}}{A(p^{2})T_{\mu\nu} - \frac{1}{\alpha}\{p^{2} - \alpha B(p^{2})\}L_{\mu\nu}} 0}{F(p^{2})\{p^{2} - \alpha B(p^{2})\}}$$
(3.110)

where use has been made of Eq. (3.100). By adopting this type of nonlocal gauge fixing condition we could make the structure of the two-point functions very simple, although in non-local gauges we have various complications in the procedure of quantization.

How about the ghost propagator in this gauge condition? First go back to the second type of the WT identity (2.119):

$$F_i^{a} \frac{\delta \widetilde{\Gamma}}{\delta K_i} + \frac{\delta \widetilde{\Gamma}}{\delta \overline{c}^{a}} = 0, \qquad (3.111)$$

which is deduced from the ghost equation of motion. In our gauge it is written as

$$\partial_{\mu} \frac{\delta \widetilde{\Gamma}}{\delta K^{+\mu}} - \alpha X^{-} \cdot \frac{\delta \widetilde{\Gamma}}{\delta K^{+}_{\chi}} + \frac{\delta \widetilde{\Gamma}}{\delta \overline{c}^{+}} = 0, \qquad (3 \cdot 112)$$

where

$$K^{\pm \mu} \equiv (K_1^{\mu} \mp i K_2^{\mu}) / \sqrt{2}$$
.

With the operation of $\delta/\delta c^{-}|_{0}$ we obtain

...

$$-ip^{2}J(p^{2}) + \alpha \widetilde{X}^{-}(p^{2})I(p^{2}) = \gamma^{-+}(p^{2}), \qquad (3.113)$$

and we have

$$\gamma^{-+}(p^2) = -iJ(p^2) \{ p^2 - \alpha B(p^2) \}, \qquad (3.114)$$

where we have used Eqs. (3.99), (3.100) and (3.109). Thus the ghost inverse propagator is

$$\Gamma_{c}^{(2)} = \frac{c^{+}}{c^{+}} \frac{c^{-}}{-J(p^{2}) \{p^{2} - \alpha B(p^{2})\}} \frac{-J^{*}(p^{2}) \{p^{2} - \alpha B(p^{2})\}}{0}$$
(3.115)

We easily see the evidence of the quartet mechanism which guarantees

the unitarity of physical S-matrix explained in § 2.2. The propagators of the four particles [the scalar component of the gauge boson $(L_{\mu\nu}$ part), the NG boson (χ) , the ghost (c) and the antighost (\bar{c})] have a pole in a common position which is the solution of

$$p^2 - \alpha B(p^2) = 0$$
. (3.116)

These quartet particles can appear in the physical subspace of the total Hilbert space only in a zero-norm combination and hence cannot be observed. The only one physical mode of this sector is the massive vector boson which is represented by the $T_{\mu\nu}$ mode in the inverse propagator.

Let us proceed to the renormalization procedure. We introduce renormalization constants:

$$W_{\mu}^{\pm}(x) \equiv Z_{3}^{1/2} W_{\mu R}^{\pm}(x), \qquad (3.117)$$

$$\chi^{\pm}(x) = Z_{\chi}^{1/2} \chi_{\mathbb{R}}^{\pm}(x), \qquad (3 \cdot 118)$$

$$M_{W0}^2 = M_W^2 + \delta M_W^2, \qquad (3.119)$$

$$\delta \overline{M}_{\mathbf{w}}^{2} \equiv \delta M_{\mathbf{w}}^{2} Z_{\mathbf{s}} . \tag{3.120}$$

From the bilinear terms of the bare Lagrangian we separate out the counter terms for $\Gamma^{(2)}$, which will be shown systematically in § 4.3. The $\tilde{\Gamma}$ parts of renormalized inverse propagators are written by the use of renormalization constants:

$$\widetilde{\Gamma}^{R}_{\mu\nu} \equiv Z_{3} \widetilde{\Gamma}_{\mu\nu} = Z_{3} A(p^{2}) T_{\mu\nu} + Z_{3} B(p^{2}) L_{\mu\nu}$$
$$\equiv A_{R}(p^{2}) T_{\mu\nu} + B_{R}(p^{2}) L_{\mu\nu}, \qquad (3.121)$$

$$\widetilde{\Gamma}^{\mathrm{R}}_{W_{\mu}\chi} \equiv Z_{\chi}^{1/2} \widetilde{\Gamma}_{W_{\mu}\chi} \equiv i p_{\mu} C_{\mathrm{R}}(p^{2}), \qquad (3.122)$$

$$\widetilde{\Gamma}_{\chi\chi}^{R} \equiv Z_{\chi}\widetilde{\Gamma}_{\chi\chi} \equiv p^{2}F_{R}(p^{2}). \qquad (3.123)$$

They are also obtained directly by differentiating $\widetilde{\Gamma}$ with respect to the renormalized fields.

Suppose that the renormalization procedure up to (n-1)-loop is completed. We calculate *n*-loop contribution which is written as

$$A_{\rm R}^{(n)}(p^2) = Z_3^{(n)}(M_{\rm W}^2 - p^2) - \delta \overline{M}_{\rm W}^{2(n)} + \overline{A}^{(n)}(p^2), \qquad (3.124)$$

$$B_{\rm R}^{(n)}(p^2) = Z_3^{(n)} M_{\rm W}^2 - \delta \overline{M}_{\rm W}^{2(n)} + \overline{B}^{(n)}(p^2), \qquad (3.125)$$

where the suffix (n) denotes the *n*-loop contribution and the first two terms of each equation are *n*-loop counter terms not yet determined. The *n*-loop momentum integration part is involved in $\overline{A}^{(n)}$ and $\overline{B}^{(n)}$, and all their inner divergences are eliminated by the renormalization up to (n-1)-loop and there remain only overall divergences. The on-shell renormalization conditions derived in Eq. (3.83), Chap. 3 On-Shell Renormalization in Electroweak Theory. I 75

$$A_{\rm R}^{(n)}(M_{\rm W}^2) = 0, \quad A_{\rm R}^{(n)\prime}(M_{\rm W}^2) = 0, \qquad (3 \cdot 126)$$

determine the counter terms:

$$\delta \overline{M}_{W}^{2(n)} = \overline{A}^{(n)}(M_{W}^{2}), \quad Z_{3}^{(n)} = \overline{A}^{(n)'}(M_{W}^{2}). \quad (3 \cdot 127)$$

The resultant invariant functions $A_{R}^{(n)}(p^{2})$ become a finite function including the $(p^{2}-M_{W}^{2})^{2}$ factor.

To prove the finiteness of $B_{\rm R}^{(n)}(p^2)$ whose counter terms are already fixed by Eq. (3.127), we should note an important relation,

$$A(0) = B(0). (3.128)$$

This relation is obtained from the condition for non-existence of any pole in one-particle irreducible Green functions, that is, inverse propagators. This condition is always satisfied in perturbation theory. The above relation is also required to give rise to the Higgs mechanism: The unphysical modes $(B^{as} \text{ and } \chi^{as})$ in the vector boson field (W^{as}_{μ}) are eliminated and the massive vector boson asymptotic field is constructed. The relation $(3 \cdot 128)$ holds in any order of loop expansion:

$$A_{\rm R}^{(n)}(0) = B_{\rm R}^{(n)}(0). \tag{3.129}$$

As is seen in Eq. (3.125), the renormalization of $B(p^2)$ is only to subtract p^2 -independent constant terms. It can be expressed as

$$B_{\rm R}^{(n)}(p^2) = \overline{B}^{(n)}(p^2) - \overline{B}^{(n)}(0) + A_{\rm R}^{(n)}(0), \qquad (3.130)$$

where we take account of Eq. (3.129). Here another point should be noted that $\overline{B}^{(n)}(p^2)$ includes the factor M_W^2 because $\overline{B}(p^2)$ is equal to zero in the case of unbroken gauge symmetry $(M_W=0)$. This indicates that $\overline{B}^{(n)}(p^2)$ has only logarithmic divergence and we obtain finite $B_R^{(n)}(p^2)$ from Eq. (3.130).

The renormalized χ propagator in *n*-loop is expressed as

$$p^{2}F_{R}^{(n)}(p^{2}) = p^{2}Z_{\chi}^{(n)} + p^{2}\overline{F}^{(n)}(p^{2}), \qquad (3.131)$$

where the first term on the right-hand side is the counter term. The *n*-loop term $\overline{F}^{(n)}(p^2)$ has only logarithmic divergence. Therefore $F_{R}^{(n)}(p^2)$ is made finite by adjusting Z_{χ} as, for example,

$$Z_{\chi}^{(n)} = -\bar{F}^{(n)}(\mu^2), \qquad (3.132)$$

where the corresponding renormalization condition is

$$F_{\rm R}^{(n)}(\mu^2) = 0$$
. (3.133)

The renormalized mixing part $C_{\mathbb{R}}(p^2)$ is related to $B_{\mathbb{R}}(p^2)$ and $F_{\mathbb{R}}(p^2)$ by the identity $(3 \cdot 100)$:

$$|C_{\rm R}^{+-}(p^2)|^2 = B_{\rm R}(p^2) F_{\rm R}(p^2). \qquad (3.134)$$

The mixing part is automatically finite after the renormalization for B and F. We have made all $W\chi$ components of $\widetilde{\Gamma}^{(2)}$ finite.

Next we discuss the contribution of gauge fixing term. We define $X_{\mathbf{R}}^{\pm}$ and $\alpha_{\mathbf{R}}$ as

$$X_{R}^{\pm}(x) \equiv Z_{\chi}^{1/2} Z_{3}^{1/2} X^{\pm}(x), \qquad (3.135)$$

$$\alpha_{\mathrm{R}} \equiv Z_{3}^{-1} \alpha \,, \qquad (3.136)$$

where α_{R} is supposed to be finite. The gauge fixing condition (3.109) leads

$$\widetilde{X}_{\mathrm{R}}^{\pm}(p^{2}) = -C_{\mathrm{R}}^{\pm\mp}(p^{2}). \qquad (3.137)$$

The gauge fixing part of the effective action,

$$\Gamma_{\rm GF} = \int \mathcal{L}_{\rm GF} d^4 x$$

= $\int \frac{-1}{\alpha_{\rm R}} (\partial^{\mu} W^+_{\mu \rm R} - \alpha_{\rm R} X^+_{\rm R} \cdot \chi^+_{\rm R}) (\partial^{\mu} W^-_{\mu \rm R} - \alpha_{\rm R} X^-_{\rm R} \cdot \chi^-_{\rm R}) d^4 x$
(3.138)

becomes finite functional of renormalized quantities. Thus two-point functions of Γ_{GF} are renormalized to be finite.

Finally, the ghost inverse propagator is written with the renormalized quantities $\alpha_{\rm R}$ and $B_{\rm R}$:

$$\gamma^{+-}(p^2) = -J^*(p^2) \{ p^2 - \alpha_{\rm R} B_{\rm R}(p^2) \}.$$
(3.139)

This includes only logarithmic divergence because of the factorization of the finite term, $\{p^2 - \alpha_R B_R(p^2)\}$. Therefore we can make it finite by adjusting Z_c factor defined by

$$c(x) \equiv Z_c c_{\mathrm{R}}(x), \qquad (3.140)$$

appropriately.

We have made the pole structure of two-point functions clear and proved explicitly the renormalizability in the charged gauge boson sector.

3.3.2 Neutral gauge boson sector

In this subsection we study the neutral gauge boson sector which includes two gauge bosons $(Z_{\mu} \text{ and } A_{\mu}, \text{ or, } W_{\mu}^{3} \text{ and } W_{\mu}^{0})$, two Higgs fields (ϕ and χ_{s}), two ghosts (c^{3} and c^{0}) and two antighosts (\bar{c}^{3} and \bar{c}^{0}). We assume CP invariance of Lagrangian. In this case the physical Higgs field ϕ decouples completely from other particles. We adopt the following gauge fixing Lagrangian,

$$\mathcal{L}_{GF}^{\text{Neutral}} = \frac{-1}{2\alpha_{\mathfrak{z}}} \left(\partial^{\mu} W_{\mu}^{\mathfrak{z}} - \alpha_{\mathfrak{z}} \omega \cdot \partial^{\mu} W_{\mu}^{\mathfrak{z}} - \alpha_{\mathfrak{z}} X_{\mathfrak{z}} \cdot \chi_{\mathfrak{z}} \right)^{\mathfrak{z}} - \frac{1}{2\alpha_{\mathfrak{z}}} \left(\partial^{\mu} W_{\mu}^{\mathfrak{z}} \right)^{\mathfrak{z}}, \qquad (3.141)$$

where $\omega \cdot \partial^{\mu} W_{\mu}^{0}$ and $X_{s} \cdot \chi_{s}$ represent convolution integrals similar to Eq. (3.102). Similar to charged gauge boson case two-point functions obtained from $\tilde{\Gamma}$ are defined as

		$W^{\mathfrak{z}}_{\mathfrak{p}}$	W^{0}_{ν}	χ3
$\widetilde{\Gamma}^{(2)} = $	$W^{\mathfrak{s}}_{\scriptscriptstyle \mu}$	$A_{s}(p^{2})T_{\mu u}+B_{s}(p^{2})L_{\mu u}$	$A_{30}(p^2) T_{\mu u} + B_{30}(p^2) L_{\mu u}$	$ip_{\mu}C_{3\chi}(p^2)$
	$W^{\scriptscriptstyle 0}_{\scriptscriptstyle \mu}$	$A_{30}(p^2) T_{\mu\nu} + B_{30}(p^2) L_{\mu\nu}$	$A_0(p^2) T_{\mu u} + B_0(p^2) L_{\mu u}$	$ip_{\mu}C_{0\chi}(p^2)$
	X3	$-ip_{\nu}C_{3\chi}(p^2)$	$-ip_{\nu}C_{0x}(p^2)$	$p^2F_3(p^2)$
				$(3 \cdot 142)$

$$\frac{\overline{c^{3}}}{\delta c^{i} \delta \overline{c}^{j}} = -\frac{\delta^{2} \widetilde{\Gamma}}{\delta \overline{c}^{j} \delta c^{i}} = \frac{\overline{c^{3}}}{c^{0}} \frac{\gamma_{33}(p^{2})}{\gamma_{03}(p^{2})} \frac{\gamma_{30}(p^{2})}{\gamma_{00}(p^{2})}$$
(3.143)

Note that γ_{s0} has nothing to do with γ_{0s} . Gauge fixing part contribution is obtained as follows:

		$W^3_{ u}$	$W^0_{ u}$	X3
$\Gamma^{(2)}_{ m GF} =$	$W^{\mathfrak{s}}_{\mu}$	$-rac{p^2}{lpha_3}L_{\mu u}$	$p^2 \widetilde{\omega}_{\mathfrak{z}}(p^2) L_{\mu u}$	$ip_{\mu}\widetilde{X}_{\mathfrak{z}}(p^{\mathfrak{d}})$
	$W^{\scriptscriptstyle 0}_{\scriptscriptstyle \mu}$	$p^2 \widetilde{\omega}_{3}(p^2) L_{\mu u}$	$-\Big(lpha_{\mathfrak{z}}\widetilde{\omega}_{\mathfrak{z}}{}^{2}(p^{2})+rac{1}{lpha_{\mathfrak{s}}}\Big)p^{2}L_{\mu u}$	$-ip_{\mu}lpha_{\mathfrak{z}}\widetilde{\omega}_{\mathfrak{z}}(p^{2})\widetilde{X}_{\mathfrak{z}}(p^{2})$
	χ3	$-ip_{ u}\widetilde{X}_{3}(p^{2})$	$ip_{\mu}lpha_{3}\widetilde{\omega}_{3}(p^{2})\widetilde{X}_{3}(p^{2})$	$-lpha_{\mathfrak{z}}\widetilde{X}_{\mathfrak{z}}^{\mathfrak{z}}(p^2)$
				(3.144)

which is to do with $L_{\mu\nu}$ part in the vector sector.

By differentiating the WT identity (3.85) with respect to various combinations of fields, we obtain the following relations among invariant functions already defined. Here we represent these identities in the momentum representation in a graphical way explained in §§ 2.4 and 2.5.

,

$$\frac{\delta^2}{\delta W^3_{\mu} \delta c^3} \operatorname{Eq.}(3.85) \bigg|_{0} : \left. \begin{array}{c} & & \\ & & \\ & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & \\ & & \\ & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & \\ & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & \\ & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & \\ & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & \\ & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & \\ & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & \\ & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & \\ & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & \\ & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & \\ & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & \\ & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & \\ & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & \\ & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & \\ & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & \\ & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & \\ & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & \\ & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & \\ & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & \\ & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & \\ & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & \\ & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & \\ & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & \\ & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & \\ & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & \\ & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & \\ & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & \\ & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & \\ & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & \\ & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & \\ & & \\ \end{array} \right|_{0} : \left. \left. \left. \right|_{0} : \left. \begin{array}{c} & & \\ & & \\ \end{array} \right|_{0} : \left. \left. \left. \right|_{0} : \left. \left. \right|_{0} : \left. \left. \right|_{0} : \left. \left. \right|_{0} : \left. \right|_{0} : \left. \left. \left. \right|_{0} : \left. \left. \left. \right|_{0} : \left. \right|_{0} : \left. \left. \left. \right|_{0} : \left. \right|_{0} : \left. \left. \left. \right|_{0} : \left. \left. \right|_{0} : \left. \right|_{0} : \left. \left. \left. \right|_{0} : \left. \right|_{0} : \left. \left. \right|_{0} : \left. \left. \right|_{0} : \left. \left. \right|_{0} : \left. \left. \left. \right|_{0} : \left. \left. \left. \right|_{0} : \left. \right|_{0} : \left. \left. \left. \right|_{0} : \left. \left. \left. \right|_{0} : \left. \right$$

$$\frac{\delta^2}{\delta W^0_{\mu} \delta c^3} \operatorname{Eq.}(3.85) \bigg|_{0} : \left. \begin{array}{c} & & & \\ & & & \\ & & & \\ & & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & & \\ & & & \\ & & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & & \\ & & & \\ & & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & & \\ & & & \\ & & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & & \\ & & & \\ & & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & & \\ & & & \\ & & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & & \\ & & & \\ & & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & & \\ & & & \\ & & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & & \\ & & & \\ & & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & & \\ & & & \\ & & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & & \\ & & & \\ & & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & & \\ & & & \\ & & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & & \\ & & & \\ & & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & & \\ & & & \\ & & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & & \\ & & & \\ & & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & & \\ & & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & & \\ & & & \\ & & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & & \\ & & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & & \\ & & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & & \\ & & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & & \\ & & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & & \\ & & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & & \\ & & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & & \\ & & & \\ \end{array} \right|_{0} : \left. \left. \begin{array}{c} & & & \\ & & & \\ \end{array} \right|_{0} : \left. \begin{array}{c} & & & \\ & & & \\ \end{array} \right|_{0} : \left. \left. \begin{array}{c} & & & \\ & & & \\ \end{array} \right|_{0} : \left. \left. \left. \begin{array}{c} & & & \\ & & & \\ \end{array} \right|_{0} : \left. \left. \left. \begin{array}{c} & & & \\ & & & \\ \end{array} \right|_{0} : \left. \left. \left. \left. \right|_{0} : \left. \left. \right|_{0} : \left. \right|_{0} : \left. \left. \left. \right|_{0} : \left. \left. \right|_{0} : \left. \right|_{0} : \left. \right|_{0} : \left. \left. \right|_{0} : \left. \left. \right|_{0} : \left. \right|_{0} : \left. \right|_{0} : \left. \left. \right|_{0} : \left. \left. \right|_{0} : \left. \right|_{0} : \left. \left. \right|_{0} : \left. \right|_{0} : \left. \right|_{0} : \left. \left. \right|_{0} : \left. \left. \right|_{0} : \left. \left. \right|_{0} : \left. \right|_{0} : \left. \right|_{0} : \left. \left. \right|_{0} : \left. \left. \right|_{0} : \left. \left. \right|_{0} : \left. \right|_{0} : \left. \right|_{0} : \left. \left. \right|_{0} : \left. \right|_{0} : \left. \right|_{0} : \left. \left. \right|_{0} : \left. \left. \right|_{0} : \left. \right|_{0} : \left. \left. \right|_{0} : \left. \left. \right|_{0} : \left. \left. \right|_{0} : \left. \right|_{0}$$

$$\frac{\delta^2}{\delta \chi^8 \delta c^8} \operatorname{Eq.}(3.85) \bigg|_{0} : \frac{1}{\chi} \underbrace{-3}_{3} \underbrace{-3}_{3} + \underbrace{-3}_{\chi} \underbrace{-3}_{\chi} \underbrace{-3}_{\chi} = 0 , \quad (3.145c)$$

$$\frac{\delta^2}{\delta W^3_{\mu} \delta c^0} \operatorname{Eq.}(3.85) \bigg|_{\mathfrak{g}} : \operatorname{add}_{3} \operatorname{cond}_{3} \operatorname{cond}_{0} + \operatorname{add}_{3} \operatorname{cond}_{0} \operatorname{ip}_{\nu} + \operatorname{add}_{\lambda} \operatorname{cond}_{\nu} = 0 ,$$

$$\frac{\delta^2}{\delta W^0_{\mu} \delta c^0} \operatorname{Eq.}(3.85) \bigg|_{0} : \left. \underset{0}{\longrightarrow} \underset{0}{\longrightarrow} \underset{1}{\longrightarrow} \underset{0}{\longrightarrow} \underset{0$$

$$(3 \cdot 145e)$$

 $(3 \cdot 145f)$

where we have used the explicit form of Green functions,

$$W^{0}_{\mu}$$
, W^{0}_{μ} , W^{0}_{μ} , W^{0}_{μ} , (3.146)

derived from the BRS transform of W_{μ}^{0} ,

$$\delta^{\text{BRS}} W^0_{\mu} = \partial_{\mu} c^0 . \qquad (3 \cdot 147)$$

With the definitions of new invariant functions,

$$W^{3}_{\mu} \longrightarrow_{i} = p_{\mu} G_{3i}(p^{2}),$$
 (3.148a)

$$X_{3} = G_{xi}(p^{2}), \qquad (3.148b)$$

Eqs. (3.145) are rewritten as follows:

$$B_{3}G_{33} + iC_{3x}G_{x3} = 0, \qquad (3.149a)$$

$$B_{30}G_{33} + iC_{0x}G_{x3} = 0, \qquad (3.149b)$$

Chap. 3 On-Shell Renormalization in Electroweak Theory. I 79

$$C_{3x}G_{33} + iF_{3}G_{x3} = 0, \qquad (3.149c)$$

$$B_{3}G_{30} + iC_{3x}G_{x3} + iB_{30} = 0, \qquad (3.149d)$$

$$B_{30}G_{30} + iC_{0x}G_{x0} + iB_0 = 0, \qquad (3.149e)$$

$$C_{3x}G_{30} + iF_{3}G_{x0} + iC_{0x} = 0. \qquad (3.149f)$$

In these identities five equations are independent of each other:

$$B_{3}F_{3} = (C_{3\chi})^{2}$$
, (3.150a)

$$B_{3}B_{0} = (B_{30})^{2}, \qquad (3 \cdot 150b)$$

$$B_{3}C_{0\chi} = B_{30}C_{3\chi}$$
, (3.150c)

$$G_{x3} = iG_{s3}B_{s}/C_{sx}$$
, (3.150d)

$$G_{\rm x0} = (iG_{\rm s0}B_{\rm s} - B_{\rm s0})/C_{\rm sx}$$
 (3.150e)

From the second type of the WT identities (2.119),

$$\partial_{\mu} \frac{\delta \widetilde{\Gamma}}{\delta K_{\mu}^{3}} - \alpha_{s} \omega \cdot \partial_{\mu} \frac{\delta \widetilde{\Gamma}}{\delta K_{\mu}^{0}} - \alpha_{s} X_{s} \cdot \frac{\delta \widetilde{\Gamma}}{\delta K_{\chi_{s}}} + \frac{\delta \widetilde{\Gamma}}{\delta \overline{c}^{3}} = 0, \qquad (3.151a)$$

$$\partial_{\mu} \frac{\delta \widetilde{\Gamma}}{\partial K_{\mu}^{0}} + \frac{\delta \widetilde{\Gamma}}{\delta \overline{c}^{0}} = 0, \qquad (3.151b)$$

we have the following identities of the ghost inverse propagator with suitable operations:

$$\frac{\delta}{\delta c^{3}} \operatorname{Eq.}(3 \cdot 151 \mathrm{a}) \bigg|_{0} : i p^{2} G_{33} + \alpha_{3} \widetilde{X}_{3} G_{x3} = \gamma_{33}, \qquad (3 \cdot 152 \mathrm{a})$$

$$\frac{\delta}{\delta c^{0}} \operatorname{Eq.}(3 \cdot 151 \mathrm{a}) \bigg|_{0} : i p^{2} G_{30} + \alpha_{x} \widetilde{\omega}_{3} p^{2} + \alpha_{3} \widetilde{X}_{3} G_{x0} = \gamma_{03}, \qquad (3 \cdot 152 \mathrm{b})$$

$$\frac{\delta}{\delta c^3} \operatorname{Eq.}(3.151 \mathrm{b}) \bigg|_{0} : \quad 0 = \gamma_{30} , \qquad (3.152 \mathrm{c})$$

$$\frac{\delta}{\delta c^{\mathfrak{o}}} \operatorname{Eq.}(3 \cdot 151 \mathrm{b}) \bigg|_{\mathfrak{o}}: - p^{2} = \gamma_{\mathfrak{o}\mathfrak{o}}. \qquad (3 \cdot 152 \mathrm{d})$$

The ghost inverse propagator is expressed as

$$\Gamma_{c}^{(2)} = \frac{\overline{c^{3}}}{c^{0}} \begin{bmatrix} \overline{c^{0}} & \overline{c^{0}} \\ i(p^{2} + \alpha_{3}\widetilde{X}_{3}B_{3}/C_{3z})G_{33} & 0 \\ i(p^{2} + \alpha_{3}\widetilde{X}_{3}B_{3}/C_{3z})(G_{30} + iB_{30}/B_{3}) & -p^{2} \\ + ip^{2}(B_{30}/B_{3} - \alpha_{3}\widetilde{\omega}) & , (3.153) \end{bmatrix}$$

where we have replaced G_{x3} and G_{x0} by the right-hand side of Eqs. (3.150d, e). This expression will play an important role in the proof of charge universality (§ 3.4).

We clarify the pole structure of unphysical mode in this sector. In order to cancel the mixing terms between vector and scalar in $\Gamma^{(2)}$,

$$\Gamma^{(2)} = \widetilde{\Gamma}^{(2)} + \Gamma_{\rm GF}^{(2)}, \qquad (3 \cdot 154)$$

we choose the following gauge fixing conditions,

$$\widetilde{X}_{\mathfrak{s}}(p^2) = -C_{\mathfrak{s}\mathfrak{s}}(p^2), \qquad (3.155a)$$

$$\widetilde{\omega}(p^2) = -C_{\mathfrak{o}\mathfrak{x}}(p^2)/\alpha_{\mathfrak{s}}C_{\mathfrak{s}\mathfrak{x}}(p^2). \qquad (3.155b)$$

In this gauge $\varGamma^{\scriptscriptstyle(2)}$ and $\varGamma^{\scriptscriptstyle(2)}_c$ become

 $\Gamma^{(2)} =$

	$W^3_{ u}$	$W^0_{ m v}$	χ3
$W^{\mathfrak{s}}_{\mu}$	$A_{3}T_{\mu\nu}+\left(B_{3}-\frac{p^{2}}{\alpha_{3}}\right)L_{\mu\nu}$	$A_{30}T_{\mu\nu} + \left(B_{30} - \frac{p^2 B_{30}}{\alpha_3 B_3}\right) L_{\mu\nu}$	0
W ⁰ _µ	$A_{30}T_{\mu u} + \left(B_{30} - rac{p^2 B_{30}}{lpha_3 B_3} ight)L_{\mu u}$	$A_{0}T_{\mu\nu} + \left(B_{0} - \frac{p^{2}B_{0}}{\alpha_{s}B_{s}} - \frac{p^{2}}{\alpha_{0}}\right)L_{\mu\nu}$	0
X3	0	0	$(p^2-\alpha_{s}B_{s})F_{s}$
			(3.156)

$$\Gamma_{c}^{(2)} = \frac{\begin{array}{c|c} & \overline{c}^{3} & \overline{c}^{0} \\ \hline \\ \hline \\ c^{0} & c^{0} & (p^{2} - \alpha_{3}B_{3})G_{33} & 0 \\ \hline \\ (p^{2} - \alpha_{3}B_{3})\left(G_{30} + i\frac{B_{30}}{B_{3}}\right) & -p^{2} \\ \hline \\ \end{array}, \qquad (3.157)$$

where use has been made of Eq. (3.150). The determinant of the invariant functions of $L_{\mu\nu}$ part is

det
$$L_{\mu\nu} = \frac{1}{\alpha_{3}\alpha_{0}} p^{2} (p^{2} - \alpha_{3}B_{3}),$$
 (3.158)

and that of the ghost part is

det
$$\Gamma_c^{(2)} = p^2 (p^2 - \alpha_3 B_3) G_{33}$$
. (3.159)

We have obtained two zeros in these unphysical modes, that is, one at $p^2 = 0$ and the other at the root of $p^2 - \alpha_s B_s(p^2) = 0$, which we call m^2 . There are two quartets in this neutral sector: One quartet at $p^2 = 0$ consists of c_A , \bar{c}_A , $L^{A}_{\mu\nu}$ and the longitudinal polarization in $T^{A}_{\mu\nu}$, and the other one at $p^{2} = m^{2}$ consists of c_{z} , \bar{c}_{z} , $L^{Z}_{\mu\nu}$ and χ_{s} .

Next we investigate the on-shell renormalization conditions for A(photon)-Z boson system. The renormalized inverse propagator of this sector is defined as follows,

$$\Gamma_{ij}^{(2)R} \equiv \operatorname{FT} \frac{\delta^2 \Gamma}{\delta R_i \delta R_j}$$
$$= (Z_{im}^{1/2})^i \operatorname{FT} \frac{\delta^2 \Gamma}{\delta B_m \delta B_n} Z_{nj}^{1/2}$$
$$= \{ (Z^{1/2})^i \Gamma^{(2)} Z^{1/2} \}_{ij}, \qquad (3 \cdot 160)$$

where R_i and B_i represent renormalized and bare fields respectively:

$$R_i: R_Z = Z, \quad R_A = A,$$

 $B_i: B_3 = W_3, \quad B_0 = W_0.$

The matrix $Z_{ij}^{1/2}$ in Eq. (3.160) connects R_i and B_i :

$$B_i = Z_{ij}^{1/2} R_j \,. \tag{3.161}$$

In the tree approximation the $Z_{ij}^{1/2}$ matrix is the well-known mixing matrix,

$$Z^{1/2} = \begin{pmatrix} \cos \theta_{\mathbf{w}} & \sin \theta_{\mathbf{w}} \\ -\sin \theta_{\mathbf{w}} & \cos \theta_{\mathbf{w}} \end{pmatrix}.$$
(3.162)

The on-shell renormalization conditions are imposed as follows (see Eq. (3.83)),

$$A_{\rm R}^{A}(0) = 0$$
, $A_{\rm R}^{ZA}(0) = A_{\rm R}^{AZ}(0) = 0$, $(3 \cdot 163a)$

$$A_{\mathbf{R}}^{\mathbf{Z}A}(M_{\mathbf{Z}}^2) = A_{\mathbf{R}}^{\mathbf{A}\mathbf{Z}}(M_{\mathbf{Z}}^2) = 0, \qquad A_{\mathbf{R}}^{\mathbf{Z}}(M_{\mathbf{Z}}^2) = 0, \qquad (3.163b)$$

$$A_{\mathbf{R}}^{\mathbf{4}\prime}(0) = -1, \qquad \qquad A_{\mathbf{R}}^{\mathbf{Z}\prime}(M_{\mathbf{Z}}^{2}) = -1, \qquad (3.164)$$

where $A_{\rm R}^{ij}(p^2)$ is invariant functions of $T_{\mu\nu}$ part of $\Gamma_{ij}^{(2){\rm R}}$ parametrized similarly in Eq. (3.142). There are six conditions while the number of adjustable renormalization constants are four of $Z_{ij}^{1/2}$ factor and one of δM_Z^2 . Therefore in order to satisfy six conditions there must be an identity between invariant functions. We first recognize the non-pole conditions in the inverse propagator,

$$A_{\mathfrak{s}}(0) = B_{\mathfrak{s}}(0),$$

$$A_{\mathfrak{s}\mathfrak{o}}(0) = B_{\mathfrak{s}\mathfrak{o}}(0),$$

$$A_{\mathfrak{o}}(0) = B_{\mathfrak{o}}(0),$$
(3.165)

which are similar to Eq. (3.128). The WT identities (3.150b) shows

det
$$\begin{vmatrix} B_{\mathfrak{s}}(0) & B_{\mathfrak{s0}}(0) \\ B_{\mathfrak{s0}}(0) & B_{\mathfrak{o}}(0) \end{vmatrix} = 0$$
, (3.166)

and hence,

det
$$\begin{vmatrix} A_{\mathfrak{s}}(0) & A_{\mathfrak{s0}}(0) \\ A_{\mathfrak{s0}}(0) & A_{\mathfrak{o}}(0) \end{vmatrix} = 0.$$
 (3.167)

This relation guarantees the existence of a massless pole in this sector. For the renormalized amplitudes the same type of identity is deduced as

det
$$\begin{vmatrix} A_{R}^{z}(0) & A_{R}^{zA}(0) \\ A_{R}^{Az}(0) & A_{R}^{A}(0) \end{vmatrix} = 0$$
, (3.168)

where we have assumed

det
$$Z_{ij}^{1/2} \neq 0$$
. (3.169)

Due to the identity (3.168) two conditions (3.163a) are not independent. Consequently the number of independent renormalization conditions is five and is equal to the degrees of freedom of counter terms. The existence of massless pole is a direct result of gauge invariance; an unbroken gauge invariance guarantees the existence of a massless gauge boson.

Let us comment on the structure of the $Z_{ij}^{1/2}$ matrix. This matrix was once defined as an orthogonal type of matrix in a previous literature [Ross and Taylor 73]. As a result it was claimed that the on-shell renormalization conditions could not be satisfied for Z-A mixing sector. We understand that



Fig. 3. 1. Schematical diagram of the relation between renormalized on-shell fields and bare fields.

the degrees of freedom of this $Z^{1/2}$ factor were not enough. We allow four degrees of freedom to $Z_{ij}^{1/2}$ and we can satisfy the complete on-shell renormalization conditions. The circumstance that renormalized on-shell fields are not obtained by rotationlike transformation of bare fields is schematically shown in Fig. 3.1. It should be mentioned that angles θ_A and θ_Z in Fig. 3.1 are not physical quantities which could be calculated perturbatively.

Let us investigate details of our renormalization scheme from another point of view which we have mentioned in § 3.2.3. We denote the matrices of the invariant functions of $T_{\mu\nu}$ part in $\Gamma_{ij}^{(2)}$ and $\Gamma_{ij}^{(2)R}$ by $T_{ij}(p^2)$ and $T_{ij}^{R}(p^2)$ respectively,

$$T_{ij}(p^{2}) = \begin{pmatrix} A_{3}(p^{2}) & A_{30}(p^{2}) \\ A_{30}(p^{2}) & A_{0}(p^{2}) \end{pmatrix},$$

$$T_{ij}^{R}(p^{2}) = \begin{pmatrix} A_{R}^{Z}(p^{2}) & A_{R}^{ZA}(p^{2}) \\ A_{R}^{ZA}(p^{2}) & A_{R}^{A}(p^{2}) \end{pmatrix}.$$
(3.170)

The on-shell renormalization conditions (3.163) are expressed in terms of $T_{ij}^{R}(p^{2})$,

$$T_{ij}^{\mathbf{R}}(0) \begin{pmatrix} 0\\ 1 \end{pmatrix}_{j} = 0, \qquad (3.171a)$$

$$T_{ij}^{\mathsf{R}}(M_Z^2) \begin{pmatrix} 1\\ 0 \end{pmatrix}_j = 0. \qquad (3 \cdot 171b)$$

These equations are rewritten as

$$T_{ij}(0) Z_{jA}^{1/2} = 0$$
, (3.172a)

$$T_{ij}(M_z^2) Z_{iz}^{1/2} = 0$$
. (3.172b)

We find that the first type of the on-shell renormalization conditions (3.163) is nothing but the eigenvalue equations of T_{ij} with zero eigenvalue. Their eigenvectors correspond to $Z_{ij}^{1/2}$ matrix. With this interpretation the second type of conditions (3.164) may be regarded as normalization conditions for eigenvectors.

The eigenvalue equations are formally solved:

$$Z_{fa}^{1/2} = f_{a} \begin{pmatrix} A_{0}(0) \\ -A_{30}(0) \end{pmatrix} = f_{a} \begin{pmatrix} B_{0}(0) \\ -B_{30}(0) \end{pmatrix} , \qquad (3.173a)$$

$$Z_{JZ}^{1/2} = f_{Z} \begin{pmatrix} A_{0}(M_{Z}^{2}) \\ -A_{30}(M_{Z}^{2}) \end{pmatrix}$$
 (3.173b)

We determine the normalization factors f_A and f_z . From the renormalization conditions (3.164) written in the forms,

$$\frac{\partial}{\partial p^2} \{ (Z^{1/2})^{t} T Z^{1/2} \}_{ZZ} \bigg|_{p^2 = M_Z^2} = -1, \qquad (3 \cdot 174a)$$

$$\frac{\partial}{\partial p^2} \{ (Z^{1/2})^t T Z^{1/2} \}_{AA} \bigg|_{p^2 = 0} = -1, \qquad (3 \cdot 174b)$$

we obtain

$$f_{z} = \{2A_{0}A_{30}A_{30}' - (A_{0})^{2}A_{3}' - (A_{30})^{2}A_{0}'\}^{-1/2}|_{p^{2}=M_{z}^{3}}, \qquad (3 \cdot 175a)$$

$$f_{A} = \{2A_{0}A_{30}A_{30}' - (A_{0})^{2}A_{3}' - (A_{30})^{2}A_{0}'\}^{-1/2}|_{p^{2}=0} \quad . \tag{3.175b}$$

Equations (3.175) correspond to the following relation in QED:

$$Z_{3}^{1/2} = \{-A_{0}'(0)\}^{-1/2}.$$
(3.176)

In this section we have investigated the structure of two-point functions and explicitly proved that the on-shell renormalization conditions defined in § 3.2 are consistently imposed even in the Z-A mixing sector with the aid of the WT identities.

§ 3.4 Charge universality

In this section the electric charge universality is investigated. We mean by the electric charge universality that the on-shell coupling constant of the photon and a particle is proportional to the bare coupling constant in a matterindependent way. In a practical case, for example, the absolute value of the bare coupling constant of the proton is equal to that of the electron. Hence in this case the charge universality means that the proton and the electron have the same physical charge which is the on-shell coupling strength to the photon. This universality should hold in any model because the agreement between the charge of the proton and that of the electron is verified experimentally with quite good accuracy.

3.4.1 What is the problem?

In QED the charge universality is obtained as follows. The Lagrangian including gauge fixing terms,

$$\mathcal{L} = -\frac{1}{4} F^{0}_{\mu\nu} F^{0\mu\nu} + B^{0} \partial_{\mu} A^{0\mu} + \frac{\alpha_{0}}{2} B^{0} B^{0} + \sum_{i} \overline{\psi}_{i}^{\ 0} (i \partial_{\mu} \gamma^{\mu} + q_{i} e^{0} A_{\mu}^{\ 0} \gamma^{\mu} - m_{i}) \psi_{i}^{\ 0}, \qquad (3.177)$$

has a global U(1) invariance and leads to the conserved current:

Chap. 3 On-Shell Renormalization in Electroweak Theory. I 85

$$j_{\mu} = \sum q_i e_0 \overline{\psi}_i^0 \gamma_{\mu} \psi_i^0, \qquad (3 \cdot 178)$$

$$\partial^{\mu} j_{\mu} = 0 . \qquad (3 \cdot 179)$$

The charge defined by

$$Q \equiv \int j_0(x) d^3x , \qquad (3.180)$$

generates a corresponding U(1) transformation of fields ψ_i^0 such that

$$[\psi_i^0, Q] = q_i e_0 \psi_i^0. \qquad (3 \cdot 181)$$

From the conservation law (3.179) a trivial identity holds:

$$\langle 0|T\partial^{\mu}j_{\mu}(x)\psi_{i}^{0}(y)\overline{\psi}_{i}^{0}(z)|0\rangle = 0. \qquad (3.182)$$

With the aid of the equations of motion and the equal time commutators, we obtain a Ward-Takahashi (WT) identity in the momentum representation [Takahashi 57],

$$\Gamma_{i}^{0\nu}k_{\nu} = q_{i}e_{0}\{S_{i}^{0-1}(p) - S_{i}^{0-1}(p')\}, \qquad (3.183)$$

where Γ^0 and S^{-1} represent the photon proper vertex and the inverse propagator, respectively. This relation is graphically expressed as follows (see § 2.5):

For renormalized amplitudes the WT identity (3.183) takes the form,

$$\Gamma_i {}^{\nu}k_{\nu} = Z_3^{1/2} q_i e_0 \{ S_i^{-1}(p) - S_i^{-1}(p') \}.$$
(3.185)

With the operation $\partial/\partial k_{\mu}|_{k=0, p^2=m_i^2}$, we obtain the on-shell photon coupling as follows [Ward 50]:

$$\Gamma_i^{\ \mu} = \gamma^{\mu} Z_3^{1/2} q_i e_0 , \qquad (3.186)$$

where use has been made of the on-shell renormalization condition for $S^{-1}(p)$:

$$\frac{\partial}{\partial p} S^{-1}(p) \bigg|_{\text{on-shell}} = -1. \qquad (3.187)$$

The expression (3.186) assures the charge universality because the proportionality factor $Z_3^{1/2}$ is independent of *i*. Thus in QED the charge universality holds in a trivial way according to the property that the photon couples to the conserved current j^{μ} which generates the U(1) gauge transformation. In the Weinberg-Salam theory the situation is different, that is, the current to which the photon couples is not conserved,

$$\partial^{\mu}j_{\mu}^{WS} = O_1 \neq 0, \qquad (3.188)$$

and it is not equal to the $U(1)_{QED}$ generator current J:

$$j_{\mu}^{WS} - J_{\mu} = O_2 \neq 0$$
. (3.189)

Therefore the proof of the charge universality based on the WT identities is non-trivial. Here it should be noted that Eqs. (3.188) and (3.189) do not bring about physical difficulties. Operators $O_{1,2}$ are expressed in any gauge as

$$O_{1,2} = \{ Q_{\rm B}, \bar{O}_{1,2} \}, \qquad (3 \cdot 190)$$

that is, the BRS transform of appropriate operators $\overline{O}_{1,2}$. Such a type of operator does not develop the expectation value in a physical state because of the physical state condition,

$$Q_{\rm B}|{\rm phys}\rangle = 0$$
. (2.47)

In other words, $O_{1,2}$ are practically equivalent to null operators.

We comment on the unitary gauge, which is given by the limit in which the gauge parameter α goes to infinity in the 't Hooft gauge (3.103). In this gauge the mass of unphysical quartet (B, χ, c, \bar{c}) is αM and diverges in this limit. Following the decoupling argument, the Lagrangian in the unitary gauge is simply obtained by neglecting all unphysical fields relevant to the broken symmetries. In this gauge O_1 and O_2 equal zero due to the local $U(1)_{\text{QED}}$ invariance of the Lagrangian except for the gauge fixing part for the photon. This local invariance does not exist in a usual covariant gauge because the gauge fixing terms for W^{\pm} bosons break it. The proof of the charge universality in this gauge is trivial as in QED apart from a little modification as follows. With the use of the relation

$$\partial^{\mu} j_{\mu}^{\rm WS} [\rm unitary] = 0, \qquad (3.191a)$$

$$j^{\text{WS}}_{\mu}$$
[unitary] = J_{μ} , (3.191b)

we have the same WT identity as Eq. (3.183),

$$\Gamma_{Ai}^{0\nu}k_{\nu} = q_{i}e_{0}\{S_{i}^{0-1}(p) - S_{i}^{0-1}(p')\}. \qquad (3.192)$$

The renormalized photon vertex is obtained as follows,

$$\begin{split} \Gamma_{Ai}^{\nu} k_{\nu} &= Z_{AA}^{1/2} q_{i} e_{0} \{ S_{i}^{-1}(p) - S_{i}^{-1}(p') \} \\ &+ Z_{ZA}^{1/2} \times [\text{bare } Z \text{ boson vertex}], \end{split}$$
(3.193)

where $Z_{AA}^{1/2}$ and $Z_{ZA}^{1/2}$ are components of the matrix renormalization constant defined in Eq. (4.27). Because of the Z-A mixing the second term in Eq.

(3.193) appears. The conservation law (3.191a) guarantees the following identities of two-point functions defined in § 4.3.2,

$$A_{AA}(0) = 0, \quad A_{ZA}(0) = 0, \quad (3.194)$$

which imply (see $\S 3.3.2$)

$$Z_{ZA}^{1/2} = 0$$
. (3.195)

Using this property, we have the charge universality from Eq. (3.193):

$$\Gamma^{\mu}_{Ai} = \gamma^{\mu} Z^{1/2}_{AA} q_i e_0 . \qquad (3.196)$$

As is seen above, the WS theory in the unitary gauge can be regarded as the simplest extension of QED to include charged vector bosons W_{μ}^{\pm} and has properties similar to QED as far as the photon sector is concerned, although there may exist some complications in the renormalizability in this gauge.

Here we comment on the relation between the "charge universality"* and the renormalizability. In §2.5 we have demonstrated the renormalizability with the use of the "charge universality" which is obtained from the WT identities. The renormalizability is related only to the divergent part of amplitudes. On the other hand, the "charge universality" is concerned with the relations among full amplitudes including all finite terms as well as infinite terms. Therefore the "charge universality" is sufficient but not necessary condition for the renormalizability. Even if the symmetry is spontaneously broken, the renormalizability holds with the same counter terms as those in symmetric theory (see § 2.5), but the "charge universality" is broken in this case; for instance, the W^{\pm} boson coupling constants are not universal, once higher order corrections are taken into account.

There are the following three cases for remaining symmetry in spontaneously broken gauge theory:

- a) non-Abelian type,
- b) pure U(1) type,

c) U(1) which includes non-Abelian components of the original theory. In Case a) the proof of the "charge universality" holds as shown in § 2.5. Case b) essentially corresponds to pure QED and the proof of the "charge universality" is trivial as in QED. The actual QED in broken WS theory corresponds to Case c), that is, the photon field is the linear combination of $W_{\mu}^{0}(U(1))$ and $W_{\mu}^{s}(SU(2))$. In this case there are some difficulties for the proof of the charge universality. The inclusion of non-Abelian component in the photon causes $O_{1,2}\neq 0$ (see Eqs. (3.188) and (3.189)). Furthermore invariant tensors do not exist because the remaining symmetry is U(1) instead of non-Abelian symmetry. Thus the charge universality in the WS theory

^{*)} We denote the charge universality in a general case which is not restricted to the photon coupling by the "charge universality" (with quotation mark).

becomes non-trivial. We comment on another type of the breakdown of the "charge universality". Consider the case where an original symmetry G is broken to leave a symmetry $G_1 \times G_2$ (non-simple). The universal charge g_1 of G_1 gauge symmetry does not necessarily equal the universal charge g_2 of G_2 . For example, the separating behavior of running coupling constants in grand unified theory corresponds to this type of the breakdown of the "charge universality".

In the following, first, we discuss the "charge universality" in the framework of S-matrix theory. Second, we give a concise proof of the charge universality with the use of the WT identities. Finally, we give an alternative straightforward proof with the aid of equations of motion and the physical state condition [Aoki 79].

3.4.2 S-matrix theoretical approach

In this subsection we give an S-matrix theoretical proof of the charge universality without referring to an explicit form of the Lagrangian. An original argument was given by Weinberg where the photon and the graviton were treated [Weinberg 64]. Here we generalize it for non-Abelian gauge bosons [Aoki 79]. A general theorem treating any massless particle with spin j (≥ 1) is given by Kugo-Uehara, which will be stated later [Kugo and Uehara 81].

Consider a G invariant theory including massless vector particles which belong to the adjoint representation of G, where G is a simple Lie group. The S-matrix element with emission of a massless vector boson is written as

$$S = \varepsilon^{**} M_{\mu}, \qquad (3.197)$$

where $\varepsilon^{*\mu}$ is a polarization vector of the massless vector particle. The Lorentz invariance of S requires

$$k^{\mu}M_{\mu}=0$$
, (3.198)

where k^{μ} is the momentum of the emitted vector boson. This is because in a covariant field theory a Lorentz transformation of $\varepsilon^{*\mu}$ must be accompanied by the inhomogeneous term proportional to k^{μ} . This inhomogeneous term is the very result coming from masslessness of the particle. The "gauge invariance" of the S-matrix, Eq. (3.198), is required by the Lorentz invariance [Kugo and Uehara 81].

We define the on-shell charge g_m of a particle whose representation of G is R(m), by referring to the on-shell S-matrix element as follows:

$$p' \sigma' i' \xrightarrow{k}_{a} \mu \varepsilon_{\mu}^{*} \xrightarrow{k' \to 0} g_{m}(T_{R(m)}^{a})_{i'i} 2p^{\mu} \varepsilon_{\mu}^{*} \delta_{\sigma\sigma'}, \qquad (3.199)$$

where $p^{(\prime)}$, $\sigma^{(\prime)}$ and $i^{(\prime)}$ represent the momentum, the spin and the component of the representation of G respectively, T_R^a is the generator matrix of the representation R and a is the component of the emitted vector boson. In Eq. (3.199) we have omitted a trivial kinematical factor. Invariant amplitudes other than (3.199) must vanish in the limit $k \rightarrow 0$, which is shown later.

Consider M_{μ} including i_m $(m=1\sim N)$ particle of R(m) representation. In the limit $k\rightarrow 0$ in Eq. (3.198), similarly in the usual soft pion technique, non-vanishing contributions come from pole terms in M:

$$0 = \sum_{m} \frac{i_{m}}{\underset{\substack{\mu \\ k_{\mu}}}{\overset{i'_{m}}{\underset{\substack{\mu \\ k_{\mu}}}{\overset{i'_{m+1}}{\underset{\substack{\mu \\ k_{\mu}}}{\overset{i'_{m+1}}{\underset{\substack{\mu \\ k_{\mu}}}{\overset{i'_{m+1}}{\underset{\substack{\mu \\ k_{\mu}}}}}}}.$$
 (3.200)

The intermediate propagator in each graph in Eq. (3.200) has a pole as

$$\frac{1}{(p_m + \eta_m k)^2 - M_m^2} \sim \frac{\eta_m}{2p_m k},$$

$$\eta_m = \begin{cases} +1: \text{ for outgoing particle,} \\ -1: \text{ for incoming particle,} \end{cases} (3.201)$$

where M_m is the mass of particle *m*. This pole cancels the external factor k_{μ} and we obtain the identity,

$$\sum_{m=1}^{N} \eta_{m} g_{m} (T^{a}_{\mathcal{B}(m)})_{i_{m} i_{m'}} \Gamma_{i_{1} \cdots i_{m'} \cdots i_{N}} = 0, \qquad (3 \cdot 202)$$

where $\Gamma_{i_1 \dots i_N}$ is the S-matrix element without vector emission:

$$\Gamma_{i_1\cdots i_N} \equiv \begin{array}{c} \overset{i_1}{\longrightarrow} \\ & & \\ & \\ & &$$

We note here that $\Gamma_{i_1\cdots i_N}$ is proportional to an invariant G tensor and is invariant under an infinitesimal G transformation:

$$\sum_{m=1}^{N} \eta_{m} (T^{a}_{R(m)})_{i_{m}i_{m'}} \Gamma_{i_{1}\cdots i_{m'}\cdots i_{N}} = 0. \qquad (3 \cdot 204)$$

By comparing Eq. (3.202) with Eq. (3.204), we conclude the *m* independence of g_m :

$$g_m = g$$
 for all m . (3.205)

This indicates the "charge universality". One can also understand that the on-shell coupling has the unique form (3.199) because other forms of ampli-

tudes must break the gauge invariance of the S-matrix (3.200). This type of argument, however, does not give any information about quartic couplings which is also indispensable for the proof of the renormalizability (see Eq. (2.199)). Of course the above S-matrix theoretical approach is not applied to unphysical particles.

If the symmetry group G is not simple but U(1), we obtain

'n

ń

$$\sum_{m=1}^{N} \eta_m g_m = 0, \qquad (3 \cdot 206)$$

$$\sum_{m=1}^{N} \eta_m g_m^0 = 0, \qquad (3 \cdot 207)$$

instead of Eqs. (3.202) and (3.204), where g_m^0 is the bare coupling constant of particle *m*. Since these equations hold for any *S*-matrix element, we may conclude

 $g_m \propto g_m^0$ (3.208)

also in this U(1) case. In the case where there is some other symmetry independent of G, however, we cannot completely deduce the charge universality (3.208). For instance, if the electron number and the muon number conserve independently, we cannot conclude the equality of the electron charge and muon charge. Furthermore even if there is $\mu \rightarrow e\gamma$ process, we cannot deduce the vanishing charge of neutrinos in the case where the fermion number conservation holds.

By treating an emission of a massless particle with spin $j \ge 1$, the general gauge invariance of the S-matrix is obtained and the existence of the corresponding conserved charge with tensorial rank j-1 is concluded [Kugo and Uehara 81]. For example, massless particles with spin 1, 3/2 and 2 (graviton) imply scalar charge Q, super charge Q_{α} and vector charge Q_{μ} (energy-momentum P_{μ}) respectively. These conserved charges are related to the corresponding symmetries in the Lagrangian formalism. The proportionality between the charges and the quantum numbers is implied in every case; especially in the case of graviton, it means the equality between the gravitational and the inertial mass.

To sum up the S-matrix theoretical argument, the Lorentz invariance of the S-matrix element including massless vector is guaranteed by the cancellation among poles existing in the contribution of one particle intermediate state, and this cancellation implies the on-shell charge conservation and further, the "charge universality".

3.4.3 Proof by the use of the WT identities

We give a proof of the charge universality in the WS theory with the use of the WT identities [Aoki 79]. As the proof is non-trivial and very lengthy as mentioned in § 3.4.1, we present it in a concise form. We start with the differentiation of the WT identity as follows:

$$\frac{\delta^{3}}{\delta \overline{\psi}_{a} \delta \psi_{b} \delta c_{A}} \quad \text{Eq.} (2 \cdot 120) \Big|_{0} , \qquad (3 \cdot 209)$$

where ψ and $\overline{\psi}$ are renormalized fields and $\delta/\delta c_A$ is defined by

$$\frac{\delta}{\delta c_A} \equiv x \frac{\delta}{\delta c_3} + y \frac{\delta}{\delta c_0}. \qquad (3.210)$$

The coefficients x and y in Eq. (3.210) are determined later. Equation (3.209) is expressed graphically as



where we have assumed for simplicity that ϕ does not mix with the photon sector. According to the renormalization of ψ and $\overline{\psi}$, we introduce the renormalized BRS source currents K_{ψ} and $K_{\overline{\psi}}$ satisfying

$$\frac{\delta}{\delta K^{0}_{\psi_{a}}} \cdot \frac{\delta}{\delta \psi_{a}^{0}} = \frac{\delta}{\delta K_{\psi_{a}}} \cdot \frac{\delta}{\delta \psi_{a}},$$

$$\frac{\delta}{\delta K^{0}_{\psi_{a}}} \cdot \frac{\delta}{\delta \overline{\psi}_{a}^{0}} = \frac{\delta}{\delta K_{\overline{\psi}_{a}}} \cdot \frac{\delta}{\delta \overline{\psi}_{a}}.$$
(3.212)

All graphs in (3.211) represent those obtained by the differentiation with respect to renormalized quantities $(\psi, \overline{\psi}, K_{\psi}, K_{\overline{\psi}})$.

First, we determine the coefficients x and y so that the sum of the first two terms in Eq. (3.211) becomes the renormalized physical photon vertex in the limit $k^2 \rightarrow 0$:

By the use of invariant functions defined in § 3.3.2, we obtain

$$x = -f_{A}\{iB_{0}(0) + B_{30}(0)G_{30}(0)\}/G_{33}(0), \qquad (3.214a)$$

$$y = f_A B_{so}(0),$$
 (3.214b)

where the formal solution of the renormalized photon channel (3.173a) is used. With the aid of the WT identities (3.150), these equations are rewritten as

$$x = i f_{a} C_{0x}(0) G_{x0}(0) / G_{ss}(0), \qquad (3 \cdot 215a)$$

$$y = -if_{a}C_{0x}(0)G_{x3}(0)/G_{33}(0). \qquad (3.215b)$$

From Eqs. $(3 \cdot 215a)$ and $(3 \cdot 215b)$ we have

$$xG_{x3}(0) + yG_{x0}(0) = 0, \qquad (3.216)$$

which represents that the scalar mixing contribution $G_{\chi A}$ (the third term in Eq. (3.211)) vanishes in the limit $k^2 \rightarrow 0$. With the use of Eq. (3.153), we have also

$$x\gamma_{33}(0) + y\gamma_{03}(0) = 0, \qquad (3.217)$$

which means that c_A corresponds to the massless channel of ghosts. In short, the renormalized photon channel is obtained by the differentiation with respect to the massless channel of ghosts c_A .

The resultant WT identity is

which resembles that of a symmetric theory (2.181). There is, however, a difference between them regarding the existence of particle mixing. This is the critical point to make the proof very lengthy. With the extensive analysis of the structure of on-shell renormalized inverse propagators and other invariant functions, one can derive the following relation,

$$\Gamma_{\text{on-shell}}^{\mu} = c_{\mathbf{A}} |_{\text{on-shell}} \cdot 2p^{\mu} . \qquad (3.219)$$

For other particles than fermions, we obtain the on-shell charge in a similar way. For example, for the W^+ boson we have

$$\Gamma_{W^{+}\text{on-shell}}^{\mu} = \Pr_{\mathbf{c}_{\mathbf{A}}}^{\nu} \stackrel{\mathbf{c}_{\mathbf{A}}}{\underset{\mathbf{c}_{\mathbf{A}}}{}} = \Pr_{\mathbf{c}_{\mathbf{A}}}^{\nu} \stackrel{\mathbf{c}_{\mathbf{A}}}{\underset{\mathbf{c}_{\mathbf{A}}}{}} = \Pr_{\mathbf{c}_{\mathbf{A}}}^{\mu} (3.220)$$

We easily generalize the above results as

$$\Gamma_{\text{on-shell}}^{\mu,ij} = \overset{i}{\longrightarrow}_{c_{A}} |_{\text{on-shell}} \cdot 2^{p^{\mu}}, \qquad (3\cdot221)$$

which we call the (renormalized) charge graph.

In order to evaluate the charge graph we proceed to a fourth derivative of the WT identity (2.120). For fermions, for instance, applying the operation,

$$\frac{\delta^4}{\delta \overline{\psi}_a \delta \psi_b \delta W_{\nu}^+ \delta c_A} \cdot \bigg|_0 , \qquad (3.222)$$

we have the following identity,



By putting all external lines of the above amplitudes on mass-shell, one finds Eq. (3.223) reduces to

$$= 0 , (3.224)$$

which is similar to Eq. $(2 \cdot 190)$. We can generalize Eq. $(3 \cdot 224)$ for the case of arbitrary number of external lines to obtain,

$$\sum_{i}^{r} \qquad (3 \cdot 225)$$

This is nothing but the on-shell charge conservation law (3.206).

To prove the charge universality, we must extract the $U(1)_{\text{QED}}$ quantum number q_i from the charge graph leaving a matter independent factor. In the symmetric theory with simple group, we have obtained the "charge universality" easily because there exists invariant tensors in that case (see Eqs. $(2\cdot191)$ and $(3\cdot204)$). In the case of pure U(1) symmetry, we can evaluate the charge graph directly as follows. The ghost c_A corresponds to the U(1)channel and hence it is a free particle, which is related to the well-known fact that QED Lagrangian does not need a ghost field. Thus the charge graph is just a constant given by the coefficient of c_A term in the BRS transform of the relevant field:

The photon in the WS theory corresponds to the U(1) symmetry which is the combination of the original U(1) and non-Abelian component. The above fact obstructs the gauge independent proof of the charge universality. We adopt the Landau gauge such as

$$\mathcal{L}_{GF} = \sum_{a=0}^{3} B^{a} \partial^{\mu} W^{a}_{\mu} . \qquad (3 \cdot 227)$$

In this gauge some amplitudes are evaluated very easily because of the transversal structure of gauge boson propagators as follows. Consider, for example, the Green function G_{x3} (bare) which is graphically expressed as

where the dot represents the BRS transform of χ_3 . The BRS transform of Higgs scalar ϕ is obtained by the following replacement in its gauge transform: gauge parameters \rightarrow corresponding ghost fields (see (2.39)),

$$\delta^{\text{BRS}} \boldsymbol{\varPhi}(x) = i \lambda g^a c^a(x) T^a \boldsymbol{\varPhi}(x). \qquad (3 \cdot 229)$$

In terms of each component of Higgs field defined in Eq. (2.65), the BRS transform is given by

$$\delta\phi = -\lambda \frac{g}{2} \chi_{a} c^{a} + \lambda \frac{g'}{2} \chi_{3} c^{0},$$

$$\delta\chi_{a} = \lambda \frac{g}{2} [(v + \phi) c^{a} + f^{abc} \chi_{b} c^{c}] + \begin{cases} \lambda \frac{g'}{2} \chi_{2} c^{0}, & (a = 1) \\ -\lambda \frac{g'}{2} \chi_{1} c^{0}, & (a = 2) \\ -\lambda \frac{g'}{2} (v + \phi) c^{0}, & (a = 3) \end{cases}$$
(3.230)

where all quantities are bare ones. With the use of Eq. (3.230), G_{x3} is separated into two parts:

$$= \frac{1}{2} g_0 v_0 + (3.231)$$

In the limit $k \rightarrow 0$ the second term including the gauge coupling with ghost fields vanishes. The reason for this is that ghost-gauge coupling constant has a factor q^{ν} (the ghost momentum), which is contracted with the gauge boson propagator in the Landau gauge,

$$\frac{g_{\mu\nu} - p_{\mu} p_{\nu} / p^2}{p^2}, \qquad (3.232)$$

and vanish in the limit $k \rightarrow 0$. Thus we obtain

$$G_{x_3}(0) = \frac{1}{2} g_0 v_0, \qquad (3 \cdot 233)$$

and in a similar way we have

$$G_{\mathbf{x}\mathbf{0}}(0) = -\frac{1}{2} g_{\mathbf{0}}' v_{\mathbf{0}} \,. \tag{3.234}$$

The differentiation with respect to c_A is, in this gauge, given by

$$\frac{\delta}{\delta c_{A}} = -if_{A}v_{0}\frac{C_{0x}(0)}{2G_{33}(0)} \left(g_{0}'\frac{\delta}{\delta c_{3}} + g_{0}\frac{\delta}{\delta c_{0}}\right), \qquad (3.235)$$

where use has been made of Eq. (3.215). The BRS transform of a field ϕ_t^0 is expressed as

$$\delta^{\text{BRS}}\phi_{i}^{0} = i\left(-if_{4}v_{0}\frac{C_{3\chi}(0)}{2G_{33}(0)}\right)g_{0}g_{0}'q_{i}c_{4}\phi_{i}^{0} + \cdots, \qquad (3 \cdot 236)$$

where q_i is the $U(1)_{QED}$ generator:

$$q_i = \left(I_s + \frac{Y}{2}\right)_i. \tag{3.237}$$

The bare charge graph of ϕ_t^{0} is easily evaluated in a way similar to the evaluation of G_{xs} and we obtain

where 1 is the unit matrix in all spin and flavour indices. The renormalized charge graph (3.221) equals the bare one (3.238) because of index-independence (1) of Eq. (3.238). The proof is now completed with this equation. Other expressions of the on-shell charge are obtained,

$$e_{\text{on-shell}}^{i} = Z_{04}^{1/2} g_{0}' q_{i}$$

= $\frac{i}{G_{33}(0)} Z_{34}^{1/2} g_{0} q_{i}$. (3.239)

They may be interesting in comparison with that of QED,

$$e_{\text{on-shell}}^{i}(\text{QED}) = Z_{3}^{1/2} e_{0} q_{i}$$
. (3.240)

3.4.4 "Maxwell" equation and charge universality

We investigate here the "Maxwell" equations, the equations of motion of gauge fields, to give another proof for the charge universality. For simplicity, gauge fixing terms are taken as

$$\mathcal{L}_{\rm GF} = -\partial_{\mu}B^{a}W^{a}_{\mu} + \frac{\alpha}{2}B^{a}B^{a}, \qquad (3\cdot 241)$$

which preserve the global $SU(2) \times U(1)$ symmetry. In this gauge the equations of motion of $W_{\mu}^{\ \alpha}$ are

$$\partial^{\nu} F^{a}_{\mu\nu} = g J^{a}_{\mu} - \{ Q_{B}, D_{\mu} \overline{c}^{a} \}$$
 for $a = 1, 2, 3,$ (3.242)

$$\partial^{\nu} F^{0}_{\mu\nu} = \frac{1}{2} g' J^{0}_{\mu} - \{ Q_{B}, \partial_{\mu} \bar{c}^{0} \}, \qquad (3.243)$$

where J_{μ}^{a} $(a=0\sim3)$ are the corresponding generator currents of the SU(2) $\times U(1)$ global symmetry. On the true vacuum only one unbroken symmetry is $U(1)_{\text{QED}}$ whose charge is formally written as

$$Q = I_{s} + \frac{1}{2}Y = Q^{s} + \frac{1}{2}Q^{0}. \qquad (3.244)$$

With this combination for the charge, "Maxwell" equations (3.242) can be rewritten as

$$\partial^{\nu} F_{\mu\nu} = e^{0} \left(J_{\mu}^{3} + \frac{1}{2} J_{\mu}^{0} \right) - N_{\mu} , \qquad (3 \cdot 245)$$

where we define

$$e^{0} \equiv \frac{gg'}{\sqrt{g^{2} + g'^{2}}},$$
 (3.246)

$$F_{\mu\nu} = \frac{1}{\sqrt{g^2 + g'^2}} (g' F^3_{\mu\nu} + g F^0_{\mu\nu}), \qquad (3.247)$$

$$N_{\mu} = \left\{ Q_{\rm B}, \, \frac{1}{\sqrt{g^2 + {g'}^2}} [g' \, (D_{\mu} \bar{c})^3 + g \partial_{\mu} \bar{c}^{\,0}] \, \right\} \,. \tag{3.248}$$

Here we should notice both of the formal charges,

$$\widetilde{Q} \equiv \int d^3x \left(J_0^3 + \frac{1}{2} J_0^0 \right) \tag{3.249}$$

and

$$N \equiv \frac{1}{e^0} \int d^8 x N_0, \qquad (3.250)$$

are conserved and equally generate the $U(1)_{QED}$ transformation, since the difference between them is a space integration of total divergence and is

replaced by a surface term which does not contribute to commutators with local fields. In fact, neither of them is well-defined because of the massless one-particle contributions. The unbroken symmetry assures only one well-defined charge which is a certain combination of \tilde{Q} and N,

$$Q = \int d^{3}x \left[\left(J_{0}^{3} + \frac{1}{2} J_{0}^{0} \right) - \frac{1}{e^{0}} \zeta N_{0} \right] \frac{1}{1 - \zeta}$$

= $\int d^{3}x J_{0}^{q}$, (3.251)

where ζ is a parameter to be determined dynamically so that the massless pole cancels out. For instance, in QED this ζ parameter is easily evaluated to be $(1-Z_s^{1/2})$ because the massless pole in this case is *B* field which is free. We obtain from Eqs. (3.245) and (3.251),

$$\frac{1}{1-\zeta}\partial^{\nu}F_{\mu\nu} = e^{\theta}J_{\mu}^{\ q} - N_{\mu}. \qquad (3.252)$$

This $1/(1-\zeta)$ factor is important to obtain the correct result in the following, although in the usual proof with differentiation of J_{μ} (see § 3.4.1), this factor is irrelevant.

Sandwiching Eq. (3.252) by physical one-particle states $|i\rangle$ and $|f\rangle$, we have

$$(1-\zeta)^{-1} \langle f | \partial^{\mu} F_{\mu\nu} | i \rangle = e^{0} \langle f | J_{\mu}^{Q} | i \rangle. \qquad (3.253)$$

Note that N_{μ} does not contribute,

$$\langle f|N_{\mu}|i\rangle = \langle f|\{Q_{\mathrm{B}},*\}|i\rangle = 0,$$
 (3.254)

due to the physical state condition:

$$Q_{\rm B}|i\rangle = 0, \quad Q_{\rm B}|f\rangle = 0. \tag{3.255}$$

We apply $\int d^4x e^{ikx}$ to both sides of Eq. (3.253) and take the limit $k \rightarrow 0$. On the left-hand side of Eq. (3.253), only the contribution coming from a massless particle in the $F_{\mu\nu}$ channel, namely, the photon A_{μ} , survives:

$$\mathbf{k}_{\mu} \bigoplus^{\mathbf{F}_{\mu\nu}} \bigcap_{i}^{\mathbf{f}} \xrightarrow{\mathbf{k} \neq \mathbf{0}} \bigoplus^{\mathbf{A}} \bigcap_{i}^{\mathbf{f}} \cdot (3 \cdot 256)$$

It should be noted that a massless scalar particle such as $B \text{ or } \chi$ cannot contribute to the $F_{\mu\nu}$ channel because $F_{\mu\nu}$ is antisymmetric in μ and ν . Thus the left-hand side reduces to the on-shell photon coupling up to a constant independent of matters *i* and *f*. The right-hand side ($\mu=0$) in this limit is nothing but the matrix element of the well-defined electromagnetic charge operator Q, that is, $q_i \delta_{fi}$, where q_i is the eigen value of the charge eigen state $|i\rangle$. Thus we have obtained the charge universality in a fairly simple way.

This simplification of the proof is due to the physical state condition (3.255). Furthermore the proof here does not limit the states $|i\rangle$ and $|f\rangle$ to those made out of elementary fields and can be applied to any physical particles including bound states. Thus we have got a proof for the equality of the electric charges of the proton and the electron, assuming that the proton is a bound state of the three quark channel, *uud*. Here we have seen an example to suggest that the canonical operator formalism with equations of motion is very useful and powerful to reveal essential structures of gauge theories.

To confirm that the above result agrees with the former results (3.238) and (3.239) obtained only by using the WT identities, we evaluate the proportionality constant of the on-shell charge to the bare charge. The onephoton contribution to the $F_{\mu\nu}$, C is estimated by the use of the asymptotic form of $F_{\mu\nu}$,

$$(F_{\mu\nu})^{\rm as} = C \left(\partial_{\mu} A_{\nu}^{\rm as} - \partial_{\nu} A_{\mu}^{\rm as} \right) + \cdots, \qquad (3 \cdot 257)$$

where A_{μ}^{as} is the asymptotic field of the physical photon and dots stand for other particles such as Z boson. Noting that Eq. (3.257) leads to

$$FT\langle 0|TF_{\mu\nu}(x)A_{\rho}(y)|0\rangle = iC(p_{\mu}g_{\nu\rho} - p_{\nu}g_{\mu\rho})\frac{1}{p^{2}}, \qquad (3.258)$$

we have

$$\lim_{k \to 0} \int d^4 x e^{ikx} \langle f | \partial^{\nu} F_{\nu\mu}(x) | i \rangle (1 - \zeta)^{-1} \\ = -C (1 - \zeta)^{-1} \lim_{p^f \to p^i} (2\pi)^4 \delta^4 (p^f - p^i) \cdot e_{fi} (p^f + p^i)_{\mu}, \quad (3 \cdot 259)$$

where the state normalization convention is

$$\langle f|i\rangle = (2\pi)^{3} 2p_0^{i} \delta_{fi} \delta^{3} (p^f - f^i). \qquad (3 \cdot 260)$$

The kinematical form $(p^{f} + p^{i})_{\mu}$ is implied by the following expression of the right-hand side $(\mu = 0)$ in Eq. (3.253),

$$\lim_{k \to 0} \int d^4x e^{ikx} \langle f | e^0 J_0^{\ Q}(x) | i \rangle$$

= $e^0 q_i \delta_{fi} \lim_{p^f \to p^i} (2\pi)^4 \delta^4 (p^f - p^i) \cdot 2p_0^i$. (3.261)

By comparing Eq. (3.259) with (3.261), the on-shell charge e_{fi} is represented by

$$e_{fi} = -C^{-1}(1-\zeta)\,\delta_{fi}e^{0}q_{i}\,. \tag{3.262}$$

The factor $C^{-1}(1-\zeta)$ is evaluated as follows. From the equation of motion $(3\cdot 252)$, this factor is given by the residue of the massless pole of the following Green function,

$$-\mathrm{FT}\langle 0|TA_{\rho}N_{\mu}|0\rangle,\qquad(3\cdot263)$$

where we have taken into account that $J_{\mu}^{\,Q}$ has no massless one-particle contribution. The Green function is easily estimated in the Landau gauge ($\alpha = 0$ in \mathcal{L}_{GF} (3.241)) and the result is

$$-C^{-1}(1-\zeta) e^{0} = Z_{04}^{1/2} g_{0}' . \qquad (3.264)$$

This relation agrees with Eq. (3.239) obtained by the use of the WT identities.

Chapter 4

On-Shell Renormalization in Electroweak Theory. II

In this chapter we summarize the necessary machinery to carry out higher-order calculations practically in the framework of the on-shell renormalization in the Weinberg-Salam theory.

§4.1 Structure of the full Lagrangian

The basic Lagrangian to start with is given by Eq. (2.86) together with Eqs. (2.18), (2.36a), (2.36b), (2.66), (2.87) and (2.88). For convenience we here present the full form of the Lagrangian once again:

$$\mathcal{L} = \mathcal{L}_{G} + \mathcal{L}_{F} + \mathcal{L}_{GF} + \mathcal{L}_{FP} + \mathcal{L}_{H} + \mathcal{L}_{M}, \qquad (4.1)$$

$$\mathcal{L}_{\mathsf{G}} = -\frac{1}{4} F^{a}_{\mu\nu} F^{\mu\nu}_{a}, \qquad (4\cdot 2)$$

$$\mathcal{L}_{\mathrm{F}} = i \sum_{i} \overline{L}_{i} \mathcal{D}_{\mathrm{L}} L_{i} + i \sum_{n=i,I} \overline{R}_{n} \mathcal{D}_{\mathrm{R}} R_{n}, \qquad (4.3)$$

$$\mathcal{L}_{\rm GF} = B^a F_i^a \phi^i + \frac{\alpha}{2} (B^a)^2, \qquad (4 \cdot 4)$$

$$\mathcal{L}_{\rm FP} = -\bar{c}^a \delta^a (F_i^{\ b} \phi^i) c^b , \qquad (4.5)$$

$$\mathcal{L}_{\mathrm{H}} = (D_{\mathrm{L}}^{\mu} \boldsymbol{\varPhi})^{\dagger} (D_{\mathrm{L}\mu} \boldsymbol{\varPhi}) + \mu^{2} \boldsymbol{\varPhi}^{\dagger} \boldsymbol{\varPhi} - \lambda (\boldsymbol{\varPhi}^{\dagger} \boldsymbol{\varPhi})^{2}, \qquad (4 \cdot 6)$$

$$\mathcal{L}_{\mathrm{M}} = -\sum_{i} f_{i} \overline{L}_{i} \mathcal{O} R_{i} - \sum_{I} f_{I} \overline{L}_{I} (i \tau_{2} \mathcal{O}^{*}) R_{I} + \mathrm{h.c.}, \qquad (4.7)$$

where the definitions of $F_{\mu\nu}^{a}$, D_{μ}^{μ} , D_{μ}^{ab} , \emptyset , $L_{i,I}$ and $R_{i,I}$ are given in Eqs. (2.20), (2.21), (2.22), (2.37a), (2.65), (2.81) and (2.84) respectively. It is understood that all the fields and parameters in this Lagrangian are bare quantities. Following the procedure described in § 2.3, we reexpress the fields in the basic Lagrangian in terms of bare physical fields (see Eq. (2.69)) and give a nonvanishing vacuum expectation value to the Higgs field. Then we choose the most convenient set of independent bare parameters. Introducing renormalization constants, we define renormalized fields and parameters and split the Lagrangian into tree parts and counter terms. At this stage we fix the gauge fixing term \mathcal{L}_{GF} so as to be convenient for the calculation of higher order corrections. Finally we determine the FP ghost term \mathcal{L}_{FP} by using the BRS transformation. The Feynman rules will be derived

in 4.2 from the tree Lagrangian and the counter terms will be tabulated in § 4.3.

The basic Lagrangian $(4 \cdot 1)$ reexpressed in terms of physical fields takes a lengthy expression which will be given in the following. Here the vacuum expectation value v of the Higgs field is introduced as a parameter. We shall discuss, in a moment, how v is determined in terms of the other parameters. The relations between the original bare fields $(W_{\mu}^{a}, \chi_{1}, \chi_{2})$ and the bare physical fields $(W_{\mu}^{\pm}, Z_{\mu}, A_{\mu}, \chi^{\pm})$ are given as in § 2.3 by

$$W_{\mu}^{\pm} = (W_{\mu}^{1} \mp i W_{\mu}^{2}) / \sqrt{2} , \qquad (4.8)$$

$$Z_{\mu} = (g W_{\mu}^{3} - g' W_{\mu}^{0}) / \sqrt{g^{2} + {g'}^{2}}, \qquad (4.9)$$

$$A_{\mu} = (g' W_{\mu}^{3} + g W_{\mu}^{0}) / \sqrt{g^{2} + {g'}^{2}}, \qquad (4.10)$$

$$\chi^{\pm} = (\chi_1 \mp i \chi_2) / \sqrt{2}$$
 (4.11)

The Lagrangian in terms of the bare physical fields takes the following expression:

$$\mathcal{L} = \mathcal{L}_{G} + \mathcal{L}_{F} + \mathcal{L}_{H} + \mathcal{L}_{M} + \mathcal{L}_{GF} + \tilde{\mathcal{L}}_{FP},$$

$$(4.12)$$

$$\mathcal{L}_{G} = -\frac{1}{2} F_{\mu\nu}^{+} F^{-\mu\nu} - \frac{1}{4} F_{\mu\nu}^{Z} F^{Z\mu\nu} - \frac{1}{4} F_{\mu\nu}^{A} F^{A\mu\nu}$$

$$+ \frac{ig}{\sqrt{g^{2} + g'^{2}}} (g^{\alpha\gamma} g^{\beta\delta} - g^{\alpha\delta} g^{\beta\gamma}) [g \{ (\partial_{\alpha} W_{\beta}^{+}) W_{7}^{-} Z_{\delta} + (\partial_{\alpha} W_{\beta}^{-}) Z_{7} W_{\delta}^{+}$$

$$+ (\partial_{\alpha} Z_{\beta}) W_{7}^{+} W_{\delta}^{-} \}$$

$$+ g' \{ (\partial_{\alpha} W_{\beta}^{+}) W_{7}^{-} A_{\delta} + (\partial_{\alpha} W_{\beta}^{-}) A_{7} W_{\delta}^{+} + (\partial_{\alpha} A_{\beta}) W_{7}^{+} W_{\delta}^{-} \}]$$

$$- \frac{g^{2}}{g^{2} + {g'}^{2}} [(g^{\alpha\beta} g^{\gamma\delta} - g^{\alpha\gamma} g^{\beta\delta}) W_{\alpha}^{+} W_{\beta}^{-} (g^{2} Z_{7} Z_{\delta} + {g'}^{2} A_{7} A_{\delta})$$

$$+ gg' (2g^{\alpha\beta} g^{\gamma\delta} - g^{\alpha\gamma} g^{\beta\delta}) W_{\alpha}^{+} W_{\beta}^{-} W_{\delta}^{-} ,$$

$$+ \frac{g^{2}}{2} (g^{\alpha\beta} g^{\gamma\delta} - g^{\alpha\gamma} g^{\beta\delta}) W_{\alpha}^{+} W_{\beta}^{-} W_{\delta}^{-} ,$$

$$(4.13)$$

where $F_{\mu\nu}^{\pm}$, $F_{\mu\nu}^{\mathbf{Z}}$ and $F_{\mu\nu}^{\mathbf{A}}$ are given by

$$F_{\mu\nu}^{\pm} = \partial_{\mu} W_{\nu}^{\pm} - \partial_{\nu} W_{\mu}^{\pm}, \text{ etc.}, \qquad (4.14)$$

$$\mathcal{L}_{F} = \mathcal{L}_{F}^{\mathrm{Kin}} + \mathcal{L}_{F}^{\mathrm{Int}}, \qquad (4.15)$$

$$\mathcal{L}_{\mathrm{F}}^{\mathrm{Kin}} = i \overline{\psi}_{n} \partial \psi_{n}, \quad (n = i, I)$$

$$(4.16)$$

$$\mathcal{L}_{\mathrm{F}}^{\mathrm{Int}} = \frac{g}{\sqrt{2}} \left(\overline{\psi}_{I} U_{Ii}^{\dagger} \partial^{\mu} \frac{1 - \gamma_{\mathfrak{s}}}{5} \psi_{i} W_{\mu}^{+} + \overline{\psi}_{i} U_{iI} \partial^{\mu} \frac{1 - \gamma_{\mathfrak{s}}}{2} \psi_{I} W_{\mu}^{-} \right) \\ + \frac{gg'}{\sqrt{g^{2} + g'^{2}}} Q_{n} \overline{\psi}_{n} \gamma^{\mu} \psi_{n} A_{\mu}$$

$$+ \frac{\sqrt{g^{2} + g'^{2}}}{2} \left\{ \overline{\psi}_{I} \gamma^{\mu} \left(\frac{1 - \gamma_{5}}{2} - \frac{2Q_{I}g'^{2}}{g^{2} + g'^{2}} \right) \psi_{I} - \overline{\psi}_{i} \gamma^{\mu} \left(\frac{1 - \gamma_{5}}{2} + \frac{2Q_{i}g'^{2}}{g^{2} + g'^{2}} \right) \psi_{i} \right\} Z_{\mu},$$

$$(4 \cdot 17)$$

$$\mathcal{L}_{H} = \mathcal{L}_{H}^{(2)} + \mathcal{L}_{H}^{(3)} + \mathcal{L}_{H}^{(4)} + \mathcal{L}_{V},$$

$$(4 \cdot 18)$$

$$\mathcal{L}_{H}^{(2)} = -\frac{1}{2} \phi \Box \phi - \frac{1}{2} \chi_{3} \Box \chi_{3} - \chi^{+} \Box \chi^{-}$$

$$+ \frac{1}{4} g^{2} v^{2} W_{\mu}^{+} W^{-\mu} + \frac{1}{2} \frac{1}{4} (g^{2} + g'^{2}) v^{2} Z_{\mu} Z^{\mu}$$

$$- \frac{1}{2} g v (W_{\mu}^{+} \partial^{\mu} \chi^{-} + W_{\mu}^{-} \partial^{\mu} \chi^{+}) - \frac{1}{2} \sqrt{g^{2} + g'^{2}} v Z_{\mu} \partial^{\mu} \chi_{3},$$

$$(4 \cdot 19)$$

$$\mathcal{L}_{\mathrm{H}}^{(3)} = \frac{1}{2} g W_{\mu}^{+} (\chi^{-} \ddot{\partial}^{\mu} \phi) + \frac{1}{2} g W_{\mu}^{-} (\chi^{+} \ddot{\partial}^{\mu} \phi) + \frac{i}{2} g W_{\mu}^{+} (\chi_{3} \ddot{\partial}^{\mu} \chi^{-}) - \frac{i}{2} g W_{\mu}^{-} (\chi_{3} \ddot{\partial}^{\mu} \chi^{+}) + \frac{1}{2} \sqrt{g^{2} + {g'}^{2}} Z_{\mu} (\chi_{3} \ddot{\partial}^{\mu} \phi) + \frac{i}{2} \frac{g^{2} - {g'}^{2}}{\sqrt{g^{2} + {g'}^{2}}} Z_{\mu} (\chi^{-} \ddot{\partial}^{\mu} \chi^{+}) + \frac{g g'}{\sqrt{g^{2} + {g'}^{2}}} A_{\mu} (\chi^{-} \ddot{\partial}^{\mu} \chi^{+}), (A \ddot{\partial}^{\mu} B \equiv A \cdot \partial^{\mu} B - \partial^{\mu} A \cdot B)$$
(4.20)

$$\mathcal{L}_{\mathrm{H}}^{(4)} = \frac{1}{4} g^{2} W_{\mu}^{+} W^{-\mu} (2v\phi + \phi\phi + 2\chi^{+}\chi^{-} + \chi_{s}\chi_{s})$$

$$+ \frac{1}{4} \frac{(g^{2} - g'^{2})^{2}}{g^{2} + g'^{2}} Z_{\mu} Z^{\mu} \chi^{+} \chi^{-} + \frac{1}{8} (g^{2} + g'^{2}) Z_{\mu} Z^{\mu} (2v\phi + \phi\phi + \chi_{s}\chi_{s})$$

$$+ \frac{g^{2} g'^{2}}{g^{2} + g'^{2}} A_{\mu} A^{\mu} \chi^{+} \chi^{-} + \frac{1}{2} \frac{gg'^{2}}{\sqrt{g^{2} + g'^{2}}} Z_{\mu} W^{+\mu} \chi^{-} (\chi_{s} + iv + i\phi)$$

$$+ \frac{1}{2} \frac{gg'^{2}}{\sqrt{g^{2} + g'^{2}}} Z_{\mu} W^{-\mu} \chi^{+} (\chi_{s} - iv - i\phi)$$

$$- \frac{1}{2} \frac{g^{2} g'}{\sqrt{g^{2} + g'^{2}}} A_{\mu} W^{+\mu} \chi^{-} (\chi_{s} + iv + i\phi)$$

$$- \frac{1}{2} \frac{g^{2} g'}{\sqrt{g^{2} + g'^{2}}} A_{\mu} W^{-\mu} \chi^{+} (\chi_{s} - iv - i\phi)$$

$$+ \frac{gg' (g^{2} - g'^{2})}{g^{2} + g'^{2}} Z_{\mu} A^{\mu} \chi^{+} \chi^{-} , \qquad (4.21)$$

$$\mathcal{L}_{\mathbf{v}} = \mu^2 \boldsymbol{\varPhi}^{\dagger} \boldsymbol{\varPhi} - \lambda (\boldsymbol{\varPhi}^{\dagger} \boldsymbol{\varPhi})^2$$
$$= v (\mu^2 - \lambda v^2) \boldsymbol{\phi} + (\mu^2 - \lambda v^2) \chi^- \chi^+$$

.

$$+ \frac{1}{2} (\mu^{2} - \lambda v^{2}) \chi_{3} \chi_{3} + \frac{1}{2} (\mu^{2} - 3\lambda v^{2}) \phi \phi$$

$$- 2v\lambda(\phi\chi^{-}\chi^{+}) - v\lambda(\phi\chi_{3}\chi_{3}) - v\lambda(\phi\phi\phi)$$

$$- \lambda(\chi^{-}\chi^{-}\chi^{+}\chi^{+}) - \lambda(\chi^{-}\chi^{+}\chi_{3}\chi_{3}) - \lambda(\phi\phi\chi^{-}\chi^{+})$$

$$- \frac{1}{4}\lambda(\chi_{3}\chi_{3}\chi_{3}\chi_{3}) - \frac{1}{4}\lambda(\phi\phi\phi\phi) - \frac{1}{2}\lambda(\phi\phi\chi_{3}\chi_{3}), \qquad (4\cdot22)$$

$$C_{M} = -\frac{f_{n}}{\sqrt{2}}v\overline{\psi}_{n}\psi_{n} - \frac{i}{2}\overline{\psi}_{I}U_{Ii}^{+} \left\{ (f_{i} - f_{I}) + (f_{i} + f_{I})\gamma_{3} \right\}\psi_{i} \cdot \chi^{+}$$

$$- \frac{i}{2}\overline{\psi}_{i}U_{iI} \left\{ (f_{I} - f_{i}) + (f_{I} + f_{i})\gamma_{3} \right\}\psi_{I} \cdot \chi^{-}$$

٦

 $-\frac{f_n}{\sqrt{2}}\overline{\psi}_n\psi_n\phi-\frac{if_n}{\sqrt{2}}\overline{\psi}_n\gamma_5\psi_n\chi_3. \qquad (4.23)$

The independent bare parameters present in the Lagrangian $(4 \cdot 1)$ are $g, g', \lambda, \mu^2, f_i$ and f_I . These are free parameters and are fixed only by confronting the theory with experiments. For this purpose it is convenient to choose, instead of the original parameters, another set of bare parameters corresponding to the following physical quantities: the unit of charge (e) and the masses of the W boson, Z boson, Higgs boson and fermions $(M_W, M_Z, m_{\phi}, m_i \text{ and } m_I)$. The new bare parameters corresponding to these physical quantities will be called the bare physical parameters and may be related to the original bare parameters $(g, g', \lambda, \mu^2, f_i, f_I)$ through the use of the vacuum expectation value v of the Higgs boson.

The vacuum expectation value v is not an independent parameter but should be determined in the theory as a function of g, g', λ , μ^2 , f_i and f_I . In practice, v is determined to be equal to the minimum point of the effective potential of ϕ to each order of perturbative expansion. The parameter v is introduced as if it were an independent parameter. The term linear in ϕ in the Lagrangian \mathcal{L}_{V} (Eq. (4.22)) represents the counter term for the tadpole. To each order of perturbation we require that the tadpole term in the effective action disappears, i.e., the coefficient of the term linear in ϕ vanishes. For later convenience we write

$$T = v \left(\mu^2 - \lambda v^2 \right). \tag{4.24}$$

Summing up, we introduce the redundant parameter v (or T) and add the condition that "tadpole = 0". The counter term corresponding to this condition is nothing but T. Allowing v to be an independent parameter, we have seven kinds of parameters g, g', v, λ , μ^2 , f_i and f_I , which corresponds six kinds of bare physical parameters e, M_W , M_Z , m_{ϕ} , m_i , m_I , and the tadpole counter term T. The relation between these two sets of parameters is readily given by Eqs. (2.68),

(2.70), (2.74), (2.89) and (4.24). Inverted relations for these parameters read

$$g = eM_{z}/\sqrt{M_{z}^{2} - M_{w}^{2}},$$

$$g' = eM_{z}/M_{w},$$

$$v = 2M_{w}\sqrt{M_{z}^{2} - M_{w}^{2}}/(eM_{z}),$$

$$\lambda = \frac{e^{2}M_{z}^{2}}{8M_{w}^{2}(M_{z}^{2} - M_{w}^{2})} \left(m_{\phi}^{2} - \frac{eTM_{z}}{M_{w}\sqrt{M_{z}^{2} - M_{w}^{2}}}\right),$$

$$\mu^{2} = m_{\phi}^{2}/2,$$

$$f_{i,I} = \sqrt{2} m_{i,I}/v,$$
(4.25)

where we have taken the sign convention $\operatorname{sgn} e = \operatorname{sgn} g = \operatorname{sgn} g'$.

Using Eq. (4.25), we can replace all the bare parameters g, g', v, λ , μ^2 , f_i and f_I in the Lagrangian by the bare physical parameters. For the later use in § 4.2 where the Feynman rule is developed, we tabulate useful relations in the following:

$$\frac{g}{\sqrt{g^{2}+g'^{2}}} = \frac{M_{W}}{M_{Z}}, \quad \sqrt{g^{2}+g'^{2}} = \frac{eM_{Z}^{2}}{M_{W}\sqrt{M_{Z}^{2}-M_{W}^{2}}}, \\
\frac{g^{2}-g'^{2}}{\sqrt{g^{2}+g'^{2}}} = \frac{e(2M_{W}^{2}-M_{Z}^{2})}{M_{Z}\sqrt{M_{Z}^{2}-M_{W}^{2}}}, \\
\mu^{2}-\lambda v^{2} = \frac{eTM_{Z}}{2M_{W}\sqrt{M_{Z}^{2}-M_{W}^{2}}}, \\
\mu^{2}-3\lambda v^{2} = -m_{\phi}^{2} + \frac{3eTM_{Z}}{2M_{W}\sqrt{M_{Z}^{2}-M_{W}^{2}}}, \\
v\lambda = \frac{eM_{Z}}{4M_{W}\sqrt{M_{Z}^{2}-M_{W}^{2}}} \left(m_{\phi}^{2} - \frac{eTM_{Z}}{M_{W}\sqrt{M_{Z}^{2}-M_{W}^{2}}}\right). \quad (4.26)$$

We now define renormalization constants and renormalized quantities in such a way that

$$W_{0}^{\pm} = Z_{W}^{1/2} W^{\pm} ,$$

$$\begin{pmatrix} Z_{0} \\ A_{0} \end{pmatrix} = (Z_{ij}^{1/2}) \begin{pmatrix} Z \\ A \end{pmatrix} ,$$

$$\psi_{0R,L}^{n} = (Z_{R,L}^{1/2})^{nm} \psi_{R,L}^{m} , \quad (Q_{n} = Q_{m})$$

$$\chi_{0}^{\pm} = Z_{\chi}^{1/2} \chi^{\pm} ,$$

$$\chi_{30} = Z_{\chi3}^{1/2} \chi^{\pm} ,$$

$$\phi_{0} = Z_{\phi}^{1/2} \phi ,$$

$$M_{W0}^{2} = M_{W}^{2} + \delta M_{W}^{2} ,$$

$$M_{Z_0}^2 = M_Z^2 + \delta M_Z^2,$$

$$m_{\phi_0}^2 = m_{\phi}^2 + \delta m_{\phi}^2,$$

$$m_{n_0} = m_n + \delta m_n,$$

$$e_0 = Ye,$$

(4.27)

where quantities with (without) suffix 0 denote bare (renormalized) quantities (so far the quantities without suffix 0 have been used to denote the bare one, but we follow the above notation henceforth). As already mentioned in §§ 3.2 and 3.3, (Z_{ij}) is a real matrix with independent four components and $Z_{R,L}$ are complex matrices. We need not deal with mixing between ψ_L and ψ_R^* since we assume the fermion number conservation.

We make loop expansions following our Lagrangian. In zeroth order of the loop expansion Z=1 and $\delta m=0$. We call the corresponding effective Lagrangian the tree Lagrangian. We split the Lagrangian into two parts: a part consisting of the zeroth order terms, i.e., the tree part, and a part consisting of higher-loop terms, i.e., the counter-term part. We choose the gauge fixing term \mathcal{L}_{GF} so that bilinear terms in the tree part, i.e., inverse propagator terms, take the simplest form; in fact, we adopt the following form to cancel the mixing terms among W and χ (Z and χ_s):

$$\mathcal{L}_{GF} = -\frac{1}{\alpha} (\partial^{\mu} W_{\mu}^{+} + \alpha M_{W} \chi^{+}) (\partial^{\mu} W_{\mu}^{-} + \alpha M_{W} \chi^{-}) -\frac{1}{2\alpha_{z}} (\partial^{\mu} Z_{\mu} + \alpha_{z} M_{z} \chi_{3})^{2} - \frac{1}{2\alpha_{A}} (\partial^{\mu} A_{\mu})^{2}, \qquad (4.28)$$

where α , α_z and α_A are the gauge parameters and, in particular, the case with $\alpha = \alpha_z = \alpha_A = 1$ is called the 't Hooft-Feynman gauge.

Introducing auxiliary fields *B*, the Lagrangian \mathcal{L}_{GF} of Eq. (4.28) may be written as

$$\mathcal{L}_{GF} = B^{+} (\partial^{\mu} W_{\mu}^{-} + \alpha M_{W} \chi^{-}) + B^{-} (\partial^{\mu} W_{\mu}^{+} + \alpha M_{W} \chi^{+}) + B^{Z} (\partial^{\mu} Z_{\mu} + \alpha_{Z} M_{Z} \chi_{3}) + B^{4} \partial^{\mu} A_{\mu} + \alpha B^{+} B^{-} + \frac{\alpha_{Z}}{2} B^{Z} B^{Z} + \frac{\alpha_{A}}{2} B^{A} B^{A}.$$
(4.29)

It is convenient to replace the renormalized quantities in Eq. (4.29) by corresponding bare ones for the purpose of finding the form of \mathcal{L}_{FP} :

$$\mathcal{L}_{GF} = B_{0}^{+} (\partial^{\mu} W_{0\mu}^{-} + \alpha_{0} M_{W0} \chi_{0}^{-}) + B^{-} (\partial^{\mu} W_{0\mu}^{+} + \alpha_{0} M_{W0} \chi_{0}^{+}) + B_{0}^{Z} (\partial^{\mu} Z_{0\mu}^{-} + \alpha_{Z0} M_{Z0} \chi_{30}) + B_{0}^{A} (\partial^{\mu} A_{0\mu}^{-} + \beta M_{Z0} \chi_{30}) + \text{terms quadratic in the } B_{0}^{*} \text{s},$$
 (4.30)

where the bare quantities B_0 , α_0 , α_{z0} and β are defined by

$$B_{0}^{\pm} = Z_{W}^{-1/2} B^{\pm} ,$$

$$\alpha_{0} = \alpha Z_{W}^{1/2} Z_{\chi}^{-1/2} M_{W} / M_{W0} ,$$

$$\alpha_{Z0} = \alpha_{Z} Z_{ZZ}^{1/2} Z_{\chi3}^{-1/2} M_{Z} / M_{Z0} ,$$

$$\beta = \alpha_{Z} Z_{AZ}^{1/2} Z_{\chi3}^{-1/2} M_{Z} / M_{Z0} ,$$

$$(4.31)$$

and the last terms in Eq. (4.30) are not written explicitly since they do not affect the FP-ghost terms according to the fact that the BRS transforms of the B's vanish. We can now easily find the corresponding FP-ghost Lagrangian \mathcal{L}_{FP} by using Eq. (2.41) resulting in

$$\mathcal{L}_{FP} = -\bar{c}^{+}{}_{0}\delta^{BRS}(\partial^{\mu}W_{0\mu}^{-} + \alpha_{0}M_{W0}\chi_{0}^{-}) - \bar{c}_{0}^{-}\delta^{BRS}(\partial^{\mu}W_{0\mu}^{+} + \alpha_{0}M_{W0}\chi_{0}^{+}) -\bar{c}_{0}^{z}\delta^{BRS}(\partial^{\mu}Z_{0\mu} + \alpha_{z0}M_{z0}\chi_{30}) - \bar{c}_{0}^{4}\delta^{BRS}(\partial^{\mu}A_{0\mu} + \beta M_{z0}\chi_{30}), \quad (4.32)$$

where the rule of the BRS transformation is given in Eqs. (2.39) and (3.229). To find the explicit form of \mathcal{L}_{FP} , it is more convenient to reexpress Eqs. (2.39) and (3.229) in terms of the physical fields:

$$\delta W_{\mu}^{\pm} = \partial_{\mu} c^{\pm} \pm \frac{ig}{\sqrt{g^{2} + g'^{2}}} [W_{\mu}^{\pm} (gc^{Z} + g'c^{A}) - (gZ_{\mu} + g'A_{\mu})c^{\pm}],$$

$$\delta Z_{\mu} = -\frac{ig^{2}}{\sqrt{g^{2} + g'^{2}}} (W_{\mu}^{+}c^{-} - W_{\mu}^{-}c^{+}) + \partial_{\mu}c^{Z},$$

$$\delta A_{\mu} = -\frac{igg'}{\sqrt{g^{2} + g'^{2}}} (W_{\mu}^{+}c^{-} - W_{\mu}^{-}c^{+}) + \partial_{\mu}c^{A},$$

$$\delta \phi = -\frac{g}{2} (\chi^{+}c^{-} + \chi^{-}c^{+}) - \frac{\sqrt{g^{2} + g'^{2}}}{2} \chi_{s}c^{Z},$$

$$\delta \chi^{\pm} = \frac{g}{2} [(v + \phi)c^{\pm} \mp \chi_{s}c^{\pm}] \pm \frac{i}{2\sqrt{g^{2} + g'^{2}}} \chi^{\pm} [(g^{2} - g'^{2})c^{Z} + 2gg'c^{A}],$$

$$\delta \chi_{s} = \frac{\sqrt{g^{2} + g'^{2}}}{2} (v + \phi)c^{Z} - \frac{ig}{2} (\chi^{+}c^{-} - \chi^{-}c^{+}),$$

(4.33)

where the superfix BRS is neglected for δ^{BRS} and we have omitted the suffix 0 for the bare quantities and the fields c^{z} and c^{A} are constructed from c^{s} and c^{0} just like the fields Z and A:

$$c^{z} = (gc^{3} - g'c^{0}) / \sqrt{g^{2} + g'^{2}},$$

$$c^{4} = (g'c^{3} + gc^{0}) / \sqrt{g^{2} + g'^{2}}.$$
(4.34)

We finally find

$$\mathcal{L}_{\rm FP} = \mathcal{L}_{\rm FP}^{(2)} + \mathcal{L}_{\rm FP}^{(3)}, \qquad (4.35)$$

Downloaded from https://academic.oup.com/ptps/article-abstract/doi/10.1143/PTPS.73.1/1848359 by guest on 06 January 2019

where

$$\mathcal{L}_{\rm FP}^{(2)} = -\bar{c}^{+} (\Box + \alpha M_{\rm w}^{2}) c^{-} - \bar{c}^{-} (\Box + \alpha M_{\rm w}^{2}) c^{+} \\ -\bar{c}^{z} (\Box + \alpha_{z} M_{z}^{2}) c^{z} - \bar{c}^{A} \Box c^{A} - \bar{c}^{A} (\beta M_{z}^{2}) c^{z}, \qquad (4\cdot36)$$

$$\mathcal{L}_{\rm FP}^{(3)} = \frac{ig^{2}}{\sqrt{g^{2} + g'^{2}}} W_{\mu}^{+} [\partial^{\mu} \bar{c}^{-} \cdot c^{z} - \partial^{\mu} \bar{c}^{z} \cdot c^{-}] + ie W_{\mu}^{+} [\partial^{\mu} \bar{c}^{-} \cdot c^{A} - \partial^{\mu} \bar{c}^{A} \cdot c^{-}]$$

$$- \frac{ig^{2}}{\sqrt{g^{2} + g'^{2}}} W_{\mu}^{-} [\partial^{\mu} \bar{c}^{+} \cdot c^{z} - \partial^{\mu} \bar{c}^{z} \cdot c^{+}] - ie W_{\mu}^{-} [\partial^{\mu} \bar{c}^{+} \cdot c^{A} - \partial^{\mu} \bar{c}^{A} \cdot c^{+}]$$

$$+ \frac{ig^{2}}{\sqrt{g^{2} + g'^{2}}} Z_{\mu} [\partial^{\mu} \bar{c}^{+} \cdot c^{-} - \partial^{\mu} \bar{c}^{-} \cdot c^{+}] + ie A_{\mu} [\partial^{\mu} \bar{c}^{+} \cdot c^{-} - \partial^{\mu} \bar{c}^{-} \cdot c^{+}]$$

$$+ \frac{ig^{2}}{\sqrt{g^{2} + g'^{2}}} Z_{\mu} [\partial^{\mu} \bar{c}^{+} \cdot c^{-} - \partial^{\mu} \bar{c}^{-} c^{+}] + ie A_{\mu} [\partial^{\mu} \bar{c}^{+} \cdot c^{-} - \partial^{\mu} \bar{c}^{-} \cdot c^{+}]$$

$$+ \frac{ig^{2}}{\sqrt{g^{2} + g'^{2}}} Z_{\mu} [\partial^{\mu} \bar{c}^{+} \cdot c^{-} - \partial^{\mu} \bar{c}^{-} c^{+}] + ie A_{\mu} [\partial^{\mu} \bar{c}^{+} \cdot c^{-} - \partial^{\mu} \bar{c}^{-} \cdot c^{+}]$$

$$+ \frac{ig^{2}}{\sqrt{g^{2} + g'^{2}}} Z_{\mu} [\partial^{\mu} \bar{c}^{+} c^{-} - \partial^{\mu} \bar{c}^{-} c^{+}] + ie A_{\mu} [\partial^{\mu} \bar{c}^{+} c^{-} - \partial^{\mu} \bar{c}^{-} c^{+}]$$

$$+ i\chi^{+} \left[\frac{-\alpha M_{w} (-g'^{2} + g^{2})}{2\sqrt{g^{2} + g'^{2}}} \bar{c}^{-} c^{z} - \alpha M_{w} e \bar{c}^{-} c^{A}$$

$$+ \frac{\beta}{2} M_{z} g \bar{c}^{z} c^{-} + \frac{\alpha}{2} M_{z} g \bar{c}^{z} c^{-} \right]$$

$$+ i\chi^{-} \left[\frac{\alpha M_{w} (-g'^{2} + g^{2})}{2\sqrt{g^{2} + g'^{2}}} \bar{c}^{+} c^{z} + \alpha M_{w} e \bar{c}^{+} c^{A}$$

$$- \frac{\beta}{2} M_{z} g \bar{c}^{A} c^{+} - \frac{\alpha}{2} M_{z} g \bar{c}^{z} c^{+} \right]$$

$$+ \frac{i\alpha}{2} M_{w} g \chi_{s} [-\bar{c}^{+} c^{-} + \bar{c}^{-} c^{+}] - \frac{\alpha}{2} M_{w} g \phi [\bar{c}^{+} c^{-} + \bar{c}^{-} c^{+}]$$

$$- \frac{\alpha^{z}}{2} M_{z} \sqrt{g^{2} + g'^{2}} \phi \bar{c}^{z} c^{z} - \frac{\beta}{2} M_{z} \sqrt{g^{2} + g'^{2}} \phi \bar{c}^{A} c^{z} . \qquad (4\cdot37)$$

The renormalization constants and renormalized fields for the FP ghosts are defined by

$$c_0^{\pm} = \widetilde{Z}_3 c^{\pm}, \quad \begin{pmatrix} c_0^Z \\ c_0^A \end{pmatrix} = (\widetilde{Z}_{ij}) \begin{pmatrix} c^Z \\ c^A \end{pmatrix},$$

$$\overline{c}_0^{\pm} = \overline{c}^{\pm}, \quad \overline{c}_0^Z = \overline{c}^Z, \quad \overline{c}_0^A = \overline{c}^A.$$
(4.38)

In Eq. (4.38) we have introduced the renormalization constants only for the c's and kept the \bar{c} 's unmodified. This is legitimate because the ghost fields appear only in internal loops in Feynman diagrams. The decomposition of the Lagrangian $\mathcal{L}_{GF} + \mathcal{L}_{FP}$ into the tree part and the counter terms may be made in the same way as before.

Summing up, in this section we have rewritten the original Lagrangian in terms of the bare physical quantities and then have decomposed the resulting Lagrangian into the tree and counter terms. The procedure may be summarized as follows,

where ψ , α and Z represent generically fields, gauge parameters and field renormalization constants respectively.

For the use in the next section, we finally present the bilinear terms $\mathcal{L}_{\text{tree}}^{(2)}$ in $\mathcal{L}_{\text{tree}}$, i.e., the inverse propagator part,

$$\mathcal{L}_{\text{tree}}^{(2)} = W_{\mu}^{+} \left[(g^{\mu\nu} \Box - \partial^{\mu} \partial^{\nu}) + \frac{1}{\alpha} \partial^{\mu} \partial^{\nu} + M_{W}^{2} g^{\mu\nu} \right] W_{\nu}^{-} + \frac{1}{2} Z_{\mu} \left[(g^{\mu\nu} \Box - \partial^{\mu} \partial^{\nu}) + \frac{1}{\alpha_{Z}} \partial^{\mu} \partial^{\nu} + M_{Z}^{2} g^{\mu\nu} \right] Z_{\nu} + \frac{1}{2} A_{\mu} \left[(g^{\mu\nu} \Box - \partial^{\mu} \partial^{\nu}) + \frac{1}{\alpha_{A}} \partial^{\mu} \partial^{\nu} \right] A_{\nu} + \chi^{+} (-\Box - \alpha M_{W}^{2}) \chi^{-} + \frac{1}{2} \phi (-\Box - m_{\phi}^{2}) \phi + \frac{1}{2} \chi_{3} (-\Box - \alpha_{Z} M_{Z}^{2}) \chi_{3} + \overline{\psi}_{i} (i \gamma^{\mu} \partial_{\mu} - m_{i}) \psi_{i} + \overline{\psi}_{I} (i \gamma^{\mu} \partial_{\mu} - m_{I}) \psi_{I} - \overline{c}^{+} (\Box + \alpha M_{W}^{2}) c^{-} - \overline{c}^{-} (\Box + \alpha M_{W}^{2}) c^{+} - \overline{c}^{Z} (\Box + \alpha_{Z} M_{Z}^{2}) c^{Z} - \overline{c}^{A} \Box c^{A} .$$

$$(4 \cdot 40)$$

§4.2 Feynman rules

In this section we present a full list of Feynman rules corresponding to the Lagrangian (4.12). The S-matrix is given in terms of the interaction Lagrangian $\mathcal{L}_{I}(x)$ by

$$S = \mathrm{T} \exp\left[i \int d^4x \mathcal{L}_{\mathrm{I}}(x)\right], \qquad (4.41)$$

where T represents an ordinary time ordering. The S-matrix element S_{fi} for specific initial (i) and final (f) states is related to the transition amplitude T_{fi} through the following equation:

$$S_{fi} = 1 + i (2\pi)^4 \delta^4 (\sum_j p_j) T_{fi} . \qquad (4 \cdot 42)$$

By using our Feynman rule which will be given in this section, one directly obtains T_{fi} defined in Eq. (4.42) for the process $i \rightarrow f$.

We define the propagators by

108
Chap. 4 On-Shell Renormalization in Electroweak Theory. II 109

$$\Delta_{ij}(k) = i \int d^4x \, e^{-ik \cdot x} \langle 0 | \mathrm{T}[\phi_i(x)\phi_j(0)] | 0 \rangle , \qquad (4 \cdot 43)$$

where ϕ_i generically represents all the relevant fields in the following. Propagators in the Feynman diagram will be denoted by wavy, dashed, full and dotted lines for gauge bosons, Higgs particles, fermions and ghosts respectively (see § 4.2.1). In some cases we put an arrow on the propagator line in order to specify the direction of the flow of a certain quantum number: for the W^+ boson and χ^+ particle the arrow represents the flow of the positive charge, for the fermion that of the fermion number and for the ghost that of the ghost number.

The vertex functions for the interactions of gauge-bosons, fermions, Higgs fields and FP ghosts are derived from \mathcal{L}_{I} (note that we do not use $i\mathcal{L}_{I}$). All the momenta of particles associated with vertices are taken to flow in. (The case of the FP-ghost is exceptional. The momentum of the FP ghost is taken to flow along the direction of the ghost number.)

The loop integration is performed by the rule

$$\int \frac{d^{p}k}{(2\pi)^{p}i},\qquad (4\cdot44)$$

where we have in mind the dimensional regularization with D the space-time dimension. To the fermion and FP-ghost loops we associate the factor (-1) to account for the anticommutation of fermion and ghost fields.

It should be stressed here that our Feynman rule is designed to minimize the number of times that imaginary unit *i* appears. Thus an extra factor *i* has been introduced in the definition of Δ_{ij} in Eq. (4.43) and also the vertices are defined through \mathcal{L}_{I} instead of $i\mathcal{L}_{I}$. For example, the fermion tree propagator is 1/(m-p) with fermion mass *m* and the photon-fermion vertex is $e\gamma_{\mu}$.

Let us explain the notations used in our Feynman rules by taking some typical examples. The propagator of the W^+ field is defined by

$$\mathcal{A}_{\mu\nu}^{+}(q) = i \int d^{4}x \, e^{-iq \cdot x} \langle 0 | \mathrm{T} [W_{\mu}^{+}(x) \, W_{\nu}^{-}(0)] | 0 \rangle. \qquad (4.45)$$

The corresponding diagram together with the tree propagator is given at the top of the list in § 4.2.1. There the arrow on the wavy line represents the flow of a positive charge so that W^+ is created at the end point denoted by W_{μ}^- . The W-W-Z proper vertex function is depicted in the first diagram of § 4.2.2(1). It should be noted that the vertex function is defined as an amputated 3- or 4-point Green function and has no propagator as external lines. The letter W^+ (or W^-) attached to the vertex diagram shows that the W^+ (or W^-) particle comes in the vertex. The same remark also applies to the gauge 4-point functions as depicted in § 4.2.2(2).

To see the virtue of our Feynman rule we present two examples of the calculation. The first example deals with the W-boson contribution to the one-loop self-energy part of Z-bosons, $\Pi^{z}_{\mu\nu}(q)$. The diagram to be considered is shown in Fig. 4.1.

$$z_{\mu} \longrightarrow z_{\nu}$$
 Fig. 4.1. W-boson contribution to the one-loop self-energy part of Z boson.

Following our Feynman rule, we have

$$\begin{split} \Pi^{z}_{\mu\nu}(q) &= \int \frac{d^{D}k}{(2\pi)^{P}i} \frac{eM_{W}}{\sqrt{M_{Z}^{2} - M_{W}^{2}}} \left\{ (k - 2q)^{\alpha}g_{\mu}^{\ \beta} + (q - 2k)_{\mu}g^{\beta\alpha} + (k + q)^{\beta}g_{\mu}^{\ \alpha} \right\} \\ &\times \frac{1}{(k^{2} - M_{W}^{2})\left((k - q)^{2} - M_{W}^{2}\right)} \\ &\times \frac{eM_{W}}{\sqrt{M_{Z}^{2} - M_{W}^{2}}} \left\{ (k + q)_{\beta}g_{\nu\alpha} + (q - 2k)_{\nu}g_{\alpha\beta} + (k - 2q)_{\alpha}g_{\beta\nu} \right\}, \quad (4 \cdot 46) \end{split}$$

where we have adopted the Feynman gauge. Obviously we have the least number of the *i*'s in Eq.(4.46) compared with the expressions for $\Pi^{Z}_{\mu\nu}(q)$ in ordinary Feynman rules. Our $\Pi^{Z}_{\mu\nu}(q)$ is normalized by the following perturbative relation,

$$\Delta_{\mu\nu}^{Z}(q) = \Delta_{\mu\nu}^{Z0}(q) + \Delta_{\mu\alpha}^{Z0}(q) \Pi^{Z,\alpha\beta}(q) \Delta_{\beta\nu}^{Z0}(q) + \cdots, \qquad (4.47)$$

where $\Delta_{\mu\nu}^{Z}$ is an unrenormalized full propagator of the Z-boson and $\Delta_{\mu\nu}^{Z0}$ is the tree propagator appearing in the Feynman rule § 4.2.1. Equation (4.46) can be calculated explicitly in a straightforward manner. The result may be found in Appendix B.

The next example is one of the correction diagrams in the one-loop order for ν_{μ} -e scattering. The diagram we consider is shown in Fig. 4.2. By the use of our Feynman rule, we obtain its contribution to the scattering amplitude $A(p_e', p_{\nu}'; p_e, p_{\nu})$,

$$\begin{split} A(p_{e}', p_{\nu}'; p_{e}, p_{\nu}) \\ &= \int \frac{d^{p}k}{(2\pi)^{p}i} \overline{u}_{\nu}(p_{\nu}') \left(\frac{eM_{Z}^{2}}{2M_{W}\sqrt{M_{Z}^{2}-M_{W}^{2}}}\right)^{2} \gamma_{\alpha} \frac{1-\gamma_{5}}{2} \frac{1}{k-p_{\nu}} \gamma_{\beta} \frac{1-\gamma_{5}}{2} u_{\nu}(p_{\nu}) \\ &\times \overline{u}_{e}(p_{e}') \left(\frac{eM_{Z}^{2}}{2M_{W}\sqrt{M_{Z}^{2}-M_{W}^{2}}}\right)^{2} \gamma^{\alpha} \left(\frac{1-\gamma_{5}}{2}-2\frac{M_{Z}^{2}-M_{W}^{2}}{M_{Z}^{2}}\right) \frac{1}{m_{e}-p_{e}-k} \\ &\times \gamma^{\beta} \left(\frac{1-\gamma_{5}}{2}-2\frac{M_{Z}^{2}-M_{W}^{2}}{M_{Z}^{2}}\right) u_{e}(p_{e}) \frac{1}{k^{2}-M_{Z}^{2}} \frac{1}{(k-q)^{2}-M_{Z}^{2}}, \quad (4.48) \end{split}$$

where we have adopted the Feynman gauge, and p_{ν} and p_{e} (p_{ν}' and p_{e}') are

initial (final) neutrino and electron momenta respectively with $q = p_e' - p_e$. Here again we observe that the imaginary unit *i* appears only once in Eq. (4.48). Note that the amplitude *A* coincides exactly with T_{fi} defined in Eq. (4.42). The loop integral in Eq. (4.48) can be performed by applying the Feynman parametrization. The resulting expression may be found in Appendix B. If one takes into account only a few leading terms in powers of $1/M_z$, the expression for the amplitude *A* is further simplified and the Feynman parameter integrals are trivially done. The final form for *A* may be easily obtained by using the formula given in Appendix C.

4.2.1 Propagators

w _µ ⁺ w _v [−]	$\frac{1}{k^{\hat{\imath}}+i\varepsilon-M_{\textit{W}}^{2}}\left[g^{_{\textit{\mu}\textit{\nu}}}-(1-\alpha)\frac{k^{_{\textit{\mu}}}k^{_{\textit{\nu}}}}{k^{^{2}}+i\varepsilon-\alpha M_{\textit{W}}^{^{2}}}\right]$
^z _μ ^z _ν	$\frac{1}{k^2 + i\varepsilon - M_z^2} \left[g^{\mu\nu} - (1 - \alpha_z) \frac{k^{\mu}k^{\nu}}{k^2 + i\varepsilon - \alpha_z M_z^2} \right]$
Α _μ , Α _ν	$\frac{g^{\mu\nu}}{k^2+i\varepsilon}-(1-\alpha_A)\frac{k^{\mu}k^{\nu}}{(k^2+i\varepsilon)^2}$
x ⁺ x ⁻	$rac{1}{lpha M_w^2 - k^2 - i arepsilon}$
X ₃ X ₃	$rac{1}{lpha_{z}M_{z}^{2}-k^{2}-iarepsilon}$
ф ф ••	$\frac{1}{m_{\phi}^2-k^2-i\varepsilon}$
ψ,i ψ,j •€ ↓ ψ,j	$\frac{1}{m-k-i\varepsilon}\Big _{ij} = \frac{m+k}{m^2-k^2-i\varepsilon}\Big _{ij}$
c ⁺ ⊕ ¯c¯ ⊷∢	$rac{1}{lpha M_w^2 - k^2 - i arepsilon}$
c ⊖ c + c + c + c + c + c + c + c + c +	$\frac{1}{\alpha M_w^2 - k^2 - i\varepsilon}$
c ^Z	$rac{1}{lpha_{z}M_{z}^{2}-k^{2}-iarepsilon}$
c ^A c ^A	$\frac{1}{-k^2-i\varepsilon}$

The arrow on the propagator line represents the direction of the flow of a certain quantum number; the charge for W^{\pm} , χ^{\pm} , the fermion number for ψ , the ghost number for c, \overline{c} . The symbol \oplus or \ominus beside charged ghost propagators represents the sign of the charge carried by the arrow.

- 4.2.2 Vertices in \mathcal{L}_{G}
 - (1) Gauge-boson three-vertices

$$\underset{q}{\overset{k}{\downarrow}^{\mathbf{Z}_{\alpha}}} \underbrace{eM_{W}}_{q} \underbrace{(k-p)^{r}g^{\alpha\beta}}_{q} + (p-q)^{\alpha}g^{\beta\gamma} + (q-k)^{\beta}g^{r\alpha} \underbrace{eM_{W}}_{\sqrt{M_{Z}^{2}-M_{W}^{2}}} \{(k-p)^{r}g^{\alpha\beta} + (p-q)^{\alpha}g^{\beta\gamma} + (q-k)^{\beta}g^{r\alpha} \underbrace{eM_{W}}_{\sqrt{M_{Z}^{2}-M_{W}^{2}}} \Big\}$$

$$W_{\beta} p q^{W_{\gamma}^{+}} e \{ (k-p)^{r} g^{\alpha\beta} + (p-q)^{\alpha} g^{\beta\gamma} + (q-k)^{\beta} g^{\gamma\alpha} \}$$

(2) Gauge-boson four-vertices



All the momenta of particles associated with vertices are taken to flow in.

4.2.3 Vertices in \mathcal{L}_{F}

$$\overline{\psi}_{\mathrm{I}} \qquad \frac{eM_{Z}}{2\sqrt{2(M_{Z}^{2}-M_{W}^{2})}}U_{Ii}^{\dagger}\gamma_{\alpha}(1-\gamma_{5})$$





$$\overline{\psi}_{I} = \frac{eM_{Z}^{2}}{2M_{W}\sqrt{M_{Z}^{2}-M_{W}^{2}}} \gamma_{\alpha} \left(\frac{1}{2}-2Q_{I}\frac{M_{Z}^{2}-M_{W}^{2}}{M_{Z}^{2}}-\frac{1}{2}\gamma_{5}\right)$$

$$\overline{\psi}_{i} = \frac{eM_{z}^{2}}{2M_{w}\sqrt{M_{z}^{2}-M_{w}^{2}}} \gamma_{\alpha} \left(-\frac{1}{2}-2Q_{s}\frac{M_{z}^{2}-M_{w}^{2}}{M_{z}^{2}}+\frac{1}{2}\gamma_{s}\right)$$

 $Q_{i,I}$ represent electromagnetic charges of fields $\psi_{i,I}$ in units of e.

- 4.2.4 Vertices in \mathcal{L}_{H}
 - (1) Gauge-boson-Higgs three-vertices



(2) Gauge-boson-Higgs four-vertices



(3) Higgs three-vertices



(4) Higgs four-vertices



4.2.5 Vertices in \mathcal{L}_{M}

$$\overline{\psi}_{I} = \frac{-ieM_{Z}}{2M_{W}\sqrt{2(M_{Z}^{2}-M_{W}^{2})}}U_{Ii}^{\dagger}\{(m_{i}-m_{I})+(m_{i}+m_{I})\gamma_{5}\}$$

$$\overline{\Psi}_{i} = \frac{-ieM_{Z}}{2M_{W}\sqrt{2(M_{Z}^{2}-M_{W}^{2})}}U_{iI}\{(m_{I}-m_{i})+(m_{I}+m_{i})\gamma_{5}\}$$

$$\overline{\psi}_n$$
 ψ_n $\frac{-em_nM_Z}{2M_W\sqrt{M_Z^2-M_W^2}}$

$$\overline{\psi}_{\mathrm{I}} \qquad \frac{-iem_{\mathrm{I}}M_{\mathrm{Z}}}{2M_{\mathrm{W}}\sqrt{M_{\mathrm{Z}}^{2}-M_{\mathrm{W}}^{2}}}\gamma_{5}$$

$$\overline{\Psi_{i}} \qquad \frac{iem_{i}M_{Z}}{2M_{W}\sqrt{M_{Z}^{2}-M_{W}^{2}}}\gamma_{z}$$

- 4.2.6 Vertices in \mathcal{L}_{FP}
 - (1) Gauge-boson-ghost three-vertices













(2) Higgs-ghost three-vertices



§ 4.3 Renormalization conditions and counter terms

In this section we tabulate the on-shell renormalization conditions and counter terms for the tadpole and for all two-, three- and four-point functions. Every counter term is presented at the place where the related renormalization condition appears.

Renormalized *n*-point functions are coefficients of the renormalized fields defined in Eq. (4.27) in the effective action. They are given by the sum of one-particle irreducible Feynman diagrams including the contribution of counter terms. In the case of two-point functions they are nothing but the inverse propagators.

As explained in § 4.1, we split the bare Lagrangian into two parts: a part consisting of the zeroth order terms of the loop expansion (the tree part) and the remaining part consisting of higher-loop terms (the counter-term part). According to this separation, we generate the loop expansion of the effective action systematically. At each order of the loop expansion we determine the counter terms so that the amplitudes satisfy the renormalization conditions.

This process is schematically shown as follows. Consider a renormalized two-point function Γ_i . In the one-loop approximation we have two kinds of diagrams for $\Gamma_i^{(1)}$,

$$\Gamma_{i}^{(1)} = - + + \frac{\mathbf{x}}{\mathbf{c}^{(1)}}, \qquad (4 \cdot 49)$$

where the second term $C^{(1)}$ on the right-hand side represents the one-loop part of the counter terms. We determine $C^{(1)}$ by the renormalization condition for Γ_i . With $C^{(1)}$ thus determined the contribution from counter terms cancels out the divergence included in the first term in Eq. (4.49) and we obtain the finite $\Gamma_i^{(1)}$. In the two-loop approximation there are three kinds of diagrams,

where $C^{(2)}$ is the two-loop part of the counter terms. The inner divergences of the first term are cancelled out by the second term and there remains the overall divergence. We determine $C^{(2)}$ to satisfy the renormalization condition for Γ_i and the third term in Eq. (4.50) cancels out the above-mentioned overall divergence to give finite $\Gamma_i^{(2)}$. We proceed to the N-loop approximation.

The N-loop part of the counter terms $C^{(N)}$ is determined by the renormalization condition and it cancels out the overall divergence in the N-loop diagrams. In this way we obtain the loop expansion of Γ_t in which every term is finite.

In the following tables we first present the renormalized *n*-point function and write down the corresponding counter terms explicitly, where the N-loop parts of these counter terms should be regarded as the last term in Eq. (4.51); then we give the on-shell renormalization conditions and the renormalization constants which are determined by these conditions. All the renormalization constants except Y are determined by the renormalization conditions for the tadpole and for the two-point functions. The constant Y is determined by the renormalization condition for the photon vertex of any of the charged particles. For all other vertices (three- and four-point functions) no more independent renormalization conditions and counter terms appear. They are already fixed by the foregoing renormalization conditions. The explicit expressions for the renormalization constants are given in Appendix E.

For mixing channels the renormalization conditions are imposed following the formulas discussed in detail in §§ 3.2 and 3.3. The renormalization conditions for diagonal elements are the same as the usual ones and those for off-diagonal elements require that amplitudes vanish at the on-shell momentum of each particle (§ 4.3.2(1)(2)).

It should be mentioned that no counter terms come out from the gauge fixing Lagrangian \mathcal{L}_{GF} as discussed in §4.1.

In what follows we arrange the tables of renormalization conditions and counter terms for amplitudes, from which their tree parts are removed, following the order of presentation of Feynman rules in the last section. On the next page we presnt abbreviated notations used in the following tables. Notations used in tables of counter terms:

$$G_{w} \equiv \sqrt{1 + \frac{\delta M_{w}^{2}}{M_{w}^{2}}},$$

$$G_{z} \equiv \sqrt{1 + \frac{\delta M_{z}^{2}}{M_{z}^{2}}},$$

$$G_{\phi} \equiv 1 + \frac{\delta m_{\phi}^{2}}{m_{\phi}^{2}},$$

$$G_{mj} \equiv 1 + \frac{\delta m_{j}^{2}}{m_{j}^{2}},$$

$$H \equiv \sqrt{1 + \frac{\delta M_{z}^{2} - \delta M_{w}^{2}}{M_{z}^{2} - M_{w}^{2}}};$$

$$G_{1} \equiv G_{W}/H,$$

$$G_{2} \equiv G_{Z}/H,$$

$$G_{3} \equiv G_{Z}/G_{W},$$

$$G_{4} \equiv \left\{ 1 + \frac{2\delta M_{W}^{2} - \delta M_{Z}^{2}}{2M_{W}^{2} - M_{Z}^{2}} \right\} \left| (G_{W}H);$$

$$\begin{split} Z_{\phi}^{1/2} &\equiv \frac{Z_{L}^{1/2} + Z_{B}^{1/2}}{2} + \frac{Z_{B}^{1/2} - Z_{L}^{1/2}}{2} \gamma^{5} , \\ Z_{\phi}^{1/2} &\equiv \frac{Z_{L}^{1/2\dagger} + Z_{B}^{1/2\dagger}}{2} + \frac{Z_{L}^{1/2\dagger} - Z_{B}^{1/2\dagger}}{2} \gamma^{5} . \end{split}$$

4.3.1 Renormalization condition and counter term for tadpole

- 4.3.2 Renormalization conditions and counter terms for two-point functions (proper self-energies) of physical particles
- (1) Gauge-bosons

$$\begin{array}{l} \overset{W^{+}_{\mu}}{\longrightarrow} & \overset{W^{-}_{\nu}}{\longrightarrow} & = \left(g_{\mu\nu} - \frac{k_{\mu}k_{\nu}}{k^{2}}\right) A^{W}(k^{2}) + \frac{k_{\mu}k_{\nu}}{k^{2}} B^{W}(k^{2}) , \\ & \overset{W^{+}_{\mu}}{\longrightarrow} & \overset{W^{-}_{\nu}}{\longrightarrow} & = \left(g_{\mu\nu} - \frac{k_{\mu}k_{\nu}}{k^{2}}\right) \left[\delta M_{W}^{2} \cdot Z_{W} + (M_{W}^{2} - k^{2}) Z_{W}\right] \\ & \quad + \frac{k_{\mu}k_{\nu}}{k^{2}} \left[\delta M_{W}^{2} \cdot Z_{W} + M_{W}^{2} \cdot Z_{W}\right]. \end{array}$$

Renormalization conditions

 $A^{W}(M_{W}^{2}) = 0$ }

Renormalization constants

9

$$\overset{\mathbf{Z}_{\mu}}{\longrightarrow} \overset{\mathbf{A}_{\nu}}{\longrightarrow} = \left(g_{\mu\nu} - \frac{k_{\mu}k_{\nu}}{k^{2}}\right)A^{ZA}(k^{2}) + \frac{k_{\mu}k_{\nu}}{k^{2}}B^{ZA}(k^{2}),$$

$$\overset{A_{\nu}}{\longrightarrow} = \left(g_{\mu\nu} - \frac{k_{\mu}k_{\nu}}{k^2}\right) \left[\left(M_Z^2 + \delta M_Z^2\right) \left(Z_{ZZ}^{1/2} Z_{ZA}^{1/2}\right)\right]$$

$$-k^{2} \{ (Z_{ZZ}^{1/2} Z_{ZA}^{1/2}) + (Z_{AZ}^{1/2}) (Z_{AA}^{1/2}) \}] \\ + \frac{k_{\mu}k_{\nu}}{k^{2}} [(M_{Z}^{2} + \delta M_{Z}^{2}) (Z_{ZZ}^{1/2} Z_{ZA}^{1/2})].$$

Renormalization conditions

i

Renormalization constants

$$\left. \begin{array}{l} A^{Z}(M_{Z}^{2}) = 0 , \quad A^{Z'}(M_{Z}^{2}) = 0 ; \\ A^{A}(0) = 0 , \qquad A^{A'}(0) = 0 ; \\ A^{ZA}(M_{Z}^{2}) = 0 , \quad A^{ZA}(0) = 0 . \end{array} \right\} \quad \Rightarrow \quad Z_{ZZ}^{1/2}, Z_{ZA}^{1/2}, Z_{AZ}^{1/2}, \delta M_{Z}^{2} .$$

In the A-Z mixing channel there are six renormalization conditions for five constants. In fact, as discussed in § 3.3, six conditions are not independent. The Ward-Takahashi identity representing the unbroken U(1) symmetry guarantees the existence of the massless pole in the A-Z channel. Two equations $A^{4}(0) = 0$ and $A^{Z4}(0) = 0$ gives an identical condition. This situation corresponds to that in the QED case: $\Pi^{\mu\nu}(k^2) = (g^{\mu\nu}k^2 - k^{\mu}k^{\nu}) \times \Pi(k^2)$. (2) Fermions

$$\stackrel{i}{\longleftarrow} \stackrel{j}{\longleftarrow} = K_1^{ij}(k^2) \mathbf{1} + K_5^{ij}(k^2) \gamma^5 + K_{\gamma}^{ij}(k^2) k + K_{\delta\gamma}^{ij}(k^2) k \gamma^5,$$

$$\begin{split} \stackrel{i}{\longrightarrow} & \stackrel{j}{\longrightarrow} \\ = \mathbf{1} \bigg[-\frac{1}{2} (Z_{L}^{1/2\dagger})_{iI} (m_{I} + \delta m_{I}) (Z_{R}^{1/2})_{IJ} - \frac{1}{2} (Z_{R}^{1/2\dagger})_{iI} (m_{I} + \delta m_{I}) (Z_{L}^{1/2})_{IJ} \bigg] \\ &+ \gamma^{5} \bigg[-\frac{1}{2} (Z_{L}^{1/2\dagger})_{iI} (m_{I} + \delta m_{I}) (Z_{R}^{1/2})_{IJ} + \frac{1}{2} (Z_{R}^{1/2\dagger})_{iI} (m_{I} + \delta m_{I}) (Z_{L}^{1/2})_{IJ} \bigg] \\ &+ k \bigg[\frac{1}{2} (Z_{L}^{1/2\dagger})_{iI} (Z_{L}^{1/2})_{IJ} + \frac{1}{2} (Z_{R}^{1/2\dagger})_{iI} (Z_{R}^{1/2})_{IJ} \bigg] \\ &+ k \gamma^{5} \bigg[-\frac{1}{2} (Z_{L}^{1/2\dagger})_{iI} (Z_{L}^{1/2})_{IJ} + \frac{1}{2} (Z_{R}^{1/2\dagger})_{iI} (Z_{R}^{1/2})_{IJ} \bigg] . \end{split}$$

Renormalization conditions

$$= j:$$

$$\begin{cases}
K^{ii}(m_i^2) u(m_i) = \overline{u}(m_i) K^{ii}(m_i^2) = 0, \\
\frac{1}{k - m_i} K^{ii}(m_i^2) u(m_i) = \overline{u}(m_i) K^{ii}(m_i^2) \frac{1}{k - m_i} = 0;
\end{cases}$$

i.e.,

$$\begin{cases} K_{1}^{ii}(m_{i}^{2}) + m_{i}K_{r}^{ii}(m_{i}^{2}) = 0, \\ K_{5}^{ii}(m_{i}^{2}) = 0, \\ K_{5r}^{ii}(m_{i}^{2}) = 0, \\ \left[\frac{\partial}{\partial k}(K_{1}^{ii}(k^{2}) + kK_{r}^{ii}(k^{2}))\right]_{k=m_{i}} = 0 \end{cases}$$

i > j:

$$\begin{cases} K^{ij}(m_j^2) u(m_j) = 0, & \text{i.e.,} \\ \overline{u}(m_i) K^{ij}(m_i^2) = 0; \\ K_5^{ij}(m_j^2) - m_j K_{57}^{ij}(m_j^2) = 0, \\ K_1^{ij}(m_i^2) - m_j K_{57}^{ij}(m_j^2) = 0, \\ K_1^{ij}(m_i^2) + m_i K_r^{ij}(m_i^2) = 0, \\ K_5^{ij}(m_i^2) + m_i K_{57}^{ij}(m_i^2) = 0. \end{cases}$$

By these conditions are determined the renormalization constants $(Z_{L}^{1/2})_{ij}$, $(Z_{R}^{1/2})_{ij}$ (complex non-unitary matrices) and δm_i . We need not impose the conditions for i < j since the renormalization conditions with i < j are not independent of those with i > j. This is because the effective action Γ is hermite due to the hermiticity of the Lagrangian as explained in § 3.2 explicitly.

In the case where the *CP*-invariance is imposed, the condition $K_5^{ii}(m_i^2) = 0$ holds automatically and no counter terms are required.

(3) Higgs particle

Renormalization conditions

Renormalization constants

$$F_{\phi}(m_{\phi}^{2}) = 0 \\ F_{\phi}'(m_{\phi}^{2}) = 0 \} \qquad \Rightarrow \qquad Z_{\phi}, \, \delta m_{\phi}^{2} \,.$$

i

4.3.3 Renormalization conditions and counter terms for two-point functions of unphysical particles

Since the unphysical particles do not appear in external lines, the finite parts of their renormalization constants are irrelevant. Hence we may impose any renormalization condition for these particles. It is only our task to eliminate divergences caused by the presence of these particles. Practically, it is most convenient to adopt the minimal subtraction procedure.

(1) Nambu-Goldstone particles

$$x^{+} \xrightarrow{\chi^{-}} = k^{2}F(k^{2}),$$

$$\chi^{+}_{---} \chi^{-}_{---} = k^{2} Z_{\chi} + T Y \frac{e M_{Z}}{2 M_{W} \sqrt{M_{Z}^{2} - M_{W}^{2}}} G_{3} H^{-1} Z_{\chi} .$$

Relevant renormalization ⇒ condition

Renormalization constant Z_{x} .

$$\chi_{3} = k^{2}F_{3}(k^{2}),$$

$$\chi_{3} = k^{2}Z_{x3} + TY \frac{eM_{z}}{2M_{w}\sqrt{M_{z}^{2} - M_{w}^{2}}}G_{3}H^{-1}Z_{x3}.$$

Relevant renormalization \Rightarrow Renormalization constant Z_{x3} . condition

(2) Faddeev-Popov ghosts

$$\begin{bmatrix} \overline{c}^{+} & & c^{-} \\ \hline c^{-} & & c^{-} \\ \hline c^{-} & & c^{+} \\ \hline c^{-} & & c^{+} \\ \hline c^{-} & & c^{+} \\ \hline c^{+} & & = \gamma_{-+}(k^{2}), \end{bmatrix}$$

$$\underbrace{\overline{c}^{\dagger}}_{\bullet\bullet\bullet\bullet\bullet\bullet\bullet} = k^2 \widetilde{Z}_{\mathfrak{s}} - \alpha M_{\mathfrak{w}}^2 G_{\mathfrak{w}} \widetilde{Z}_{\mathfrak{s}} Z_{\mathfrak{w}}^{1/2} Z_{\mathfrak{x}}^{-1/2} .$$

Relevant renormalization \Rightarrow Renormalization constant \widetilde{Z}_s . condition

$$\overbrace{c^{\mathbb{Z}}}^{c^{\mathbb{Z}}} - \overbrace{c^{\mathbb{Z}}}^{c^{\mathbb{Z}}} = \gamma_{ZZ}(k^2),$$



(3) Two-point functions between physical and unphysical particles

We have determined all the renormalization constants except the constant Y. In terms of these renormalization constants determined already the counter terms are specified for the following two-point functions:





4.3.4 Charge renormalization constant Y

We have exhausted all the renormalization constants and the counter terms for the two-point functions. Only one remaining renormalization constant Y can be determined as follows:

$$e^{-\Gamma^{\mu}(p,k)}$$

See § 4.3.6 on the counter term.

Renormalization condition		Renormalization	constant	
$\overline{u}(m_e)\Gamma^{\mu}u(m_e)\big _{k^{\mu}=0}=0$	⇔	Y		

We may, in fact, impose the above renormalization condition on any charged particle and obtain the same value for the constant Y, since the universality of the on-shell charge (coupling constant with the photon) has been proved in § 3.4.

In the rest of tables we are concerned with the counter terms for the three- and four-point functions at least with one loop. In these expressions for counter terms, it should be understood that the diagrams in parentheses indicate the tree vertices given by the relevant Feynman rules and that their tree parts are omitted.

4.3.5 Counter terms for three- and four-point functions in \mathcal{L}_{G}

(1) Gauge-boson three-point functions





4.3.6 Counter terms for three-point functions in \mathcal{L}_{F}



- 4.3.7 Counter terms for three- and four-point functions in \mathcal{L}_{H}
 - (1) Gauge-boson-Higgs three-point functions





(2) Gauge-boson-Higgs four-point functions



134



(3) Higgs three-point functions

$$(\phi_{\chi_3})_{\chi^-} \times \chi^+(\chi_3\phi) = G_2 G_W^{-1} G_\phi Z_\phi^{1/2} Z_\chi \left(1 - \frac{Y G_2 T}{G_\phi G_W} \frac{e M_Z}{m_\phi^2 M_W \sqrt{M_Z^2 - M_W^2}} \right)$$
$$\times \left(\begin{pmatrix} \phi_{\chi_3} \\ (\phi_{\chi_3}) \chi^- \\ (\phi_{\chi_3}) \chi^- \\ \chi^+(\chi_3\phi) \end{pmatrix} \right)$$

(4) Higgs four-point functions



Downloaded from https://academic.oup.com/ptps/article-abstract/doi/10.1143/PTPS.73.1/1848359 by guest on 06 January 2019

136

4.3.8 Counter terms for three-point functions in \mathcal{L}_{M} Fermion-Higgs three-point functions









4.3.9 Counter terms for three-point functions in \mathcal{L}_{FP}

(1) Gauge-boson-ghost three-point functions



(2) Higgs-ghost three-point functions

$$\begin{array}{c} \overset{\chi^{\pm}}{e^{\frac{1}{4}}} & \overset{\chi^{\pm}}{e^{\frac{1}{4}}} = YG_{i}G_{w}^{-1}Z_{w}^{1/2} \begin{pmatrix} \overset{\chi^{\pm}}{e^{\frac{1}{4}}} & \overset{\chi^{\pm}}{e^{\frac{1}{4}}} \end{pmatrix} \tilde{Z}_{zz} + YZ_{w}^{1/2} \begin{pmatrix} \overset{\chi^{\pm}}{e^{\frac{1}{4}}} & \overset{\chi^{\pm}}{e^{\frac{1}{4}}} \end{pmatrix} \tilde{Z}_{4z} .$$

$$\begin{array}{c} \overset{\chi^{\pm}}{e^{\frac{1}{4}}} & \overset{\chi^{\pm}}{e^{\frac{1}{4}}} & \overset{\chi^{\pm}}{e^{\frac{1}{4}}} \end{pmatrix} \tilde{Z}_{44} + YG_{i}G_{w}^{-1}Z_{w}^{1/2} \begin{pmatrix} \overset{\chi^{\pm}}{e^{\frac{1}{4}}} & \overset{\chi^{\pm}}{e^{\frac{1}{4}}} \end{pmatrix} \tilde{Z}_{zz} .$$

$$\begin{array}{c} \overset{\chi^{\pm}}{e^{\frac{1}{4}}} & \overset{\chi^{\pm}}{e^{\frac{1}{4}}} & \overset{\chi^{\pm}}{e^{\frac{1}{4}}} \end{pmatrix} \tilde{Z}_{44} + YG_{i}G_{w}^{-1}Z_{w}^{1/2} \begin{pmatrix} \overset{\chi^{\pm}}{e^{\frac{1}{4}}} & \overset{\chi^{\pm}}{e^{\frac{1}{4}}} \end{pmatrix} \tilde{Z}_{zz} .$$

$$\begin{array}{c} \overset{\chi^{\pm}}{e^{\frac{1}{4}}} & \overset{\chi^{\pm}}{e^{\frac{1}{4}}} & \overset{\chi^{\pm}}{e^{\frac{1}{4}}} \end{pmatrix} \tilde{Z}_{w}^{1/2}Z_{w}^{1/2} Z_{w}^{1/2} Z_{w}^{1/2} \\ \begin{pmatrix} \overset{\chi^{\pm}}{e^{\frac{1}{4}}} & \overset{\chi^{\pm}}{e^{\frac{1}{4}}} \end{pmatrix} .$$

$$\begin{array}{c} \overset{\chi^{\pm}}{e^{\frac{1}{4}}} & \overset{\chi^{\pm}}{e^{\frac{1}{4}}} & \overset{\chi^{\pm}}{e^{\frac{1}{4}}} \end{pmatrix} \tilde{Z}_{w}^{1/2} Z_{w}^{1/2} Z_{w}^{1/2} Z_{w}^{-1/2} \\ \begin{pmatrix} \overset{\chi^{\pm}}{e^{\frac{1}{4}}} & \overset{\chi^{\pm}}{e^{\frac{1}{4}}} \end{pmatrix} .$$

$$\begin{array}{c} \overset{\chi^{\pm}}{e^{\frac{1}{4}}} & \overset{\chi^{\pm}}{e^{\frac{1}{4}}} Z_{w}^{1/2} Z_{w}^{1/2} Z_{w}^{1/2} Z_{w}^{-1/2} \\ \overset{\chi^{\pm}}{e^{\frac{1}{4}}} & \overset{\chi^{\pm}}{e^{\frac{1}{4}}} \end{pmatrix} .$$

$$\begin{array}{c} \overset{\chi^{\pm}}{e^{\frac{1}{4}}} & \overset{\chi^{\pm}}{e^{\frac{1}{4}}} & \overset{\chi^{\pm}}{e^{\frac{1}{4}}} \\ \overset{\chi^{\pm}}{e^{\frac{1}{4}}} & \overset{\chi^{\pm}}{e^{\frac{1}{4}}} \end{pmatrix} \tilde{Z}_{w}^{1/2} Z_{w}^{1/2} Z$$

The last two counter terms are obtained from the corresponding couplings proportional to β in $\mathcal{L}_{FP}^{(3)}$. (See Eqs.(4.31) and (4.37).) These couplings have no tree parts.

Chapter 5

Electroweak Radiative Corrections to Leptonic Processes

In this chapter we treat concrete examples of application of the on-shell renormalization procedure which has been explained in the foregoing chapters. We will calculate the electroweak radiative corrections of order α to the pure leptonic processes: the neutrino (anti-neutrino)-electron scattering, $\nu_{\mu}(\bar{\nu}_{\mu})e^{-}$ $\rightarrow \nu_{\mu}(\bar{\nu}_{\mu})e^{-}$, the inverse muon decay, $\nu_{\mu}e^{-}\rightarrow\mu^{-}\nu_{e}$, and the muon decay, $\mu^{-}\rightarrow e^{-}\nu_{\mu}\bar{\nu}_{e}$. We mention briefly the outline of calculation in our renormalization scheme (§ 5.2) and give the explicit expressions of results in every step of the calculation for the above processes (§§ 5.3 and 5.4).

§ 5.1 Leptonic processes and the Weinberg-Salam theory

As we mentioned already on the occasion of setting the Weinberg-Salam model (§ 1. 2), the comparison of its predictions with various experiments gives almost the same values to the parameters of the Weinberg-Salam (WS) theory, the Weinberg angle θ_W and the coupling constant g (and consequently to the masses M_W , M_Z of the gauge bosons). The argument was, however, in the tree level; then it seems worthwhile to perform the systematic calculation of the higher order effects and to examine closely the aspect of the WS theory as the renormalizable field theory. This situation reminds us of the quantum electrodynamics (QED) in the fourties. In those days the renormalizability of QED was found and the investigations of the radiative corrections revealed out the abundant contents of the theory. Now we proceed similar calculations in the WS theory by making use of the on-shell renormalization procedure which we regard as the most natural extension of the renormalization prescription in QED.

In these calculations we will consider only the leptonic processes with non-hadronic targets which come out of the purely leptonic currents to avoid dynamical complications due to the relevant strong interactions as far as possible.

From this point of view the experimentally most feasible and well-defined process is the annihilation of e^+e^- into l^+l^- pair. This process is mediated by the photon in the tree level, and the Z boson exchange becomes effective in high energy region. The electroweak radiative corrections of order α to this process were given in [Passarino and Veltman 79] for $\mu^+\mu^-$ pair and [Consoli 79] for e^+e^- pair.

Next to the above process the leptonic reactions of our interests are

$$\nu_e(\bar{\nu}_e)e^- \rightarrow \nu_e(\bar{\nu}_e)e^-, \quad \nu_\mu(\bar{\nu}_\mu)e^- \rightarrow \nu_\mu(\bar{\nu}_\mu)e^-,$$

the inverse muon decay $\nu_{\mu}e^{-} \rightarrow \mu^{-}\nu_{e} (\bar{\nu}_{e}e^{-} \rightarrow \mu^{-}\bar{\nu}_{\mu})$

and

the muon decay
$$\mu^-(\mu^+) \rightarrow e^- \nu_\mu \bar{\nu}_e (e^+ \bar{\nu}_\mu \nu_e)$$
.

These processes are all mediated through only the gauge boson exchange in the tree level.

The elastic scattering of the neutrino (anti-neutrino) on the electron is caused mainly by the neutral leptonic current which is one of the characteristics of the WS theory, and these elastic processes were early considered by 't Hooft as a test of the WS theory ['t Hooft 71a]. The electron-neutrino (antineutrino)-electron scattering is possible even in the usual charged current V-A theory (the Feynman—Gell-Mann theory) but with different magnitudes of cross sections from those in the WS theory. The observed cross sections of $\bar{\nu}_{e}$ -e scattering in the reactor experiment were in favor of the WS theory [Reines et al. 76]. The muon-neutrino (anti-neutrino)-electron scattering is mediated by the neutral leptonic current only and the confirmation of these reactions [Hasert et al. 73, see also Mo 82] was the most persuasive evidence for the existence of the neutral leptonic current and for the WS theory before the advent of the polarized electron experiment [Prescott et al. 78 and 79].

Then we will calculate the electroweak radiative corrections of order α to these processes $\nu_{\mu}e^{-} \rightarrow \nu_{\mu}e^{-}$ and $\bar{\nu}_{\mu}e^{-} \rightarrow \bar{\nu}_{\mu}e^{-}$ as examples of the neutral current processes [Salomonson and Ueda 75, Green and Veltman 80, Aoki et al. 81, Bardin et al. 82]. For the comparison with observations in the future, it may be relevant to calculate the processes for $\nu_{\mu}(\bar{\nu}_{\mu})$ -beams which are more easily feasible in accelerator experiments.

In the reactions $\nu_{\mu}e^{-} \rightarrow \mu^{-}\nu_{e}$ and $\bar{\nu}_{e}e^{-} \rightarrow \mu^{-}\bar{\nu}_{\mu}$ take part the charged leptonic currents.^{*)} These reactions were discussed at the tree level by Jarlskog based on the Feynman—Gell-Mann theory [Jarlskog 70]. She suggested to observe the ratio of the forward and backward cross sections as a test of the V-A structure of the charged leptonic current. It was, however, only recent years that these reactions were investigated experimentally because of their rather high threshold energies which amount to 10.9 GeV in the incident neutrino energy and of the exceedingly low cross sections [Armenise

^{*)} The corresponding process induced by the muon-anti-neutrino $\bar{\nu}_{\mu}e^{-} \rightarrow \mu^{-}\bar{\nu}_{e}$ is possible only when one assumes the multiplicative conservation of the leptonic numbers [Feinberg and Weinberg 61, see also Derman and Jones 77, Derman 79]. The observation gives the ratio of cross sections $\sigma(\bar{\nu}_{\mu}e^{-} \rightarrow \mu^{-}\bar{\nu}_{e})/\sigma(\nu_{\mu}e^{-} \rightarrow \mu^{-}\nu_{e}) < 0.09$ at 90% confidence level [Jonker et al. 80].

et al. 79, Jonker et al. 80]. As an example of charged current processes we will take up the reaction $\nu_{\mu}e^- \rightarrow \mu^-\nu_e$. Once we have the amplitude of this process, the amplitude of the muon decay can be easily obtained by making use of the crossing relation. In this way we present the electroweak radiative corrections of order α both to the reaction $\nu_{\mu}e^- \rightarrow \mu^-\nu_e$ [Aoki et al. 81, Aoki and Hioki 81, Hioki 82a] and to the muon decay $\mu^- \rightarrow e^-\nu_{\mu}\bar{\nu}_e$ [Ross 73, Appelquist et al. 72 and 73, Hioki 82a and 82b].

§ 5.2 Scenario of calculation

When we intend to perform some calculation in the WS theory, we find with perplexed eyes that discouragingly great many Feynman diagrams should be taken into account in contrast with those in QED. This situation is brought about from the structure of the WS theory in which we have many fields to be taken into care. This is the case in our calculations. By adopting the relevant approximations explained in what follows, we can leave considerable numbers of diagrams out of account, but it is still desirable to arrange systematically the order of calculations. We give the outline of a plan for this purpose in Tables 5.1 and 5.2.

We will carry out the calculations of the electroweak radiative corrections to the cross sections of the above-mentioned leptonic processes in the one-loop order $(O(\alpha))$. In these calculations we assume the simplest version of the WS theory with three generations of quarks and leptons and with one Higgs scalar doublet. The masses of the electron- and muon-neutrino are considered as to be zero. For experimentally available energies E_{μ} of the incident neutrino in the laboratory frame, the squared momentum transfers q^2 in such processes are negligible in comparison with the square of gauge boson mass $M^2 = M_W^2$ or M_z^2 (the mean value of the squared momentum transfer in our calculation is estimated, e.g., to be $\overline{q^2} = 4 \sim 5 (\text{GeV})^2$ at $E_y = 10^4 \text{ GeV}$). This is the case for the lepton mass $m = m_e$ or m_μ compared with the gauge boson mass M. Then we neglect all the quantities of $O(q^2/M^2)$ and of O(m/M) in our following calculations. As a result, we may leave out of consideration every Feynman diagram with the Higgs-lepton vertex from the beginning, because this coupling is proportional to the ratio m/M. The Feynman amplitudes associated with our leptonic processes in the one-loop order and the formulas on 1/M expansion of the amplitudes are found in Appendix B and Appendix C, respectively.

In these Feynman amplitudes we encounter inevitably both ultraviolet and infrared divergent integrals. In order to arrive at the well-defined cross sections we should calculate the divergent integrals by adopting relevant regularization method and separate out the finite parts of amplitudes. We adopt the method of dimensional regularization for the ultraviolet divergent integrals

Steps		What is to be calculated		Numbers of Feynman diagrams	Renormalization constants and counter terms to be determined		Renormali- zation conditions	
and IR-divergent for (*))		1. Two-point functions						
	Self-energy corrections	1-1	$A_{\mathbf{R}}^{\mathbf{Z}}(q^2)$:	Z boson self-energy part	13(Fig. 5. 3)	δM_Z^2 , Z	$Z_{ZZ}^{1/2}; A_{\rm c}^{\rm Z}$	§4.3.2 (1)
		1-2	$A_{\mathbf{R}}^{\mathbf{Z}\mathbf{A}}(q^{2})$:	ZA transition self- energy part	9(Fig. 5. 4)	$Z_{ZA}^{1/2}, Z_{A}^{1/2}$	$a_z^2; A_c^{ZA}$	§4.3.2 (1)
		1-3	$A^{\mathtt{A}}_{\mathtt{R}}(q^{\mathtt{2}})$:	Photon self-energy part	9(Fig. 5. 5)	$Z_{AA}^{1/2};$	Ac [▲]	§4.3.2 (1)
		1-4	$A_{\mathbb{R}}^{W}(q^2)$:	W boson self-energy part (*)	19(Fig. 5. 6)	$\delta M_{W}^{2};$	Ac ^w	§4.3.2 (1)
		1–5	$\Sigma^l_{\mathbf{R}}(q)$:	Charged lepton self-energy part (*)	4(Fig. 5.7)	$\delta m_l, Z_L^l$, $Z_{\mathbf{R}}^{i}$; $\Sigma_{\mathbf{C}}^{i}$	§4.3.2 (2)
		1-6	${\Sigma}^{ u}_{ m R}(q)$:	Neutrino self- energy part	3(Fig. 5.7 (2)(3))	Ζ ι ";	Σ_{c}	§4.3.2 (2)
žt (Vertex corrections	2. Th	ree-point	functions				
UV-divergen		2-1	$\Gamma_{R^{\alpha}}^{eeA}(p_{e}',$	p_e): eeA vertex function (*)	4(Fig. 5.8)	Υ;	$\Gamma^{eeA}_{C\alpha}$	§4. 3. 4
		2-2	$\Gamma_{\mathrm{R}\alpha}^{\mathrm{eeZ}}(p_{\mathrm{e}}',$	p_{e}): eeZ vertex function (*)	5(Fig. 5.9)		$\Gamma^{eeZ}_{C^{lpha}}$	§4. 3. 6
		2–3	$\Gamma^{\mu\nu Z}_{R\alpha}(p_{\nu}',$	p_{ν}): $\forall v Z$ vertex function	4(Fig. 5. 10)		$\Gamma_{C\alpha}^{\nu\nu Z}$	§4. 3. 6
		2-4	$\Gamma^{\nu\nu}_{\mathbf{R}^{\alpha}}(p_{\nu}',$	$(p_{\nu}): v vertex$ function	3(Fig. 5. 11)		Γ ^{νν Δ} Cα	§4. 3. 6
	3.		ur-point fu	unctions				······
			Box diag	rams	3(Fig. 5.1 (7)(8)(9))	No UV-	and IR-d	ivergences
	4. Infrared divergences							
liv.	Rea S H		Real photon emissions Soft photon effect		2(Fig. 5. 12)			
IR-6								
			Hard p	photon effect				
5. Cross sections		Arrangement of the invariant amplitudes \rightarrow Numerical computation \rightarrow d σ /dt						

Table 5.1. A scenario of one-loop calculation $(\nu e \rightarrow \nu e)$.

['t Hooft and Veltman 72a]. For the infrared divergent integrals we calculate them by using both the dimensional regularization method and non-gauge invariant conventional method in which the fictitious photon mass λ is introduced. The infinite constants in the *D*-dimensional regularization are denoted as

$$C_{\rm UV} \equiv (4\pi\mu^2)^{\varepsilon_{\rm U}} \Gamma(\varepsilon_{\rm U}) \simeq (1/\varepsilon_{\rm U}) - \gamma + \ln(4\pi\mu^2), \qquad (5\cdot1)$$

$$C_{\rm IR} \equiv (4\pi\mu^2)^{\varepsilon_{\rm I}} \Gamma(\varepsilon_{\rm I}) \simeq (1/\varepsilon_{\rm I}) - \gamma + \ln(4\pi\mu^2), \qquad (5\cdot 2)$$

Steps		ps	What is to be calculated	Numbers of Feynman diagrams	Renormalization constants and counter terms to be determined	Renormali- zation conditions
		1. Tw	vo-point functions			
nt (and IR-divergent for (*))	Self-energy corrections	1-1	$A_{\mathbf{R}}^{\mathbf{Z}}(q^2): Z$ boson self-energy part	13(Fig. 5. 3)	δM_{z}^{2} , A_{c}^{z}	§4.3.2 (1)
		1-2	$A_{\mathbf{R}}^{W}(q^{2})$: W boson self-energy part (*)	19(Fig. 5. 6)	δM_W^2 , Z_W ; A_C^W	§4.3.2 (1)
		1–3	$A_{\mathbb{R}}^{\mathbb{Z}A}(q^2)$: ZA transition self- energy part	9(Fig. 5. 4)	$Z_{ZA}^{1/2}, Z_{AZ}^{1/2}; A_0^{ZA}$	§4.3.2 (1)
		1-4	$A_{\mathbf{R}}^{\mathbf{A}}(q^2)$: Photon self-energy part	9(Fig. 5. 5)	$Z_{AA}^{1/2}; A_{C}^{A}$	§4.3.2 (1)
		1–5	$\Sigma_{\mathbf{R}}{}^{l}(q)$: Charged lepton self-energy part (*)	4(Fig. 5.7)	$\delta m_l, Z_L^l, Z_R^l; \Sigma_0^l$	§4.3.2 (2)
		1-6	$\Sigma_{R}^{\nu}(q)$: Neutrino self- energy part	3(Fig. 5.7 (2)(3))	Z_{L} "; Σ_{C} "	§4.3.2 (2)
erge		2. Th	ree-point functions			
UV-dive	corrections	2-1	$\Gamma_{R\alpha}^{eeA}(p_e', p_e): eeA \text{ vertex} $ function (*)	4(Fig. 5.8)	$Y; \qquad \Gamma_{C\alpha}^{eeA}$	§4. 3. 4
		2-2	$\Gamma_{\mathrm{K}\alpha}^{\nu l W}(p_{\nu}', p_{l}): \nu l W \text{ vertex} $ function $(l=e, \mu)$	5(Fig. 5. 13)	$\Gamma^{\nu l W}_{C lpha}$	§4. 3. 6
	rtex	3. Fo	ur-point functions			
gent	Ve		Box diagrams	$5(Fig. 5. 2, (5) \sim (9))$	No UV-divergence	e
iver	4. Infrared divergences					
R-di	Real photon emissions		$(3) \rightarrow 2 (Fig. 5. 14)$			
Π			Soft photon effect			
			Hard photon effect			
5. Cross sections			oss sections	Arrangement of invariant amplitudes \rightarrow Numerical computation \rightarrow $d\sigma/dt$		

Table 5.2. A scenario of one-loop calculation $(\nu e \rightarrow \mu \nu)$.

$$4 - D \equiv 2\varepsilon_{\rm U} \quad \text{or} \quad 2\varepsilon_{\rm I} \tag{5.3}$$

for ultraviolet and infrared divergences, respectively, where γ is the Euler constant and μ is introduced on the dimensional reason. We have added the suffix U or I to ε as a convention to distinguish ultraviolet and infrared divergences. In our following calculations we choose $\mu^2 = 1$ without the loss of generality. As for the infrared divergent integrals which appear in our one-loop calculations, it can be seen that the correspondence

$$C_{\rm IR} \leftrightarrow \ln(\lambda^2)$$
 (5.4)

between the infinite constants obtained by the dimensional regularization method
and those by the conventional method, and that for the finite parts of integrals both the methods give the completely same results (see also [Marciano and Sirlin 75]).

To the one-loop corrections for the $\nu_{\mu}(\bar{\nu}_{\mu})$ -e scattering contribute the Feynman diagrams of the types shown in Fig. 5.1, where diagram (1) is the tree diagram and the blobs in the other diagrams mean all possible one-loop corrections after reducing the number of diagrams by the above-mentioned approximation. These corrections consist of the self-energy corrections (two-point functions, Fig. 5.1 (2) (3)) and the vertex corrections which include the three-point functions (Fig. 5.1 (4) (5) (6)) and the four-point functions (Fig. 5.1 (7) (8) (9)). We must determine the renormalization constants to carry out our renormalization program on these functions. For the two-point functions the following renormalization constants come to be necessary as one can see from § 4.3.2 (1):

for the Z boson self-energy part (Fig. 5.1 (2)),

 δM_{Z}^{2} , $Z_{ZZ}^{1/2}$ and $Z_{ZA}^{1/2}$;

for the ZA transition self-energy part (Fig. 5.1 (3)),

 δM_Z^2 , $Z_{ZZ}^{1/2}$, $Z_{ZA}^{1/2}$, $Z_{AZ}^{1/2}$ and $Z_{AA}^{1/2}$.

For the three-point functions we should have more renormalization constants in our hand $(\S 4.3.6)$:

for eeZ vertex function (Fig. 5.1 (4)),^{*)}

 δM_z^2 , δM_w^2 , $Z_{ZZ}^{1/2}$, $Z_{AZ}^{1/2}$, Z_L^e , Z_R^e and Y;

for $\nu\nu Z$ vertex function (Fig. 5.1 (5)),



Fig. 5.1. Relevant diagrams of one-loop corrections for the processes $\nu_{\mu}e \rightarrow \nu_{\mu}e$ and $\overline{\nu}_{\mu}e \rightarrow \overline{\nu}_{\mu}e$.

^{*)} In the present case, we need only the diagonal part of the renormalization constants of leptons $(Z_L^{L^2})_{nm}$ since we do not consider the lepton mixing (we set $m_r=0$). We express $(Z_L^{L^2})_{ll}$ as $(Z_L^{L})_{ll}$.

 δM_z^2 , δM_w^2 , $Z_{zz}^{1/2}$, Z_L^{ν} and Y;

for $\nu\nu A$ vertex function (Fig. 5.1 (6)),

 $\delta M_{Z}^{2}, \ \delta M_{W}^{2}, \ Z_{ZA}^{1/2}, \ Z_{L}^{\nu} \ \text{and} \ Y.$

First we determine the renormalization constants δM_z^2 , $Z_{ZZ}^{1/2}$, $Z_{ZZ}^{1/2}$, $Z_{AZ}^{1/2}$ and $Z_{AA}^{1/2}$ in Steps 1-1~1-3 of Table 5.1 by referring to the renormalization conditions in § 4.3.2 (1). For determining the charge renormalization constant Y from the *eeA*-vertex we need moreover the renormalization constants Z_L^e , Z_R^e and δM_W^2 in addition to those known in the above (§ 4.3.4). Then we prepare the renormalization constants δM_W^2 and Z_W in Step 1-4 (Z_W is for the later use) and Z_L^i , Z_R^i ($l=e, \mu$) and Z_L^v in Steps 1-5 and 1-6 in consideration of the renormalization of the above-mentioned vertex functions. Now we can determine the charge renormalization constants. The explicit forms of renormalization constants are given in Appendix E. In terms of these constants we fix the counter terms for three-point functions in Steps 2-2~2-4. For the four-point functions in Step 3 we have neither ultraviolet nor infrared divergences. In this way we arrive at the ultraviolet divergence free amplitude for the $\nu_{\mu}(\bar{\nu}_{\mu})$ -e scattering in the one-loop order.

In Step 4 we calculate the cross section for the real photon emission $\nu_{\mu}(\bar{\nu}_{\mu}) e \rightarrow \nu_{\mu}(\bar{\nu}_{\mu}) e \gamma$. It should be added to the cross section for the elastic scattering to cancel out the infrared divergences which remained in the transition amplitude. The work that sums up every part calculated up to this and makes the square is left to the computer's care in the final step 5 (Chapter 6).

The Feynman diagrams of the types seen in Fig. 5.2 contribute to the electroweak radiative corrections for the reaction $\nu_{\mu}e^{-} \rightarrow \mu^{-}\nu_{e}$ in the one-loop order, where the first one is the tree diagram. The renormalization constants which come to be necessary for our purpose in this case are



Fig. 5.2. Relevant diagrams of one-loop corrections for the process $\nu_{\mu}e \rightarrow \mu\nu_{e}$ (and $\mu \rightarrow e\nu_{\mu}\overline{\nu}_{e}$).

Chap. 5 Electroweak Radiative Corrections to Leptonic Processes 147

$$\delta M_{W}^{2}, Z_{W}; \ \delta M_{Z}^{2}, Z_{L}^{l} \ (l = e, \mu), Z_{L}^{\nu} \text{ and } Y$$

The procedures which are necessary for determining these constants, i.e., the steps above the dotted line in Table 5.2 are already completed in the calculation of ν -e scattering. Then substantially new jobs are Step 2-2 (Fig. 5.2 (3) (4)) and Step 3 (Fig. 5.2 (5) \sim (9)) only. In the calculation of the four-point functions we should take care of the infrared divergence which comes out of diagram (7) in Fig. 5.2. In Step 4 we can neglect the contribution from diagram (2) in Fig. 5.15 in our approximation and apply the foregoing results to this case.

In what follows we will give the renormalized results in every step of our scenario.

§ 5.3 Neutral current processes: $\nu_{\mu}e^- \rightarrow \nu_{\mu}e^-$ and $\bar{\nu}_{\mu}e^- \rightarrow \bar{\nu}_{\mu}e^-$

The transition amplitudes for the processes $\nu_{\mu}e^- \rightarrow \nu_{\mu}e^-$ and $\bar{\nu}_{\mu}e^- \rightarrow \bar{\nu}_{\mu}e^-$ are of the general forms

$$\langle \nu_{\mu}e | T | \nu_{\mu}e \rangle = \overline{u}_{e}(p_{e}') \left[\gamma^{\mu} \{A(t) - B(t)\gamma_{5}\} + C(t) \frac{(p_{e}' + p_{e})^{\mu}}{2m_{e}} \right] u_{e}(p_{e}) \\ \times \overline{u}_{\nu}(p_{\nu}')\gamma_{\mu}(1 - \gamma_{5})u_{\nu}(p_{\nu})$$
(5.5a)

and

 $\langle \bar{\nu}_{\mu} e | T | \bar{\nu}_{\mu} e \rangle$

= the same expression as the above except the replacement $u_{\mu} \rightarrow v_{\mu}$,

(5·5b)

respectively, where the transition amplitudes T is defined by Eq. (4.42) and we adopt the normalization $\bar{u}u=2m$. In terms of these invariant amplitudes A, B and C, which are the functions of the squared momentum transfer q^2 or $t \equiv (p_e' - p_e)^2$, the cross sections are written as

$$\frac{d\sigma}{dt} = \frac{1}{2\pi} \bigg[|A + \hat{\varsigma}B|^2 + |A - \hat{\varsigma}B|^2 \Big(\frac{u - m_e^2}{s - m_e^2} \Big)^2 + 2(|A|^2 - |B|^2) \frac{m_e^2 t}{(s - m_e^2)^2} + 4\operatorname{Re}(AC^*) \bigg\{ 1 + \frac{st}{(s - m_e^2)^2} \bigg\} \bigg],$$
(5.6)

where $\xi = 1$ for $\nu_{\mu} e$ scattering and $\xi = -1$ for $\bar{\nu}_{\mu} e$ scattering.

Then we must find the well-defined expressions of the invariant amplitudes A, B and C. We begin with the determination of the renormalization constants δM_z^2 , $Z_{ZZ}^{1/2}$, $Z_{$



Fig. 5.3. The Z boson self-energy diagrams.

carry out the calculations in the Feynman gauge. The explicit forms of renormalization constants are all given collectively in Appendix E.

Step 1-1 Z boson self-energy part

For the blob of diagram (2) in Fig. 5.1 are thirteen Feynman diagrams shown in Fig. 5.3 where ψ stands for all relevant fermions and the last one expresses the counter term. By using formulas in Appendix B.1, we obtain the transverse part of the renormalized Z boson self-energy $A_{\rm R}^{Z}(q^2)$

$$A_{\mathsf{R}}^{Z}(q^{2}) = \sum_{n=1}^{12} A_{n}^{Z}(q^{2}) + A_{\mathsf{C}}^{Z}(q^{2}), \qquad (5\cdot7)$$

where $A_c^z(q^2)$ is the counter term corresponding to the last diagram in Fig. 5.3. The terms on the right-hand side are explicitly given by

$$\begin{split} &= \frac{e^2}{32\pi^2 M_W^2 (M_Z^2 - M_W^2)} \left[\sum_i M_Z^4 \left\{ -\frac{1}{6} q^2 (\eta_i^2 + 1) + m_i^2 \right\} \right. \\ &\quad - 2M_Z^2 (M_Z^4 + 2M_Z^2 M_W^2 - 4M_W^4) \\ &\quad - \frac{1}{3} q^2 (M_Z^4 - 2M_Z^2 M_W^2 - 18M_W^4) \right] C_{\rm UV} \\ &\quad + \frac{e^2}{32\pi^2 M_W^2 (M_Z^2 - M_W^2)} \left[\sum_i M_Z^4 \{ (\eta_i^2 + 1) q^2 F(m_i, m_i, q^2) \right. \\ &\quad - m_i^2 F_0(m_i, m_i, q^2) \} - \frac{q^2}{3} (M_Z^4 - 2M_Z^2 M_W^2 + 4M_W^4) \\ &\quad + M_W^2 (M_Z^4 - 4M_Z^2 M_W^2 + 16M_W^4) \ln M_W^2 \\ &\quad + \frac{1}{2} M_Z^4 (M_Z^2 \ln M_Z^2 + m_\phi^2 \ln m_\phi^2) - 10M_W^4 q^2 F_0(M_W, M_W, q^2) \\ &\quad + (M_Z^4 - 4M_Z^2 M_W^2 + 24M_W^4) q^2 F(M_W, M_W, q^2) \\ &\quad + M_W^2 (3M_Z^4 - 4M_Z^2 M_W^2 - 16M_W^4) F_0(M_W, M_W, q^2) \end{split}$$

1

Chap. 5 Electroweak Radiative Corrections to Leptonic Processes 149

$$+ M_{Z}^{4} \{ 2M_{Z}^{2} F_{0}(m_{\phi}, M_{Z}, q^{2}) - M_{Z}^{2} F_{1}(m_{\phi}, M_{Z}, q^{2}) + q^{2} F(m_{\phi}, M_{Z}, q^{2}) \} - m_{\phi}^{2} M_{Z}^{4} F_{1}(M_{Z}, m_{\phi}, q^{2}) \Big].$$
(5.8)

In the one-loop order, the counter term given in § 4.3.2 (1) is reduced to

$$A_{\rm c}{}^{\rm z}(q^2) = \delta M_{\rm z}{}^2 + 2Z_{\rm zz}^{1/2}(M_{\rm z}{}^2 - q^2).$$
(5.9)

On the right-hand side of Eq. (5.8), \sum_i denotes the sum on the species of fermions and the sum over the color degrees of freedom must be included. The η_i 's should be understood as

$$\begin{split} \eta_{\nu} &\equiv 1 , \\ \eta_{e,\,\mu,\,\tau} &\equiv (4M_W^2 - 3M_Z^2) / M_Z^2 , \\ \eta_{u,\,e,\,\iota} &\equiv (8M_W^2 - 5M_Z^2) / 3M_Z^2 , \\ \eta_{d,\,s,\,b} &\equiv (4M_W^2 - M_Z^2) / 3M_Z^2 . \end{split}$$
(5.10)

The functions F_n (n=0, 1, 2) and F are defined by

$$F_n(M_1, M_2, q^2) \equiv \int_0^1 x^n \ln \{M_1^2(1-x) + M_2^2 x - q^2 x (1-x)\} dx,$$

(n=0, 1, 2)

$$F(M_1, M_2, q^2) \equiv F_1(M_1, M_2, q^2) - F_2(M_1, M_2, q^2).$$
(5.11)

Their explicit forms are found in Appendix D.

The renormalization conditions in the one-loop order are

$$\sum_{n=1}^{12} A_n^{\ z} (M_z^{\ 2}) + \delta M_z^{\ 2} = 0 \tag{5.12}$$

and

$$\sum_{n=1}^{12} A_n^{Z'}(M_Z^2) - 2Z_{ZZ}^{1/2} = 0, \qquad (5.13)$$

from which are determined the renormalization constants δM_z^2 and $Z_{zz}^{1/2}$, respectively.

Step 1-2 ZA transition self-energy part

For diagram (3) in Fig. 5.1 are nine Feynman diagrams which are shown in Fig. 5.4. We call them the ZA transition self-energy part. The contributions to the transverse part from these diagrams are given by

$$A_{R}^{ZA}(q^{2}) = \sum_{n=1}^{8} A_{n}^{ZA}(q^{2}) + A_{C}^{ZA}(q^{2}), \qquad (5 \cdot 14)$$



Fig. 5.4. The transition self-energy diagrams between the Z boson and the photon.

where

$$\begin{split} \sum_{n=1}^{8} A_{n}^{ZA}(q^{2}) \\ &= \frac{e^{2}M_{W}}{16\pi^{2}\sqrt{M_{Z}^{2}-M_{W}^{2}}} \bigg[\left\{ \frac{q^{2}}{6} \Big(-\sum_{i} \zeta_{i} + \frac{M_{Z}^{2}}{M_{W}^{2}} + 18 \Big) + 2M_{Z}^{2} \right\} C_{\mathrm{UV}} \\ &+ (-M_{Z}^{2} + 8M_{W}^{2}) \ln M_{W}^{2} - (M_{Z}^{2} + 8M_{W}^{2}) F_{0}(M_{W}, M_{W}, q^{2}) \\ &+ q^{2} \bigg\{ \sum_{i} \zeta_{i} F(m_{i}, m_{i}, q^{2}) - \frac{M_{Z}^{2} - 12M_{W}^{2}}{M_{W}^{2}} F(M_{W}, M_{W}, q^{2}) \\ &- 5F_{0}(M_{W}, M_{W}, q^{2}) + \frac{M_{Z}^{2} - 4M_{W}^{2}}{6M_{W}^{2}} \bigg\} \bigg]$$
(5.15)

and

$$A_{c}^{ZA}(q^{2}) = Z_{ZA}^{1/2}(M_{Z}^{2} - q^{2}) - Z_{AZ}^{1/2}q^{2}$$
(5.16)

is the counter term. In Eq. (5.15) the ζ_i 's are

$$\begin{aligned} \zeta_{\nu} &\equiv 0 , \\ \zeta_{e,\mu,\tau} &\equiv 2 \left(4M_{W}^{2} - 3M_{Z}^{2} \right) / M_{W}^{2} , \\ \zeta_{u,c,t} &\equiv 4 \left(8M_{W}^{2} - 5M_{Z}^{2} \right) / 9M_{W}^{2} \end{aligned}$$

and

$$\zeta_{d,s,b} = 2(4M_W^2 - M_Z^2)/9M_W^2. \qquad (5.17)$$

According to § 4.3.2 (1), the renormalization constants $Z_{AZ}^{1/2}$ and $Z_{ZA}^{1/2}$ are determined by the conditions

$$\sum_{n=1}^{8} A_n^{ZA} (M_Z^2) - M_Z^2 Z_{AZ}^{1/2} = 0$$
(5.18)

and

$$\sum_{n=1}^{8} A_n^{ZA}(0) + M_Z^2 Z_{ZA}^{1/2} = 0, \qquad (5.19)$$

respectively.

Step 1-3 Photon self-energy part

Relevant Feynman diagrams in this case are shown in Fig. 5.5.



Fig. 5.5. The photon self-energy diagrams.

The transverse part of the renormalized photon self-energy $A_{\mathbf{R}}^{A}(q^{2})$ is given by

$$A_{R}^{4}(q^{2}) = \sum_{n=1}^{8} A_{n}^{4}(q^{2}) + A_{C}^{4}(q^{2}), \qquad (5 \cdot 20)$$

where

$$\sum_{n=1}^{8} A_n^{\ 4}(q^2) = \frac{e^2}{\pi^2} q^2 \left(-\frac{2}{9} N + \frac{3}{16} \right) C_{\text{UV}} \\ + \frac{e^2}{16\pi^2} \left[8 \sum_i Q_i^2 q^2 F(m_i, m_i, q^2) \\ + 8M_W^2 \{ \ln M_W^2 - F_0(M_W, M_W, q^2) \} \\ + q^2 \left\{ -\frac{2}{3} + 12F(M_W, M_W, q^2) - 5F_0(M_W, M_W, q^2) \right\} \right]$$

(N: the number of generations. We take N=3.) (5.21)

and the counter terms

$$A_{\rm c}{}^{\rm A}(q^2) = -2q^2 Z_{\rm AA}^{1/2} \,. \tag{5.22}$$

It can be seen from Eqs. (5.21) and (5.22) that one of the renormalization conditions $A_R^A(0) = 0$ is satisfied automatically, by observing $F_0(M_W, M_W, 0) = \ln M_W^2$. This is guaranteed by the remaining U(1) symmetry in the theory. The other renormalization condition $A_R^{A'}(0) = 0$ is reduced to

$$\sum_{n=1}^{8} A_n^{A'}(0) - 2Z_{AA}^{1/2} = 0, \qquad (5 \cdot 23)$$

from which is determined the renormalization constant $Z_{AA}^{1/2}$. As seen in Steps 1-1 to 1-3, in the one-loop order we can determine independently every



Fig. 5.6. The W boson self-energy diagrams.

renormalization constant in the neutral gauge boson sector. Then we go on the preparation to determine the charge renormalization constant Y, i.e., on the determination of the renormalization constants δM_{W}^{2} , Z_{W} , Z_{L}^{i} , Z_{R}^{i} and Z_{L}^{ν} .

Step 1-4 W boson self-energy part

The transverse part of the renormalized W boson self-energy $A_{\mathbb{Z}}^{\mathbb{W}}(q^2)$ (see Fig. 5.6) is given by

$$A_{R}^{W}(q^{2}) = \sum_{n=1}^{18} A_{n}^{W}(q^{2}) + A_{C}^{W}(q^{2}).$$
 (5.24)

The renormalization constants δM_W^2 and Z_W are determined by

$$\sum_{n=1}^{18} A_n^{W}(M_W^2) + \delta M_W^2 = 0, \qquad (5 \cdot 25)$$

$$\sum_{n=1}^{18} A_n^{W'}(M_W^2) - Z_W = 0, \qquad (5 \cdot 26)$$

where

$$\begin{split} A_{\mathcal{C}}^{\mathsf{W}}(q^2) &= \delta M_{\mathsf{W}}^2 + Z_{\mathsf{W}}(M_{\mathsf{W}}^2 - q^2),\\ \sum_{n=1}^{18} A_n{}^{\mathsf{W}}(q^2) &= \frac{e^2 M_Z{}^2}{32\pi^2 (M_Z{}^2 - M_{\mathsf{W}}{}^2)} \left\{ \sum_{(I,i)} |U_{Ii}|^2 \left(m_i{}^2 + m_I{}^2 - \frac{2}{3}q^2 \right) \right. \\ &\quad -2 \left(M_Z{}^2 - 2M_{\mathsf{W}}{}^2 \right) + \frac{19}{3}q^2 \right\} C_{\mathsf{UV}} \\ &\quad + \frac{e^2}{32\pi^2 (M_Z{}^2 - M_{\mathsf{W}}{}^2)} \left[2M_Z{}^2 \sum_{(I,i)} |U_{Ii}|^2 \left\{ 2q^2 F\left(m_I, m_i, q^2 \right) \right. \\ &\quad -m_i{}^2 F_1(m_I, m_i, q^2) - m_I{}^2 F_1(m_i, m_I, q^2) \right\} \\ &\quad -M_Z{}^2 \left\{ q^2 - 7M_{\mathsf{W}}{}^2 \ln M_{\mathsf{W}}{}^2 - \left(6M_{\mathsf{W}}{}^2 + \frac{M_Z{}^2}{2} \right) \ln M_Z{}^2 - \frac{1}{2}m_\phi{}^2 \ln m_\phi{}^2 \right\} \end{split}$$

$$+ (2M_{Z}^{4} - 5M_{Z}^{2}M_{W}^{2} - 14M_{W}^{4})F_{0}(M_{W}, M_{Z}, q^{2}) - 10M_{W}^{2}q^{2}F_{0}(M_{W}, M_{Z}, q^{2}) - (M_{Z}^{4} + 15M_{Z}^{2}M_{W}^{2} - 16M_{W}^{4})F_{1}(M_{W}, M_{Z}, q^{2}) + (M_{Z}^{2} + 20M_{W}^{2})q^{2}F(M_{W}, M_{Z}, q^{2}) - 2(M_{Z}^{2} - M_{W}^{2})\{7M_{W}^{2}F_{0}(M_{W}, \lambda, q^{2}) + 5q^{2}F_{0}(M_{W}, \lambda, q^{2}) - 8M_{W}^{2}F_{1}(M_{W}, \lambda, q^{2}) - 10q^{2}F(M_{W}, \lambda, q^{2})\} - m_{\phi}^{2}M_{Z}^{2}\{F_{0}(m_{\phi}, M_{W}, q^{2}) - F_{1}(m_{\phi}, M_{W}, q^{2})\} + M_{Z}^{2}M_{W}^{2}\{2F_{0}(m_{\phi}, M_{W}, q^{2}) - F_{1}(m_{\phi}, M_{W}, q^{2})\} + M_{Z}^{2}q^{2}F(m_{\phi}, M_{W}, q^{2})],$$

$$(5.27)$$

where λ is the small photon mass introduced to regularize the infrared divergence, and U_{It} is the Cabibbo-like mixing matrix. (For the practical calculation we consider only the Cabibbo angle.)

Step 1-5 Charged lepton self-energy part

٦

The relevant Feynman diagrams are shown in Fig. 5.7. The contributions of graphs (1) \sim (3) are

$$\begin{split} \sum_{i=1}^{3} & \Sigma_{i}^{l}(q) \\ &= \frac{e^{2}}{16\pi^{2}} \frac{1}{8M_{w}^{2}(M_{z}^{2} - M_{w}^{2})} \left[\left\{ -16m_{l}M_{z}^{2}(M_{z}^{2} - M_{w}^{2}) + M_{z}^{2}(5M_{z}^{2} - 2M_{w}^{2})q\gamma_{5} \right\} C_{\mathrm{UV}} \\ &+ M_{z}^{2}(5M_{z}^{2} - 2M_{w}^{2})q\gamma_{5} \right\} C_{\mathrm{UV}} \\ &+ 8m_{e}(M_{z}^{2} - M_{w}^{2}) \left\{ M_{z}^{2} + 4M_{w}^{2}F_{0}(m_{e}, \lambda, q^{2}) + 2(M_{z}^{2} - 2M_{w}^{2})F_{0}(m_{e}, M_{z}, q^{2}) \right\} \mathbf{1} \\ &- \left\{ M_{z}^{2}(5M_{z}^{2} - 2M_{w}^{2})F_{0}(m_{e}, \lambda, q^{2}) + 4M_{z}^{2}M_{w}^{2}F_{1}(0, M_{w}, q^{2}) \right. \\ &+ 2(5M_{z}^{4} - 12M_{z}^{2}M_{w}^{2} + 8M_{w}^{4})F_{1}(m_{e}, M_{z}, q^{2}) \right\} q \\ &- M_{z}^{2} \left\{ 3(M_{z}^{2} - 2M_{w}^{2}) - 4M_{w}^{2}F_{1}(0, M_{w}, q^{2}) \right. \\ &+ 2(3M_{z}^{2} - 4M_{w}^{2})F_{1}(m_{e}, M_{z}, q^{2}) \right\} q \\ &+ 2(3M_{z}^{2} - 4M_{w}^{2})F_{1}(m_{e}, M_{z}, q^{2}) \right\} q \\ &+ 2(3M_{z}^{2} - 4M_{w}^{2})F_{1}(m_{e}, M_{z}, q^{2}) \right\} q \\ &+ 2(3M_{z}^{2} - 4M_{w}^{2})F_{1}(m_{e}, M_{z}, q^{2}) \right\} q \\ &+ 2(3M_{z}^{2} - 4M_{w}^{2})F_{1}(m_{e}, M_{z}, q^{2}) \right\} q \\ &+ 2(3M_{z}^{2} - 4M_{w}^{2})F_{1}(m_{e}, M_{z}, q^{2}) \right\} q \\ &+ 2(3M_{z}^{2} - 4M_{w}^{2})F_{1}(m_{e}, M_{z}, q^{2}) \right\} q \\ &+ 2(3M_{z}^{2} - 4M_{w}^{2})F_{1}(m_{e}, M_{z}, q^{2}) \right\} q \\ &+ 2(3M_{z}^{2} - 4M_{w}^{2})F_{1}(m_{e}, M_{z}, q^{2}) \right\} q \\ &+ 2(3M_{z}^{2} - 4M_{w}^{2})F_{1}(m_{e}, M_{z}, q^{2}) \right\} q \\ &+ 2(3M_{z}^{2} - 4M_{w}^{2})F_{1}(m_{e}, M_{z}, q^{2}) \right\} q \\ &+ 2(3M_{z}^{2} - 4M_{w}^{2})F_{1}(m_{e}, M_{z}, q^{2}) \right\} q \\ &+ 2(3M_{z}^{2} - 4M_{w}^{2})F_{1}(m_{e}, M_{z}, q^{2}) \right\} q \\ &+ 2(3M_{z}^{2} - 4M_{w}^{2})F_{1}(m_{e}, M_{z}, q^{2}) \right\} q \\ &+ 2(3M_{z}^{2} - 4M_{w}^{2})F_{1}(m_{e}, M_{z}, q^{2}) \right\} q \\ &+ 2(3M_{z}^{2} - 4M_{w}^{2})F_{1}(m_{e}, M_{z}, q^{2}) \right\} q \\ &+ 2(3M_{z}^{2} - 4M_{w}^{2})F_{1}(m_{e}, M_{z}, q^{2}) \right\} q \\ &+ 2(3M_{z}^{2} - 4M_{w}^{2})F_{1}(m_{e}, M_{z}, q^{2}) \right\} q \\ &+ 2(3M_{z}^{2} - 4M_{w}^{2})F_{1}(m_{e}, M_{z}, q^{2}) \right\} q \\ &+ 2(3M_{z}^{2} - 4M_{w}^{2})F_{1}(m_{e}, M_{z}, q^{2}) \right\} q \\ &+ 2(3M_{z}^{2} - 4M_{w}^{2})F_{1}(m_{e}, M_{z}, q^{2}) \right\} q \\ &+ 2(3M_{z}^{2} - 4M_{w}^{2})F_{1}(m_{e}$$



Fig. 5.7. The fermion self-energy diagrams.

By using the counter terms

$$\Sigma_{\rm C}{}^{l}(q) = -\delta m_{l} \mathbf{1} + \frac{1}{2} (Z_{\rm L}{}^{l} + Z_{\rm R}{}^{l}) (q - m_{l}) \mathbf{1} - \frac{1}{2} (Z_{\rm L}{}^{l} - Z_{\rm R}{}^{l}) q \gamma_{5}, \qquad (5 \cdot 29)$$

renormalization constants Z_{L}^{l} , Z_{R}^{l} and δm_{l} are determined from the following conditions:

$$\begin{split} & \Sigma_{1}^{l}(m_{i}^{2}) + m_{i}\Sigma_{r}^{l}(m_{i}^{2}) = 0 , \\ & \Sigma_{5r}^{l}(m_{i}^{2}) = 0 , \\ & 2m_{i}\{(\Sigma_{1}^{l'}(m_{i}^{2}) + m_{i}\Sigma_{r}^{l'}(m_{i}^{2})\} + \Sigma_{r}^{l}(m_{i}^{2}) = 0 . \end{split}$$

$$(5\cdot30)$$

Here we have expressed the renormalized self-energy as

$$\Sigma_{\rm R}^{\ l}(q) = \Sigma_{\rm 1}^{\ l}(q^2) \, \mathbf{1} + \Sigma_{\rm r}^{\ l}(q^2) \, q + \Sigma_{\rm 5r}^{\rm l}(q^2) \, q \gamma_{\rm 5} \,. \tag{5.31}$$

Step 1-6 Neutrino self-energy part

The relevant diagrams for neutrino self-energy are the same as those in Fig 5.7, but except for graph (1). Renormalized self-energy is

$$\Sigma_{R^{\nu}}(q) = \sum_{n=2,3} \Sigma_{n^{\nu}}(q) + \Sigma_{C^{\nu}}(q), \qquad (5\cdot32)$$

where

$$\sum_{n=2,3} \Sigma_{n}^{\nu}(q) = \frac{e^{2}}{16\pi^{2}} \frac{M_{Z}^{2}}{8M_{W}^{2}(M_{Z}^{2} - M_{W}^{2})} \{ (M_{Z}^{2} + 2M_{W}^{2}) C_{UV} - (M_{Z}^{2} + 2M_{W}^{2}) - 4M_{W}^{2}F_{1}(M_{l}, M_{W}, q^{2}) - 2M_{Z}^{2}F_{1}(0, M_{Z}, q^{2}) \} q (1 - \gamma_{5}),$$

$$\Sigma_{C}^{\nu}(q) = \frac{1}{2} Z_{L}^{\nu}q (1 - \gamma_{5}). \qquad (5.33)$$

 Z_{L} is determined by

$$\Sigma_{\mathbf{R}^{\nu}}(q)|_{q^{2}=0}=0.$$
 (5.34)

Step 2-1 eeA vertex function

Using the renormalization constants which have so far been calculated, we determine the charge renormalization constants Y through eeA vertex.

The relevant diagrams are shown in Fig. 5.8.



Fig. 5.8. Relevant diagrams of eeA vertex.

154

Chap. 5 Electroweak Radiative Corrections to Leptonic Processes 155

$$\Gamma_{R\alpha}^{eeA}(p_e', p_e) = \sum_{n=1}^{3} \Gamma_{n\alpha}^{eeA}(p_e', p_e) + \Gamma_{C\alpha}^{eeA}, \qquad (5.35)$$

where

$$\Gamma_{1\alpha}^{eea}(p_{e}', p_{e}) = \frac{e^{2}}{16\pi^{2}} \left\{ F(q^{2})\gamma_{\alpha} + G(q^{2}) \frac{(p_{e} + p_{e}')_{\alpha}}{2m_{e}} \right\},$$
(5.36)

$$F(q^{2}) = C_{\text{UV}} - \ln m_{e}^{2} + \frac{3}{a} \ln \frac{1+a}{1-a} - \frac{1+a^{2}}{a}$$
$$\times \left\{ \frac{1}{2} \ln \frac{4}{1-a^{2}} \ln \frac{1+a}{1-a} - (C_{\text{IR}} - \ln m_{e}^{2}) \ln \frac{1+a}{1-a} - \phi(a) \right\}, \quad (5.36a)$$

$$G(q^2) = -\frac{1-a^2}{a} \ln \frac{1+a}{1-a} \,. \tag{5.36b}$$

$$\begin{split} \sum_{n=2}^{3} \Gamma_{n\alpha}^{ee4}(p_{e}', p_{e}) &= -\frac{e^{2}}{16\pi^{2}} \frac{eM_{z}^{2}}{8M_{w}^{2}(M_{z}^{2}-M_{w}^{2})} \\ &\times \left\{ \left[\left(5M_{z}^{2}-6M_{w}^{2}+8\frac{M_{w}^{4}}{M_{z}^{2}} \right) \gamma_{\alpha} + \left(3M_{z}^{2}-10M_{w}^{2} \right) \gamma_{\alpha} \gamma_{5} \right] C_{\mathrm{UV}} \right. \\ &- \left[\left(\frac{5}{2}M_{z}^{2}-5M_{w}^{2}+4\frac{M_{w}^{4}}{M_{z}^{2}} \right) + 6M_{w}^{2} \ln M_{w}^{2} \right. \\ &+ \left(5M_{z}^{2}-12M_{w}^{2}+8\frac{M_{w}^{4}}{M_{z}^{2}} \right) \ln M_{z}^{2} \right] \gamma_{\alpha} \\ &+ \left[\left(-\frac{3}{2}M_{z}^{2}+3M_{w}^{2} \right) + 6M_{w}^{2} \ln M_{w}^{2} - \left(3M_{z}^{2}-4M_{w}^{2} \right) \ln M_{z}^{2} \right] \gamma_{\alpha} \gamma_{5} \right\}, \end{split}$$

$$(5\cdot37)$$

$$\Gamma_{C\alpha}^{eeA} = -e\gamma_{\alpha} \left(Y + Z_{AA}^{1/2} + \frac{Z_{L}^{e} + Z_{R}^{e}}{2} - \frac{Z_{L}^{e} - Z_{R}^{e}}{2} \gamma_{5} \right) + \frac{e}{4M_{W} \sqrt{M_{Z}^{2} - M_{W}^{2}}} Z_{ZA}^{1/2} \gamma_{\alpha} \left\{ (3M_{Z}^{2} - 4M_{W}^{2}) + M_{Z}^{2} \gamma_{5} \right\}.$$
(5.38)

In Eq. (5.36a), we have used the following function,

$$\phi(a) = \operatorname{Sp}\left(\frac{1+a}{2}\right) - \operatorname{Sp}\left(\frac{1-a}{2}\right), \qquad (5\cdot39)$$
$$\left(a = \sqrt{\frac{-q^2}{-q^2 + 4m_e^2}}\right)$$

and

$$\operatorname{Sp}(x) \equiv -\int_{0}^{x} \frac{1}{t} \ln(1-t) dt \,. \quad (\operatorname{Spence function}) \quad (5\cdot40)$$



Fig. 5.9. The eeZ vertex diagrams.

Some useful formulas on the Spence function are given in Appendix F. Step 2-2 eeZ vertex function

Now we can go on to evaluate the correction for eeZ vertex (Fig. 5.1 (4)) with the initial electron momentum p_e and the final one p_e' . Only five Feynman diagrams shown in Fig. 5.9 should be considered on our approximation. This correction is expressed as

$$\Gamma_{R\alpha}^{eeZ}(p_e', p_e) = \sum_{i=1}^{4} \Gamma_{i\alpha}^{eeZ}(p_e', p_e) + \Gamma_{C\alpha}^{eeZ}, \qquad (5.41)$$

where $\Gamma_{i\alpha}^{eez}(p_{e'}, p_{e})$ are contributions from four diagrams (i=1, 2, 3, 4) respectively and $\Gamma_{C\alpha}^{eez}$ is the relevant counter terms. Their explicit forms are given as follows:

$$\Gamma_{1\alpha}^{eeZ}(p_{e}', p_{e}) = \frac{e^{3}}{64\pi^{2}M_{W}\sqrt{M_{Z}^{2} - M_{W}^{2}}} \times \left[(3M_{Z}^{2} - 4M_{W}^{2}) \left\{ F(q^{2})\gamma_{\alpha} + G(q^{2}) \frac{(p_{e}' + p_{e})_{\alpha}}{2m_{e}} \right\} + M_{Z}^{2} \left\{ F_{5}(q^{2})\gamma_{\alpha}\gamma_{5} + G_{5}(q^{2}) \frac{(p_{e}' - p_{e})_{\alpha}}{2m_{e}} \gamma_{5} \right\} \right].$$
(5.42)

In Eq. (5.42), $F(q^2)$ and $G(q^2)$ are found in Eqs. (5.36a) and (5.36b) respectively,

$$F_{\mathfrak{s}}(q^{2}) = C_{\text{UV}} - \ln m_{e}^{2} + \frac{1+2a^{2}}{a} \ln\left(\frac{1+a}{1-a}\right)$$
$$-\frac{1+a^{2}}{a} \left\{ \frac{1}{2} \ln\left(\frac{4}{1-a^{2}}\right) \ln\left(\frac{1+a}{1-a}\right) - (C_{\text{IR}} - \ln m_{e}^{2}) \ln\left(\frac{1+a}{1-a}\right) - \phi(a) \right\}$$
(5.42a)

and

$$G_{\mathfrak{s}}(q^2) = \frac{1-a^2}{a^2} \left\{ \frac{1+2a^2}{a} \ln\left(\frac{1+a}{1-a}\right) - 2 \right\}.$$
 (5.42b)

$$\begin{split} &\sum_{i=2}^{4} \Gamma_{i\alpha}^{eeZ}(p_{e}', p_{e}) \\ &= \frac{e^{3}}{64\pi^{2}} \frac{M_{Z}^{2}}{(M_{Z}^{2} - M_{W}^{2})\sqrt{M_{Z}^{2} - M_{W}^{2}}} \\ &\times \left[\frac{-M_{Z}^{4}}{16M_{W}^{3}\xi^{3}} \left(-C_{UV} + \frac{1}{2} + \ln M_{Z}^{2} \right) \gamma_{\alpha} \left\{ (1 + 3\xi^{2}) + (3\xi + \xi^{3}) \gamma_{5} \right\} \right. \\ &+ \left\{ -\frac{M_{Z}^{2}}{2M_{W}} \left(-C_{UV} + \frac{1}{2} + \ln M_{W}^{2} \right) \right. \\ &+ \left. \left. + \left\{ -\frac{M_{Z}}{2M_{W}} \left(-C_{UV} + \frac{1}{2} + \ln M_{W}^{2} \right) \right\} \gamma_{\alpha} (1 - \gamma_{5}) \right], \\ &\left. \left(\xi \equiv \frac{M_{Z}^{2}}{3M_{Z}^{2} - 4M_{W}^{2}} \right) \right\} \end{split}$$
(5.43)

 $\Gamma_{C\alpha}^{eeZ} = -eZ_{AZ}^{1/2} \gamma_{\alpha}$ $-\frac{eM_{Z}^{2}}{4M_{m}\sqrt{M_{c}^{2}-M_{m}^{2}}}$

$$\frac{1}{4M_{w}\sqrt{M_{z}^{2}-M_{w}^{2}}} \times \left\{Y + \frac{M_{z}^{2}-2M_{w}^{2}}{2(M_{z}^{2}-M_{w}^{2})} \left(\frac{\delta M_{z}^{2}}{M_{z}^{2}} - \frac{\delta M_{w}^{2}}{M_{w}^{2}}\right) + Z_{zz}^{1/2} + Z_{L}^{e}\right\}\gamma_{\alpha}(1-\gamma_{5}) + \frac{e}{M_{w}}\sqrt{M_{z}^{2}-M_{w}^{2}}\gamma_{\alpha}\left\{Y + \frac{M_{z}^{2}}{2(M_{z}^{2}-M_{w}^{2})} \left(\frac{\delta M_{z}^{2}}{M_{z}^{2}} - \frac{\delta M_{w}^{2}}{M_{w}^{2}}\right) + Z_{zz}^{1/2} + \frac{Z_{L}^{e}+Z_{R}^{e}}{2} - \frac{Z_{L}^{e}-Z_{R}^{e}}{2}\gamma_{5}\right\}.$$
(5.44)

Steps 2-3 and 2-4 $\nu\nu Z$ and $\nu\nu A$ vertex functions

עע A-vertices which correspond to Figs. 5.10 and 5.11 are given as follows:

$$\Gamma_{R\alpha}^{\nu\nu Z}(p_{\nu}', p_{\nu}) = \sum_{i=1}^{3} \Gamma_{i\alpha}^{\nu\nu Z}(p_{\nu}', p_{\nu}) + \Gamma_{C\alpha}^{\nu\nu Z},$$

$$= \frac{-e^{3}M_{Z}^{2}}{64\pi^{2}(M_{Z}^{2} - M_{W}^{2})\sqrt{M_{Z}^{2} - M_{W}^{2}}} \left\{ -\frac{(M_{Z}^{2} + M_{W}^{2})^{2} + 7M_{W}^{4}}{4M_{W}^{3}}C_{UV} \right\}$$

$$= \frac{-e^{3}M_{Z}^{2}}{64\pi^{2}(M_{Z}^{2} - M_{W}^{2})\sqrt{M_{Z}^{2} - M_{W}^{2}}} \left\{ -\frac{(M_{Z}^{2} + M_{W}^{2})^{2} + 7M_{W}^{4}}{4M_{W}^{3}}C_{UV} \right\}$$

$$= \frac{V}{(1)} = \sum_{i=1}^{2} \left(-\frac{W_{Z}^{2} + M_{W}^{2}}{(1)} + \frac{W_{W}^{2}}{(1)} + \frac{W_{W}$$





K-I. Aoki, Z. Hioki, R. Kawabe, M. Konuma and T. Muta

$$+ \frac{M_{Z}^{4}}{4M_{W}^{3}} \left(\frac{1}{2} + \ln M_{Z}^{2}\right) + \frac{M_{Z}^{2} - 2M_{W}^{2}}{2M_{W}} \left(\frac{1}{2} + \ln M_{W}^{2}\right)$$

$$+ M_{W} \left(\frac{1}{2} + 3\ln M_{W}^{2}\right) \right\} \gamma_{\alpha} (1 - \gamma_{5}), \qquad (5 \cdot 46)$$

$$\Gamma_{C\alpha}^{\nu\nu Z} = \frac{eM_Z^2}{4M_W\sqrt{M_Z^2 - M_W^2}} \left\{ Y + \frac{M_Z^2 - 2M_W^2}{2(M_Z^2 - M_W^2)} \left(\frac{\delta M_Z^2}{M_Z^2} - \frac{\delta M_W^2}{M_W^2} \right) + Z_{ZZ}^{1/2} + Z_L^{\nu} \right\} \gamma_{\alpha} (1 - \gamma_5) , \qquad (5 \cdot 47)$$

$$\begin{split} \Gamma_{R\alpha}^{\nu\nu A}(p_{\nu}', p_{\nu}) &= \sum_{i=1}^{2} \Gamma_{i\alpha}^{\nu\nu A}(p_{\nu}', p_{\nu}) + \Gamma_{C\alpha}^{\nu\nu A}, \qquad (5.48) \\ \sum_{i=1}^{2} \Gamma_{i\alpha}^{\nu\nu A}(p_{\nu}', p_{\nu}) &= \frac{-e^{3}M_{Z}^{2}}{96\pi^{2}(M_{Z}^{2} - M_{W}^{2})} \\ &\times \left[-3C_{\rm UV} + 3\ln M_{W}^{2} + \frac{q^{2}}{M_{W}^{2}} \{1 + \ln M_{W}^{2} - 6F(m_{e}, m_{e}, q^{2})\} \right] \gamma_{\alpha}(1 - \gamma_{5}), \qquad (5.49) \end{split}$$

where q^2/M_W^2 terms must be maintained in connection with the $1/q^2$ term in the photon propagator. The counter term is

$$\Gamma_{C\alpha}^{\nu\nu\lambda} = \frac{eM_Z^2}{4M_W\sqrt{M_Z^2 - M_W^2}} Z_{ZA}^{1/2} \gamma_\alpha (1 - \gamma_5).$$
 (5.50)

Step 3 Box diagrams

.

.

The four-point functions in our processes come out of the three box diagrams in the one-loop calculation (in Fig. 5.1 (7) (8) (9)). They have no ultraviolet divergence and exchange gauge bosons with large masses, so that the results are very simple under the approximations we have adopted. Referring to Appendix B.3, we have for the sum of contributions from these three diagrams

$$A(p_{e}', p_{\nu}'; p_{e}, p_{\nu})_{(7)+(8)+(9)}$$

$$= \left(\frac{e^{2}}{4\pi}\right)^{2} \frac{M_{Z}^{4}}{64M_{W}^{4}(M_{Z}^{2}-M_{W}^{2})^{2}} \overline{u}_{\nu}\gamma^{\mu}(1-\gamma_{5}) u_{\nu}$$

$$\times \overline{u}_{e}\gamma_{\mu} \left\{ (28M_{W}^{2}-9M_{Z}^{2}) - \left(15M_{Z}^{2}-20M_{W}^{2}+\frac{24M_{W}^{4}}{M_{Z}^{2}}\right)\gamma_{5} \right\} u_{e}. \quad (5\cdot51)$$

In this way we have calculated every necessary part of one-loop corrections. Then we can arrive at the cross section for the $\nu_{\mu}(\bar{\nu}_{\mu}) e \rightarrow \nu_{\mu}(\bar{\nu}_{\mu}) e$ scattering free from ultraviolet divergence through Eq. $(5 \cdot 5)$.

Step 4 Infrared divergence

There still remains, however, a job of which we must make in order to obtain well-defined cross sections. Our amplitudes calculated in the foregoing steps have the infrared singularity and the collinear singularity, too. We can get rid of this infrared singularity by calculating cross sections for final states with finite energy resolution, that is, for those in which a photon of energy k smaller than some maximum energy ω (a soft photon) is emitted, and by adding them to the elastic cross sections (the well-known Kinoshita-Lee-Nauenberg theorem [Kinoshita 62, Lee and Nauenberg 64]; see also [Nakanishi 58]). In the same way one can regulate the collinear singularity by introducing an angular resolution on the direction of outgoing charged lepton and by computing transition rates to nearly degenerate states which include the charged lepton and a collinear photon (not necessarily soft) [Berman 58, Kinoshita and Sirlin 59, Tsai 60, 65, Nieuwenhuizen 71, Berends et al. 73a and 73b, Byers et al. 79, Green and Veltman 80, Aoki and Hioki 81, Hioki 82a, Paschos and Wirbel 82, Berends et al. 82]. (See also [Sterman and Weinberg 77] as an example in QCD.) The numerical values of the energy resolution and of the angular resolution should be determined depending on individual experimental apparatus.

For these purposes we make beforehand the square of the matrix element for bremsstrahlung $\nu e \rightarrow \nu l r$. (For later convenience we deal with the neutral and charged current processes together.) The results corresponding to Fig. 5.12, and Fig. 5.15 (1) and (3) are

$$M(\nu e \rightarrow \nu l \gamma) = G_{B} \bar{\nu} (p_{\nu}') \gamma^{\rho} (1 - \gamma_{5}) \nu (p_{\nu})$$

$$\times \bar{l} (p_{\iota}') \left\{ \varepsilon(k, \lambda) \frac{m_{\iota} + k + p_{\iota}'}{m_{\iota}^{2} - (k + p_{\iota}')^{2}} \gamma_{\rho} (\hat{\varsigma} - \gamma_{5}) + \gamma_{\rho} (\hat{\varsigma} - \gamma_{5}) \frac{m_{e} + p_{e} - k}{m_{e}^{2} - (p_{e} - k)^{2}} \varepsilon(k, \lambda) \right\} e(p_{e}), \qquad (5.52)$$

where $\varepsilon(k, \lambda)$ is the polarization vector of photon,

$$G_{B} = \frac{-e^{3}M_{Z}^{2}}{16M_{W}^{2}(M_{Z}^{2} - M_{W}^{2})}, \quad \xi = \frac{4M_{W}^{2} - 3M_{Z}^{2}}{M_{Z}^{2}}$$

for $\nu e \rightarrow \nu e \gamma$ $(l = e),$ (5.53a)

$$G_{B} = \frac{e^{s} M_{z}^{2}}{8M_{w}^{2} (M_{z}^{2} - M_{w}^{2})}, \quad \hat{\varsigma} = 1$$
for $\gamma e \to \mu \gamma \gamma \ (l = \mu)$. (5.53b)

(We have neglected the contribution of Fig. 5.15 (2). See the comments after Eq. (5.64h).)

The process, $\nu e \rightarrow \mu \nu \gamma$, will be used in § 5.4. After taking a sum over final spin states and an average over initial spin states of the charged lepton, we have

$$\frac{1}{2} \sum_{s_1} \sum_{s_1} |M|^2 = -32G_B^2 \left[\frac{f_1}{(kp_l')^2} + \frac{f_2}{(kp_\ell)^2} - \frac{f_3}{(kp_l')(kp_\ell)} \right], \quad (5.54)$$

$$f_{1} = m_{i} f_{0} + m_{i} \{ (1+\xi)^{2} (p_{e}p_{\nu}) (kp_{\nu}) + (1-\xi)^{2} (p_{e}p_{\nu}) (kp_{\nu}) \} - (kp_{i}') [(1+\xi)^{2} (p_{e}p_{\nu}) \{ (p_{i}'p_{\nu}') + (kp_{\nu}') \} + (1-\xi)^{2} (p_{e}p_{\nu}') \{ (p_{i}'p_{\nu}) + (kp_{\nu}) \}], \qquad (5.54a)$$

$$f_{2} = m_{e}^{2} f_{0} - m_{e}^{2} \{ (1 + \hat{\varsigma})^{2} (p_{i}' p_{\nu}') (k p_{\nu}) + (1 - \hat{\varsigma})^{2} (p_{i}' p_{\nu}) (k p_{\nu}') \}$$

+ $(k p_{e}) [(1 + \hat{\varsigma})^{2} (p_{i}' p_{\nu}') \{ (p_{e} p_{\nu}) - (k p_{\nu}) \}$
+ $(1 - \hat{\varsigma})^{2} (p_{i}' p_{\nu}) \{ (p_{e} p_{\nu}') - (k p_{\nu}') \}], \qquad (5.54b)$

$$\begin{aligned} f_{3} &= 2(p_{i}'p_{e})f_{0} \\ &+ 2(1+\hat{\varsigma}^{2})\left\{(kp_{e})(p_{i}'p_{\nu})(p_{i}'p_{\nu}') - (kp_{i}')(p_{e}p_{\nu})(p_{e}p_{\nu}')\right\} \\ &+ 2(1-\hat{\varsigma}^{2})m_{i}m_{e}(kp_{\nu}')(kp_{\nu}) \\ &+ (p_{i}'p_{e})\left[(1+\hat{\varsigma})^{2}\left\{(kp_{\nu}')(p_{e}p_{\nu}) - (kp_{\nu})(p_{i}'p_{\nu}')\right\} \\ &+ (1-\hat{\varsigma})^{2}\left\{(kp_{\nu})(p_{e}p_{\nu}') - (kp_{\nu}')(p_{i}'p_{\nu})\right\}\right], \end{aligned}$$
(5.54c)

where

$$f_{0} = (1+\hat{\xi})^{2} (p_{i}'p_{\nu}') (p_{e}p_{\nu}) + (1-\hat{\xi})^{2} (p_{i}'p_{\nu}) (p_{e}p_{\nu}') + (1-\hat{\xi}^{2}) m_{l}m_{e}(p_{\nu}'p_{\nu})$$

$$(5\cdot55)$$

is of the same form as the spin sum and average of the squared matrix element for the tree diagram (Fig. 5.1(1)):

$$\frac{1}{2} \sum_{s_1} \sum_{s_f} |M_{\text{tree}}|^2 = (32G_B^2/e^2) f_0. \qquad (5.56)$$

First we take the soft photon approximation in Eq. (5.28), i.e., we take the limit $k\rightarrow 0$ in the f_i 's (i=1,2,3) and integrate over the two-body phase



Fig. 5.12. Diagrams of a real-photon emission in the neutral current processes.

space restricted by the maximum photon energy ω . Then only the f_0 terms remain in Eqs. $(5.54a) \sim (5.54c)$ and the cross section of soft photon emission is expressed by the factorized form

$$d\sigma^{\text{soft}} = -\frac{e^2}{16\pi^3} \int \frac{d^3k}{k_0} \left[\frac{m_i^2}{(kp_i')^2} + \frac{m_e^2}{(kp_e)^2} - \frac{2(p_i'p_e)}{(kp_i')(kp_e)} \right] \cdot d\sigma_0, \quad (5.57)$$

where $d\sigma_0$ is the Born cross section derived from Eq. (5.56). In the one-loop approximation the addition of the soft photon emission cross section is effectively equivalent to adding the finite part M_{soft} of Eq. (5.57) to the amplitude (5.4).^{*'} M_{soft} is given by

$$M_{\rm soft} = C_{\rm soft} \times (\text{tree amplitude}),$$
 (5.58)

where

$$\begin{split} C_{\text{soft}} &= \frac{e^2}{16\pi^2} \left\{ \frac{s + m_i^2}{s - m_i^2} \ln\left(\frac{s}{m_i^2}\right) + \frac{s + m_e^2}{s - m_e^2} \ln\left(\frac{s}{m_e^2}\right) \\ &\quad -2\left(m_i^2 + m_e^2 - \eta\right) \int_0^1 dx \frac{1}{\varepsilon^2 a \sqrt{1 - a}} \ln\frac{1 + \sqrt{1 - a}}{\sqrt{a}} \right\}, \quad (5 \cdot 58a) \\ &\quad \eta \equiv \left(p_i' - p_e\right)^2, \\ &\quad \varepsilon \equiv E_i' \left(1 - x\right) + E_e x, \\ &\quad a \equiv \frac{1}{\varepsilon^2} \{m_i^2 (1 - x) + m_e^2 x - \eta x (1 - x)\}, \end{split}$$

in the center of mass frame, and

$$C_{\text{soft}} = \frac{e^{2}}{8\pi^{2}} \left[1 + \frac{E_{i}'}{|\mathbf{p}_{i}'|} \ln\left(\frac{E_{i}' + |\mathbf{p}_{i}'|}{m_{i}}\right) \left\{ 1 - 2\ln 2 - \ln\frac{|\mathbf{p}_{i}'|^{2}}{m_{i}(E_{i}' + |\mathbf{p}_{i}'|)} \right\} - \frac{E_{i}'}{|\mathbf{p}_{i}'|} \left\{ \frac{\pi^{2}}{6} - \operatorname{Sp}\left(\frac{E_{i}' - |\mathbf{p}_{i}'|}{E_{i}' + |\mathbf{p}_{i}'|}\right) \right\} \right], \qquad (5.58b)$$

in the laboratory frame.

The cross section calculated so far including soft photon emission effect is free from UV and IR divergence. However, it is still unrealistic in the sense that, in actual experiments, it is difficult to detect an extra photon with ω much smaller than the charged lepton mass. (Soft photon approximation is only valid under the condition $\omega \ll m_i$ in the Lab-frame.) Further, there remains theoretical dissatisfaction that the terms which diverge in the limit $m_i \rightarrow 0$ still exist in the results (collinear or mass singularity).

^{*)} In the case where one regulates the soft photon divergence by introducing the fictitious photon mass λ , one should replace the photon mass λ in the expression (5.4) by twice the maximum photon energy, 2ω .

In order to eliminate them, we have to add the exact bremsstrahlung cross section,

$$\frac{d^{4}\sigma}{dE_{i}'d\cos\theta_{1}d\cos\theta_{2}d\phi}(\nu e \rightarrow \nu l\gamma)$$

$$=\frac{|\mathbf{p}_{i}'|E_{r}}{32(2\pi)^{4}(s-m_{e}^{2})(\sqrt{s}-E_{i}'+|\mathbf{p}_{i}'|\cos\theta_{1})}\sum_{\mathbf{s}_{1}}\sum_{\mathbf{s}_{1}}|M|^{2}, \quad (5\cdot59)$$

where the variables are shown in Fig. 5.13. Taking this hard photon contribution into account, we get the cross section which can directly be compared with experiments. Various numerical results are presented in the next chapter.



Fig. 5.13. Axes and angles used in Eq. (5.59).

§ 5.4 Charged current processes: $\nu_{\mu}e^{-} \rightarrow \mu^{-}\nu_{e}$ and $\mu^{-} \rightarrow e^{-}\nu_{\mu}\bar{\nu}_{e}$

We can calculate the $O(\alpha)$ correction to the charged current processes in a way similar to the neutral current processes. As mentioned in § 5.2, we need only to calculate νlW vertex function and box diagrams in Fig. 5.2. We again proceed according to Table 5.2.

Step 2-2 VIW vertex function

The relevant diagrams are in Fig. 5.14.



Fig. 5.14. The $\nu l W$ vertex diagrams $(l=e, \mu)$.

$$\Gamma_{R\alpha}^{\nu lW}(p_{\nu}', p_{l}) = \sum_{n=1}^{4} \Gamma_{n\alpha}^{\nu lW}(p_{\nu}', p_{l}) + \Gamma_{C\alpha}^{\nu lW}, \quad (l = e \text{ or } \mu)$$
(5.60)

$$\begin{split} &\sum_{n=1}^{4} \Gamma_{n\alpha}^{\nu l \mathcal{W}}(p_{\nu}', p_{l}) \\ = & \frac{e^{3}M_{Z}^{3}(M_{Z}^{2} + 10M_{W}^{2})}{128\pi^{2}M_{W}^{2}(M_{Z}^{2} - M_{W}^{2})\sqrt{2(M_{Z}^{2} - M_{W}^{2})}} C_{UV} \\ &+ \frac{e^{3}M_{Z}}{64\pi^{2}\sqrt{2(M_{Z}^{2} - M_{W}^{2})}} \left[5 - 6\ln M_{W}^{2} + \frac{M_{W}^{2}}{(M_{Z}^{2} - M_{W}^{2})^{2}} \right] \\ &\times \left\{ 5(M_{Z}^{2} - M_{W}^{2}) - 6(M_{Z}^{2}\ln M_{Z}^{2} - M_{W}^{2}\ln M_{W}^{2}) \right\} \\ &- \frac{M_{Z}^{2}(M_{Z}^{2} - 2M_{W}^{2})}{4M_{W}^{2}(M_{Z}^{2} - M_{W}^{2})} (1 + 2\ln M_{Z}^{2}) \left] \gamma_{\alpha}(1 - \gamma_{5}), \quad (5 \cdot 61) \end{split}$$

$$\begin{split} \Gamma_{C\alpha}^{\nu \iota W} &= \frac{eM_Z}{4\sqrt{2}(M_Z^2 - M_W^2)} \\ &\times \Big(\frac{\delta M_Z^2}{M_Z^2} - \frac{\delta M_Z^2 - \delta M_W^2}{M_Z^2 - M_W^2} + 2Y + Z_L^{\ \iota} + Z_L^{\ \nu} + Z_W \Big) \gamma_{\alpha} (1 - \gamma_5) \ . \ (5 \cdot 62) \end{split}$$

Step 3 Box diagrams

The contributions from box diagrams in Fig. 5.2 except the photon exchange graph are collected as

$$A(p_{\mu}', p_{\nu}'; p_{e}, p_{\nu})_{(5)+(6)+(8)+(9)} = \left(\frac{e^{2}}{4\pi}\right)^{2} \frac{M_{z}^{2}(3M_{z}^{4}-6M_{z}^{2}M_{w}^{2}-2M_{w}^{4})}{16M_{w}^{2}(M_{z}^{2}-M_{w}^{2})^{3}} \ln\left(\frac{M_{z}^{2}}{M_{w}^{2}}\right) \overline{u}_{\mu}\gamma_{\lambda}(1-\gamma_{5}) u_{\nu} \cdot \overline{u}_{\nu}\gamma^{\lambda}(1-\gamma_{5}) u_{e}.$$
(5.63)

The photon exchange box diagram (Fig. 5.2 (7)) gives, after the Fierz transformation, following contribution:

$$\overline{u}_{\mu}(p_{\mu}') \left[A(u) \gamma_{\lambda} (1-\gamma_{5}) + B(u) \gamma_{\lambda} (1+\gamma_{5}) + C(u) (1-\gamma_{5}) (p_{e})_{\lambda} \right. \\ \left. + D(u) (1+\gamma_{5}) \right] u_{e}(p_{e})_{\lambda} \cdot \overline{u}_{\nu}(p_{\nu}') \gamma^{\lambda} (1-\gamma_{5}) u_{\nu}(p_{\nu}), \qquad (5 \cdot 64)$$

$$A(u) = G\left[\frac{m_{\mu}^{2} + m_{e}^{2} - u}{R} \left\{ 2RT \ln\left(\frac{R}{2\lambda^{2}}\right) + \ln\left(\frac{4m_{\mu}^{2}}{-u}\right) \ln\left(\frac{m_{e}^{2} - m_{\mu}^{2} - u + R}{2m_{\mu}\sqrt{-u}}\right) + \ln\left(\frac{4m_{e}^{2}}{-u}\right) \ln\left(\frac{m_{\mu}^{2} - m_{e}^{2} - u + R}{2m_{e}\sqrt{-u}}\right) - \phi\left(\frac{R}{-u}, \frac{m_{\mu}^{2} - m_{e}^{2}}{-u}\right) \right\} - \frac{1}{2} - \ln\left(\frac{M_{w}^{2}}{m_{\mu}m_{e}}\right) - \frac{m_{\mu}^{2} - m_{e}^{2}}{u} \ln\left(\frac{m_{\mu}}{m_{e}}\right) + \frac{(m_{\mu}^{2} - m_{e}^{2})^{2} - 4(m_{\mu}^{2} + m_{e}^{2})u + 3u^{2}}{u}T \right], \qquad (5.64a)$$

$$B(u) = -4m_{\mu}m_{e}GT, \qquad (5.64b)$$

K-I. Aoki, Z. Hioki, R. Kawabe, M. Konuma and T. Muta

$$C(u) = \frac{2m_{\mu}G}{u} \left\{ \ln\left(\frac{m_{\mu}}{m_{e}}\right) + (m_{e}^{2} - m_{\mu}^{2} + u)T \right\}, \qquad (5.64c)$$

$$D(u) = \frac{2m_e G}{u} \left\{ \ln\left(\frac{m_e}{m_{\mu}}\right) + (m_{\mu}^2 - m_e^2 + u) T \right\}, \qquad (5.64d)$$

$$G = \left(\frac{e^2}{4\pi}\right)^2 \frac{M_Z^2}{8M_W^2 (M_Z^2 - M_W^2)},$$
 (5.64e)

$$R \equiv \sqrt{(m_{\mu}^{2} - m_{e}^{2} + u)^{2} - 4m_{\mu}^{2}u}, \qquad (5.64f)$$

$$T = \frac{1}{R} \ln\left(\frac{m_{\mu}^{2} + m_{e}^{2} - u + R}{2m_{\mu}m_{e}}\right), \qquad (5 \cdot 64g)$$

$$\phi(a,b) \equiv \operatorname{Sp}\left(\frac{a+1+b}{2a}\right) - \operatorname{Sp}\left(\frac{a-1+b}{2a}\right) + \operatorname{Sp}\left(\frac{a+1-b}{2a}\right) - \operatorname{Sp}\left(\frac{a-1-b}{2a}\right).$$
(5.64h)

Infrared divergence is removed by the soft-photon emission in the same way as in the case of the neutral current processes except the following point: In Fig. 5.15 graph (2) has no contribution to the infrared divergence and gives very small magnitude. We do not take account of it.



Fig. 5.15. Diagrams of a real-photon emission in the charged current processes.

Collecting all the parts, we obtain the one-loop corrected amplitudes of the form (5.63), where all the contributions except the photon exchange box diagram are collected into the term A. Then the cross section is written as

$$\frac{d\sigma}{dt} = \frac{2}{\pi (s - m_e^2)^2} [|A|^2 (s - m_\mu^2) (s - m_e^2) + 2m_\mu m_e \operatorname{Re}(AB^*) u + (m_\mu^2 m_e^2 - st) \operatorname{Re} A^* (m_\mu C + m_e D)].$$
(5.65)

So far, we have calculated the amplitude for $\nu_{\mu}e \rightarrow \mu\nu_{e}$, but the amplitude for μ decay can be obtained by the same function.

$$\frac{d^{2}\Gamma}{dE_{e}d\cos\theta} = \frac{|\mathbf{p}_{e}| (m_{\mu}^{2} + m_{e}^{2} - 2m_{\mu}E_{e})^{2}}{4\pi^{3} (m_{\mu} - E_{e} + |\mathbf{p}_{e}|\cos\theta)^{4}} \times [|A|^{2} \{ (m_{\mu}^{2} + m_{e}^{2})E_{e} + (m_{\mu}^{2} - m_{e}^{2}) |\mathbf{p}_{e}|\cos\theta - 2m_{\mu}m_{e}^{2} \}$$

Chap. 5 Electroweak Radiative Corrections to Leptonic Processes 165

$$-2m_{e}\operatorname{Re}\left(A^{*}B\right)\left(m_{\mu}-E_{e}+|\boldsymbol{p}_{e}|\cos\theta\right)^{2}$$
$$+m_{\mu}\operatorname{Re}\left\{A^{*}\left(m_{\mu}C+m_{e}D\right)\right\}|\boldsymbol{p}_{e}|^{2}\sin^{2}\theta\right].$$
(5.66)

Finally, we obtain the complete cross section and decay width by adding the hard photon emission effect as done for the neutral current processes. The cross section, $\sigma(\nu e \rightarrow \mu \nu \gamma)$ has been evaluated in Eqs. (5.54) and (5.59). The width $\Gamma(\mu \rightarrow e \nu \bar{\nu} \gamma)$ can be calculated with slight modification of Eq. (5.54) as

$$\frac{d^{\mathfrak{d}}\Gamma}{dE_{\mathfrak{e}}dE_{\mathfrak{r}}d\cos\theta}(\mu \to e\nu\bar{\nu}\gamma) = \frac{1}{32(2\pi)^{\mathfrak{g}}m_{\mu}}E_{\mathfrak{r}}|\boldsymbol{p}_{e}|I_{\rho\sigma}M^{\rho\sigma}, \qquad (5\cdot67)$$

(θ : angle between e and γ)

$$I_{\rho\sigma} = \frac{1}{6} \pi \{ 2Q_{\rho}Q_{\sigma} + g_{\rho\sigma}Q^{2} \} \quad (Q \equiv p_{\mu} - p_{e} - k),$$

$$M^{\rho\sigma} = -256G_{B}^{2} \left[\frac{1}{(kp_{\mu})^{2}} \{ m_{\mu}^{2} + (kp_{\mu}) \} (p_{e})^{\rho} (p_{\mu} - k)^{\sigma} \right]$$
(5.67a)

$$L(kp_{\mu})^{s} - \frac{1}{(kp_{\mu})(kp_{e})} \{ 2(p_{\mu}p_{e})(p_{e})^{\rho}(p_{\mu})^{\sigma} + (p_{\mu}p_{e})k^{\rho}(p_{\mu} - p_{e})^{\sigma} + (kp_{\mu})(p_{e})^{\rho}(p_{e})^{\sigma} - (kp_{e})(p_{\mu})^{\rho}(p_{\mu})^{\sigma} \} + \frac{1}{(kp_{e})^{2}} \{ m_{e}^{2} - (kp_{e}) \} (p_{\mu})^{\rho}(p_{e} + k)^{\sigma}].$$
 (5.67b)

Chapter 6

Order α Corrections to Physical Quantities

We have calculated $O(\alpha)$ electroweak radiative corrections to $\nu_{\mu}e \rightarrow \nu_{\mu}e$, $\bar{\nu}_{\mu}e \rightarrow \bar{\nu}_{\mu}e$, $\nu_{\mu}e \rightarrow \mu\nu_{e}$ and $\mu \rightarrow e\nu_{\mu}\bar{\nu}_{e}$ as functions of the renormalized parameters e, M_{W} , M_{Z} , m_{f} and m_{ϕ} . Our next step is to determine the values of these parameters from experimental data. After the determination of the values of the parameters, it becomes possible to give predictions for various quantities.

In this chapter, we go into these studies. At first we consider the relation between the higher order effects and predictions for W^{\pm} and Z boson masses (§ 6.1). Next, we study the $O(\alpha)$ radiative corrections to the tree cross sections $\sigma(\nu_{\mu}e\rightarrow\nu_{\mu}e), \cdots$, and decay width $\Gamma(\mu\rightarrow e\nu_{\mu}\bar{\nu}_{e})$, and present detailed numerical results for them (§ 6.2).

§ 6.1 Higher order effects and W^{\pm} , Z boson masses

Recent development of high energy accelerators has made it possible to expect that W^{\pm} and Z bosons (and also other heavy particles) will be observed in the near future. As for the masses of W^{\pm} and Z bosons, the values, $M_W \simeq 77 \,(\text{GeV})$ and $M_Z \simeq 88 \,(\text{GeV})$ have been predicted in the tree level analyses of low energy experimental data.

In this section, we consider the improvement for these predictions by including the $O(\alpha)$ correction. At the first step we summarize the tree level analyses.

According to the general discussion in § 3.1, we must first choose input data in order to determine the values of the parameters M_W and M_Z . Usually, experimental data on the muon decay Γ^{\exp} and on a ratio of cross sections of ν and $\bar{\nu}$ induced neutral current processes (we take here $R^{\exp} = \sigma (\bar{\nu}_{\mu} e \rightarrow \bar{\nu}_{\mu} e)$ $/\sigma (\nu_{\mu} e \rightarrow \nu_{\mu} e)$) are used. Then, the tree level values of W^{\pm} and Z boson masses should be determined by

$$\Gamma^{(0)}(M_{W}, M_{Z}) = \Gamma^{\exp},$$

 $R^{(0)}(M_{W}, M_{Z}) = R^{\exp},$ (6.1)

where $\Gamma^{(0)}$ and $R^{(0)}$ express the corresponding quantities calculated as functions of M_W and M_Z at the tree level in the Weinberg-Salam theory. The oneloop analyses are carried out in a similar way: Replace the left-hand side of Eq. (6.1) by the one-loop corrected quantities $\Gamma^{(1)}$ (M_W, M_Z) and $R^{(1)}$ (M_W, M_Z) as Chap. 6 Order α Corrections to Physical Quantities 167

$$\Gamma^{(1)}(M_{W}, M_{Z}) = \Gamma^{\exp},$$

 $R^{(1)}(M_{W}, M_{Z}) = R^{\exp}.$ (6.2)

However, the traditional analyses on the muon decay are somewhat different from the above line of argument. The quantity which is used in the traditional analyses is not $\Gamma^{(0)}$ but $\Gamma_{\text{fermi}} \cdot (1 + \Delta_{\text{EM}})$. Here Γ_{fermi} is the following muon decay width calculated in the framework of the four-fermion interaction,

$$\Gamma_{\rm fermi} = \frac{G_{\rm F}^2 m_{\mu}^{\ 5}}{192 \pi^3} \left(1 - \frac{8 m_{e}^{\ 2}}{m_{\mu}^{\ 2}} \right), \tag{6.3}$$

and $\Delta_{\rm EM}$ is the QED-radiative correction to this interaction (Fig. 6.1+soft and hard photon emission effects) [Berman 58, Kinoshita and Sirlin 59, Hioki 82a],

$$\mathcal{\Delta}_{\rm EM} = \frac{\alpha}{2\pi} \left(\frac{25}{4} - \pi^2 \right) \,. \tag{6.4}$$



Fig. 6.1. The electromagnetic correction to the four-fermion interaction $\mu \rightarrow e\nu\overline{\nu}$.

That is, the following equation is used instead of the first equation in Eq. $(6 \cdot 1)$,

$$\Gamma_{\text{fermi}} \cdot (1 + \Delta_{\text{EM}}) = \Gamma^{\text{exp}} . \tag{6.5}$$

Then, the parameters of the Weinberg-Salam theory M_W , M_Z are determined by

$$\Gamma^{(0)}(M_{W}, M_{Z}) = \Gamma_{\text{fermi}}(=\Gamma^{\text{exp}} \cdot (1 - \mathcal{A}_{\text{EM}})), \qquad (6 \cdot 6a)$$

$$R^{(0)}(M_{w}, M_{Z}) = R^{\exp} . (6.6b)$$

We can summarize these as follows. There are two steps in the traditional analyses in order to determine M_W and M_Z at "tree level": First, the parameter of the four-fermion interaction G_F is determined, and the parameters M_W , M_Z are fixed by Eq. (6.6a, b). The well-known values $M_W \simeq 77 \,(\text{GeV})$ and $M_Z \simeq 88 \,(\text{GeV})$ are not the solutions of Eq. (6.1) but of Eq. (6.6). (Hereafter we express these solutions as $M_W^{(0)}$ and $M_Z^{(0)}$.) Such complicated determinations have been adopted from the historical reason that i) the QED radiative corrections to the four-fermion interaction $\mu \rightarrow e\nu\bar{\nu}$ (Fig. 6.1) had been found to be ultraviolet-convergent, and ii) the Weinberg-Salam theory was constructed to reproduce this four-fermion interaction effectively in a low-energy limit on the charged current sector. Let us call these analyses tree analyses here, although they are not "real tree" analyses.

These predictions $(M_w^{(0)} \simeq 77 \text{ GeV} \text{ and } M_z^{(0)} \simeq 88 \text{ GeV})$ are improved by including the $O(\alpha)$ electroweak corrections. The one-loop condition on the muon decay width is

$$\Gamma^{(1)}(M_{W}, M_{Z}) (\equiv \Gamma^{(0)}(M_{W}, M_{Z}) (1 + \mathcal{A}_{EW})) = \Gamma^{exp}, \qquad (6.7)$$

where Δ_{EW} denotes "electroweak correction". By the use of Eq. (6.6a), Eq. (6.7) is reduced to

$$\Gamma^{(W)}(M_{W}, M_{Z}) = \Gamma^{(0)}(M_{W}^{(0)}, M_{Z}^{(0)}), \qquad (6 \cdot 8a)$$

where

$$\Gamma^{(W)}(M_{W}, M_{Z}) \equiv \Gamma^{(0)}(M_{W}, M_{Z}) (1 + \mathcal{I}_{W}),$$

$$\mathcal{I}_{W} \equiv \mathcal{I}_{EW} - \mathcal{I}_{EM}.$$
(6.8b)

Therefore Δ_{W} becomes necessary in the practical calculations.^{*}) We conventionally call Δ_{W} "weak" part and Δ_{EM} "electromagnetic" or "QED" part respectively in the whole electroweak correction Δ_{EW} .

Next, let us consider R. Strictly speaking, one-loop-corrected R ($R^{(1)}(M_w, M_z)$) depends on the incident $\nu(\bar{\nu})$ energy and consequently the average of $R^{(1)}$ over $\nu(\bar{\nu})$ energy spectrum is necessary. Fortunately, however, the energy

$$M_{\rm fermi} = M_{\rm box} + \frac{1}{2} \left\{ Z^{\rm e} + Z^{\mu} - \frac{\alpha}{4\pi} (1 + 2 \ln M_{\rm W}^2) \right\} M_0 \ .$$

Here M_0 is the tree amplitude, and Z^i is (the finite part of) the photonic correction part of Z_L^i given in Appendix E,

$$Z^{i} = -\frac{\alpha}{4\pi} (4 - 3 \ln m_{i}^{2} + 2C_{\mathrm{IR}}).$$

(It should be noted that this type of definition of the QED part is no longer possible for the process, e.g., $\nu d \rightarrow \mu u$. For such processes, the photonic correction to the four-fermion interaction is not ultraviolet-convergent.)

On the other hand, it is possible to define the QED part in the electroweak correction for $\nu_{\mu}(\bar{\nu}_{\mu})e \rightarrow \nu_{\mu}(\bar{\nu}_{\mu})e$ processes by taking the contribution of Fig. 5.9(1) +appropriate counter terms.

^{*)} In the Weinberg-Salam theory, it is not possible for the charged current interaction to define the QED part properly by selecting some diagrams in Fig. 5.2. It is a mere convention to call $\Delta_{\rm EM}$ the QED part of $\Delta_{\rm EW}$. Actually, however, the QED-corrected renormalized amplitude of the four-fermion interaction $M_{\rm fermi}$ (Fig. 6.1+counter terms) has the following simple relation to the amplitude $M_{\rm box}$ corresponding to Fig. 5.2 (7),

dependence of R and the pure QED correction to R are found to be negligible. (Concerning the energy dependence of $R^{(1)}$, see \mathcal{A}_R in Table 6.3. The pure QED corrections to $\sigma(\nu_{\mu}e\rightarrow\nu_{\mu}e)$ and $\sigma(\bar{\nu}_{\mu}e\rightarrow\bar{\nu}_{\mu}e)$ (Fig. 5.9(1)) are non-negligible, but almost equal each other. See Table 6.2 or 6.3. Then the QED corrections to R become very small.) Therefore, we can take the condition on R similar to the case of Γ at some definite E_{ν} . (We take $E_{\nu}=5 \text{ GeV.}$) That is, by denoting the ratio R including the weak one-loop correction as $R^{(W)}(M_W, M_Z)$, one-loopcorrected values $M_W^{(1)}$ and $M_Z^{(1)}$ are obtained as the solutions of the equations

$$\Gamma^{(W)}(M_{W}, M_{Z}) = \Gamma^{(0)}(M_{W}^{(0)}, M_{Z}^{(0)}),$$

$$R^{(W)}(M_{W}, M_{Z})|_{E_{\nu}(p) = 5 \text{ GeV}} = R^{(0)}(M_{W}^{(0)}, M_{Z}^{(0)}). \quad (6.9)$$

Here we can see the advantage of the on-shell renormalization scheme. That is, renormalized M_W and M_Z are always the real part of the poles of the W and Z propagators at any order of perturbation.

The analytical expression of the one-loop radiative correction is very complicated, so solving Eq. (6.9) numerically, we obtain

$$M_{W}^{(1)} \simeq 79.17$$
 (GeV),
 $M_{z}^{(1)} \simeq 90.54$ (GeV). (6.10)

(As for the other parameters, see Eq. $(6\cdot13)$.) These solutions give the answer to the question: In what energy region should the search for W^{\pm} and Z bosons be performed? For example, concerning the Z boson search, the solutions $(6\cdot10)$ imply that a prominent peak will be observed at $\sqrt{s} \simeq 90.5 (GeV)$, not at $\sqrt{s} \simeq 88 (GeV)$ in e^+e^- collider (or dimuon invariant mass $M_{\mu\mu} \simeq 90.5 (GeV)$ in hadron-hadron reactions). It may be rather surprising that there appear considerable deviations (2.8% correction to the tree prediction). This is due to the large logarithmic terms $\sim \alpha \ln(M_{W,Z}^2/(m_r^2))$ or q^2) which exist, e.g., in the charge renormalization constant, Y. It is possible to sum up these logarithmic terms by renormalization group equation as is well known in perturbative QCD. We will argue this method in Chapter 7.

Several authors have also predicted W^{\pm} and Z boson masses at one-loop level. [Antonelli et al. 80 and 81a, Veltman 80, Aoki et al. 81, Sirlin and Marciano 81, Bardin et al. 82, Wheater and Llewellyn Smith 82]. Our results are consistent with the others although there are small differences which are considered to occur due to the different input parameters, e.g., quark masses.

Finally, let us consider another possibility to predict M_W once M_Z is experimentally measured. Among the input data used to determine the values

of the parameters, R^{\exp} includes a rather large experimental error. Present experimental uncertainties in R^{\exp} lead to ambiguities at least $\sim \pm (2.5-3.0)$ GeV in the determination of $M_{W}^{(0)}$ and $M_{Z}^{(0)}$ (consequently, for $M_{W}^{(1)}$ and $M_{Z}^{(1)}$).

On the other hand, M_z^{\exp} is expected to be measured with high accuracy (within ~0.1% [Davier 79]) in e^+e^- collider. Therefore, under such circumstances, it is worthwhile using the input data, (assumed) M_z^{\exp} , instead of R^{\exp} . ([Hioki 82b]. This possibility was also indicated in [Sirlin 80].) The use of this new input will improve the prediction for W^{\pm} boson mass. We examine this improvement.

The one-loop-corrected value $M_W^{(1)}$, and tree value $M_W^{(0)}$, are derived from

$$\Gamma^{(W)}(M_{W}^{(1)}, M_{Z})|_{M_{Z}=M_{Z}} = \Gamma^{(0)}(M_{W}^{(0)}, M_{Z})|_{M_{Z}=M_{Z}} = \Gamma^{\exp} \cdot (1 - \mathcal{A}_{\rm EM}).$$

$$(6.11)$$

We present the results in Table 6.1. We should keep in mind that uncertainties in this case are small enough to clarify the higher order effects, $|M_{W}^{(1)} - M_{W}^{(0)}|$.

M_z^{exp}	$M_{w}^{(0)}$	$M_{w}^{(1)}$
90.0	79.49	78.49
91.0	80.71	79.75
92.0	81.92	80.99
93.0	83.12	82.22
94.0	84.31	83.44
95.0	85.49	84.64
96.0	86.66	85.64

Table 6.1. Prediction for W^{\pm} boson mass (in GeV unit) by using M_z^{exp} and Γ^{exp} .

§ 6.2 Order α corrections to cross sections and decay width

In § 6.1 we have shown that there are rather large corrections for the predictions of W^{\pm} and Z boson masses. In this section, we study the size of the correction to various cross sections and decay width. As mentioned in Chapter 1, the Weinberg-Salam theory has succeeded in explanation of various phenomena by using tree level approximation. Therefore, it is important to study how the above success of the Weinberg-Salam theory is affected by the inclusion of the higher order effects. We deal with the pure QED part and the weak part separately in order to clarify the effect of weak part which is important for the field theoretical test of the Weinberg-Salam theory.

As a first step of calculating the radiative correction, the values of the renormalized parameters, e, M_W , M_Z , m_f and m_{ϕ} have to be fixed. If all the values of the masses of W^{\pm} , Z bosons, fermions and Higgs boson are measured, it is quite natural and straightforward to substitute the experimental data $M_W^{\exp p}$, $M_Z^{\exp p}$, $m_f^{\exp p}$ and $m_{\phi}^{\exp p}$ for the corresponding parameters respectively. Another possibility is to use the input data, $\Gamma^{\exp p}$ and $R^{\exp p}$ instead of $M_W^{\exp p}$ and $M_Z^{\exp p}$ as mentioned in § 6.1. Below we study the radiative correction according to the above two cases.

a) (M_w, M_z) case

In this case, we can immediately obtain the output (correction). We denote the weak radiative corrections as

$$\mathcal{A}_{W} = \{ \sigma^{(W)}(M_{W}^{exp}, M_{Z}^{exp}) - \sigma^{(0)}(M_{W}^{exp}, M_{Z}^{exp}) \} / \sigma^{(0)}(M_{W}^{exp}, M_{Z}^{exp}) .$$
(6.12)

Here we assume simply $M_W^{exp} = 77 \text{ GeV}$ and $M_Z^{exp} = 88 \text{ GeV}$. The results are shown in Table 6.2. (The results are found to change scarcely even if we take the values given in Eq. (6.10) for M_W^{exp} and M_Z^{exp} .) As for the pure QED correction, we have shown there two kinds of corrections $\mathcal{J}_{\text{EM}}^{\text{soft}}$ and \mathcal{J}_{EM} . $\mathcal{J}_{\text{em}}^{\text{soft}}$ is the correction only with contribution of soft-photon emission, which depends on the maximum soft-photon energy, ω . \mathcal{J}_{EM} is the correction including not only soft-photon effect but also hard-photon contribution. A detailed description on these pure QED corrections is seen in [Byers et al. 79, Green and Veltman 80, Aoki and Hioki 81, Hioki 82a].

In the actual calculation we have used the following values for the parameters^{*)} (in GeV unit for masses):

$$m_{\nu_e} = m_{\nu_{\mu}} = m_{\nu_{\tau}} = 0, \quad m_e = 5.11 \times 10^{-4},$$

$$m_{\mu} = 1.057 \times 10^{-1}, \quad m_{\tau} = 1.782, \quad m_u = m_d = m_s = 0.1,$$

$$m_e = 1.5, \quad m_b = 4.7, \quad m_t = 30 \quad \text{and} \quad m_{\phi} = 10,$$

$$\sin^2 \theta_{\rm C} \text{ (Cabibbo angle)} = 0.0562. \quad (6.13)$$

We have neglected all the other mixings.

There appear large weak corrections. We, however, notice that these large corrections are almost universal and energy independent for all the cross sections and the decay width. This fact implies that the most part of these effects may be absorbed into the re-normalization of parameters which are related to the overall normalization of the cross sections and width, that is,

^{*)} Concerning the values of $m_{u,d,s}$, see [Marciano and Sirlin 81].

Table 6.2. $O(\alpha)$ corrections with input M_{W}^{exp} and M_{z}^{exp} to the tree cross sections $\sigma(\nu_{\mu}e\rightarrow\nu_{\mu}e)$, ..., and decay width $\Gamma(\mu\rightarrow e\nu_{\mu}\overline{\nu}_{e})$. A_{W} is the weak correction, and $\mathcal{A}_{\rm EM}(\mathcal{A}_{\rm EM}^{ooft})$ is the pure QED correction with (without) hard-photon contribution. $\mathcal{A}_{\rm EM}^{\rm soft}$ depends on ω , the maximum soft-photon energy.

E,	⊿w	$\Delta_{\rm EM}^{\rm soft}(\omega=1~{\rm keV})$	$\Delta_{\rm EM}^{\rm soft}(\omega \!=\! 100 {\rm keV})$	$\varDelta_{\rm EM}$
0.1(GeV)	14.1(%)	-14.2	- 6.0	-1.1
1	14.1	-25.7	-12.6	-1.4
5	14.1	-35.2	-18.7	-1.7
15	14.1	-42.4	-23.6	-1.9
50	14.1	-51.0	-29.5	-2.0
100	14.1	-56.2	-33.3	-2.1
500	14.2	-69.2	-42.8	-2.4
1000	14.2	-75.2	-47.3	-2.5
5000	14.2	-89.9	-58.6	-2.8
10000	14.3	-96.6	-63.8	-2.9
				1

vµe→vµe

	$\overline{\mathcal{V}}_{\mu}e ightarrow \overline{\mathcal{V}}_{\mu}e$			
Eν	⊿w	$\Delta_{\rm EM}^{\rm soft}(\omega=1{\rm keV})$	$\int_{\rm EM}^{\rm soft} (\omega = 100 {\rm ~keV})$	⊿ _{EM}
0.1(GeV)	13.2(%)	-14.0	- 5.9	-1.0
1	13.2	-25.3	-12.4	-1.4
5	13.2	-34.8	-18.5	-1.6
15	13.2	-42.0	-23.4	-1.9
50	13.2	-50.6	-29.3	-2.0
100	13.2	-55.8	-33.0	-2.1
500	13.1	-68.8	-42.6	-2.4
1000	13.1	-74.7	-47.0	-2.5
5000	13.1	-89.4	-58.3	-2.7
10000	13.1	-96.1	-63.5	-2.9

ve-	>	UV.	
μu	1	MPE.	

E _v	⊿ _w	$\Delta_{\rm EM}^{\rm soft}(\omega=1~{\rm keV})$	$\Delta_{\rm EM}^{\rm soft}(\omega \!=\! 100 \rm keV)$	⊿ _{em}
15(GeV)	12.6(%)	-21.7	-12.1	-4.5
50	12.6	-27.0	-15.7	-2.6
100	12.6	-30.6	-18.2	-2.4
500	12.6	-40.8	-25.3	-2.5
1000	12.6	-45.8	-28.9	-2.6
5000	12.6	-58.4	-38.1	-2.8
10000	12.6	-64.2	-42.5	-2.9

⊿ _w	$\Delta_{\rm EM}^{\rm soft}(\omega=1{\rm keV})$	$\Delta_{\rm EM}^{\rm soft}(\omega = 100 {\rm keV})$	⊿ _{em}
12.6(%)	-17.9	-9.5	-0.4

 M_W and M_Z . This point has close connection to the calculation using the input data Γ^{\exp} and R^{\exp} as will be seen below.

b) (Γ, R) case

When we use the data Γ^{exp} and R^{exp} as input, we must first determine the values of the parameters, M_W and M_Z . This has been done in §6.1. The results are

$$M_{
m W}{}^{(0)}{\simeq}77~{
m (GeV)}$$
 , $M_{
m z}{}^{(0)}{\simeq}88~{
m (GeV)}$,

at tree level, and

$$M_{W}^{(1)} \simeq 79.17 \text{ (GeV)},$$

 $M_{Z}^{(1)} \simeq 90.54 \text{ (GeV)},$

at one-loop level.

Using these results, we can calculate the radiative correction as

$$\mathcal{A}_{\mathbf{W}} = \{ \mathcal{G}^{(\mathbf{W})}(M_{\mathbf{w}^{(1)}}, M_{\mathbf{z}^{(1)}}) - \mathcal{G}^{(0)}(M_{\mathbf{w}^{(0)}}, M_{\mathbf{z}^{(0)}}) \} / \mathcal{G}^{(0)}(M_{\mathbf{w}^{(0)}}, M_{\mathbf{z}^{(0)}}), \quad (6.14)$$

where $M_{W,Z}^{(0)}$ and $M_{W,Z}^{(1)}$ satisfy the equations

$$\Gamma^{(W)}(M_{W}^{(1)}, M_{Z}^{(1)}) = \Gamma^{(0)}(M_{W}^{(0)}, M_{Z}^{(0)}),$$

$$R^{(W)}(M_{W}^{(1)}, M_{Z}^{(1)})|_{E_{y}=5 \text{ GeV}} = R^{(0)}(M_{W}^{(0)}, M_{Z}^{(0)}).$$
(6.15)

The magnitudes of these corrections are very important for the test of the Weinberg-Salam theory because various tree predictions by using the input Γ^{exp} and R^{exp} are in good agreement with experimental data. If the Weinberg-Salam theory is the correct theory of electroweak interaction, the radiative corrections are expected to be small. We show the results in Table 6.3.

In the results, the weak corrections to μ decay width and $R \equiv \sigma(\bar{\nu}_{\mu}e \rightarrow \bar{\nu}_{\mu}e) / \sigma(\nu_{\mu}e \rightarrow \nu_{\mu}e)$ at $E_{\nu,\nu} = 5 \text{ GeV}$ are exactly zero from the condition, Eq. (6.15). Concerning the weak corrections, we have given the numerical results up to second decimal place to clarify their energy dependence.

The results show that the absolute values of the weak corrections, $|\mathcal{A}_{W}|$, are smaller than $|\mathcal{A}_{EM}|$ in this case. That is, the large corrections in Table 6.2 have been actually absorbed into the shifts $\mathcal{A}M_{W,Z} = M_{W,Z}^{(1)} - M_{W,Z}^{(0)}$. This supports the validity of the tree approximation.

 $\Delta_{\rm R}$ denotes the weak correction to the ratio R. The fact that the energy dependence of $\Delta_{\rm R}$ is weak shows that the calculations of $M_W^{(1)}$ and $M_Z^{(1)}$ (Eq. (6.9)) are not affected severely by the change of the energy where $R^{(1)}$ is calculated.

The experimental data of the neutral current processes are usually expressed in terms of the Weinberg angle, $\sin^2 \theta_w$. We have not used this para-

meter since it is not a convenient parameter once higher order effects are included. However, we can define *E*-dependent Weinberg angle, $\sin^2 \theta_W(E)$ according to the tree relation

$$R^{(0)} = \frac{16 \sin^4 \theta_{\rm W} - 4 \sin^2 \theta_{\rm W} + 1}{16 \sin^4 \theta_{\rm W} - 12 \sin^2 \theta_{\rm W} + 3},$$

as follows:

$$R^{(1)} = \frac{16 \sin^4 \theta_{\rm W}(E) - 4 \sin^2 \theta_{\rm W}(E) + 1}{16 \sin^4 \theta_{\rm W}(E) - 12 \sin^2 \theta_{\rm W}(E) + 3}.$$
(6.16)

Let us express the relation between $\sin^2\theta_W$ and $\sin^2\theta_W(E)$ as

$$\sin^2 \theta_{\mathbf{W}}(E) = \sin^2 \theta_{\mathbf{W}} \times (1 + \mathcal{A}_{\mathbf{SW}}). \tag{6.17}$$

That is, Δ_{sw} is the weak correction to $\sin^2\theta_w$. By combining Eq. (6.17) with

Table 6.3. $O(\alpha)$ corrections with input Γ^{\exp} and R^{\exp} to the tree cross sections and the width. $\mathcal{A}_{\mathbf{R}}$ is the weak correction to the ratio $R \equiv \sigma(\overline{\nu}_{\mu} e \rightarrow \overline{\nu}_{\mu} e) / \sigma(\nu_{\mu} e \rightarrow \nu_{\mu} e)$. The meaning of $\mathcal{A}_{\mathbf{EM}}$ is the same as those in Table 6.2.

 $\nu_{\mu}e \rightarrow \nu_{\mu}e$

E_{ν}	⊿ _w	⊿ _{EM}	$\Delta_{\rm EW}(=\Delta_{\rm EM}+\Delta_{\rm W})$
0.1(GeV)	0.85(%)	-1.1	-0.2
1	0.86	-1.4	-0.5
5	0.86	-1.7	-0.8
15	0.87	-1.9	-1.0
50	0.88	-2.0	-1.1
100	0.89	-2.1	-1.2
500	0.92	-2.4	-1.5
1000	0.93	-2.5	-1.6
5000	0.97	-2.8	-1.8
10000	0.99	-2.9	-1.9



$E\mathfrak{p}$	⊿w	⊿ _{EM}	$\Delta_{\rm EW}(=_{\rm EM}+\Delta_{\rm W})$
0.1(GeV)	0.88(%)	-1.0	-0.1
1	0.87	-1.4	-0.5
5	0.86	-1.6	-0.7
15	0.86	-1.9	-1.0
50	0.85	-2.0	-1.1
100	0.84	-2.1	-1.3
500	0.81	-2.4	-1.6
1000	0.80	-2.5	-1.7
5000	0.76	-2.7	-1.9
10000	0.74	-2.9	-2.2

(continued)

$E_{\nu_{\cdot}(\overline{\nu})}$	⊿ _R
0.1(GeV)	0.03(%)
1	0.01
5	0 (input)
15	-0.01
50	-0.03
100	-0.05
500	-0.11
1000	-0.13
5000	-0.21
10000	-0.25

$$\nu_{\mu}e \rightarrow \mu \nu_{e}$$

⊿w	⊿ _{em}	$\Delta_{\rm EW}(=\Delta_{\rm EM}+\Delta_{\rm W})$
0.00(%)	-4.5	-4.5
0.00	-2.6	-2.6
0.00	-2.4	-2.4
0.00	-2.5	-2.5
0.00	-2.6	-2.6
0.00	-2.8	-2.8
-0.01	-2.9	-2.9
	⊿ _w 0.00(%) 0.00 0.00 0.00 0.00 0.00 0.00 −0.01	Δ_{W} Δ_{EM} $0.00(\%)$ -4.5 0.00 -2.6 0.00 -2.4 0.00 -2.5 0.00 -2.6 0.00 -2.8 -0.01 -2.9

 $\mu \rightarrow e \nu_{\mu} \overline{\nu}_{e}$

⊿w	⊿ _{EM}	$\Delta_{\rm EW}(=\Delta_{\rm EM}+\Delta_{\rm W})$
0(%) (input)	-0.4	-0.4

Eq. (6.16), Δ_{sw} is explicitly given in terms of Δ_{R} as

$$\begin{aligned} \mathcal{A}_{\rm SW} &= -\frac{(16\sin^4\theta_{\rm W} - 4\sin^2\theta_{\rm W} + 1)(16\sin^4\theta_{\rm W} - 12\sin^2\theta_{\rm W} + 3)}{64\sin^4\theta_{\rm W}(2\sin^2\theta_{\rm W} - 1)} \mathcal{A}_{\rm R} \\ &\simeq 0.54 \mathcal{A}_{\rm R} \,. \end{aligned}$$
(6.18)

We can see that the effective Weinberg angle $\sin^2\theta_W(E)$ decreases as energy becomes higher. However, the magnitude of Δ_{SW} is so small that it will be difficult to check this behavior in ν -experiments. The effective Weinberg angle like this has been studied by many authors who have calculated the electroweak correction to the neutral current processes. (See § 7.1.)

Finally, concerning the dependence of the above results on unknown parameters, m_t and m_{ϕ} , we have examined the change of the results under the replacement, $m_t: 30 \rightarrow 100$ (GeV) and $m_{\phi}: 10 \rightarrow 100$ (GeV). For example, $\Delta M_{W,z}$ and the correction Δ_W to $\nu_{\mu}e \rightarrow \nu_{\mu}e$ at $E_{\nu} = 100$ (GeV) change as follows: K-I. Aoki, Z. Hioki, R. Kawabe, M. Konuma and T. Muta

$$\begin{split} \Delta M_{w} &= M_{w}^{(1)} - M_{w}^{(0)} : 2.167 \, (\text{GeV}) \rightarrow 2.221 \quad (m_{t} \rightarrow 100) \\ & \rightarrow 2.224 \quad (m_{\phi} \rightarrow 100) , \\ \Delta M_{z} &= M_{z}^{(1)} - M_{z}^{(0)} : 2.535 \, (\text{GeV}) \rightarrow 2.416 \quad (m_{t} \rightarrow 100) \\ & \rightarrow 2.634 \quad (m_{\phi} \rightarrow 100) , \\ \Delta_{w} : & 0.89 \, (\%) \quad \rightarrow 1.41 \quad (m_{t} \rightarrow 100) \\ & \rightarrow 0.82 \quad (m_{\phi} \rightarrow 100) . \end{split}$$

In conclusion, we have shown that the success of the Weinberg-Salam theory at tree analyses is not affected by the inclusion of higher order effects.

Note added in proof: After the submission of this article for publication, the authors were informed that i) L. Maiani arrived at the same conclusion on the test of the electroweak higher order effects by using the input data, α^{\exp} , Γ^{\exp} and M_{z}^{\exp} (L. Maiani, Roma Univ. Preprint n304, July 1982), and ii) S. Sarantakos et al. calculated the electroweak one-loop correction to νe scattering and obtained consistent results on the E.M. correction (including hard photon contribution) with the present ones (S. Sarantokos, A. Sirlin and W. J. Marciano, Preprint October 1982). The authors are grateful to Professor L. Maiani and Professor A. Sirlin for their information.

Chapter 7

Brief Survey of Studies on Electroweak Radiative Corrections

In Chapters 5 and 6 we have studied electroweak higher order effects in purely leptonic weak processes. The study of such effects is, in fact, important as a precise test of the Weinberg-Salam theory. Similar studies have also been performed by other authors and the results are presented in a variety of renormalization schemes. In this chapter we summarize the present status of these studies in order to understand the interrelation among them.

We classify these studies on electroweak radiative corrections into two categories: (1) studies based on conventional perturbation in powers of a fixed coupling constant and (2) studies based on leading-log approximation in the renormalization group method. We shall, in the following, discuss these two approaches separately.

§ 7.1 Conventional perturbation

The leptonic processes which have been studied with higher order corrections in the conventional perturbation are listed as follows:

$\nu_{\mu}e \rightarrow \nu_{\mu}e$ and $\bar{\nu}$	$\bar{\nu}_{\mu}e \rightarrow \bar{\nu}_{\mu}e$ [Salomonson and Ueda 75, Veltman 80, Antonelli
	et al. 80, Green and Veltman 80, Aoki et al. 80,
	Aoki et al. 81, Antonelli et al. 81a, Bardin et al.
	82]
$\nu_{\mu}e \rightarrow \mu\nu_{e}$	[Aoki et al. 81, Bardin et al. 82]
$\nu_{\mu}q \rightarrow \nu_{\mu}q$	[Marciano and Sirlin 80, Sakakibara 81, Llewellyn Smith
	and Wheater 81, Wheater and Llewellyn Smith 82]
$ u_{\mu}q \rightarrow \mu q'$	[Sirlin and Marciano 81, Paschos and Wirbel 82]
$eq \rightarrow eq$	[Llewellyn Smith and Wheater 81, Wheater and Llewel-
	lyn Smith 82]
$e^+e^- \rightarrow e^+e^-$	[Consoli 79]
$e^+e^- \rightarrow \mu^+\mu^-$	[Passarino and Veltman 79]
$e^+e^- \rightarrow W^+W^-$	[Lemoine and Veltman 80, Phillippe 82]
$\mu^- \rightarrow e^- \nu_\mu \bar{\nu}_e$	[Ross 73, Appelquist et al. 72 and 73, Bardin et al. 82,
	Hioki 82b]
$W^- \rightarrow e \bar{\nu}_e$	[Inoue et al. 80]
$Z \rightarrow f\bar{f}$	[Antonelli et al. 81b]
H (Higgs boson) $\rightarrow \tau^+ \tau^-$, $W^+ W^-$, ZZ [Fleischer and legerlehner 81]	

In these calculations diverse renormalization schemes are adopted and various

Downloaded from https://academic.oup.com/ptps/article-abstract/doi/10.1143/PTPS.73.1/1848359 by guest on 06 January 2019

choices of the set of independent parameters are made. Such differences in the calculational procedure, however, do not lead to essentially different physical predictions. In comparing the results of the calculations, the following two points should be clearly distinguished and carefully examined: (1) choice of independent parameters and (2) choice of input data needed to fix the values of independent parameters.

(1) Independent parameters

The choice of independent parameters is, in principle, arbitrary. It occurs, however, that some of the choices are more convenient in the sense that the definition of the parameters is unambiguous to all orders in perturbation series. Let us consider, as a typical example, the problem of the definition of the Weinberg angle $\theta_{\rm W}$.^{*)} The parameter $\sin \theta_{\rm W}$ is now widely used in analyzing neutral current phenomena. In fact $\sin \theta_{\rm W}$ is convenient at the tree level in the sense that the structure of amplitudes relevant to the neutral current takes a simple universal expression in terms of this parameter. This simplicity, however, does not persist once higher order corrections are taken into account. With the higher order corrections included, there is no unique and proper definition of $\sin \theta_{\rm W}$ and hence it is not convenient to adopt $\sin \theta_{\rm W}$ as one of the independent parameters. In the following we explain, by taking two examples of different definitions of $\sin \theta_{\rm W}$, how the simple tree relation $M_{\rm W} = M_Z \cos \theta_{\rm W}$ is modified by higher order corrections.

Example 1. A set of parameters^{**}) e, g and M_w was first employed by Appelquist et al. [Appelquist et al. 72 and 73] and later by Salomonson and Ueda [Salomonson and Ueda 75]. Here e is the electric charge renormalized at $q^2 = 0$, M_w the physical W-boson mass corresponding to the pole of the renormalized W-boson propagator, and g is the SU(2) coupling constant renormalized on the mass shell with the $W-\mu-\nu$ vertex function. The Weinberg angle is defined by***)

$$\sin \theta_{\rm W} = e/g \,. \tag{7.1}$$

The Z-boson mass is not an independent parameter and is calculated as a pole position in the Z-boson propagator, i.e., the solution of

$$q^2 - M_{Z_0}^2 - A_0(q^2) = 0 , \qquad (7 \cdot 2)$$

where M_{Z0} is the bare mass and $A_0(q^2)$ is the unrenormalized Z-boson selfenergy part. The counter term for the Z-boson self-energy part is settled

^{*)} In our scheme this problem is irrelevant since we do not take this parameter (also in [Inoue et al. 80] and [Bardin et al. 82]).

^{**)} We consider only three parameters in the present discussion and suppress all other independent parameters like μ^2 and f_i .

^{***)} Note that $\cos^2\theta_W$ is denoted as R in [Appelquist et al. 72 and 73].

as a function of e, g and M_w and Eq. (7.2) reduces to the following form at the one-loop level:

$$q^{2} - M_{W}^{2} / (1 - e^{2}/g^{2}) - A(q^{2}) = 0$$
(7.3)

with $A(q^2)$ the renormalized Z-boson self-energy part. Solving Eq. (7.3), one finds

$$M_{Z} = M_{W} / \sqrt{1 - e^{2}/g^{2}} + \Delta(e, g, M_{W}), \qquad (7 \cdot 4)$$

where Δ is a function of e, g and M_w calculable once $A(q^2)$ is known and is of order e^2 (or g^2). Thus the relation $M_w = M_z \cos \theta_w$ is violated in this renormalization scheme.

It should also be noted that the W- μ - ν vertex function on the mass shell includes the infrared divergence and hence g depends on the cutoff of photon energy after the cancellation of the infrared divergence. In this sense the scheme discussed in this example is rather inconvenient.

Example 2. In the renormalization scheme employed by Sirlin [Sirlin 80], g, g' and v are chosen as independent parameters, where g is the SU(2) coupling constant as before, g' the U(1) coupling constant and v the vacuum expectation value of the Higgs scalar. Other parameters $\theta_{\rm W}, M_{\rm W}, M_{\rm Z}$ and e are defined as functions of g, g' v such that

$$\tan \theta_{\rm W} \equiv g'/g ,$$

$$M_{\rm W}^2 \equiv v^2 g^2/4 ,$$

$$M_{\rm Z}^2 \equiv v^2 (g^2 + g'^2)/4 ,$$

$$e \equiv gg'/(g^2 + g'^2)^{1/2} = g \sin \theta_{\rm W} . \qquad (7.5)$$

In dealing with loop corrections, counter terms for g, g' and v are determined by equating M_W and M_z (as defined by Eq. (7.5)) to the pole positions in the W- and Z-boson propagators respectively. The constant e defined by Eq. (7.5) is identified with the electric charge. As a result the relation $M_W = M_z \cos \theta_W$ holds to any order in the perturbation series.

With the above two examples we now recognize that the actual meaning of the parameter θ_{W} differs from scheme to scheme once the loop corrections are taken into account.

(2) Input data

The numerical values of independent parameters are determined by using experimental data on some measurable quantities. The resulting numerical values are affected by the choice of the quantities as experimental data as far as the truncated perturbation theory is concerned.

Let us assume that radiative corrections to cross sections of processes A

and B, σ_A and σ_B , respectively in a theory with coupling constant g. Denote the tree cross section by $\sigma_A^0(g)$ ($\sigma_B^0(g)$), the cross section with one-loop correction by $\sigma_A^{-1}(g)$ ($\sigma_B^{-1}(g)$), and so on. One may think of two possibilities of determining the value of g.

One way is to take the experimental data on process A (denoted by σ_{A}^{exp}) to fix the value of g in such a way that

$$\sigma_{\mathrm{A}}^{0}(g_{\mathrm{A}}^{0}) = \sigma_{\mathrm{A}}^{\exp}, \quad \sigma_{\mathrm{A}}^{1}(g_{\mathrm{A}}^{1}) = \sigma_{\mathrm{A}}^{\exp}, \quad \cdots, \quad (7 \cdot 6)$$

where g_{A^0} , g_{A^1} , \cdots represent values of g determined by the condition (7.6). With this coupling constant determined by Eq. (7.6), we can calculate radiative corrections in process B, e.g.,

$$\left[\sigma_{\rm B}^{-1}(g_{\rm A}^{-1}) - \sigma_{\rm B}^{0}(g_{\rm A}^{0})\right] / \sigma_{\rm B}^{0}(g_{\rm A}^{0}) . \tag{7.7}$$

This correction is, in general, nonzero while the correction for process A vanishes exactly because of Eq. $(7 \cdot 6)$.

The other way is to use the data on process B and the procedure is essentially the same as above if we exchange A for B. We denote by g_{B^0} , g_{B^1} , ... the coupling constants corresponding to the previous ones g_{A^0} , g_{A^1} ,

The important observation here is that, as far as we rely on the truncated perturbation theory, the value of the coupling constant determined as above depends on how one determines it, i.e., $g_A{}^i \neq g_B{}^i$ $(i=0, 1, 2\cdots)$. The difference between $g_A{}^i$ and $g_B{}^i$ lies in higher order terms of order g^{i+1} .

§ 7.2 Leading-log calculation

In the presence of two hierarchically different mass scales μ and μ' with $\mu' \gg \mu$, large logarithmic terms of the form $(\alpha \ln (\mu'/\mu))^n$ show up in conventional perturbation series where α is an expansion parameter. A typical example of such large-log terms is already met in QCD, where summing up the logarithmic terms to all orders is necessary as the expansion parameter α is not so small and the renormalization group method is very useful to deal with this problem.

In electroweak theory we also have two hierarchically different sets of mass scales in low energy phenomena, i.e., (M_w, M_z) and low energy variables like incident energy and momentum transfer. Here the coupling constant α is very small and so the logarithmic terms do not cause any serious problem. It is, however, still useful to use the renormalization group method to pick up dominant contributions. A study in this direction was put forward by Marciano in the discussion of electroweak higher order effects [Marciano 79]. The formulation based on operator product expansion was recently proposed by Rome group [Antonelli and Maiani 81, Bellucci et al. 81] and by Harvard group [Dawson et al. 81]. We shall briefly survey these works in the
following.

The effective electroweak Hamiltonian for low energy processes characterized by mass scale μ may be expressed by

$$H(\mu) = \sum_{i} C_{i} \left(\frac{M}{\mu}, \alpha(\mu) \right) O_{i}, \qquad (7.8)$$

where O_i are composite operators relevant to the process under consideration and M is a mass scale of order M_W (or M_Z) and $\alpha(\mu)$ is the fine structure constant defined at the scale μ . In Eq. (7.8) all corrections suppressed by inverse powers of M are neglected [Kazama and Yao 80a, 80b, 80c, 82]. As we consider here only the effective theory, the W- and Z-boson propagators are contracted and radiative corrections due to the photon propagator are relevant.

The coefficient function $C_i(M/\mu, \alpha(\mu))$ satisfies the renormalization group equation

$$\left[\mu\frac{\partial}{\partial\mu}+\beta(\alpha)\frac{\partial}{\partial\alpha}-\gamma_{i}(\alpha)\right]C_{i}\left(\frac{M}{\mu},\alpha(\mu)\right)=0, \qquad (7\cdot9)$$

where $\beta(\alpha)$ is the so-called β -function for the electromagnetic coupling and $\gamma_i(\alpha)$ is the anomalous dimension of operator O_i . They are defined as follows:

$$\beta(\alpha) = \mu \frac{d\alpha}{d\mu}, \quad \gamma_i(\alpha) = \frac{\mu}{Z_i} \frac{\partial Z_i}{\partial \mu}\Big|_{\text{bare } \alpha \text{ fixed}}$$
(7.10)

with O_i (renormalized) = $Z_i^{-1}O_i$. Here we consider only the case where there is no operator mixing. Solving the renormalization group equation for the coefficient function $C_i(M/\mu, \alpha(\mu))$, we obtain

$$C_{i}\left(\frac{M}{\mu},\alpha(\mu)\right) = C_{i}(1,\alpha(M)) \exp\left[-\int_{\alpha(\mu)}^{\alpha(M)} dx \frac{\gamma_{i}(x)}{\beta(x)}\right]. \quad (7.11)$$

Calculating γ_i and β perturbatively

$$\gamma_{i}(\alpha) = \gamma_{i}^{0} \alpha + \gamma_{i}^{1} \alpha^{2} + \cdots,$$

$$\beta(\alpha) = \beta_{0} \alpha^{2} + \beta_{1} \alpha^{3} + \cdots,$$
(7.12)

we obtain an improved perturbation series for the coefficient function,

$$C_{i}\left(\frac{M}{\mu}, \alpha(\mu)\right) = C_{i}(1, \alpha(M)) \left[\frac{\alpha(M)}{\alpha(\mu)}\right]^{-\gamma_{i}^{0}/\beta_{0}} \times \left[1 + \frac{\beta_{i}}{\beta_{0}}\left(\frac{\gamma_{i}^{0}}{\beta_{0}} - \frac{\gamma_{i}^{1}}{\beta_{1}}\right)\alpha(M) + \cdots\right].$$
(7.13)

Here $C_i(1, \alpha(M))$ can also be calculated perturbatively in $\alpha(M)$ with no

large logarithmic term.

In charged-current leptonic processes $\nu_{\mu}e \rightarrow \mu\nu_{e}$ and $\mu \rightarrow e\nu_{\mu}\bar{\nu}_{e}$, the relevant operators are

$$O_{1} = \bar{\nu}_{\mu} \gamma_{\lambda} (1 - \gamma_{5}) \nu_{e} \cdot \bar{e} \gamma^{\lambda} \mu ,$$

$$O_{2} = \bar{\nu}_{\mu} \gamma_{\lambda} (1 - \gamma_{5}) \nu_{e} \cdot \bar{e} \gamma^{\lambda} \gamma_{5} \mu . \qquad (7.14)$$

The coefficient functions corresponding to these operators are calculated to lowest order and the anomalous dimensions γ_1 and γ_2 to one-loop order. The relevant diagrams in one-loop order contributing to γ_1 and γ_2 are shown in Fig. 7.1. It is well-known that the contribution of Fig. 7.1 is ultraviolet convergent and hence there is no renormalization of operators O_1 and O_2 . Accordingly $\gamma_1 = \gamma_2 = 0$. Thus we have

$$C_{i}\left(\frac{M}{\mu}, \alpha(\mu)\right) = C_{i}(1, \alpha(M)) \simeq C_{i}(1, 0). \qquad (7.15)$$

The explicit calculation shows that

$$C_1(1,0) = -C_2(1,0) = \frac{g^2(M)}{8M_w^2} = \frac{G_F}{\sqrt{2}}.$$
(7.16)

In neutral-current processes $\nu_{\mu}f \rightarrow \nu_{\mu}f$ (f = fermions) and charged-current process $\nu_{\mu}d \rightarrow \mu u$, the anomalous dimensions are not necessarily zero.



Fig. 7.1. The photonic correction for the four fermion interaction $\nu_{\mu}e \rightarrow \nu \mu_e$.

Let us consider the $\nu_{\mu}d \rightarrow \mu u$ process first. The relevant operators are

$$O_{1} = \overline{\mu} \gamma_{\lambda} (1 - \gamma_{5}) \nu_{\mu} \cdot \overline{u} \gamma^{\lambda} d,$$

$$O_{2} = \overline{\mu} \gamma_{\lambda} (1 - \gamma_{5}) \nu_{\mu} \cdot \overline{u} \gamma^{\lambda} \gamma_{5} d.$$
(7.17)

Feynman diagrams contributing to the one-loop anomalous dimension are shown in Fig. 7.2. The full contribution of Fig. 7.2 is ultraviolet divergent and requires a renormalization.

In the case of the process $\nu_{\mu}f \rightarrow \nu_{\mu}f$, the relevant operators are

$$O^{i} = \bar{\nu}_{\mu} \gamma_{\lambda} (1 - \gamma_{5}) \nu_{\mu} \cdot \bar{f} \gamma^{\lambda} A^{(i)} f,$$

$$O_{5}^{i} = \bar{\nu}_{\mu} \gamma_{\lambda} (1 - \gamma_{5}) \nu_{\mu} \cdot \bar{f} \gamma^{\lambda} \gamma_{5} A_{5}^{(i)} f,$$



Fig. 7.2. The photonic correction for the four fermion interaction $\nu_{\mu}d \rightarrow \mu u$.



Fig. 7.3. The photonic correction for the four fermion interaction $\nu_{\mu}f \rightarrow \nu_{\mu}f$.

where $A^{(i)}$ and $A_5^{(i)}$ denote τ_s , Q, 1, etc. The relevant diagrams are shown in Fig. 7.3. In Fig. 7.3 the contributions of the first three diagrams are ultraviolet convergent while the final diagram creates a new ultraviolet divergence (the fourth diagram for operator O_5^i is convergent due to the presence of γ_5).

In order to see the significance of the leading-log summation in electroweak theory, we quote gauge-boson mass shifts calculated by Antonelli and Maiani in Table 7.1 [Antonelli and Maiani 81]. Here the leading-log correction ΔM_W (leading-log), is obviously very important while the correction of order α^2 , $\Delta M_W(\alpha^2 \ln^2 M)$, is small and comparable to the non-logarithmic correction of order α , $\Delta M_W(\alpha)$. It should be mentioned here that to incorpolate the nonlogarithmic terms into the renormalization group argument, one should deal with the next-to-leading terms as seen in Eq. (7.13) and calculate two-loop anomalous dimensions.

Га	bl	e	7	•	1	•
----	----	---	---	---	---	---

$M_{\scriptscriptstyle W}^{\scriptscriptstyle (0)}$	$\begin{array}{c c} & \Delta M_{W} \\ (\text{leading-log}) \end{array}$	$\Delta M_{W}(\alpha)$	$\Delta M_{W}(\alpha^{2} \ln^{2} M)$
79.48 (GeV)	3.50	-0.70	0.20
77.74	3.27	-0.65	0.18
76.10	3.07	-0.61	0.16
$M_Z^{(0)}$	$\frac{\Delta M_z}{(\text{leading-log})}$	$\Delta M_{z}(\alpha)$	$\Delta M_Z(\alpha^2 {\ln^2} M)$
90.00 (GeV)	3.83	-0.40	0.24
88.59	3.64	-0.36	0.21
87.29	3.46	-0.33	0.20

Chapter 8

Conclusions and Outlook

In the present review article, we have developed the on-shell renormalization procedure in the Weinberg-Salam theory and given a proof of the charge universality. The procedure has been applied to the study of higher order effects in purely leptonic weak processes $\nu_{\mu}e \rightarrow \nu_{\mu}e$, $\bar{\nu}_{\mu}e \rightarrow \bar{\nu}_{\mu}e$, $\nu_{\mu}e \rightarrow \mu\nu_{e}$ and $\mu \rightarrow e\nu_{\mu}\bar{\nu}_{e}$. In the course of these investigations we have met with several interesting problems of theoretical and phenomenological importance. We shall briefly recapitulate these problems here.

(1) On-shell renormalization We have proposed the on-shell renormalization scheme of the Weinberg-Salam theory as a most convenient and physically significant scheme and have developed the renormalization procedure in this scheme. The consistency of this scheme has been explicitly proved by using the Ward-Takahashi identities especially in the neutral gauge boson sector. The structure of the theory has been clarified paying special attention to the relation between physical particles and corresponding Heisenberg fields.

(2) Charge universality While the proof of the charge universality is straightforward in quantum electrodynamics, the proof is nontrivial in gauge theories with spontaneous symmetry breakdown such as the Weinberg-Salam theory. One would naturally expect that the charge universality be satisfied in the Weinberg-Salam theory. By the lack of the proof, however, one could not safely perform higher order calculations. Hence it is worth spending a considerable amount of time to prove it. We have given a lengthy but transparent proof of the charge universality in §3.4 by the use of the Ward-Takahashi identity.

(3) Scheme dependence There is a variety of the renormalization schemes employed by many authors. The difference of the schemes should not lead to different physical predictions since the choice of a scheme is a matter of convention which is unphysical. In practical calculations, however, we truncate perturbation series and obtain expressions differing from each other by neglected higher-order terms. This situation creates some confusion when one tries to make a comparison of results obtained in different schemes. We have tried to clarify this point as much as possible in §§ 3.1 and 7.1. In particular there are two important ingredients in the problem of renormalization schemes: the definition of renormalized parameters and the choice of input data. In choosing independent parameters, one could pick up a parameter (or parameters) which is not directly measurable. For example, the Weinberg angle θ_{w} is not a directly measurable quantity although it is quite

a convenient parameter at the tree level. Beyond the tree level the definition of $\sin \theta_W$ is not unique and its definition must be clearly stated when two results obtained by different schemes are compared. We have adopted the onshell renormalization scheme throughout the present review article with the choice of independent parameters α , M_W and M_Z (and also fermion and Higgs masses which we omit for a moment). Since our parameters are all directly measurable in experiments, we have no such ambiguity of the definition of parameters.

In order to fix these independent parameters, we must choose the sufficient number of input experimental data. If the data on M_W and M_Z were known, our scheme would be the most convenient to perform higher order calculations. In the present situation one is forced to take some other choices of input data. Different choices of input data may lead to quite a large discrepancy in the size of radiative corrections. This apparent disagreement can, however, be consistently explained by careful examination on the choice of parameters fixed in the calculation of radiative corrections. The careful analysis of this problem has been made in §§ 6.1 and 6.2.

(4) Large logarithmic corrections As discussed in Chapter 6, we have separated the electroweak radiative correction into two parts, the electromagnetic part and the rest. We have obtained, in our scheme, rather large corrections concerning the latter part. The origin of these large corrections may be traced back to the large logs of the type $lpha \ln{(M^2/q^2)}$ and $lpha \ln{(M^2/m_f^2)}$ where Mstands for M_W or M_Z and m_f for light fermion masses. The large logs of this type are harmful to perturbative calculations and so they have to be summed to all orders. The renormalization group method is quite useful for this purpose just as it was in perturbative QCD. The renormalization group analyses of the leading log contribution to electroweak processes were carried out by some groups as described in §7.2. According to the analysis by the Rome group mentioned there, the magnitudes of the leading log terms of order α , $\left[\alpha \ln \left(M^2/q^2(m_f^2)\right)\right]$, those of the non-log terms of order α , $\left[\alpha\right]$, those of the leading log terms of order α^2 , $[\alpha^2 \ln^2(M^2/q^2(m_f^2))]$, etc., are in the following order,

$$[\alpha \ln(M^2/q^2(m_f^2))] > [\alpha] > [\alpha^2 \ln^2(M^2/q^2(m_f^2))].$$

This means that the effect of the leading log sum to all orders is less important than that of non-log terms. Thus the conventional perturbation to one-loop order which takes care of terms $[\alpha \ln M]$ and $[\alpha]$ is complementary to the leading log sum by renormalization group method which takes into account terms of $[\alpha \ln M]$, $[\alpha^2 \ln^2 M]$, It would be interesting to estimate the next-to-leading log contribution in the renormalization group method where two-loop anomalous dimensions of relevant operators must be calculated.

(5) Real photon emission The calculation of radiative corrections to

exclusive electroweak processes necessarily includes infrared (soft) divergences due to the vanishing photon mass. There is another source of divergences if some of the charged lepton masses vanish. This is the so-called mass singularity. Actually the lepton has small but nonvanishing mass m_l and the singularity takes the form $\ln^2(q^2/m_l^2)$ and $\ln(q^2/m_l^2)$. As is well-known, both of the soft divergence and the mass singularity are removed by adding real photon emission effects. The soft divergence is cancelled by the low energy part of the real photon effect while the mass singularity is cancelled by the collinear configuration of the emitted real photon and the lepton. The hard photon effect, i.e., the collinear part of the real photon effect is more complicated to calculate than the low energy part of the real photon effect. In order to compare the calculated results with experimental data, however, it is indispensable to include the hard photon correction. We have performed this calculation in processes including neutrinos and presented the result in § 6.2.

We have summarized here important problems some of which deserve further discussion. We believe that our investigation in the present review article is quite inclusive and presents all the necessary tools to discuss radiative corrections to electroweak processes.

By using low energy data, main independent parameters are determined to one-loop order. In fact the values of M_W and M_Z to one-loop order are now known,

 $M_w = 79.2 \text{ GeV}$, $M_z = 90.5 \text{ GeV}$.

With a proper definition of the Weinberg angle $\theta_{\rm W}$, we can also fix the value of $\sin^2 \theta_{\rm W}$ at the mass scale M. Thus we have settled the necessary information on the values of relevant parameters at mass scale M in grand unification theories. Once the values of the parameters at mass scale M are fixed, the corresponding values at the grand unification mass scale are easily calculated by the renormalization group method.

Throughout the paper the Higgs particles are treated as elementary scalar fields. If the spontaneous breakdown of the $SU(2) \times U(1)$ symmetry is generated dynamically, the Higgs particles are composite systems of some fundamental object. This possibility has been studied by many authors and has led to models of composite quarks and leptons. If this is the case, our analysis in the present article is considered to be an effective low energy theory for electroweak interactions.

We have confined ourselves to the Weinberg-Salam theory of electroweak interactions and expected the existence of the W- and Z-boson. Experimentally it is of course an open problem whether the W- and Z-boson will be

observed or not. Anticipated experiment in search for W and Z is a great challenge to the present electroweak theory. If W- and Z-boson will not be observed, there will emerge ages for theorists.

Acknowledgements

The authors wish to thank Professor T. Kugo for valuable advice and encouraging discussions. They are grateful to Professor T. Kinoshita, Professor M. Kobayashi and Dr. S. Sakakibara for useful comments and to Professor Y. Nagashima for experimental information. Advice of Dr. T. Niuya on numerical calculations proved most helpful. The authors also thank Professor M. T. Veltman and Dr. M. Green for correspondences concerning hard photon corrections. Two of the authors (K-I. A. and Z. H.) acknowledge the financial support provided by the Japan Society for the Promotion of Science.

Appendix A

Dimensional Regularization

Dimensional regularization ['t Hooft and Veltman 72a] is widely used in calculations of radiative corrections. Since an analytic continuation in the space-time dimensions is not unique, there is a variety of conventions in this method. We adopt the following convention: $\text{Tr}(\mathbf{1}) = 4$, $\{\gamma_{\mu}, \gamma_{5}\} = 0$ and $\int d^{p}k/(2\pi)^{p}i$ in *D*-dimensional space. Below we list various formulas in our convention.

A.1 γ matrix algebra in *D*-dimensional space

The basic algebra is

$$\{\gamma_{\mu},\gamma_{\nu}\}=2g_{\mu\nu}. \qquad (\mathbf{A}\cdot\mathbf{1})$$

Metric tensor $g_{\mu\nu}$ satisfies

$$g_{\mu\nu}g^{\mu\nu} = D . \qquad (\mathbf{A} \cdot 2)$$

Combining $(A \cdot 1)$ and $(A \cdot 2)$, we obtain

$$\gamma_{\lambda}\gamma^{\lambda} = D, \qquad (A \cdot 3)$$

$$\gamma_{\lambda}\gamma_{\mu}\gamma^{\lambda} = (2-D)\gamma_{\mu}, \qquad (A\cdot 4)$$

$$\gamma_{\lambda}\gamma_{\mu}\gamma_{\nu}\gamma^{\lambda} = 4g_{\mu\nu} + (D-4)\gamma_{\mu}\gamma_{\nu}, \qquad (A\cdot 5)$$

$$\gamma_{\lambda}\gamma_{\mu}\gamma_{\nu}\gamma_{\rho}\gamma^{\lambda} = -2\gamma_{\rho}\gamma_{\nu}\gamma_{\mu} + (4-D)\gamma_{\mu}\gamma_{\nu}\gamma_{\rho} . \qquad (A\cdot 6)$$

Further, by using our convention on unit matrix 1,

$$\mathrm{Tr}(\mathbf{1}) = 4, \qquad (\mathbf{A} \cdot 7)$$

we find

$$\operatorname{Tr}\left(\gamma_{\mu}\gamma_{\nu}\right) = 4g_{\mu\nu}, \qquad (\mathbf{A}\cdot\mathbf{8})$$

$$\operatorname{Tr}\left(\gamma_{\mu}\gamma_{\nu}\gamma_{\lambda}\gamma_{\rho}\right) = 4\left(g_{\mu\nu}g_{\lambda\rho} + g_{\mu\rho}g_{\nu\lambda} - g_{\mu\lambda}g_{\nu\rho}\right). \tag{A.9}$$

As mentioned above, the γ_5 matrix is defined so that it satisfies

$$\{\gamma_{\mu}, \gamma_5\} = 0. \qquad (\mathbf{A} \cdot \mathbf{10})$$

There occurs no trouble concerning the γ_5 matrix in the present case, since the Weinberg-Salam theory is an anomaly-free theory.

189

A.2 Momentum integral

Momentum integral is extended as

$$\int \frac{d^4k}{(2\pi)^4 i} \rightarrow \int \frac{d^p k}{(2\pi)^p i}. \qquad (A.11)$$

This Minkowski-space integral is performed after Wick rotation into Euclidean space, $k_0 = iK_0$.

Appendix A

$$\int d^{p} K = \int dK_{0} dK_{1} \cdots dK_{D-1}$$
$$= \int K^{D-1} dK d\Omega_{D}, \quad (K^{2} \equiv \sum_{i} K_{i} K^{i})$$
(A·12)

$$\int d\mathcal{Q}_{D} = \int_{0}^{\pi} d\theta_{1} \sin^{D-2}\theta_{1} \int_{0}^{\pi} d\theta_{2} \sin^{D-3}\theta_{2} \cdots \int_{0}^{2\pi} d\theta_{D-1}$$
$$= 2\pi^{D/2} \left[\Gamma\left(\frac{D}{2}\right) \right]^{-1}.$$
(A·13)

We list below several typical integral formulas:

$$\int d^{\mathcal{D}}k \cdot k_{\mu} f(k^2) = 0 . \qquad (A \cdot 14)$$

$$\int d^{D}k \cdot k_{\mu}k_{\nu}f(k^{2}) = \frac{1}{D}g_{\mu\nu}\int d^{D}k \cdot k^{2}f(k^{2}).$$
 (A·15)

$$\int d^{\mathcal{D}}k \frac{1}{(-k^2+Z)^{\alpha}} = i\pi^{\mathcal{D}/2} \frac{\Gamma\left(\alpha - (D/2)\right)}{\Gamma\left(\alpha\right)} Z^{((\mathcal{D}/2)-\alpha)}.$$
(A·16)

$$\int d^{D}k \frac{-k^{2}}{(-k^{2}+Z)^{\alpha}} = i\pi^{D/2} \frac{D}{2} \frac{\Gamma(\alpha-1-(D/2))}{\Gamma(\alpha)} Z^{(1+(D/2)-\alpha)} .$$
(A·17)

$$\int d^{D}k \frac{1}{\left(-k^{2}\right)^{\alpha}\left\{-\left(k+q\right)^{2}\right\}^{\beta}} = i\pi^{D/2}\left(-q^{2}\right)^{-\varepsilon+2-\alpha-\beta} \frac{\Gamma\left(\varepsilon-2+\alpha+\beta\right)}{\Gamma\left(\alpha\right)\Gamma\left(\beta\right)}$$

$$\times B(2-\alpha-\varepsilon,2-\beta-\varepsilon).$$
 (A·18)

$$\int d^{D}k \frac{k_{\mu}}{(-k^{2})^{\alpha} \{-(k+q)^{2}\}^{\beta}}$$

= $-i\pi^{D/2}q_{\mu}(-q^{2})^{-\varepsilon+2-\alpha-\beta} \frac{\Gamma(\varepsilon-2+\alpha+\beta)}{\Gamma(\alpha)\Gamma(\beta)}B(3-\alpha-\varepsilon,2-\beta-\varepsilon).$
(A·19)

$$\int d^{D}k \frac{k_{\mu}k_{\nu}}{(-k^{2})^{\alpha} \{-(k+q)^{2}\}^{\beta}} = i\pi^{D/2} (Ag_{\mu\nu} + Bq_{\mu}q_{\nu}), \qquad (A \cdot 20)$$

K-I. Aoki, Z. Hioki, R. Kawabe, M. Konuma and T. Muta

$$A = -(-q^{2})^{-\varepsilon+3-\alpha-\beta} \frac{\Gamma(\varepsilon-3+\alpha+\beta)}{2\Gamma(\alpha)\Gamma(\beta)} B(3-\alpha-\varepsilon,3-\beta-\varepsilon), \qquad (A\cdot 20a)$$

$$B = (-q^2)^{-\varepsilon+2-\alpha-\beta} \frac{\Gamma(\varepsilon-2+\alpha+\beta)}{\Gamma(\alpha)\Gamma(\beta)} B(4-\alpha-\varepsilon,2-\beta-\varepsilon).$$
 (A·20b)

 $\Gamma(x)$ and B(x, y) represent the Gamma and Beta functions respectively.

$$\varepsilon = \frac{1}{2} (4 - D) \rightleftharpoons \frac{D}{2} \equiv 2 - \varepsilon,$$
 (A·21)

$$\Gamma(\varepsilon) = \frac{1}{\varepsilon} - \gamma + O(\varepsilon), \qquad (A \cdot 22)$$

$$\Gamma(\varepsilon - 1) = -\frac{1}{\varepsilon} - (1 - \gamma) + O(\varepsilon), \qquad (A \cdot 23)$$

(γ is the Euler constant)

$$B(x, y) = \frac{\Gamma(x)\Gamma(y)}{\Gamma(x+y)} = \int_{0}^{1} d\xi \,\xi^{x-1} (1-\xi)^{y-1}$$
$$= \int_{0}^{\infty} dt \frac{t^{y-1}}{(1+t)^{x+y}} = \int_{0}^{\infty} dt \frac{t^{x-1}}{(1+t)^{x+y}}.$$
 (A·24)

Appendix B

Formulas for One-Loop Feynman Amplitudes

We give one-loop Feynman amplitudes expressed in the form of Feynman parameter integration. They are obtained by performing the loop momentum integral. We drop some overall factors in order to make formulas simple.

In the following, we express the ultraviolet divergent part as

$$C_{\rm UV} \equiv \frac{1}{\varepsilon} - \gamma + \ln 4\pi \tag{B.1}$$

and we set $D \rightarrow 4$ for the convergent parts.

B.1 Two-point functions

Here we use the following function,

$$D_2(x) = m_1^2(1-x) + m_2^2 x - q^2 x (1-x), \qquad (B\cdot 2)$$

$$\gamma(A, B) \equiv (1 + AB) - (A + B)\gamma_5, \qquad (B \cdot 3)$$

where m_1 , m_2 , and q are shown in Fig. B.1.



Fig. B.1. Typical diagrams of two-point functions. k is the loop momentum and q is the external momentum. m_1, m_2 and m denote the masses of particles of internal lines. x and (1-x) in (a) show our choice of Feynman parameter.

$$\sum \left\{ \begin{array}{l} \Sigma(q) \\ \sim \int \frac{d^{D}k}{(2\pi)^{D}i} \gamma_{\alpha} (1+A\gamma_{5}) \frac{m_{2}+q-k}{m_{2}^{2}-(q-k)^{2}} \gamma_{\beta} (1+B\gamma_{5}) \frac{g^{\alpha\beta}}{k^{2}-m_{1}^{2}} \\ = -\frac{1}{8\pi^{2}} m_{2} \gamma(A, -B) \left\{ 2C_{\rm UV} - 1 - 2 \int_{0}^{1} dx \ln(D_{2}(x)) \right\} \\ + \frac{1}{16\pi^{2}} q \gamma(-A, -B) \left\{ C_{\rm UV} - 1 - 2 \int_{0}^{1} dx (1-x) \ln(D_{2}(x)) \right\}.$$

$$(B \cdot 4)^{*}$$

$$\sum \left\{ \begin{array}{l} \sum \left\{ q \right\} \\ \sim \int \frac{d^{D}k}{(2\pi)^{D}i} (1 + A\gamma_{5}) \frac{m_{2} + q - k}{m_{2}^{2} - (q - k)^{2}} (1 + B\gamma_{5}) \frac{1}{m_{1}^{2} - k^{2}} \\ = \frac{1}{16\pi^{2}} m_{2}\gamma \left(-A, -B \right) \left\{ C_{\rm UV} - \int_{0}^{1} dx \ln \left(D_{2}(x) \right) \right\} \\ + \frac{1}{32\pi^{2}} q\gamma \left(A, -B \right) \left\{ C_{\rm UV} - 2 \int_{0}^{1} dx \left(1 - x \right) \ln \left(D_{2}(x) \right) \right\}. (B \cdot 5)^{**} \right\}$$

$$\begin{cases} \overset{\alpha}{\rightarrow} \prod_{\alpha\beta}(q) \\ \sim (-1) \times \int \frac{d^{D}k}{(2\pi)^{D}i} \operatorname{Tr} \left\{ \gamma_{\alpha}(1+A\gamma_{5}) \frac{m_{1}+k}{m_{1}^{2}-k^{2}} \\ \times \gamma_{\beta}(1+B\gamma_{5}) \frac{m_{2}+k-q}{m_{2}^{2}-(k-q)^{2}} \right\} \\ = \frac{1}{2\pi^{2}} (1+AB) \left(g_{\alpha\beta}q^{2}-q_{\alpha}q_{\beta} \right) \left\{ -\frac{1}{6}C_{UV} + \int_{0}^{1} dx x (1-x) \ln \left(D_{2}(x) \right) \right\} \\ -\frac{1}{4\pi^{2}} g_{\alpha\beta}C_{UV} \left\{ m_{1}m_{2}(1-AB) - \frac{1}{2} \left(m_{1}^{2}+m_{2}^{2} \right) (1+AB) \right\} \end{cases}$$

^{*)} A solid line and a wavy line represent a fermion and a gauge boson respectively.

^{**)} A dashed line represents a scalar particle.

K-I. Aoki, Z. Hioki, R. Kawabe, M. Konuma and T. Muta

$$+\frac{1}{4\pi^2}g_{\alpha\beta}\int_0^1 dx [m_1m_2(1-AB) - (1+AB)\overline{m^2}]\ln(D_2(x)).$$
$$(\overline{m^2} \equiv m_1^2(1-x) + m_2^2x)$$
(B·6)

$$\begin{array}{cccc}
& & \int \frac{d^{p}k}{(2\pi)^{p}i} \{(-q+k-q)_{\rho}g_{a\sigma} + (-k+q-k)_{a}g_{\rho\sigma} + (k+q)_{\sigma}g_{a\rho}\} \\
& & \sim \int \frac{d^{p}k}{(2\pi)^{p}i} \{(-q+k-q)_{\rho}g_{a\sigma} + (-k+q-k)_{a}g_{\rho\sigma} + (k+q)_{\sigma}g_{a\rho}\} \\
& & \times \frac{g^{\rho\rho'}}{k^{2}-m_{1}^{2}} \frac{g^{\sigma\sigma'}}{(k-q)^{2}-m_{2}^{2}} \\
& & \times \{(q+k)_{\sigma'}g_{\beta\rho'} + (-k-k+q)_{\beta}g_{\rho'\sigma'} + (k-q-q)_{\rho'}g_{\sigma'\beta}\} \\
& & = \frac{-1}{48\pi^{2}}q_{a}q_{\beta}(11C_{UV} + 2) \\
& & + \frac{1}{16\pi^{2}}g_{a\beta}\left\{\frac{3}{2}(m_{1}^{2}+m_{2}^{2})(3C_{UV} + 1) + \frac{1}{6}q^{2}(19C_{UV} - 3)\right\} \\
& & + \frac{1}{8\pi^{2}}q_{a}q_{\beta}\int_{0}^{1}dx \left\{1 + 5x(1-x)\right\}\ln(D_{2}(x)) \\
& & - \frac{1}{16\pi^{2}}g_{a\beta}\int_{0}^{1}dx \left\{9\overline{m^{2}} + q^{2}(11x^{2} - 11x + 5)\right\}\ln(D_{2}(x)). \quad (B\cdot7)
\end{array}$$

$$\begin{cases} \overset{\alpha}{\longrightarrow} & \Pi_{\alpha\beta}(q) \\ & \sim \int \frac{d^{D}k}{(2\pi)^{D}i} g_{\alpha\sigma} \frac{g^{\sigma\rho}}{(k-q)^{2} - m_{2}^{2}} g_{\rho\beta} \frac{1}{m_{1}^{2} - k^{2}} \\ & = \frac{-1}{16\pi^{2}} g_{\alpha\beta} \Big\{ C_{\rm UV} - \int_{0}^{1} dx \ln(D_{2}(x)) \Big\}. \end{aligned}$$
(B·8)

$$\left\{ \begin{array}{c} \int_{\alpha}^{\alpha} & \Pi_{\alpha\beta}(q) \\ & \sim \int \frac{d^{D}k}{(2\pi)^{D}i} (-k+q-k)_{\alpha} \frac{1}{m_{1}^{2}-k^{2}} \frac{1}{m_{2}^{2}-(k-q)^{2}} (-k-k+q)_{\beta} \\ & = \frac{1}{16\pi^{2}} \left\{ \frac{1}{3} C_{UV} q_{\alpha} q_{\beta} + g_{\alpha\beta} (C_{UV}+1) \left(m_{1}^{2}+m_{2}^{2}-\frac{q^{2}}{3} \right) \right\} \\ & - \frac{1}{16\pi^{2}} \int_{0}^{1} dx \left\{ (1-2x)^{2} q_{\alpha} q_{\beta} + 2g_{\alpha\beta} D_{2}(x) \right\} \ln(D_{2}(x)). \quad (B \cdot 9)$$

$$\begin{array}{c} & \Pi_{\alpha\beta}(q) \\ & & \sim (-1) \times \int \frac{d^{D}k}{(2\pi)^{D}i} (-k+q)_{\alpha} \frac{1}{m_{1}^{2}-k^{2}} \frac{1}{m_{2}^{2}-(k-q)^{2}} (-k)_{\beta} \\ & & = \frac{-1}{16\pi^{2}} \left\{ \frac{1}{4} (C_{UV}+1) g_{\alpha\beta} \left(m_{1}^{2}+m_{2}^{2}-\frac{q^{2}}{3} \right) - \frac{1}{6} C_{UV} q_{\alpha} q_{\beta} \right\} \end{array}$$

$$-\frac{1}{16\pi^2} \int_0^1 dx \left\{ -\frac{1}{2} g_{\alpha\beta} D_2(x) + x (1-x) q_\alpha q_\beta \right\} \ln(D_2(x)).$$
(B·10)*)

$$\begin{cases} \alpha & \Pi_{\alpha\beta}(q) \\ & \sim \int \frac{d^{D}k}{(2\pi)^{D}i} g_{\alpha\beta} \frac{1}{m^{2} - k^{2}} \\ & = -\frac{1}{16\pi^{2}} g_{\alpha\beta} m^{2} (C_{UV} + 1 - \ln m^{2}). \end{cases}$$
(B·12)

$$\begin{array}{c} \overset{\alpha}{\longleftarrow} & \Pi_{\alpha}(q) \\ & \sim (-1) \times \int \frac{d^{p}k}{(2\pi)^{p}i} \operatorname{Tr} \left\{ \gamma_{\alpha}(1+A\gamma_{5}) \frac{m_{1}+k}{m_{1}^{2}-k^{2}} (1+B\gamma_{5}) \frac{m_{2}+k-q}{m_{2}^{2}-(k-q)^{2}} \right\} \\ & = \frac{-1}{4\pi^{2}} q_{\alpha} \int_{0}^{1} dx \left\{ C_{\mathrm{UV}} - \ln\left(D_{2}(x)\right) \right\} \\ & \times \left\{ (1-AB) m_{2}x - (1+AB) m_{1}(1-x) \right\}. \quad (B\cdot13)
\end{array}$$

$$\begin{array}{c} & \int_{\alpha} \Pi_{\alpha}(q) \\ & \sim \int \frac{d^{D}k}{(2\pi)^{D}i} \{(-q+k-q)_{\rho}g_{\alpha\sigma} + (-k+q-k)_{\alpha}g_{\rho\sigma} + (k+q)_{\sigma}g_{\alpha\rho}\} \\ & \times \frac{g^{\rho\rho'}}{k^{2}-m_{1}^{2}}g_{\rho'\sigma'}\frac{g^{\sigma'\sigma}}{(k-q)^{2}-m_{2}^{2}} \\ & = -\frac{3}{16\pi^{2}}q_{\alpha}\int_{0}^{1}dx(1-2x)\ln(D_{2}(x)). \end{array}$$
(B·14)

$$\begin{cases} \overset{\alpha}{\longleftarrow} & II_{\alpha}(q) \\ & \sim \int \frac{d^{p}k}{(2\pi)^{p}i} g_{\alpha \rho} \frac{g^{\rho \sigma}}{(k-q)^{2} - m_{2}^{2}} (q+k)_{\sigma} \frac{1}{m_{1}^{2} - k^{2}} \\ & = -\frac{1}{16\pi^{2}} q_{\alpha} \Big\{ \frac{3}{2} C_{\mathrm{UV}} - \int_{0}^{1} dx (1+x) \ln(D_{2}(x)) \Big\}. \tag{B.15}$$

*) A dotted line represents an FP-ghost.

$$\begin{array}{c} & \Pi_{\alpha}(q) \\ & & \sim \int \frac{d^{D}k}{(2\pi)^{D}i} (k+k-q)_{\alpha} \frac{1}{m_{1}^{2}-k^{2}} \frac{1}{m_{2}^{2}-(k-q)^{2}} \\ & & = \frac{1}{16\pi^{2}} q_{\alpha} \int_{0}^{1} dx (1-2x) \ln(D_{2}(x)). \end{array}$$
 (B·16)

$$\begin{array}{c} & \Pi_{\alpha}(q) \\ & & \sim (-1) \times \int \frac{d^{D}k}{(2\pi)^{D}i} (-k+q)_{\alpha} \frac{1}{m_{1}^{2}-k^{2}} \frac{1}{m_{2}^{2}-(k-q)^{2}} \\ & & = \frac{-1}{16\pi^{2}} q_{\alpha} \Big\{ \frac{1}{2} C_{\text{UV}} - \int_{0}^{1} dx (1-x) \ln(D_{2}(x)) \Big\}. \end{array}$$
 (B·17)

$$\begin{array}{c} & \prod \left(q\right) \\ & \sim (-1) \times \int \frac{d^{p}k}{(2\pi)^{p}i} \operatorname{Tr}\left\{(1+A\gamma_{5})\frac{m_{1}+k}{m_{1}^{2}-k^{2}}(1+B\gamma_{5})\frac{m_{2}+k-q}{m_{2}^{2}-(k-q)^{2}}\right\} \\ & = \frac{-1}{4\pi^{2}} \int_{0}^{1} dx \left[\left\{(1+AB)m_{1}m_{2}\right. \\ & - (1-AB)x(1-x)q^{2}\right\} \left\{C_{\mathrm{UV}}-\ln\left(D_{2}(x)\right)\right\} \\ & + (1-AB) \left\{2C_{\mathrm{UV}}+1-2\ln\left(D_{2}(x)\right)\right\} D_{2}(x)\right]. \quad (B\cdot18)
\end{array}$$

$$\begin{array}{c} & \prod \left(q\right) \\ & \sim \int \frac{d^{D}k}{(2\pi)^{D}i} g_{\rho\rho'} \frac{g^{\rho'\sigma'}}{(k-q)^{2} - m_{2}^{2}} g_{\sigma'\sigma} \frac{g^{\sigma\rho}}{k^{2} - m_{1}^{2}} \\ & = \frac{1}{8\pi^{2}} \left\{ 2C_{\rm UV} - 1 - 2 \int_{0}^{1} dx \ln\left(D_{2}(x)\right) \right\}. \end{array} \tag{B.19}$$

$$\begin{array}{c} & \prod(q) \\ & \sim \int \frac{d^{p}k}{(2\pi)^{p}i} (k+q)_{\rho} \frac{g^{\rho\sigma}}{(k-q)^{2} - m_{2}^{2}} (q+k)_{\sigma} \frac{1}{m_{1}^{2} - k^{2}} \\ & = \frac{-1}{16\pi^{2}} \int_{0}^{1} dx [q^{2} (1+x)^{2} \{C_{\text{UV}} - \ln(D_{2}(x))\} \\ & + D_{2}(x) \{2C_{\text{UV}} + 1 - 2\ln(D_{2}(x))\}]. \end{array}$$
(B·20)

$$\bigvee_{i=1}^{\prod (q)} \sim \int \frac{d^{p}k}{(2\pi)^{p}i} \frac{1}{m_{1}^{2} - k^{2}} \frac{1}{m_{2}^{2} - (k-q)^{2}} \\ = \frac{1}{16\pi^{2}} \Big\{ C_{\text{UV}} - \int_{0}^{1} dx \ln(D_{2}(x)) \Big\}.$$
 (B·21)

Downloaded from https://academic.oup.com/ptps/article-abstract/doi/10.1143/PTPS.73.1/1848359 by guest on 06 January 2019

195

$$\prod_{i=1}^{H(q)} \sim \int \frac{d^{D}k}{(2\pi)^{D}i} g_{\rho\sigma} \frac{g^{\sigma\rho}}{k^{2} - m^{2}} = \frac{1}{8\pi^{2}} m^{2} (2C_{\rm UV} + 1 - 2\ln m^{2}). \quad (B.23)$$

$$= \frac{1}{16\pi^2} m^2 (C_{\text{UV}} + 1 - \ln m^2).$$
 (B·24)

B.2 Three-point functions

Momentum flow and masses are shown in Fig. B.2.



Fig. B.2. Typical diagrams of three-point functions. k is the loop momentum and m_1, m_2 and M denote the masses of propagating particles.

We take the Feynman parametrization as

$$\frac{1}{ABC} = \int_0^1 dx \frac{1}{\{A(1-x) + Bx\}^2} \frac{1}{C}$$
$$= \int_0^1 dx \int_0^1 dy \frac{2y}{[\{A(1-x) + Bx\}y + C(1-y)]^3},$$

where

$$A \equiv m_1^2 - (p_1 - k)^2,$$

$$B \equiv m_2^2 - (p_2 - k)^2,$$

$$C \equiv M^2 - k^2.$$

We use the following functions,

$$D_{\mathfrak{z}}(x, y) \equiv (1-y) M^2 + y \{\overline{m^2} - q^2 x (1-x)\} - y (1-y) \overline{p}^2,$$

$$(\overline{m^2} \equiv (1-x) m_1^2 + x m_2^2, \quad \overline{p} \equiv (1-x) p_1 + x p_2) \qquad (B.25)$$

$$\gamma(A, B, C) \equiv (1 + AB + BC + CA) - (A + B + C + ABC)\gamma_{\mathfrak{s}} \cdot (B \cdot 26)$$

$$\begin{split} & \prod_{\alpha} \Gamma_{\alpha}(p_{2}, p_{1}) \\ & \sim \int \frac{d^{D}k}{(2\pi)^{D}i} \gamma_{\rho} \left(1 + A\gamma_{\delta}\right) \frac{m_{2} + p_{2} - k}{m_{2}^{2} - (p_{2} - k)^{2}} \gamma_{\alpha} (1 + B\gamma_{\delta}) \\ & \qquad \times \frac{m_{1} + p_{1} - k}{m_{1}^{2} - (p_{1} - k)^{2}} \gamma_{\sigma} \left(1 + C\gamma_{\delta}\right) \frac{g^{\rho\sigma}}{k^{2} - M^{2}} \\ & = \frac{1}{8\pi^{2}} m_{1} m_{2} \gamma \left(A, -B, C\right) \gamma_{\alpha} \int_{0}^{1} \int_{0}^{1} \frac{dx dy}{D_{\delta}(x, y)} dx dy \\ & - \frac{1}{4\pi^{2}} m_{1} \gamma \left(A, B, -C\right) \int_{0}^{1} \int_{0}^{1} \frac{dx dy}{D_{\delta}(x, y)} y \left(p_{2} - y\overline{p}\right)_{\alpha} \\ & - \frac{1}{4\pi^{2}} m_{2} \gamma \left(A, -B, -C\right) \int_{0}^{1} \int_{0}^{1} \frac{dx dy}{D_{\delta}(x, y)} y \left(p_{1} - y\overline{p}\right)_{\alpha} \\ & + \frac{1}{8\pi^{2}} \gamma \left(A, B, C\right) \int_{0}^{1} \int_{0}^{1} \frac{dx dy}{D_{\delta}(x, y)} y \left(p_{1} - y\overline{p}\right) \gamma_{\alpha} \left(p_{2} - y\overline{p}\right) \\ & + \frac{1}{16\pi^{2}} \gamma \left(A, B, C\right) \gamma_{\alpha} \Big\{ C_{UV} - 2 - 2 \int_{0}^{1} \int_{0}^{1} dx dy y \ln \left(D_{\delta}(x, y)\right) \Big\}. \end{split}$$

$$(B.27)$$

$$\begin{split} & \prod_{n=1}^{\Gamma(p_{2}, p_{1})} \\ & \sim \int \frac{d^{D}k}{(2\pi)^{D}i} \gamma_{\rho} \left(1 + A\gamma_{5}\right) \frac{m_{2} + p_{2} - k}{m_{2}^{2} - (p_{2} - k)^{2}} (1 + B\gamma_{5}) \\ & \qquad \times \frac{m_{1} + p_{1} - k}{m_{1}^{2} - (p_{1} - k)^{2}} \gamma_{\sigma} \left(1 + C\gamma_{5}\right) \frac{g^{\rho\sigma}}{k^{2} - M^{2}} \\ & = \frac{-1}{4\pi^{2}} m_{1}m_{2}\gamma \left(A, B, -C\right) \int_{0}^{1} \int_{0}^{1} \frac{y}{D_{3}(x, y)} dx dy \\ & \qquad + \frac{1}{8\pi^{2}} m_{1}\gamma \left(A, -B, C\right) \int_{0}^{1} \int_{0}^{1} \frac{dx dy}{D_{3}(x, y)} y\left(p_{2} - y\overline{p}\right) \\ & \qquad + \frac{1}{8\pi^{2}} m_{2}\gamma \left(A, B, C\right) \int_{0}^{1} \int_{0}^{1} \frac{dx dy}{D_{3}(x, y)} y\left(p_{1} - y\overline{p}\right) \\ & \qquad + \frac{1}{4\pi^{2}} \gamma \left(A, -B, -C\right) \left[C_{UV} - 1\right] \end{split}$$

$$-\int_{0}^{1}\int_{0}^{1}dxdy y\left\{\frac{(\not p_{1}-y\overline{\not p})(\not p_{2}-y\overline{\not p})}{\dot{D}_{s}(x,y)}+2\ln(D_{s}(x,y))\right\}\right]. \quad (B\cdot 28)$$

$$\frac{\Gamma_{\alpha}(p_{2}, p_{1})}{\sim \int \frac{d^{p}k}{(2\pi)^{p}i} (1+A\gamma_{5}) \frac{m_{2}+p_{2}-k}{m_{2}^{2}-(p_{2}-k)^{2}} \gamma_{\alpha}(1+B\gamma_{5})} \\
\times \frac{m_{1}+p_{1}-k}{m_{1}^{2}-(p_{1}-k)^{2}} (1+C\gamma_{5}) \frac{1}{M^{2}-k^{2}} \\
= \frac{1}{16\pi^{2}} m_{1}m_{2}\gamma(-A, B, C)\gamma_{\alpha} \int_{0}^{1} \int_{0}^{1} \frac{dxdy}{D_{3}(x, y)} dxdy \\
+ \frac{1}{16\pi^{2}} m_{1}\gamma(-A, -B, -C) \int_{0}^{1} \int_{0}^{1} \frac{dxdy}{D_{3}(x, y)} y(p_{2}-y\overline{p})\gamma_{\alpha} \\
+ \frac{1}{16\pi^{2}} m_{2}\gamma(-A, B, -C)\gamma_{\alpha} \int_{0}^{1} \int_{0}^{1} \frac{dxdy}{D_{3}(x, y)} y(p_{2}-y\overline{p})\gamma_{\alpha}(p_{1}-y\overline{p}) \\
+ \frac{1}{16\pi^{2}}\gamma(-A, -B, C) \int_{0}^{1} \int_{0}^{1} \frac{dxdy}{D_{3}(x, y)} y(p_{2}-y\overline{p})\gamma_{\alpha}(p_{1}-y\overline{p}) \\
+ \frac{1}{32\pi^{2}}\gamma(-A, -B, C)\gamma_{\alpha} \left\{ C_{UV} - 1 - 2 \int_{0}^{1} \int_{0}^{1} dxdy y \ln(D_{3}(x, y)) \right\}.$$
(B·29)

$$\begin{split} & \Gamma\left(p_{2}, p_{1}\right) \\ & \sim \int \frac{d^{p}k}{(2\pi)^{p}i} (1 + A\gamma_{6}) \frac{m_{2} + p_{2} - k}{m_{2}^{2} - (p_{2} - k)^{2}} (1 + B\gamma_{6}) \\ & \qquad \times \frac{m_{1} + p_{1} - k}{m_{1}^{2} - (p_{1} - k)^{2}} (1 + C\gamma_{6}) \frac{1}{M^{2} - k^{2}} \\ & = \frac{1}{16\pi^{2}} m_{1}m_{2}\gamma\left(-A, -B, -C\right) \int_{0}^{1} \int_{0}^{1} \frac{dxdy}{D_{3}(x, y)} dxdy \\ & \qquad + \frac{1}{16\pi^{2}} m_{1}\gamma\left(-A, B, C\right) \int_{0}^{1} \int_{0}^{1} \frac{dxdy}{D_{3}(x, y)} y\left(p_{2} - y\overline{p}\right) \\ & \qquad + \frac{1}{16\pi^{2}} m_{2}\gamma\left(-A, -B, C\right) \int_{0}^{1} \int_{0}^{1} \frac{dxdy}{D_{3}(x, y)} y\left(p_{1} - y\overline{p}\right) \\ & \qquad + \frac{1}{16\pi^{2}} \gamma\left(-A, B, -C\right) \int_{0}^{1} \int_{0}^{1} \frac{dxdy}{D_{3}(x, y)} y\left(p_{2} - y\overline{p}\right) \left(p_{1} - y\overline{p}\right) \\ & \qquad - \frac{1}{32\pi^{2}}\gamma\left(-A, B, -C\right) \left\{2C_{UV} - 1 - 4 \int_{0}^{1} \int_{0}^{1} dxdy y\ln\left(D_{3}(x, y)\right)\right\}. \end{split}$$
(B·30)

$$\begin{split} & \Gamma_{\alpha}(p_{2}, p_{1}) \\ & \sim \int \frac{d^{p}k}{(2\pi)^{p}i} \tilde{\gamma}_{\rho} (1 + A\gamma_{5}) \frac{M + k}{M^{2} - k^{2}} \tilde{\gamma}_{\sigma} (1 + B\gamma_{5}) \\ & \times \frac{g^{\rho\rho'}}{(p_{2} - k)^{2} - m_{2}^{2}} \{ (q - p_{1} + k)_{\rho'} g_{a\sigma'} + (p_{1} - k + p_{2} - k)_{\alpha} g_{\rho'\sigma'} \\ & + (-p_{2} + k - q)_{\sigma'} g_{a\rho'} \} \frac{g^{\sigma\sigma'}}{(p_{1} - k)^{2} - m_{1}^{2}} \\ & = \frac{3}{16\pi^{2}} M \tilde{\gamma} (A, -B) \int_{0}^{1} \int_{0}^{1} \frac{dxdy}{D_{\delta}(x, y)} y \{ 2(p_{2} - y\bar{p})_{\alpha} - q\tilde{\gamma}_{\alpha} \} \\ & + \frac{1}{16\pi^{2}} \tilde{\gamma} (A, B) \int_{0}^{1} \int_{0}^{1} \frac{dxdy}{D_{\delta}(x, y)} y^{2} \\ & \times \{ (p_{2} - 2p_{1} + y\bar{p}) \bar{p} \tilde{\gamma}_{\alpha} - 2(p_{1} + p_{2} - 2y\bar{p})_{\alpha} \bar{p} + \tilde{\gamma}_{\alpha} \bar{p} (p_{1} - 2p_{2} + y\bar{p}) \} \\ & - \frac{1}{16\pi^{2}} \tilde{\gamma} (A, B) \tilde{\gamma}_{\alpha} \Big\{ 3C_{UV} - 2 - 6 \int_{0}^{1} \int_{0}^{1} dxdyy \ln (D_{\delta}(x, y)) \Big\} . \end{split}$$

$$(B \cdot 31)^{*} \end{split}$$

$$\Gamma_{\alpha}(p_{2}, p_{1}) = \sqrt{\int \frac{d^{p}k}{(2\pi)^{p}i}} \gamma_{\rho} (1 + A\gamma_{\delta}) \frac{M + k}{M^{2} - k^{2}} (1 + B\gamma_{\delta}) \\
\times \frac{g^{\rho\sigma}}{(p_{2} - k)^{2} - m_{2}^{2}} g_{\sigma\alpha} \frac{1}{m_{1}^{2} - (p_{1} - k)^{2}} \\
= \frac{1}{16\pi^{2}} M\gamma (A, B) \gamma_{\alpha} \int_{0}^{1} \int_{0}^{1} \frac{y}{D_{\delta}(x, y)} dx dy \\
+ \frac{1}{16\pi^{2}} \gamma (A, -B) \gamma_{\alpha} \int_{0}^{1} \int_{0}^{1} \frac{dx dy}{D_{\delta}(x, y)} y^{2} \overline{p} .$$
(B·32)

$$\Gamma(p_{2}, p_{1})
\sim \int \frac{d^{D}k}{(2\pi)^{D_{1}}} \gamma_{\rho} (1 + A\gamma_{5}) \frac{M + k}{M^{2} - k^{2}} \gamma_{\sigma} (1 + B\gamma_{5})
\times \frac{g^{\rho \rho'}}{(p_{2} - k)^{2} - m_{2}^{2}} g_{\rho' \sigma'} \frac{g^{\sigma' \sigma}}{(p_{1} - k)^{2} - m_{1}^{2}}
= \frac{1}{4\pi^{2}} M\gamma(A, -B) \int_{0}^{1} \int_{0}^{1} \frac{y}{D_{3}(x, y)} dx dy
- \frac{1}{8\pi^{2}} \gamma(A, B) \int_{0}^{1} \int_{0}^{1} \frac{dx dy}{D_{3}(x, y)} y^{2} \overline{p}.$$
(B·33)

*) $\gamma(A, B)$ is defined in Eq. (B.3).

$$\Gamma(p_{2}, p_{1})$$

$$\sim \int \frac{d^{D}k}{(2\pi)^{D}i} \gamma_{\rho} (1 + A\gamma_{5}) \frac{M + k}{M^{2} - k^{2}} (1 + B\gamma_{5})$$

$$\times \frac{g^{\rho\sigma}}{(p_{2} - k)^{2} - m_{2}^{2}} (p_{1} - k - q)_{\sigma} \frac{1}{m_{1}^{2} - (p_{1} - k)^{2}}$$

$$= \frac{-1}{16\pi^{2}} M\gamma(A, B) \int_{0}^{1} \int_{0}^{1} \frac{dxdy}{D_{3}(x, y)} y(2p_{1} - p_{2} - y\overline{p})$$

$$- \frac{1}{16\pi^{2}} \gamma(A, -B) \int_{0}^{1} \int_{0}^{1} \frac{dxdy}{D_{3}(x, y)} y^{2}(2p_{1} - p_{2} - y\overline{p}) \overline{p}$$

$$- \frac{1}{32\pi^{2}} \gamma(A, -B) \left\{ 2C_{UV} - 1 - 4 \int_{0}^{1} \int_{0}^{1} dxdy y \ln(D_{3}(x, y)) \right\}.$$

$$(B \cdot 34)$$

$$\begin{split} & \int_{\alpha} \Gamma_{\alpha}(p_{2}, p_{1}) \\ & \sim \int \frac{d^{D}k}{(2\pi)^{D}i} (1 + A\gamma_{5}) \frac{M + k}{M^{2} - k^{2}} (1 + B\gamma_{5}) \\ & \qquad \times \frac{1}{m_{2}^{2} - (p_{2} - k)^{2}} (p_{1} - k + p_{2} - k)_{\alpha} \frac{1}{m_{1}^{2} - (p_{1} - k)^{2}} \\ & = \frac{1}{16\pi^{2}} M\gamma (-A, -B) \int_{0}^{1} \int_{0}^{1} \frac{dxdy}{D_{3}(x, y)} y (p_{1} + p_{2} - 2y\overline{p})_{\alpha} \\ & \qquad + \frac{1}{16\pi^{2}} \gamma (-A, B) \int_{0}^{1} \int_{0}^{1} \frac{dxdy}{D_{3}(x, y)} y^{2} (p_{1} + p_{2} - 2y\overline{p})_{\alpha} \overline{p} \\ & \qquad + \frac{1}{32\pi^{2}} \gamma (-A, B) \gamma_{\alpha} \Big\{ C_{UV} - 2 \int_{0}^{1} \int_{0}^{1} dxdy y \ln(D_{3}(x, y)) \Big\}. \end{split}$$

$$(B.35)$$

$$\Gamma(p_{2}, p_{1})$$

$$\sim \int \frac{d^{D}k}{(2\pi)^{D}i} (1 + A\gamma_{5}) \frac{M + k}{M^{2} - k^{2}} (1 + B\gamma_{5}) \frac{1}{m_{2}^{2} - (p_{2} - k)^{2}} \frac{1}{m_{1}^{2} - (p_{1} - k)^{2}}$$

$$= \frac{1}{16\pi^{2}} M\gamma(-A, -B) \int_{0}^{1} \int_{0}^{1} \frac{y}{D_{3}(x, y)} dx dy$$

$$+ \frac{1}{16\pi^{2}} \gamma(-A, B) \int_{0}^{1} \int_{0}^{1} \frac{dx dy}{D_{3}(x, y)} y^{2} \overline{p} . \qquad (B \cdot 36)$$

B.3 Four-point functions

Masses and momentum flow are shown in Fig. B.3. Box type diagrams do not include UV divergence. Therefore we set $D\rightarrow 4$ in the following.



Fig. B.3. Typical diagram of four-point function. k is the loop momentum and $m_{1,2}$ and $M_{1,2}$ denote the masses of propagating particles.

Feynman parametrization is performed as

$$\begin{aligned} \frac{1}{ABCD} &= \int_0^1 dx \int_0^1 dy \frac{1}{\{A(1-x) + Bx\}^2} \frac{1}{\{C(1-y) + Dy\}^2} \\ &= \int_0^1 dx \int_0^1 dy \int_0^1 dz \frac{6z(1-z)}{[\{A(1-x) + Bx\}z + \{C(1-y) + Dy\}(1-z)]^4}, \end{aligned}$$

where

$$A \equiv M_{1}^{2} - (p_{1} - k)^{2},$$

$$B \equiv M_{2}^{2} - (p_{2} - k)^{2},$$

$$C \equiv m_{1}^{2} - k^{2},$$

$$D \equiv m_{2}^{2} - (p_{2} + p_{4} - k)^{2}.$$

And we use the following function:

$$D_{4}(x, y, z) = \{(1-x)M_{1}^{2} + xM_{2}^{2}\}z + \{ym_{2}^{2} + (1-y)m_{1}^{2}\}(1-z) - (1-x)z(1-(1-x)z)p_{1}^{2} - \{xz+y(1-z)\}\{1-xz-y(1-z)\}p_{2}^{2} - y(1-z)\{1-y(1-z)\}p_{4}^{2} - 2y(1-z)\{1-xz-y(1-z)\}p_{2}p_{4} + 2z(1-x)\{y(1-z)+zx\}p_{1}p_{2} + 2(1-x)yz(1-z)p_{1}p_{4}.$$
 (B·37)

In Eqs. (B·38) \sim (B·40), spinors $\bar{u}(p)$, u(p) are included in order to avoid confusion. However, on-shell relations, pu(p) = mu(p), etc., have not been used in the derivation of these formulas.

$$\frac{1}{\sum_{i=1}^{d^{4}k} A(p_{4}, p_{2}; p_{3}, p_{1})} \\
\sim \int \frac{d^{4}k}{(2\pi)^{4}i} \overline{u}(p_{4}) \gamma_{\rho} (1 + D\gamma_{5}) \frac{m_{2} - k + p_{2} + p_{4}}{m_{2}^{2} - (k - p_{2} - p_{4})^{2}} \gamma_{\sigma} (1 + C\gamma_{5}) u(p_{3})$$

$$\times \frac{g^{\rho\rho'}}{(p_{2}-k)^{2}-M_{2}^{2}} \frac{g^{\sigma\sigma'}}{(p_{1}-k)^{2}-M_{1}^{2}} \times \overline{u}(p_{2})\gamma_{\rho'}(1+B\gamma_{5}) \frac{m_{1}+k}{m_{1}^{2}-k^{2}}\gamma_{\sigma'}(1+A\gamma_{5})u(p_{1}) = \frac{1}{16\pi^{2}} \int_{0}^{1} dx \int_{0}^{1} dy \int_{0}^{1} dz \frac{z(1-z)}{\{D_{4}(x,y,z)\}^{2}} \times \overline{u}(p_{4})\gamma^{\rho}[m_{2}\gamma(C,-D)-\gamma(-C,-D)\{z(1-x)p_{1} +(zx+y(1-z)-1)p_{2}+(y(1-z)-1)p_{4}\}]\gamma^{\sigma}u(p_{3}) \times \overline{u}(p_{2})\gamma_{\rho}[m_{1}\gamma(A,-B)+\gamma(-A,-B)\{z(1-x)p_{1} +(zx+y(1-z))p_{2}+y(1-z)p_{4}\}]\gamma_{\sigma}u(p_{1}) +\frac{1}{32\pi^{2}} \int_{0}^{1} dx \int_{0}^{1} dy \int_{0}^{1} dz \frac{z(1-z)}{D_{4}(x,y,z)}\overline{u}(p_{4})\gamma(C,D)\gamma^{\rho}\gamma^{\alpha}\gamma^{\sigma}u(p_{3}) \times \overline{u}(p_{2})\gamma(A,B)\gamma_{\rho}\gamma_{\alpha}\gamma_{\sigma}u(p_{1}).$$
 (B·38)*)



^{*)} $\gamma(A, B)$ is defined at the beginning of B.1, Eq. (B.3).

$$\sim \int \frac{d^{4}k}{(2\pi)^{4}i} \overline{u} (p_{4}) (1+D\gamma_{5}) \frac{m_{2}-k+p_{2}+p_{4}}{m_{2}^{2}-(k-p_{2}-p_{4})^{2}} (1+C\gamma_{5}) u (p_{3})$$

$$\times \frac{1}{M_{2}^{2}-(k-p_{2})^{2}} \cdot \frac{1}{M_{1}^{2}-(k-p_{1})^{2}}$$

$$\times \overline{u} (p_{2}) (1+B\gamma_{5}) \frac{m_{1}+k}{m_{1}^{2}-k^{2}} (1+A\gamma_{5}) u (p_{1})$$

$$= \frac{1}{16\pi^{2}} \int_{0}^{1} dx \int_{0}^{1} dy \int_{0}^{1} dz \frac{z(1-z)}{\{D_{4}(x,y,z)\}^{2}}$$

$$\times \overline{u} (p_{4}) [m_{2}\gamma (-C, -D) - \gamma (C, -D) \{z(1-x)p_{1} + (zx+y(1-z)-1)p_{2} + (y(1-z)-1)p_{4}\}] u (p_{5})$$

$$\times \overline{u} (p_{2}) [m_{1}\gamma (-A, -B) + \gamma (A, -B) \{z(1-x)p_{1} + (zx+y(1-z))p_{2} + y(1-z)p_{4}\}] u (p_{5})$$

$$\times \overline{u} (p_{4}) [m_{2}\gamma (-C, D)\gamma^{p} u (p_{5}) \cdot \overline{u} (p_{2})\gamma (A, -B)\gamma_{p} u (p_{1}).$$

$$(B.40)$$

Appendix C

1/M Expansion of Some Feynman Amplitudes

One-loop Feynman amplitudes relevant to our calculation consist of the two-point (self-energy part), three-point (vertex part) and four-point (box diagram) functions. These functions, denoted by A_2 , A_3 and A_4 respectively, are expressed by the following integral forms,

$$A_{2} = \int_{0}^{1} dx \, I_{2}(x, M, m, q) , \qquad (C \cdot 1)$$

$$A_{s} = \int_{0}^{1} dx dy \ I_{s}(x, y, M, m, p_{1}, p_{2}), \qquad (C \cdot 2)$$

$$A_{4} = \int_{0}^{1} dx dy dz I_{4}(x, y, z, M, m, p_{1}, p_{2}, p_{3}), \qquad (C \cdot 3)$$

where M and m generically represent large masses (like M_w and M_z) and small masses respectively, and q, p_1 , p_2 and p_3 are momenta of external lines.

Downloaded from https://academic.oup.com/ptps/article-abstract/doi/10.1143/PTPS.73.1/1848359 by guest on 06 January 2019

Appendix C

It is well-known that in one-loop order the Feynman integrals can be performed analytically except for an unintegrable special function, Spence function (or dilogarithm) ['t Hooft and Veltman 79, Lewin 58]. Actually the integral (C·1) may be performed explicitly in terms of elementary functions while the integrals (C·2) and (C·3) are expressed in terms of elementary functions and the Spence function. It is, however, important to note that $M \gg m$, |q|, $|p_i|$ (i=1,2,3) as far as low energy phenomena are concerned and hence useful approximate formulas may be derived for Eqs. (C·1), (C·2) and (C·3).

We consider the amplitudes $(C\cdot 1)$, $(C\cdot 2)$ and $(C\cdot 3)$ for Feynman diagrams including at least one large-mass particle with mass M and derive the formulas in which these amplitudes are expanded in powers of 1/M. The leading term in this expansion will be found to give a good approximation to the full amplitude and to have a simple form.

C.1 Two-point functions (self-energy part)

The integrals for which we wish to derive the 1/M expansion formulas have the following forms,

$$\int_0^1 dx \, x^n \ln D_2 \,, \qquad (C \cdot 4)$$

$$\int_0^1 dx \, x^n \ln D_2' \,, \qquad (C \cdot 5)$$

where n=0, 1, 2 and

$$D_2 = M^2(1-x) + m^2 x - q^2 x (1-x), \qquad (C \cdot 6)$$

$$D_{2}' = M_{1}^{2}(1-x) + M_{2}^{2}x - q^{2}x(1-x)$$
 (C·7)

with two different large masses M_1 and M_2 .

If one expands the integral $(C \cdot 4)$ in powers of 1/M by separating

$$\ln D_2 = \ln M^2 + \ln (1-x) + \ln \left[1 + \frac{m^2 x - q^2 x (1-x)}{M^2 (1-x)}\right],$$

one immediately see that the coefficients of the expansion in 1/M are divergent integrals (divergent at x=1). Hence we use the following form:

$$D_2 = [M^2(1-x) + m^2 x] (1+f), \qquad (C \cdot 8)$$

where

$$f = \frac{-q^2 x (1-x)}{M^2 (1-x) + m^2 x}, \qquad (C \cdot 9)$$

and expand $\ln D_2$ in powers of f. This expansion in f is uniformly convergent and so the term-by-term integration is justified,

$$\int_{0}^{1} dx \ x^{n} \ln D_{2}$$

$$= \int_{0}^{1} dx \ x^{n} \ln [M^{2}(1-x) + m^{2}x] - \int_{0}^{1} dx \ x^{n} \frac{x(1-x) q^{2}}{M^{2}(1-x) + m^{2}x}$$

$$+ \int_{0}^{1} dx \ x^{n} \frac{x^{2}(1-x)^{2}q^{4}}{2[M^{2}(1-x) + m^{2}x]^{2}} + \cdots .$$
(C·10)

We easily see that the *l*-th order term in Eq. (C.10) is of order $(1/M^2)^{l-1}$ up to $\ln(m^2/M^2)$. Hence Eq. (C·10) is the desired expansion. We list here the explicit expansion formulas for n=0, 1, 2 to order $1/M^2$,

$$\int_{0}^{1} dx \ln D_{2} = \ln M^{2} - 1 - \frac{1}{M^{2}} \left(\frac{q^{2}}{2} + m^{2} \ln \frac{m^{2}}{M^{2}} \right), \qquad (C \cdot 11)$$

$$\int_{0}^{1} dx \, x \ln D_{2} = \frac{1}{2} \ln M^{2} - \frac{3}{4} - \frac{1}{M^{2}} \left(\frac{q^{2}}{3} + \frac{m^{2}}{2} + m^{2} \ln \frac{m^{2}}{M^{2}} \right), \qquad (C \cdot 12)$$

$$\int_{0}^{1} dx \ x^{2} \ln D_{2} = \frac{1}{3} \ln M^{2} - \frac{11}{18} - \frac{1}{M^{2}} \left(\frac{q^{2}}{4} + \frac{5}{6} m^{2} + m^{2} \ln \frac{m^{2}}{M^{2}} \right). \quad (C \cdot 13)$$

For the integrals of type $(C \cdot 5)$, we use the device

$$D_{2}' = [M_{1}^{2}(1-x) + M_{2}^{2}x](1+f'),$$

$$f' = \frac{-q^{2}x(1-x)}{M_{1}^{2}(1-x) + M_{2}^{2}x},$$
 (C·14)

and expand the integrals in powers of f'. We find the following formulas,

$$\begin{split} &\int_{0}^{1} dx \ln D_{2}' = \ln M_{2}^{2} - \frac{M_{1}^{2}}{M_{1}^{2} - M_{2}^{2}} \ln \frac{M_{2}^{2}}{M_{1}^{2}} - 1 \\ &+ \frac{q^{2}}{M_{1}^{2} - M_{2}^{2}} \left[\frac{1}{2} - \frac{M_{1}^{2}}{M_{1}^{2} - M_{2}^{2}} - \frac{M_{1}^{2}M_{2}^{2}}{(M_{1}^{2} - M_{2}^{2})^{2}} \ln \frac{M_{2}^{2}}{M_{1}^{2}} \right], \quad (C \cdot 15) \\ &\int_{0}^{1} dx \ x \ln D_{2}' = \frac{1}{2} \ln M_{2}^{2} - \frac{1}{4} - \frac{1}{2} \left(\frac{M_{1}^{2}}{M_{1}^{2} - M_{2}^{2}} \right)^{2} \ln \frac{M_{2}^{2}}{M_{1}^{2}} - \frac{M_{1}^{2}}{2(M_{1}^{2} - M_{2}^{2})} \\ &+ \frac{q^{2}}{M_{1}^{2} - M_{2}^{2}} \left[\frac{1}{6} + \frac{M_{1}^{2}}{2(M_{1}^{2} - M_{2}^{2})} - \left(\frac{M_{1}^{2}}{M_{1}^{2} - M_{2}^{2}} \right)^{2} \\ &- \frac{M_{1}^{4}M_{2}^{2}}{(M_{1}^{2} - M_{2}^{2})^{8}} \ln \frac{M_{2}^{2}}{M_{1}^{2}} \right], \quad (C \cdot 16) \\ &\int_{0}^{1} dx \ x^{2} \ln D_{2}' = \frac{1}{3} \ln M_{2}^{2} - \frac{1}{9} - \frac{1}{3} \left(\frac{M_{1}^{2}}{M_{1}^{2} - M_{2}^{2}} \right)^{8} \ln \frac{M_{2}^{2}}{M_{1}^{2}} \end{split}$$

•

Appendix C

$$-\frac{M_{1}^{2}}{6(M_{1}^{2}-M_{2}^{2})} - \frac{1}{3} \left(\frac{M_{1}^{2}}{M_{1}^{2}-M_{2}^{2}}\right)^{2'} \\ + \frac{q^{2}}{M_{1}^{2}-M_{2}^{2}} \left[\frac{1}{12} + \frac{M_{1}^{2}}{6(M_{1}^{2}-M_{2}^{2})} + \frac{1}{2} \left(\frac{M_{1}^{2}}{M_{1}^{2}-M_{2}^{2}}\right)^{2} \\ - \left(\frac{M_{1}^{2}}{M_{1}^{2}-M_{2}^{2}}\right)^{3} - \frac{M_{1}^{6}M_{2}^{2}}{(M_{1}^{2}-M_{2}^{2})^{4}} \ln \frac{M_{2}^{2}}{M_{1}^{2}}\right]. \quad (C \cdot 17)$$

C.2 Three-point functions (vertex part)

We wish to derive the 1/M expansion formulas for the integrals of the form,

$$\int_{0}^{1} dx \int_{0}^{1} dy y^{n} \ln D_{3}^{(i)}, \quad (n = 0, 1)$$
 (C·18)

$$\int_{0}^{1} dx \int_{0}^{1} dy \ y^{n} / D_{3}^{(i)}, \quad (n = 0, 1, 2, 3)$$
 (C·19)

where i=1 or 2 and

$$D_{\mathfrak{z}^{(1)}} = (1-y) M^2 + y [\overline{m^2} - q^2 x (1-x)] - \overline{p}^2 y (1-y), \qquad (C \cdot 20)$$

$$D_{\mathbf{s}^{(2)}} = (1-y) \, m^2 + y \left[\overline{M^2} - q^2 x \, (1-x) \right] - \overline{p}^2 y \, (1-y) \tag{C.21}$$

with

$$\overline{m^{2}} \equiv (1-x) m_{1}^{2} + x m_{2}^{2},$$

$$\overline{M^{2}} \equiv (1-x) M_{1}^{2} + x M_{2}^{2},$$

$$\overline{p} \equiv (1-x) p_{1} + x p_{2}.$$
(C·22)

For this purpose it is sufficient to consider the case where the D_s ⁽ⁱ⁾'s are replaced by

$$D = (1 - y)C + yB - y(1 - y)A$$
 (C·23)

with $C \gg B$, A. The formulas for the case $B \gg C$, A may be obtained from the above case by changing the integration variable y to 1-y.

The formulas for the integral $(C \cdot 18)$ are essentially the same as Eqs. $(C \cdot 11)$ and $(C \cdot 12)$. The 1/M expansion formulas for the first integration of the integral of type $(C \cdot 19)$ are given by

$$\int_{0}^{1} dy \frac{1}{D} = -\frac{1}{C} \ln \frac{B}{C} - \frac{1}{C^{2}} \left[B \ln \frac{B}{C} + A \left(2 + \ln \frac{B}{C} \right) \right], \qquad (C \cdot 24)$$

$$\int_{0}^{1} dy \frac{y}{D} = -\frac{1}{C} \left[1 + \ln \frac{B}{C} \right] - \frac{1}{C^{2}} \left[B \left(1 + 2 \ln \frac{B}{C} \right) + A \left(\frac{5}{2} + \ln \frac{B}{C} \right) \right], \qquad (C \cdot 25)$$

K-I. Aoki, Z. Hioki, R. Kawabe, M. Konuma and T. Muta

$$\int_{0}^{1} dy \frac{y^{2}}{D} = -\frac{1}{C} \left[\frac{3}{2} + \ln \frac{B}{C} \right] - \frac{1}{C^{2}} \left[B \left(\frac{5}{2} + 3 \ln \frac{B}{C} \right) + A \left(\frac{17}{6} + \ln \frac{B}{C} \right) \right],$$
(C·26)
$$\int_{0}^{1} dy \frac{y^{3}}{D} = -\frac{1}{C} \left[\frac{11}{6} + \ln \frac{B}{C} \right] - \frac{1}{C^{2}} \left[B \left(\frac{13}{3} + 4 \ln \frac{B}{C} \right) + A \left(\frac{37}{12} + \ln \frac{B}{C} \right) \right].$$
(C·27)

The remaining x-integration can be performed by applying the formulas already obtained.

C.3 Four-point functions (box diagram)

We consider the function A_4 in Eq. (C·3) and pay attention to the zintegration. The relevant integrals are of the form,

$$\int_{0}^{1} dz \frac{z^{n}}{D_{4}}, \quad \int_{0}^{1} dz \frac{z^{n}}{D_{4}^{2}}, \quad (C \cdot 28)$$

where

$$D_4 = Cz + B + Az \tag{C.29}$$

with $C \gg B$, A and

$$C = (1-x) M_1^2 + x M_2^2 . \qquad (C \cdot 30)$$

We expand Eq. (C·28) in powers of 1/C. For this purpose we use the following form of D_4 ,

$$D_4 = (Cz+B)(1+f),$$

$$f = \frac{Az}{Cz+B}.$$
 (C·31)

Expanding the integrands in Eq. (C·28) in powers of f, we obtain for $n \ge 1$

$$\int_{0}^{1} dz \frac{z^{n}}{D_{4}} = \begin{cases} \frac{1}{nC} - \frac{A}{nC^{2}} - \frac{1}{n-1} \frac{B}{C^{2}}, & (n \ge 2) \\ \frac{1}{C} - \frac{A}{nC^{2}} + \frac{B}{C^{2}} \ln \frac{B}{C}, & (n = 1) \end{cases}$$

$$\int_{0}^{1} dz \frac{z^{n}}{D_{4}^{2}} = \begin{cases} \frac{1}{n-1} \frac{1}{C^{2}}, & (n \ge 2) \\ -\frac{1}{C^{2}} - \frac{1}{C^{2}} \ln \frac{B}{C}, & (n = 1) \end{cases}$$
(C·32)
(C·33)

The remaining x- and y-integration may be performed in the same way as before.

For the box diagram with the photon and W boson as internal lines (see Fig. 5.2), the function D_4 takes the form,

$$D_4 = Czx + B + Azx, \qquad (C \cdot 34)$$

where $C = M^2 \gg B$, A. As before we rewrite D_4 such that

$$D_4 = (Czx + B) (1 + f)$$
 (C·35)

with

$$f = \frac{Azx}{Czx + B} \,. \tag{C.36}$$

We expand the necessary integrals in powers of f and perform the z-integrations term by term. The remaining x- and y-integration, however, are not so simple. The full expression is given in Eq. (5.64).

Appendix D

F_n and G_n Functions

As has already been mentioned in Appendix C, the integration appearing in the calculation of the two-point functions may be performed explicitly. It is convenient to define the following functions F_n and F:

$$F_n(M_1, M_2, q^2) \equiv \int_0^1 dx \, x^n \ln[M_1^2(1-x) + M_2^2x - q^2x(1-x)], \quad (D \cdot 1)$$

$$F(M_1, M_2, q^2) \equiv F_1(M_1, M_2, q^2) - F_2(M_1, M_2, q^2) , \qquad (D \cdot 2)$$

where n=0, 1, 2. The analytic expressions for F_0 , F_1 and F_2 are given by

$$F_{0}(M_{1}, M_{2}, q^{2}) = \ln M_{2}^{2} - 2 - \frac{1}{2} (1 + \delta) \ln \left(\frac{M_{2}^{2}}{M_{1}^{2}}\right) + \frac{1}{2} r \ln \rho , \qquad (D \cdot 3)$$

$$F_{1}(M_{1}, M_{2}, q^{2}) = \frac{1}{2} \ln M_{2}^{2} - 1 - \frac{1}{2} \left\{ \frac{1}{2} (1+\delta)^{2} - \frac{M_{1}^{2}}{q^{2}} \right\} \ln \left(\frac{M_{2}^{2}}{M_{1}^{2}} \right)$$
$$- \frac{1}{2} \delta + \frac{1}{4} (1+\delta) r \ln \rho, \qquad (D \cdot 4)$$

$$F_{2}(M_{1}, M_{2}, q^{2}) = \frac{1}{3} \ln M_{2}^{2} - \frac{2}{9} - \frac{1}{6} (1+\delta) \left\{ (1+\delta)^{2} - \frac{3M_{1}^{2}}{q^{2}} \right\} \ln \left(\frac{M_{2}^{2}}{M_{1}^{2}} \right)$$
$$- \frac{1}{6} (1+\delta) (3+2\delta) + \frac{2M_{1}^{2}}{3q^{2}}$$

K-I. Aoki, Z. Hioki, R. Kawabe, M. Konuma and T. Muta

$$+\frac{1}{6}\left\{(1+\delta)^2 - \frac{M_1^2}{q^2}\right\} r \ln \rho , \qquad (D.5)$$

where

$$\delta = (M_1^2 - M_2^2)/q^2, \qquad (D \cdot 6)$$

$$r \equiv \sqrt{(1+\delta)^2 - 4M_1^2/q^2},$$
 (D.7)

$$\rho \equiv \frac{M_1^2 + M_2^2 - (1+r)q^2}{M_1^2 + M_2^2 - (1-r)q^2}.$$
 (D.8)

The function G_n is defined by

$$G_{n}(M_{1}, M_{2}, q^{2}) \equiv \frac{\partial}{\partial q^{2}} F_{n}(M_{1}, M_{2}, q^{2}),$$

= $-\int_{0}^{1} dx \, x^{n+1} (1-x) / [M_{1}^{2}(1-x) + M_{2}^{2}x - q^{2}x(1-x)]$
(D.9)

Performing the integration (D.9), we find for G_0 , G_1 and G_2 ,

$$G_{0}(M_{1}, M_{2}, q^{2}) = \frac{1}{q^{2}} \left[1 + \frac{1}{2} \delta \ln \frac{M_{2}^{2}}{M_{1}^{2}} - \frac{1}{2} \left(\delta^{2} - \frac{M_{1}^{2} + M_{2}^{2}}{q^{2}} \right) \frac{\ln \rho}{r} \right],$$
(D.10)

$$G_{1}(M_{1}, M_{2}, q^{2}) = \frac{1}{q^{2}} \left[\frac{1}{2} + \delta + \frac{1}{2} \left(\delta^{2} - \frac{M_{2}^{2}}{q^{2}} \right) \ln \frac{M_{2}^{2}}{M_{1}^{2}} + \frac{1}{2} \left(\frac{M_{2}^{2}}{q^{2}} + \frac{M_{1}^{2} + 2M_{2}^{2}}{q^{2}} \delta - \delta^{3} \right) \frac{\ln \rho}{r} \right], \qquad (D.11)$$

$$\begin{aligned} G_{2}(M_{1}, M_{2}, q^{2}) &= \frac{1}{q^{2}} \bigg[\frac{1}{3} + \delta \left(1 + \delta \right) - \frac{M_{1}^{2} + M_{2}^{2}}{2q^{2}} \\ &+ \frac{1}{2} \left\{ \delta \left(1 + \delta \right)^{2} - \frac{M_{1}^{2}}{q^{2}} (1 + 2\delta) \right\} \ln \frac{M_{2}^{2}}{M_{1}^{2}} \\ &+ \frac{1}{2} \left\{ \frac{M_{1}^{2}}{q^{2}} (1 + \delta) \left(1 + 4\delta \right) - \delta \left(1 + \delta \right)^{3} - \frac{2M_{1}^{4}}{q^{4}} \right\} \frac{\ln q}{r} \bigg]. \end{aligned}$$
(D·12)

Appendix E

Renormalization Constants

We first summarize all the relevant renormalization constants defined in § 4.3:

$$Z_{ij}^{1/2} (i, j = A, Z), \quad \delta M_Z^2,$$

$$Z_W, \quad \delta M_W^2, \quad T,$$

$$(Z_L^{1/2})_{nm}, (Z_R^{1/2})_{nm}, \delta m_{i,I},$$

$$Z_\chi, \quad Z_{\chi_s}, \quad Z_{\phi}, \quad \delta m_{\phi}^2,$$

$$\tilde{Z}_{ij} (i, j = A, Z), \quad \tilde{Z}_3, \quad Y.$$
(E.1)

These constants are determined on the mass shell by the use of the renormalization conditions described in § 4.3. In actual calculations in Chapter 5, we need only the following subset of the renormalization constants $(E \cdot 1)$ according to our approximations and assumptions:

$$Z_{ij}^{1/2}, \quad \delta M_{z}^{2}, \quad Z_{w}, \quad \delta M_{w}^{2}, \quad (Z_{L,R}^{l})^{1/2} \ (= (Z_{L,R}^{1/2})_{ll}),$$

$$\delta m_{l} \ (l = e \text{ or } \mu), \quad Y. \qquad (E \cdot 2)$$

We here present the explicit expressions for the above constants $(E \cdot 2)$ in one-loop order in the on-shell renormalization scheme.

E.1 Gauge-boson field renormalization constants

The constants $Z_{ZZ}^{1/2}$, $Z_{AZ}^{1/2}$, $Z_{ZA}^{1/2}$, $Z_{AA}^{1/2}$ and Z_W are calculated using the Feynman diagrams Figs. 5.3, 5.4, 5.5 in §5.3 and Fig. 5.6 in §5.4 respectively. We list the results in the following:

$$Z_{ZZ}^{1/2} = 1 - \frac{e^2}{24\pi^2} \frac{C_{\text{UV}}}{M_w^2 (M_z^2 - M_w^2)} \left[\frac{N}{3} (8M_w^4 - 10M_w^2 M_z^2 + 5M_z^4) - \frac{1}{8} (18M_w^4 + 2M_w^2 M_z^2 - M_z^4) \right], \quad (E \cdot 3)$$

$$Z_{AZ}^{1/2} = \frac{e^2}{12\pi^2} \frac{C_{\rm UV}}{M_w \sqrt{M_z^2 - M_w^2}} \left[-\frac{N}{3} (8M_w^2 - 5M_z^2) + \frac{1}{8} (30M_w^2 + M_z^2) \right],$$
(E·4)

$$Z_{ZA}^{1/2} = -\frac{e^2}{8\pi^2} \frac{M_W}{\sqrt{M_Z^2 - M_W^2}} (C_{\rm UV} - \ln M_W^2) , \qquad (E \cdot 5)$$

$$Z_{AA}^{1/2} = 1 - \frac{e^2}{8\pi^2} \left[\frac{1}{3} \sum_i Q_i^2 (C_{\text{UV}} - \ln m_i^2) - \frac{3}{4} (C_{\text{UV}} - \ln M_{\text{W}}^2) - \frac{1}{6} \right], \quad (E \cdot 6)$$

$$Z_{W} = 1 + \frac{e^{2}}{12\pi^{2}} \frac{M_{Z}^{2}}{M_{Z}^{2} - M_{W}^{2}} C_{UV} \left(-N + \frac{19}{8} \right), \qquad (E \cdot 7)$$

where the summation on *i* runs over all fermions and color degrees of freedom, N is the number of generations and C_{UV} is given in Eq. (5.1) with $\mu = m$: $C_{UV} = 1/\varepsilon - \gamma + \ln 4\pi$. It should be noted here that we have discarded all finite pieces in Eqs. (E·3), (E·4) and (E·7). The reason for this is that $Z_{ZZ}^{1/2}$, $Z_{AZ}^{1/2}$ and Z_W are cancelled out when all the Feynman amplitudes are summed up and only the divergent pieces are necessary to get finite amplitudes in the intermediate stage. The renormalization constant $Z_{AZ}^{1/2}$ is necessary only in determining the constant Y.

E.2 Gauge-boson mass shifts

The mass shifts δM_z^2 and δM_{W^2} are obtained by considering Fig. 5.3 in §53 and Fig. 5.6 in §5.4 respectively. Utilizing the F_n and F functions defined in Appendix D and η_i defined in Eq. (5.10), we present the results for δM_z^2 and δM_W^2 in the following:

$$\delta M_{w}^{2} = -\frac{e^{2}M_{Z}^{2}}{96\pi^{2}(M_{Z}^{2} - M_{W}^{2})} \bigg[\sum_{(I,i)} |U_{Ii}|^{2} (3m_{I}^{2} + 3m_{i}^{2} - 2M_{W}^{2}) + 31M_{W}^{2} - 6M_{Z}^{2} \bigg] C_{UV} + \delta M_{Wf}^{2}, \qquad (E \cdot 10)$$

$$\delta M_{Wf}^{2} = \frac{-e^{2}}{32\pi^{2} (M_{Z}^{2} - M_{W}^{2})} \bigg[2M_{Z}^{2} \sum_{(I,i)} |U_{Ii}|^{2} \{ 2M_{W}^{2}F(m_{I}, m_{i}, M_{W}^{2}) - m_{i}^{2}F_{1}(m_{I}, m_{i}, M_{W}^{2}) - m_{I}^{2}F_{1}(m_{i}, m_{I}, M_{W}^{2}) \bigg\}$$

$$-M_{Z}^{2}M_{W}^{2} + 7M_{Z}^{2}M_{W}^{2}\ln M_{W}^{2} + M_{Z}^{2}\left(6M_{W}^{2} + \frac{M_{Z}^{2}}{2}\right)\ln M_{Z}^{2}$$

$$+ \left\{\left(M_{Z}^{4} - 20M_{Z}^{2}M_{W}^{2} - 8M_{W}^{4}\right)F_{0}\left(M_{Z}, M_{W}, M_{W}^{2}\right)\right.$$

$$+ \left(M_{Z}^{4} + 16M_{Z}^{2}M_{W}^{2} + 4M_{W}^{4}\right)F_{1}\left(M_{Z}, M_{W}, M_{W}^{2}\right)$$

$$- M_{W}^{2}\left(M_{Z}^{2} + 20M_{W}^{2}\right)F_{2}\left(M_{Z}, M_{W}, M_{W}^{2}\right)$$

$$+ 4M_{W}^{2}\left(M_{Z}^{2} - M_{W}^{2}\right)\left\{-2F_{0}\left(0, M_{W}, M_{W}^{2}\right)$$

$$+ F_{1}\left(0, M_{W}, M_{W}^{2}\right) - 5F_{2}\left(0, M_{W}, M_{W}^{2}\right)\right\}$$

$$+ m_{\phi}^{2}M_{Z}^{2}\left\{\frac{1}{2}\ln m_{\phi}^{2} - F_{1}\left(M_{W}, m_{\phi}, M_{W}^{2}\right)\right\}$$

$$+ M_{Z}^{2}M_{W}^{2}\left\{2F_{0}\left(m_{\phi}, M_{W}, M_{W}^{2}\right) - F_{2}\left(m_{\phi}, M_{W}, M_{W}^{2}\right)\right\}\left]. \quad (E \cdot 11)$$

E.3 Lepton field renormalization constants

The constants Z_{L}^{ν} , Z_{L}^{l} and Z_{R}^{l} are calculated through the Feynman diagrams Fig. 5.7 in § 5.3. In the calculation we have neglected the Higgs contribution since its effect is of order m_l/M_W . The result is as follows:

$$Z_{L}^{\nu} = 1 - \frac{e^{2}}{16\pi^{2}} \frac{M_{Z}^{2}}{4M_{W}^{2}(M_{Z}^{2} - M_{W}^{2})} [(2M_{W}^{2} + M_{Z}^{2})(C_{UV} - 1) - 4M_{W}^{2}F_{1}(m_{l}, M_{W}, 0) - 2M_{Z}^{2}F_{1}(0, M_{Z}, 0)], \quad (E \cdot 12)$$

$$\simeq 1 - \frac{e^2}{16\pi^2} \frac{M_Z^2}{4M_W^2 (M_Z^2 - M_W^2)} \left[(2M_W^2 + M_Z^2) \left(C_{\rm UV} - \frac{1}{2} \right) - 2M_W^2 \ln M_W^2 - M_Z^2 \ln M_Z^2 \right],$$
 (E·13)

$$Z_{L}^{l} = 1 - \frac{e^{2}}{16\pi^{2}} \left[C_{UV} - 1 - 2F_{1}(m_{l}, 0, m_{l}^{2}) + 4m_{l}^{2} \left\{ 2G_{0}(m_{l}, \lambda, m_{l}^{2}) - G_{1}(m_{l}, \lambda, m_{l}^{2}) \right\} \right] - \frac{e^{2}}{16\pi^{2}} \frac{M_{Z}^{2}}{2(M_{Z}^{2} - M_{W}^{2})} \left[C_{UV} - 1 - 2F_{1}(0, M_{W}, m_{l}^{2}) - 2m_{l}^{2} \cdot G_{1}(0, M_{W}, m_{l}^{2}) \right] - 2m_{l}^{2} \cdot G_{1}(0, M_{W}, m_{l}^{2}) \right] - \frac{e^{2}}{16\pi^{2}} \left[\frac{(M_{Z}^{2} - 2M_{W}^{2})^{2}}{4M_{W}^{2}(M_{Z}^{2} - M_{W}^{2})} \left\{ C_{UV} - 1 - 2F_{1}(m_{l}, M_{Z}, m_{l}^{2}) \right\} \right. + 4m_{l}^{2} \left\{ \frac{M_{Z}^{2} - 2M_{W}^{2}}{M_{W}^{2}} G_{0}(m_{l}, M_{Z}, m_{l}^{2}) - \frac{8(M_{Z}^{2} - M_{W}^{2})^{2} + M_{Z}^{2}(4M_{W}^{2} - 3M_{Z}^{2})}{8M_{W}^{2}(M_{Z}^{2} - M_{W}^{2})} G_{1}(m_{l}, M_{Z}, m_{l}^{2}) \right\} \right], \quad (E \cdot 14)$$

$$\simeq 1 - \frac{e^2}{16\pi^2} \bigg[C_{\rm UV} + 2C_{\rm IR} + 4 - 3\ln m_i^2 + \frac{1}{2} \frac{M_Z^2}{M_Z^2 - M_W^2} \Big(C_{\rm UV} - \frac{1}{2} - \ln M_W^2 \Big) \\ + \frac{1}{4} \frac{(M_Z^2 - 2M_W^2)^2}{M_W^2 (M_Z^2 - M_W^2)} \Big(C_{\rm UV} - \frac{1}{2} - \ln M_Z^2 \Big) \bigg], \qquad (E \cdot 15)$$

$$Z_{R}^{l} = 1 - \frac{e^{2}}{16\pi^{2}} [C_{UV} - 1 - 2F_{1}(m_{l}, 0, m_{l}^{2}) + 4m_{l}^{2} \{2G_{0}(m_{l}, \lambda, m_{l}^{2}) - G_{1}(m_{l}, \lambda, m_{l}^{2})\}] + \frac{e^{2}}{16\pi^{2}} \frac{M_{Z}^{2}}{M_{Z}^{2} - M_{W}^{2}} m_{l}^{2}G_{1}(0, M_{W}, m_{l}^{2}) - \frac{e^{2}}{16\pi^{2}} \left[\frac{M_{Z}^{2} - M_{W}^{2}}{M_{W}^{2}} \{C_{UV} - 1 - 2F_{1}(m_{l}, M_{Z}, m_{l}^{2})\} + 4m_{l}^{2} \left\{ \frac{M_{Z}^{2} - 2M_{W}^{2}}{M_{W}^{2}} G_{0}(m_{l}, M_{Z}, m_{l}^{2}) - \frac{8(M_{Z}^{2} - M_{W}^{2})^{2} + M_{Z}^{2}(4M_{W}^{2} - 3M_{Z}^{2})}{8M_{W}^{2}(M_{Z}^{2} - M_{W}^{2})} G_{1}(m_{l}, M_{Z}, m_{l}^{2}) \right\} \right],$$
(E · 16)
$$\simeq 1 - \frac{e^{2}}{16\pi^{2}} \left[C_{UV} + 2C_{IR} + 4 - 3\ln m_{l}^{2} + \frac{M_{Z}^{2} - M_{W}^{2}}{M_{W}^{2}} \left(C_{UV} - \frac{1}{2} - \ln M_{Z}^{2}\right) \right].$$

E.4 Lepton mass shifts

The lepton mass-renormalization constant δm_t is given by

$$\delta m_{i} = \frac{e^{2}}{16\pi^{2}} \frac{m_{i}}{8M_{w}^{2}(M_{z}^{2} - M_{w}^{2})} \left[-(11M_{z}^{4} - 14M_{z}^{2}M_{w}^{2})C_{UV} + 3M_{z}^{4} - 6M_{z}^{2}M_{w}^{2} + 16M_{w}^{2}(M_{z}^{2} - M_{w}^{2}) \left\{ 2F_{0}(m_{i}, 0, m_{i}^{2}) - F_{1}(m_{i}, 0, m_{i}^{2}) \right\} - 4M_{z}^{2}M_{w}^{2}F_{1}(0, M_{w}, m_{i}^{2}) + 16(M_{z}^{2} - M_{w}^{2})(M_{z}^{2} - 2M_{w}^{2})F_{0}(m_{i}, M_{z}, m_{i}^{2}) - 2(5M_{z}^{4} - 12M_{z}^{2}M_{w}^{2} + 8M_{w}^{4})F_{1}(m_{i}, M_{z}, m_{i}^{2}) \right].$$
(E·18)

E.5 Vertex renormalization constant Y

To calculate the constant Y, we first evaluate the *eeA* vertex function at $q^2=0$ (see Fig. 5.8 in § 5.3) and then add the counter term found in § 4.3 with $Z_{AA}^{1/2}$, $Z_{ZA}^{1/2}$, Z_{R}^{e} and Z_{R}^{e} as given in E.1 and E.3. We find

$$Y = 1 - \frac{e^2}{16\pi^2} \left[\frac{7}{2} (C_{\rm UV} - \ln M_{\rm W}^2) + \frac{1}{3} - \frac{2}{3} \sum_i Q_i^2 (C_{\rm UV} - \ln m_i^2) \right]. \quad (E \cdot 19)$$

(E·17)

Appendix F

Spence Function

Feynman parameter integrals appearing in one-loop amplitudes can be performed analytically leaving one special function unintegrated ['t Hooft and Veltman 79]. The special function is called the dilogarithm or the Spence function [Lewin 58]. In this appendix we list some useful properties of the Spence function. The Spence function is defined by Eq. (5.40), i.e.,

$$\operatorname{Sp}(z) = -\int_{0}^{z} dt \frac{\ln(1-t)}{t} \,. \tag{F.1}$$

It is an analytic function of complex z with a branch cut from z=1 to ∞ . To see this property more clearly, we reexpress Eq. (F.1) by making integration by parts with a suitable change of the integration variable:

$$\operatorname{Sp}(z) = \int_0^1 dt \frac{\ln t}{t - 1/z} \,. \tag{F.2}$$

Obviously in Eq. (F·2) we observe that the branch cut lies in the interval $0 \le 1/z \le 1$.

There are some useful relations among the Spence functions with different arguments:

$$\operatorname{Sp}(z) + \operatorname{Sp}(1-z) = \frac{\pi^2}{6} - \ln z \ln(1-z)$$
, (F·3)

$$\operatorname{Sp}\left(-\frac{1}{z}\right) + \operatorname{Sp}\left(-z\right) = -\frac{\pi^{2}}{6} - \frac{1}{2} (\ln z)^{2},$$
 (F·4)

$$\operatorname{Sp}(x) + \operatorname{Sp}\left(-\frac{x}{1-x}\right) = -\frac{1}{2}(\ln(1-x))^2, \quad x < 1.$$
 (F·5)

The series expansion of Sp(z) in powers of z reads

$$\operatorname{Sp}(z) = \sum_{n=1}^{\infty} \frac{z^n}{n^2}, \quad |z| \le 1.$$
 (F·6)

At some special values of z, Sp(z) can be expressed in terms of the known constants:

$$Sp(1) = \pi^2/6$$
, (F·7)

$$Sp(-1) = -\pi^2/12$$
, (F·8)

$$Sp(1/2) = \pi^2/12 - (1/2) (\ln 2)^2.$$
 (F·9)

Appendix G

Formulas for the Generating Functional and the Effective Action

We derive formulas for the generating functional W[J, K] defined in Eq. (2.95) and the effective action $\Gamma[\phi, K]$ defined in Eq. (2.99). We deal with both commuting and anti-commuting field variables simultaneously and give a compact proof of formulas [Aoki 79].

We start by recalling the definition of the generating functional W[J, K],

$$\exp iW[J,K] \equiv \langle 0|T \exp iS[J,K]|0\rangle, \qquad (G\cdot 1)$$

$$S[J,K] \equiv \int d^4x [J_i(x)\hat{\phi}_i(x) + K_i(x)\delta^{\text{BRS}}\hat{\phi}_i(x)], \qquad (G\cdot 2)$$

where $\hat{\phi}_i(x)$ represent field operators and $J_i(x)$ are corresponding *c*-number sources with the same commutativity as that of $\hat{\phi}_i(x)$. The BRS sources K_i are also introduced and its commutativity is the same as that of corresponding $\delta^{\text{BRS}} \hat{\phi}_i(x)$.

First of all we introduce the connected part of a Green function. The connected part is defined graphically with the aid of the corresponding Feynman diagram. The vacuum expectation values of operators should be subtracted in the connected part. We give some examples. It is convenient to introduce \bar{O} defined by

$$\bar{O}_i \equiv O_i - \langle 0 | O_i | 0 \rangle . \tag{G.3}$$

With the aid of these operators \bar{O}_i , we have

$$\langle 0|TO_1O_2|0\rangle^c = \langle 0|TO_1O_2|0\rangle - \langle 0|O_1|0\rangle \langle 0|O_2|0\rangle$$
$$= \langle 0|T\overline{O}_1\overline{O}_2|0\rangle, \qquad (G \cdot 4a)$$

$$\langle 0|TO_1O_2O_3|0\rangle^c = \langle 0|TO_1O_2O_3|0\rangle, \qquad (G\cdot 4b)$$

$$\langle 0 | TO_1O_2O_3O_4 | 0 \rangle^c = \langle 0 | T\overline{O}_1\overline{O}_2\overline{O}_3\overline{O}_4 | 0 \rangle$$

$$- \langle 0 | T\overline{O}_1\overline{O}_2 | 0 \rangle \langle 0 | T\overline{O}_3\overline{O}_4 | 0 \rangle$$

$$- \langle 0 | T\overline{O}_1\overline{O}_3 | 0 \rangle \langle 0 | T\overline{O}_2\overline{O}_4 | 0 \rangle$$

$$- \langle 0 | T\overline{O}_1\overline{O}_4 | 0 \rangle \langle 0 | T\overline{O}_2\overline{O}_3 | 0 \rangle , \qquad (G \cdot 4c)$$

where we have assumed that O_i are commuting operators.

Consider the term with S^n in the expansion of the right-hand side of Eq. (G·1), $\langle 0|(1/n!) T(iS)^n|0\rangle$. This *n*-point amplitude is a sum of various products of connected amplitudes:

$$\langle 0|\frac{1}{n!}\mathrm{T}(iS)^{n}|0\rangle = \sum_{\{\{n_{k}\}\mid \bigcup_{k=0}^{\infty}kn_{k}=n, n_{k}\geq 0\}} \frac{1}{n!}\mathrm{P}(\{n_{k}\}) \prod_{k=0}^{\infty}(\langle 0|\mathrm{T}(iS)^{k}|0\rangle^{c})^{n_{k}}.$$
(G·5)

Appendix G 215

We characterize a product of connected amplitudes by a set $\{n_k\}$ where n_k is the number of k-point amplitudes in the product of connected amplitudes. The factor $P(\{n_k\})$ in Eq. (G.5) is the number of combinations to separate n points into parts where the number of k points is n_k set. This factor is evaluated as

$$P(\lbrace n_k \rbrace) = n! / \left(\prod_{k=0}^{\infty} (k!)^{n_k} n_k! \right). \qquad (G \cdot 6)$$

Substituting Eqs. (G.6) and (G.5) into Eq. (G.1), we have

$$\exp iW = \sum_{n=0}^{\infty} \langle 0|\frac{1}{n!} \mathrm{T}(iS)^{n}|0\rangle$$

$$= \sum_{n=0}^{\infty} \sum_{\{\{n_{k}\}\mid \sum_{k=0}^{\infty} kn_{k}=n, n_{k}\geq 0\}} \prod_{k=0}^{\infty} \frac{1}{n_{k}!} (\langle 0|\frac{1}{k!} \mathrm{T}(iS)^{k}|0\rangle^{c})^{n_{k}}$$

$$= \prod_{k=0}^{\infty} \sum_{n_{k}=0}^{\infty} \frac{1}{n_{k}!} (\langle 0|\frac{1}{k!} \mathrm{T}(iS)^{k}|0\rangle^{c})^{n_{k}}$$

$$= \prod_{k=0}^{\infty} \exp \langle 0|\frac{1}{k!} \mathrm{T}(iS)^{k}|0\rangle^{c}$$

$$= \exp \langle 0|\mathrm{T} \exp iS|0\rangle^{c}, \qquad (G.7)$$

Thus the functional iW is the connected part of the Green function,

$$iW = \langle 0|T \exp iS|0 \rangle^{c}$$
. (G·8)

Before proceeding to the formulas for differentiation of W, it is necessary to fix a convention for the differentiation with respect to anticommuting variables. We adopt the so-called left-differentiation. In this convention the distribution formula is

$$\frac{\delta}{\delta\phi}(AB) = \left(\frac{\delta A}{\delta\phi}\right)B + (-1)_{\phi}^{A}A\left(\frac{\delta B}{\delta\phi}\right), \qquad (G\cdot9)$$

where $(-1)_{\phi}^{A}$ is +1 (-1) when A is commutative (anticommutative) with ϕ . The formulas for differentiation of W with respect to sources are

$$\frac{1}{i^{n-1}} \frac{\delta^n W}{\delta S_1 \delta S_2 \cdots \delta S_n} = \langle 0 | TO_1 O_2 \cdots O_n \exp iS | 0 \rangle^c, \qquad (G \cdot 10)$$

where we have used the following simplified notations:

$$S[J, k] \equiv J_i \cdot \hat{\phi}_i + K_i \cdot \delta^{\text{BRS}} \hat{\phi}_i \equiv S_i \cdot O_i, \qquad (G \cdot 11a)$$

$$S_i \equiv \{J, K\}, \quad O_i \equiv \{\hat{\phi}, \delta^{BRS} \hat{\phi}\}.$$
 (G·11b)

It should be mentioned that the order of differentiation and that of operators are correlated with each other. Setting all sources to be zero, we have K-I. Aoki, Z. Hioki, R. Kawabe, M. Konuma and T. Muta

$$\frac{1}{i^{n-1}} \frac{\delta^n W}{\delta S_1 \delta S_2 \cdots \delta S_n} \bigg|_0 = \langle 0 | \mathrm{T} O_1 O_2 \cdots O_n | 0 \rangle^c \,. \tag{G.12}$$

Now let us proceed to the effective action defined by

$$\Gamma[\phi, K] = W[J, K] - J_i \cdot \phi_i, \qquad (G \cdot 13)$$

$$\phi_i \equiv \frac{\delta}{\delta J_i} W[J, K] . \tag{G.14}$$

In Eq. (G.13) J_i is understood to be a functional of ϕ (and K). In fact J_i is expressed as

$$J_i = \varepsilon^i \frac{\delta \Gamma}{\delta \phi_i}, \qquad (G \cdot 15)$$

where ε^i is a sign factor defined by

 $\varepsilon^i = -1(+1)$ for commuting ϕ_i (anticommuting ϕ_i). (G.16)

We start with the following identity:

$$\frac{\delta \phi_i}{\delta \phi_k} = \delta_{ik} . \tag{G.17}$$

By using the definition $(G \cdot 14)$ and the formula $(G \cdot 10)$, we have

$$\frac{\delta}{\delta\phi_k} \frac{\delta W}{\delta J_i} = \frac{\delta}{\delta\phi_k} \langle 0 | \mathrm{T}\hat{\phi}_i \exp iS | 0 \rangle^c = \delta_{ik} . \qquad (G \cdot 18)$$

This is expressed graphically as

$$\frac{\delta}{\delta\phi_k} \cdot \bigodot^{\phi_i} = \delta_{ik} . \tag{G.19}$$

Thus the operator $\delta/\delta\phi_k$ amputates the "leg" $\langle 0|T\hat{\phi}_i \exp iS|0\rangle^c$.

We rewrite Eq. $(G \cdot 18)$ as

$$\delta_{ik} = \frac{\delta J_j}{\delta \phi_k} \frac{\delta^2 W}{\delta J_j \delta J_i} = \varepsilon^j \frac{\delta^2 \Gamma}{\delta \phi_k \delta \phi_j} \frac{\delta^2 W}{\delta J_j \delta J_i}$$
$$= \varepsilon^j \frac{i \delta^2 \Gamma}{\delta \phi_k \delta \phi_j} \langle 0 | \mathrm{T} \hat{\phi}_j \hat{\phi}_i \exp i S | 0 \rangle^{\mathrm{c}} . \qquad (G \cdot 20)$$

By setting all field (source) variables to their vacuum expectation values (zero): $\phi_i = \langle 0 | \hat{\phi}_i | 0 \rangle$, $K_i = J_i = 0$ (this setting is represented by the symbol $|_0$), we have

$$\frac{1}{i} \left. \frac{\delta^2 \Gamma}{\delta \phi_j \delta \phi_k} \right|_0 \cdot \langle 0 | \mathrm{T} \hat{\phi}_j \hat{\phi}_i | 0 \rangle^{\mathrm{c}} = \delta_{ik} , \qquad (\mathrm{G} \cdot 21)$$
where use has been made of

$$\varepsilon^{j} \frac{\delta^{2} \Gamma}{\delta \phi_{k} \delta \phi_{j}} \bigg|_{0} = -\frac{\delta^{2} \Gamma}{\delta \phi_{j} \delta \phi_{k}} \bigg|_{0} . \tag{G.22}$$

This relation is obtained on the assumption that any anticommuting field has vanishing vacuum expectation value. Equations (G.20) and (G.21) indicate that the second derivative of Γ is the inverse propagator.

We make further differentiation of Eq. (G·20) with respect to ϕ_l and we have

$$\varepsilon^{j} \frac{\delta^{3} \Gamma}{\delta \phi_{i} \delta \phi_{k} \delta \phi_{j}} \cdot \frac{\delta^{2} W}{\delta J_{j} \delta J_{i}} + \varepsilon_{i}^{k j} \varepsilon^{j} \frac{\delta^{2} \Gamma}{\delta \phi_{k} \delta \phi_{j}} \cdot \frac{\delta}{\delta \phi_{i}} \frac{\delta^{2} W}{\delta J_{j} \delta J_{i}} = 0.$$
 (G·23)

The factor ε_i^{kj} is the sign factor depending on the commutativity of ϕ_i with $\phi_k \phi_j$,

$$\varepsilon_{\iota}^{kj} = \begin{cases} +1 & \text{for } \phi_{\iota} \text{ commutative with the product } \phi_{k}\phi_{j}, \\ -1 & \text{for } \phi_{\iota} \text{ anticommutative with the product } \phi_{k}\phi_{j}. \end{cases}$$
(G·24)

In Eq. (G.23) we have used the fact that W and Γ are commutative quantities. We have another useful relation,

$$\frac{\delta^2 W}{\delta J_k \delta J_j} \frac{\delta^2 \Gamma}{\delta \phi_j \delta \phi_i} \varepsilon^i = \delta_{ik} , \qquad (G \cdot 25)$$

which is obtained from the identity

$$\frac{\delta J_i}{\delta J_k} = \delta_{ik} . \tag{G.26}$$

By the use of Eq. (G.25), we rewrite Eq. (G.23) as

$$\frac{i\delta^{3}\Gamma}{\delta\phi_{i}\delta\phi_{k}\delta\phi_{m}} = -\varepsilon_{i}^{kj}\varepsilon^{j}\frac{i\delta^{2}\Gamma}{\delta\phi_{k}\delta\phi_{j}}\left(\frac{\delta}{\delta\phi_{i}}\frac{\delta^{2}W}{i\delta J_{j}\delta J_{i}}\right)\frac{i\delta^{2}\Gamma}{\delta\phi_{i}\delta\phi_{m}}.$$
 (G·27)

The right-hand side of Eq. $(G \cdot 27)$ is understood to be the amputated threepoint vertex with the aid of graphic considerations as follows:

$$\frac{\delta^2 W}{i \delta J_j \delta J_i} = \frac{j}{1 - \frac{1}{1 - \frac{$$

$$\frac{\delta}{\delta\phi_l} \cdot \frac{\mathbf{j}}{\mathbf{0}} \cdot \frac{\mathbf{j}}{\mathbf{0}} = \frac{\mathbf{j}}{\mathbf{0}} \cdot \frac{\mathbf{i}}{\mathbf{0}} \cdot \mathbf{0} \cdot$$

$$\varepsilon_{l}^{kj}\varepsilon_{j}\frac{i\delta^{2}\Gamma}{\delta\phi_{k}\delta\phi_{j}}\cdot\underbrace{\mathbf{j}}_{\mathbf{k}}\underbrace{\mathbf{j}}_{\mathbf{k}}\cdot\underbrace{\mathbf{j}}\cdot\underbrace{\mathbf{j}}_{\mathbf{k}}\cdot\underbrace{\mathbf{j}}\cdot\underbrace{\mathbf{j}}\cdot\underbrace$$

Thus we have

The further differentiation $\delta/\delta\phi_n$ of Eq. (G.29) gives,

$$\frac{i\delta^4\Gamma}{\delta\phi_n\delta\phi_k\delta\phi_m} = -\varepsilon^m \, \iota \, \underbrace{\stackrel{\mathbf{k}}{\underset{\mathbf{n}}{\overset{\mathbf{k}}{\longrightarrow}}} \mathbf{m}}_{\mathbf{n}} \quad . \tag{G.30}$$

Generally we have

$$\frac{i\delta^{n}\Gamma}{\delta\phi_{1}\delta\phi_{2}\cdots\delta\phi_{n}} = -\varepsilon^{n} \qquad (G\cdot31)$$

The *n*-th derivative of Γ is the one-particle irreducible *n*-point functions. Setting all the sources equal to zero and treating the sign factors including ε^n carefully, we obtain the final result:

$$\langle 0 | T \hat{\phi}_{i} \hat{\phi}_{i'} | 0 \rangle^{c} \langle 0 | T \hat{\phi}_{j} \hat{\phi}_{j'} | 0 \rangle^{c} \cdots \langle 0 | T \hat{\phi}_{n} \hat{\phi}_{n'} | 0 \rangle^{c} \frac{i \delta^{N} \Gamma}{\delta \phi_{i'} \delta \phi_{j'} \cdots \delta \phi_{n'} |_{0} }$$

$$= \langle 0 | T \hat{\phi}_{i} \hat{\phi}_{j} \cdots \hat{\phi}_{n} | 0 \rangle^{c}_{\text{proper}}, \quad N \ge 3, \qquad (G \cdot 32)$$

where we have used Eq. (G·22). It should be noted that Eq. (G·32) with N=2 is not equal to Eq. (G·21).

The differentiation of Γ with respect to sources K is easily transformed into that of W:

$$\frac{\delta\Gamma}{\delta K_{i}}\Big|_{\phi \text{ fixed}} = \frac{\delta W}{\delta K_{i}}\Big|_{\phi \text{ fixed}} - \frac{\delta J}{\delta K_{i}}\Big|_{\phi \text{ fixed}} \cdot \phi$$
$$= \frac{\delta W}{\delta K_{i}}\Big|_{J \text{ fixed}}$$
$$= \langle 0|T \ \delta^{\text{BRS}} \widehat{\phi}_{i} \exp iS|0 \rangle . \qquad (G.33)$$

The further differentiation is evaluated as follows:

$$\frac{i^{-N+1}\delta^{M+N}\Gamma}{\delta\phi_{i_1}\delta\phi_{i_2}\cdots\delta\phi_{i_N}\delta K_{j_1}\delta K_{j_2}\cdots\delta K_{j_N}} = \phi_{i_1}^{\phi_{i_2}} \phi_{i_1}^{\phi_{i_M}} , \qquad (G\cdot 34)$$

218

Appendix G

$$\langle 0 | T \hat{\phi}_{i_1} \hat{\phi}_{i_{1'}} | 0 \rangle^c \langle 0 | T \hat{\phi}_{i_2} \hat{\phi}_{i_{2'}} | 0 \rangle^c \cdots \langle 0 | T \hat{\phi}_{i_M} \hat{\phi}_{i_{M'}} | 0 \rangle^c \times \frac{i^{-N+1} \delta^{M+N} \Gamma}{\delta \phi_{i_1'} \delta \phi_{i_{2'}} \cdots \delta \phi_{i_{M'}} \delta K_{j_1'} \delta K_{j_{2'}} \cdots \delta K_{j_{N'}}} = \langle 0 | T \hat{\phi}_{i_1} \hat{\phi}_{i_2} \cdots \hat{\phi}_{i_M} \delta^{\text{BRS}} \hat{\phi}_{j_1} \delta^{\text{BRS}} \hat{\phi}_{j_2} \cdots \delta^{\text{BRS}} \hat{\phi}_{j_M} | 0 \rangle^c_{\text{proper}} .$$
 (G·35)

It should be mentioned that Eqs. (G.34) and (G.35) do not hold for the linear part of fields in $\delta^{BRS}\hat{\phi}$. For example, we take

$$\delta^{\text{BRS}} \hat{\phi}_i = \hat{\phi}_j + \cdots . \tag{G.36}$$

In this case the differentiation of \varGamma with respect to K_i is

$$\frac{\delta \Gamma}{\delta K_{i}} = \langle 0 | \mathrm{T} \hat{\phi}_{j} \exp i S | 0 \rangle^{c} + \cdots$$
$$= \frac{\delta W}{\delta J_{j}} + \cdots$$
$$= \phi_{j} + \cdots . \tag{G.37}$$

For the first term on the right-hand side, further differentiation gives trivial results,

$$\frac{\delta}{\delta\phi_k}\phi_j = \delta_{kj}, \qquad \frac{\delta}{\delta K_k}\phi_j = 0. \qquad (G.38)$$

220

References

Abers F S and B W Lee 1073	
Phys. Reports 9C. 1.	2, 1, 2, 2, 2, 5
Abbott, L. F. and R. M. Barnett, 1978a	
Phys. Rev. Letters 40, 1303.	1.2
Abbott, L. F. and R. M. Barnett, 1978b	
Phys. Rev. D18, 3214.	1.2
Antonelli, F., M. Consoli and G. Corbò, 1980	
Phys. Letters 91B, 90.	6.1, 7.1
Antonelli, F., G. Corbò, M. Consoli and O. Pellegrino, 1981a	
Nucl. Phys. B183, 195.	6.1, 7.1
Antonelli, F., M. Consoli and G. Corbò, 1981b	
Phys. Letters 99B , 475.	7.1
Antonelli, F. and L. Maiani, 1981	
Nucl. Phys. B186 , 269.	7.2
Aoki, K-I., 1979	1
Soryushiron Kenkyu (Kyoto) 59, 383 (in Japanese).	1.3, 3.3, 3.4, G
Aoki, K-1. and Z. Hioki, 1979	1.0
Soryushiron Kenkyu (Kyoto) 00 , I (in Japanese).	1.3
Aoki, K-I., Z. Hioki, K. Kawabe, M. Konuma and I. Muta, 1980	9171
Acki KI Z Hicki P Kowaho M Konuma and T Muta 1081	3.1, 7.1
Prog. Theor. Phys. 65 1001	1 2 2 1 5 1 7 1
Acki K-I and 7 Hicki 1081	1.3, 3.1, 3.1, 7.1
Prog Theor Phys 66 2234	515362
Appelquist, T. W. I. R. Primack and H. R. Quinn, 1972	0.1, 0.0, 0.2
Phys. Rev. D6, 2998.	5.1.7.1
Appelquist, T. W., I. R. Primack and H. R. Ouinn, 1973	0.1, 1.1
Phys. Rev. D7, 2998.	5.1.7.1
Armenise, N. et al., 1979	
Phys. Letters 84B, 137.	5.1
Baltay, C., 1979	
Proceedings of the 19th International Conference on High Energy	Physics, Tokyo
1978, ed. by S. Homma, M. Kawaguchi and H. Miyazawa (Phys	sical Society of
Japan Tokyo), 882.	1.2
Bardeen, W. A., A. J. Buras, D. W. Duke and T. Muta, 1978	
Phys. Rev. D18, 3998.	3.1
Bardin, D. Yu., P. Ch. Christova and O. M. Fedorenko, 1982	
Nucl. Phys. B197, 1.	1.3, 6.2, 7.1
Becchi, C., A. Rouet and B. Stora, 1975	
Comm. Math. Phys. 42, 127.	1.3, 2.2
Becchi, C., A. Rouet and B. Stora, 1976	
Ann. of Phys. 98, 287.	1.3, 2.2
Bellucci, S., M. Lusignoli and L. Maiani, 1981	
Nucl. Phys. B189, 329.	7.2
Berends, F. A., K. J. F. Gaemers and R. Gastmans, 1973a, b	
Nucl. Phys. B57 , 381; B63 , 381.	5.3
Berends, F. A., R. Kleiss and S. Jadach, 1982	
Nucl. Phys. B202 , 63.	5.3
Berman, S. M., 1958	
rnys. Kev. 112, 267.	6.1

	221

Readinistic Quantum Plaus (DicGraw-Hil).1.4Bjorken, J. D. and S. L. Glashow, 19641.2Phys. Letters 11, 255.1.2Bogoliubov, N. N. and O. S. Parasiuk, 19572.5Acta. Math. 97, 117.2.5Buras, A. J. 1981Proceedings of the 1981 International Symposium on Lepton and Photon Interactions at High Energies, Universität Bonn 1981, ed. by W. Pfeil (Physikalishe Institut Universität Bonn), 636.1.1Byers, N., R. Rückl and A. Yano, 19795.3, 6.2Cabibbo, N. 19635.3, 6.2Cabibbo, N. 19636.3, 1.2Phys. Rev. Letters 10, 537.1.2Celmaster, W. and R. J. Gonsalves, 19791.2Phys. Rev. Job (201.2.5Consoli, M., 19791.1Nucl. Phys. Bil0, 208.5.1, 7.1Davier, M., 1979Nucl. Phys. Bil0, 208.Proceedings of the 21st International Conference on High Energy Physics, ParisJuly 1982, ed. by P. Petiau and M. Porneuf, J. de Phys. 43, Colloque C3-471.Phys. Rev. D23, 2666.7.2de Witt, B. S., 19671.3, 2.2Phys. Rev. 102, 1195, 1239.1.3, 2.2Derman, E. and D. R. T. Jones, 19775.1Prix Letters 70B, 449.5.1Derman, E., 19791.3, 2.2Phys. Rev. 109, 193.5.1Faddeev, L. D. and A. A. Slavnov, 19805.1Gauge Fi	Bjorken, J. D. and S. D. Drell, 1965		
Pijorient, J. D. and S. L. Organization, 1964 1.2 Bogolubov, N. N. and O. S. Parasiuk, 1957 2.5 Acta. Math. 97, 117. 2.5 Buras, A. J., 1981 Proceedings of the 1981 International Symposium on Lepton and Photon Interac- tions at High Energies, Universität Bonn 1981, ed. by W. Pfeil (Physikalishe Institut Universität Bonn), 636. 1.1 Byers, N., R. Rucki and A. Yano, 1979 Physice 96A, 163. 5.3, 6.2 Cabibbo, N., 1963 5.3, 6.2 2.2 Phys. Rev. Letters 10, 537. 1.2 Celmaster, W. and R. J. Gonsalves, 1979 1.2 Phys. Rev. Letters 42, 1435. 3.1 Collins, J. C. and A. J. Macfarlane, 1974 2.5 Proceedings of the LEP Summer Study (CERN 79-01), Vol 1, 61. 6.1 Davier, M., 1979 Proceedings of the LEP Summer Study (CERN 79-01), Vol 1, 61. 6.1 Davier, M., 1982 Proceedings of the LEP Summer Study (CERN 79-01), Vol 1, 61. 6.1 Davier, M., 1982 Proceedings of the LEP Summer Study (CERN 79-01), Vol 1, 61. 6.1 Davier, M., 1982 Proceedings of the LEP Summer Study (CERN 79-01), Vol 1, 61. 6.1 Davier, M., 1982 Proceedings of the S1st International Conference on High Energy Physics, Paris July 1982, ed. by P. Petiau and M. Porneuf, J. de Phys. 43, Colloque C3-471. <	Relativistic Quantum Fields (McGraw-Filli).		1.4
1195. Letters 70B, 429. 1.2 Bogoliubov, N. N. and O. S. Parasiuk, 1957 Acta. Math. 97, 117. 2.5 Buras, A. J., 1981 Proceedings of the 1981 International Symposium on Lepton and Photon Interactions at High Emergies, Universität Bonn 1981, ed. by W. Pfeil (Physikalishe Institut Universität Bonn), 636. 1.1 Byers, N., R. Rückl and A. Yano, 1979 5.3, 6.2 Physica 96A, 163. 5.3, 6.2 Cabibbo, N., 1963 5.3, 6.2 Phys. Rev. Letters 10, 537. 1.2 Celmaster, W. and R. J. Gonsalves, 1979 Phys. Rev. Letters 21, 435. Collins, J. C. and A. J. Macfarlane, 1974 2.5 Consoli, M., 1979 2.5 Proceedings of the LEP Summer Study (CERN 79-01), Vol 1, 61. 6.1 Davier, M., 1982 Proceedings of the 21st International Conference on High Emergy Physics, Paris July 1982, ed. by P. Petiau and M. Porneuf, J. de Phys. 43, Colloque C3:471. 1.2 Dawoon, S., J. S. Hagelin and L. Hall, 1981 Phys. Rev. D23, 2666. 7.2 de Witt, B. S., 1967 1.3, 2.2 Phys. Rev. D19, 319. 5.1 Prosecters 20B, 29. 1.3, 2.2 Faddeev, L. D. and V. N. Popov, 1967 2.4 Phys. Rev. D19, 319. 5.1 Feinberg, G. and S. Weinberg, 1961 <td>Djorken, J. D. and S. L. Glasnow, 1964</td> <td></td> <td>1 0</td>	Djorken, J. D. and S. L. Glasnow, 1964		1 0
Day Muttow, N. N. and G. S. I attachuk, 1907 2.5 Buras, A. J., 1981 2.5 Proceedings of the 1981 International Symposium on Lepton and Photon Interactions at High Emergies, Universität Bonn, 636. 1.1 Byers, N., R. Rückl and A. Yano, 1979 1.1 Physica 96A, 163. 5.3, 6.2 Cabibbo, N, 1963 5.3 Phys. Rev. Letters 10, 537. 1.2 Celmaster, W. and R. J. Gonsalves, 1979 3.1 Phys. Rev. Letters 42, 1435. 3.1 Collins, J. C. and A. J. Macfarlane, 1974 2.5 Phys. Rev. D10, 1201. 2.5 Consoli, M., 1979 2.5 Proceedings of the LEP Summer Study (CERN 79-01), Vol 1, 61. 6.1 Davier, M., 1982 Proceedings of the LEP Summer Study (CERN 79-01), Vol 1, 61. 6.1 Davier, M., 1982 Proceedings of the LEP Summer Study (CERN 79-01), Vol 1, 61. 6.1 Davier, M., 1982 Petiau and M. Porneuf, J. de Phys. 43, Colloque C3-471. 1.2 Derman, E., 1979 1.3 2.2 Phys. Rev. D23, 2666. 7.2 6 Witt, B. S., 1967 1.3 2.2 Priman, E., 1979 1.3 2.2 Phys. Letters 70B, 429.	Bogoliubov N N and O S Paragink 1057		1.4
2.3. Buras, A. J. 1981 Proceedings of the 1981 International Symposium on Lepton and Photon Interac- tions at High Energies, Universität Bonn 1981, ed. by W. Pfeil (Physikalishe Institut Universität Bonn), 636. Byers, N., R. Rückl and A. Yano, 1979 Physice 96A, 163. Cabibbo, N., 1963 Phys. Rev. Letters 10, 537. Celmaster, W. and R. J. Gonsalves, 1979 Phys. Rev. Letters 42, 1435. Collins, J. C. and A. J. Macfarlane, 1974 Phys. Rev. D10, 100. Davier, M., 1979 Nucl. Phys. B160, 208. Davier, M., 1979 Proceedings of the LEP Summer Study (CERN 79-01), Vol 1, 61. Davier, M., 1979 Proceedings of the 21st International Conference on High Energy Physics, Paris July 1982, ed. by P. Petiau and M. Porneuf, J. de Phys. 43, Colloque C3-471. Dawson, S., J. S. Hagelin and L. Hall, 1981 Phys. Rev. D23, 2666. Question, E., 1977 Phys. Rev. D19, 319. Phys. Rev. D19, 319. Phys. Rev. D19, 319. Phys. Rev. D19, 319. Faddeev, L. D. and V. N. Popov, 1967 Phys. Rev. D19, 319. Faddeev, L. D. and V. N. Popov, 1967 Phys. Rev. Letters 26B, 29. Phys. Rev. Tetters 6, 581.	$\Delta ata Math 97 117$		25
Datase Proceedings of the 1981 International Symposium on Lepton and Photon Interac- tions at High Energies, Universität Bonn 1981, ed. by W. Pfeil (Physikalishe Institut Universität Bonn), 636. 1.1 Byers, N., R. Rickl and A. Yano, 1979 Physica 96A, 163. 5.3, 6.2 Cabibbo, N., 1963 5.3, 6.2 Phys. Rev. Letters 10, 537. 1.2 Celmaster, W. and R. J. Gonsalves, 1979 1.2 Phys. Rev. Letters 42, 1435. 3.1 Collins, J. C. and A. J. Macfarlane, 1974 2.5 Phys. Rev. D10, 1201. 2.5 Consoli, M., 1979 2.1 Proceedings of the LEP Summer Study (CERN 79-01), Vol. 1, 61. 6.1 Davier, M., 1982 Proceedings of the 21st International Conference on High Energy Physics, Paris July 1982, ed. by P. Petiau and M. Porneuf, J. de Phys. 43, Colloque C3-471. 1.2 Dawson, S., J. S. Hagelin and L. Hall, 1981 Phys. Rev. D23, 2666. 7.2 de Witt, B. S., 1967 1.3, 2.2 2.1 Derman, E. and D. R. T. Jones, 1977 1.3, 2.2 Phys. Rev. D19, 319. 5.1 7.1 Faddeev, L. D. and V. N. Popov, 1967 7.1 7.1 Phys. Letters 263, 29. 1.3, 2.2 7 Faddeev, L. D. and A. A. Slavnov, 1980 <td< td=""><td><math display="block">\mathbf{R}_{\text{Ham}} = \mathbf{A} \mathbf{I} 1021</math></td><td></td><td>2.0</td></td<>	$\mathbf{R}_{\text{Ham}} = \mathbf{A} \mathbf{I} 1021$		2.0
 International Difference on Linear Symposium on Lippon and Parlow and Physical Berlin Physical Benergies, Universität Bonn 1981, ed. by W. Pfeil (Physikalishe Institut Universität Bonn), 636. Byers, N., R. Ruckl and A. Yano, 1979 Physica 66A, 163. Cabibbo, N., 1963 Phys. Rev. Letters 10, 537. Celmaster, W. and R. J. Gonsalves, 1979 Phys. Rev. Letters 42, 1435. Collins, J. C. and A. J. Macfarlane, 1974 Phys. Rev. D10, 1201. Consoli, M., 1979 Nucl. Phys. Bio, 208. Davier, M., 1982 Proceedings of the LEP Summer Study (CERN 79-01), Vol. 1, 61. Davier, M., 1982 Proceedings of the 21st International Conference on High Energy Physics, Paris July 1982, ed. by P. Petiau and M. Porneuf, J. de Phys. 43, Colloque C3-471. Davson, S., J. S. Hagelin and L. Hall, 1981 Phys. Rev. D23, 2666. Witt, B. S. 1967 Phys. Rev. 105, 1195, 1239. Derman, E. and D. R. T. Jones, 1977 Phys. Rev. 109, 319. Phys. Rev. D19, 319. Pradeev, L. D. and A. A. Slavnov, 1980 Gauge Fields; Introduction to Quantum Theory (Benjamin, Massachusetts and London). Ferniker, 1934 Z. Phys. Rev. 74, 1430. Feynman, R. P., 1948 Phys. Rev. 74, 1430. Feynman, R. P., 1949 Phys. Rev. 74, 1430. Feynman, R. P., 1963 Acta Phys. Pol, 20, 206. Phys. Rev. 74, 1430. Feynman, R. P., 1963 Acta Phys. Pol, 20, 2001. Fritsch, H., M. Gell-Mann and H. Leutwyler, 1973 	Duras, A. J., 1501 Proceedings of the 1981 International Symposium on Lepton and Photon Interac	•_	
Institut Universität Bonn, 636. 1.1 Byers, N., R. Rukkl and A. Yano, 1979 5.3, 6.2 Physica 96A, 163. 5.3, 6.2 Cabibbo, N. 1963 5.3, 6.2 Phys. Rev. Letters 10, 537. 1.2 Celmaster, W. and R. J. Gonsalves, 1979 1.2 Phys. Rev. Letters 42, 1435. 3.1 Collins, J. C. and A. J. Macfarlane, 1974 2.5 Phys. Rev. D10, 1201. 2.5 Consoli, M., 1979 5.1, 7.1 Davier, M., 1979 5.1, 7.1 Davier, M., 1979 5.1, 7.1 Davier, M., 1982 5.1, 7.1 Proceedings of the 21st International Conference on High Energy Physics, Paris 1.1 July 1982, ed. by P. Petiau and M. Porneuf, J. de Phys. 43, Colloque C3-471. 1.2 Dawson, S., J. S. Hagelin and L. Hall, 1981 7.2 64 Phys. Rev. D23, 2666. 7.2 64 Ge Witt, B. S., 1967 1.3, 2.2 2.5 Proteedings of the 21st International Conference on High Energy Physics, Paris 1.3 July 1982, ed. by P. Petiau and M. Porneuf, J. de Phys. 43, Colloque C3-471. 1.2 Dawson, S., J. S. Hagelin and L. Hall, 1981 7.2 64 Phys. Rev.	tions at High Energies Universität Bonn 1981 ed by W Pfeil (Physikalish	 _	
Instruct Original Description 11 Byers, N., R. Rickl and D. Yano, 1979 Physica 96A, 163. Physica 96A, 163. 5.3, 6.2 Cabibbo, N., 1963 1.2 Phys. Rev. Letters 10, 537. 1.2 Celmaster, W. and R. J. Gonsalves, 1979 1.2 Phys. Rev. Letters 42, 1435. 3.1 Collins, J. C. and A. J. Macfarlane, 1974 2.5 Phys. Rev. D10, 1201. 2.5 Consoli, M., 1979 Nucl. Phys. B160, 208. 5.1, 7.1 Davier, M., 1982 5.1, 7.1 Proceedings of the LEP Summer Study (CERN 79-01), Vol. 1, 61. 6.1 Davier, M., 1982 1.1 1.2 Proceedings of the 21st International Conference on High Energy Physics, Paris July 1982, ed. by P. Petiau and M. Porneuf, J. de Phys. 43, Colloque C3-471. 1.2 Dawson, S., J. S. Hagelin and L. Hall, 1981 Phys. Rev. D19, 2666. 7.2 de Witt, B. S., 1967 1.3, 2.2 1.3, 2.2 Derman, E. and D. R. T. Jones, 1977 1.3, 2.2 1.3, 2.2 Phys. Letters 70B, 449. 5.1 1.3 Derman, E., 1979 Phys. Rev. D19, 319. 5.1 Faddeev, L. D. and V. N. Popov, 1967 Phys. 2.2	Institut Universität Bonn) 636	C	1 1
5)CUS NO, N. 1963 5.3, 6.2 Cabibbo, N., 1963 5.3, 6.2 Phys. Rev. Letters 10, 537. 1.2 Celmaster, W. and R. J. Gonsalves, 1979 1.2 Phys. Rev. Letters 42, 1435. 3.1 Collins, J. C. and A. J. Macfarlane, 1974 1.9 Phys. Rev. D10, 1201. 2.5 Consoli, M., 1979 Nucl. Phys. B160, 208. 5.1, 7.1 Davier, M., 1979 State S	Byers N R Rückl and A Vano 1970		1.1
Cabibbo, N. 1963 0.19 Cabibbo, N. 1963 Phys. Rev. Letters 10, 537. 1.2 Celmaster, W. and R. J. Gonsalves, 1979 Phys. Rev. Letters 42, 1435. 3.1 Collins, J. C. and A. J. Macfarlane, 1974 Phys. Rev. D10, 1201. 2.5 Consoli, M., 1979 2.5 2.5 Nucl. Phys. B160, 208. 5.1, 7.1 Davier, M., 1979 7 2.5 Proceedings of the LEP Summer Study (CERN 79-01), Vol. 1, 61. 6.1 Davier, M., 1982 Proceedings of the 21st International Conference on High Energy Physics, Paris July 1982, ed. by P. Petiau and M. Porneuf, J. de Phys. 43, Colloque C3-471. 1.2 Dawson, S., J. S. Hagelin and L. Hall, 1981 7.2 Phys. Rev. D23, 2666. 7.2 de Witt, B. S., 1967 1.3, 2.2 Perman, E. and D. R. T. Jones, 1977 1.3, 2.2 Phys. Rev. D19, 319. 5.1 Faddeev, L. D. and V. N. Popov, 1967 1.3, 2.2 Pathere, G. and S. Weinberg, 1961 2. Phys. Rev. Letters 6, 381. 5.1 Ferinberg, G. and S. Weinberg, 1961 1.1 Phys. Rev. 76, 769. 1.1 Feynman, R. P., 1948 1.1 <t< td=""><td>$\frac{1}{2}$</td><td>3</td><td>62</td></t<>	$\frac{1}{2}$	3	62
Color M, W. 2007 1.2 Phys. Rev. Letters 10, 537. 1.2 Celmaster, W. and R. J. Gonsalves, 1979 1.2 Phys. Rev. Letters 42, 1435. 3.1 Collins, J. C. and A. J. Macfarlane, 1974 2.5 Phys. Rev. D10, 1201. 2.5 Consoli, M., 1979 5.1, 7.1 Davier, M., 1979 5.1, 7.1 Davier, M., 1979 5.1, 7.1 Proceedings of the LEP Summer Study (CERN 79-01), Vol. 1, 61. 6.1 Davier, M., 1982 Proceedings of the 21st International Conference on High Energy Physics, Paris July 1982, ed. by P. Petiau and M. Porneuf, J. de Phys. 43, Colloque C3-471. 1.2 Dawson, S., J. S. Hagelin and L. Hall, 1981 Phys. Rev. D23, 2666. 7.2 de Witt, B. S., 1967 1.3, 2.2 Phys. Rev. D23, 2666. 7.2 de Witt, B. S., 1967 1.3, 2.2 Perman, E. and D. R. T. Jones, 1977 1.3, 2.2 Perman, E. and D. R. T. Jones, 1977 Phys. Letters 70B, 449. 5.1 Derman, E. 1979 Phys. Letters 2B, 29. 1.3, 2.2 Faddeev, L. D. and A. A. Slavnov, 1980 Gauge Fields; Introduction to Quantum Theory (Benjamin, Massachusetts and London). 2.5 Feinberg, G. and S. Weinberg, 1961 Phys. Rev. 76, 769.	Cabibbo N 1963	• 0,	0.2
Celmaster, W. and R. J. Gonsalves, 1979 Phys. Rev. Letters 42, 1435. 3.1 Collins, J. C. and A. J. Macfarlane, 1974 Phys. Rev. Letters 42, 1435. 3.1 Collins, J. C. and A. J. Macfarlane, 1974 Phys. Rev. D10, 1201. 2.5 Consoli, M., 1979 Nucl. Phys. B160, 208. 5.1, 7.1 Davier, M., 1979 Proceedings of the LEP Summer Study (CERN 79-01), Vol. 1, 61. 6.1 Davier, M., 1982 Proceedings of the 21st International Conference on High Energy Physics, Paris July 1982, ed. by P. Petiau and M. Porneuf, J. de Phys. 43, Colloque C3-471. 1.2 Dawson, S., J. S. Hagelin and L. Hall, 1981 Phys. Rev. D23, 2666. 7.2 de Witt, B. S., 1967 1.3, 2.2 Phys. Rev. D23, 2666. 7.2 de Witt, B. S., 1967 1.3, 2.2 Phys. Rev. 162, 1195, 1239. 1.3, 2.2 Derman, E. and D. R. T. Jones, 1977 Phys. Letters 70B, 449. Phys. Letters 25B, 29. 1.3, 2.2 Faddeev, L. D. and A. A. Slavnov, 1980 Gauge Fields; Introduction to Quantum Theory (Benjamin, Massachusetts and London). Z. Phys. 88, 161. 1.1 Feynman, R. P., 1948 1.1 Phys. Rev. 74, 1430. 1.1 Feynman, R. P., 1948 1.1	Phys. Rev. Letters 10, 537.		1.2
Phys. Rev. Letters 42, 1435. 3.1 Collins, J. C. and A. J. Macfarlane, 1974 9 Phys. Rev. D10, 1201. 2.5 Consoli, M., 1979 2.5 Nucl. Phys. B160, 208. 5.1, 7.1 Davier, M., 1979 7 Proceedings of the LEP Summer Study (CERN 79-01), Vol. 1, 61. 6.1 Davier, M., 1982 7 Proceedings of the 21st International Conference on High Energy Physics, Paris 1.4 July 1982, ed. by P. Petiau and M. Porneuf, J. de Phys. 43, Colloque C3-471. 1.2 Dawson, S., J. S. Hagelin and L. Hall, 1981 Phys. Rev. D23, 2666. 7.2 de Witt, B. S., 1967 1.3, 2.2 Derman, E., and D. R. T. Jones, 1977 1.3, 2.2 Phys. Rev. D19, 319. 5.1 Faddeev, L. D. and V. N. Popov, 1967 5.1 Phys. Letters 25B, 29. 1.3, 2.2 Faddeev, L. D. and V. N. Popov, 1967 2.2 Phys. Rev. Letters 6, 381. 5.1 Fermi, E., 1979 2.4 Phys. Rev. Letters 6, 381. 5.1 Fermi, E., 1934 2.2 Feinberg, G. and S. Weinberg, 1961 2.2 Phys. Rev. 76, 769. 1.1 <t< td=""><td>Celmaster, W. and R. I. Gonsalves, 1979</td><td></td><td></td></t<>	Celmaster, W. and R. I. Gonsalves, 1979		
Collins, J. C. and A. J. Macfarlane, 1974 2.5 Consoli, M., 1979 2.5 Nucl. Phys. Bi60, 208. 5.1, 7.1 Davier, M., 1979 7.0 Proceedings of the LEP Summer Study (CERN 79-01), Vol. 1, 61. 6.1 Davier, M., 1982 7.0 Proceedings of the 21st International Conference on High Energy Physics, Paris July 1982, ed. by P. Petiau and M. Porneuf, J. de Phys. 43, Colloque C3-471. Dawson, S., J. S. Hagelin and L. Hall, 1981 Phys. Rev. D23, 2666. 7.2 de Witt, B. S., 1967 Phys. Rev. D23, 2666. 7.2 de Witt, B. S., 1967 1.3, 2.2 Perman, E. and D. R. T. Jones, 1977 1.3, 2.2 Derman, E., 1979 5.1 Phys. Rev. D19, 319. 5.1 Faddeev, L. D. and V. N. Popov, 1967 1.3, 2.2 Faddeev, L. D. and A. A. Slavnov, 1980 2.4 Gauge Fields; Introduction to Quantum Theory (Benjamin, Massachusetts and London). 2.2 Fermiberg, G. and S. Weinberg, 1961 1.1 Phys. Rev. 76, 769. 1.1 Feynman, R. P., 1948 1.1 Phys. Rev. 76, 769. 1.1 Feynman, R. P., 1963 1.1	Phys. Rev. Letters 42, 1435.		3.1
Phys. Rev. D10, 1201. 2.5 Consoli, M., 1979 Nucl. Phys. B160, 208. 5.1, 7.1 Davier, M., 1979 Focceedings of the LEP Summer Study (CERN 79-01), Vol. 1, 61. 6.1 Davier, M., 1982 Proceedings of the 21st International Conference on High Energy Physics, Paris July 1982, ed. by P. Petiau and M. Porneuf, J. de Phys. 43, Colloque C3-471. 1.2 Dawson, S., J. S. Hagelin and L. Hall, 1981 Phys. Rev. D23, 2666. 7.2 de Witt, B. S., 1967 Phys. Rev. 162, 1195, 1239. 1.3, 2.2 Derman, E. and D. R. T. Jones, 1977 Phys. Letters 70B, 449. 5.1 Perman, E. and D. R. T. Jones, 1977 Phys. Rev. D19, 319. 5.1 Faddeev, L. D. and V. N. Popov, 1967 Phys. Rev. D19, 319. 5.1 Faddeev, L. D. and V. N. Popov, 1980 Gauge Fields; Introduction to Quantum Theory (Benjamin, Massachusetts and London). 2.2 Fernberg, G. and S. Weinberg, 1961 Phys. Rev. Letters 6, 381. 5.1 Fermine, R. P., 1948 1.1 1.1 Phys. Rev. 76, 769. 1.1 1.1 Feynman, R. P., 1963 1.1 1.1 Feynman, R. P., 1963 1.1 1.1 Feynman, R. P., 1963 1.3 2.2 Phys.	Collins, J. C. and A. I. Macfarlane, 1974		
Consoli, M., 1979 Nucl. Phys. B160, 208. 5. 1, 7. 1 Davier, M., 1979 Proceedings of the LEP Summer Study (CERN 79-01), Vol. 1, 61. 6. 1 Davier, M., 1982 Proceedings of the 21st International Conference on High Energy Physics, Paris July 1982, ed. by P. Petiau and M. Porneuf, J. de Phys. 43, Colloque C3-471. 1.2 Dawson, S., J. S. Hagelin and L. Hall, 1981 Phys. Rev. D23, 2666. 7.2 de Witt, B. S., 1967 1.3, 2.2 Phys. Rev. D23, 2666. 7.2 de Witt, B. S., 1967 1.3, 2.2 Derman, E. and D. R. T. Jones, 1977 Phys. Letters 70B, 449. Derman, E., 1979 5.1 Phys. Letters 70B, 449. 5.1 Paddeev, L. D. and A. A. Slavnov, 1980 Gauge Fields; Introduction to Quantum Theory (Benjamin, Massachusetts and London). Feinberg, G. and S. Weinberg, 1961 Phys. Rev. Letters 6, 381. Fermi, E., 1934 5.1 Feynman, R. P., 1948 1.1 Phys. Rev. 76, 769. 1.1 Feynman, R. P., 1949 1.1 Phys. Rev. 76, 769. 1.3 Feynman, R. P., 1963 1.3 Acta Phys. Polon. 26, 697. 1.3, 2.2 Fleischer, J. and F. Jergerlehner, 1981 7.1	Phys. Rev. D10, 1201.		2.5
Nucl. Phys. B160, 208. 5.1, 7.1 Davier, M., 1979 Proceedings of the LEP Summer Study (CERN 79-01), Vol. 1, 61. 6.1 Davier, M., 1982 Proceedings of the 21st International Conference on High Energy Physics, Paris July 1982, ed. by P. Petiau and M. Porneuf, J. de Phys. 43, Colloque C3-471. 1.2 Dawson, S., J. S. Hagelin and L. Hall, 1981 Phys. Rev. D23, 2666. 7.2 de Witt, B. S., 1967 1.3, 2.2 Phys. Rev. D23, 2666. 7.2 de Witt, B. S., 1967 1.3, 2.2 Derman, E. and D. R. T. Jones, 1977 Phys. Rev. D23, 2666. Phys. Rev. D19, 319. 5.1 Faddeev, L. D. and V. N. Popov, 1967 Phys. Rev. D19, 319. Faddeev, L. D. and A. A. Slavnov, 1980 Gauge Fields; Introduction to Quantum Theory (Benjamin, Massachusetts and London). Feinberg, G. and S. Weinberg, 1961 Phys. Rev. Letters 6, 381. Fermi, E., 1934 5.1 Z. Phys. 88, 161. 1.1 Feynman, R. P., 1948 Phys. Rev. 76, 769. Phys. Rev. 76, 769. 1.3 Feynman, R. P., 1963 1.3 Phys. Rev. 76, 769. 1.3 Feynman, R. P., 1963 1.3 Phys. Rev. 76, 697. 1.3 Fleischer, J	Consoli, M., 1979		2.0
Nature 1979 Proceedings of the LEP Summer Study (CERN 79-01), Vol 1, 61. 6.1 Davier, M., 1982 Proceedings of the 21st International Conference on High Energy Physics, Paris July 1982, ed. by P. Petiau and M. Porneuf, J. de Phys. 43, Colloque C3-471. 1.2 Dawson, S., J. S. Hagelin and L. Hall, 1981 Phys. Rev. D23, 2666. 7.2 de Witt, B. S., 1967 1.3, 2.2 Perman, E. and D. R. T. Jones, 1977 1.3, 2.2 Derman, E., 1979 5.1 Phys. Rev. D19, 319. 5.1 Faddeev, L. D. and V. N. Popov, 1967 5.1 Phys. Rev. D19, 319. 5.1 Faddeev, L. D. and A. A. Slavnov, 1980 Gauge Fields; Introduction to Quantum Theory (Benjamin, Massachusetts and London). Perimiter, E., 1934 5.1 Phys. Rev. Letters 6, 381. 5.1 Feynman, R. P., 1948 5.1 Phys. Rev. 74, 1430. 1.1 Feynman, R. P., 1948 1.1 Phys. Rev. 76, 769. 1.1 Feynman, R. P., 1963 1.3 Acta Phys. Rev. 76, 697. 1.3, 2.2 Fleischer, J. and F. Jergerlehner, 1981 7.3 Phys. Rev. 109, 193. 1.1 Feynman, R. P., 1963 1.3 <	Nucl Phys B160 208 5	. 1.	7.1
Derived first of the LEP Summer Study (CERN 79-01), Vol. 1, 61. 6.1 Davier, M., 1982 Proceedings of the 21st International Conference on High Energy Physics, Paris July 1982, ed. by P. Petiau and M. Porneuf, J. de Phys. 43, Colloque C3-471. 1.2 Dawson, S., J. S. Hagelin and L. Hall, 1981 Phys. Rev. D23, 2666. 7.2 de Witt, B. S., 1967 Phys. Rev. D23, 2666. 7.2 de Witt, B. S., 1967 1.3, 2.2 Derman, E. and D. R. T. Jones, 1977 Phys. Letters 70B, 449. 5.1 Derman, E., 1979 5.1 Phys. Rev. D19, 319. 5.1 Faddeev, L. D. and V. N. Popov, 1967 1.3, 2.2 Phys. Letters 25B, 29. 1.3, 2.2 Faddeev, L. D. and V. N. Popov, 1967 1.3, 2.2 Feinberg, G. and S. Weinberg, 1961 2.2 Fermi, E., 1934 2.2 Z. Phys. 88, 161. 1.1 Feynman, R. P., 1948 1.1 Feynman, R. P., 1948 1.1 Feynman, R. P., 1948 1.1 Feynman, R. P., 1943 1.1 Feynman, R. P., 1943 1.1 Feynman, R. P., 1963 1.1 Acta Phys. Rev. 109, 193. 1.1 Feynm	Davier M. 1979	• _,	
Davier, M., 1982 Proceedings of the 21st International Conference on High Energy Physics, Paris July 1982, ed. by P. Petiau and M. Porneuf, J. de Phys. 43, Colloque C3-471. 1.2 Dawson, S., J. S. Hagelin and L. Hall, 1981 Phys. Rev. D23, 2666. 7.2 de Witt, B. S., 1967 1.3, 2.2 Phys. Rev. D23, 2666. 7.2 de Witt, B. S., 1967 1.3, 2.2 Derman, E. and D. R. T. Jones, 1977 Phys. Letters 70B, 449. Derman, E., 1979 5.1 Phys. Letters 70B, 449. 5.1 Paddeev, L. D. and D. N. Popov, 1967 1.3, 2.2 Faddeev, L. D. and A. A. Slavnov, 1980 Gauge Fields; Introduction to Quantum Theory (Benjamin, Massachusetts and London). Feinberg, G. and S. Weinberg, 1961 Phys. Rev. Letters 6, 381. Fermi, E., 1934 5.1 Z. Phys. 88, 161. 1.1 Feynman, R. P., 1948 1.1 Phys. Rev. 74, 1430. 1.1 Feynman, R. P., 1949 1.1 Phys. Rev. 109, 193. 1.1 Feynman, R. P., 1963 1.3, 2.2 Acta Phys. Rev. 109, 193. 1.3 Feynman, R. P., 1963 1.3, 2.2 Fleischer, J. and F. Jergerlehner, 1981 7.1	Proceedings of the LEP Summer Study (CERN 79-01). Vol. 1, 61.		6.1
Proceedings of the 21st International Conference on High Energy Physics, Paris July 1982, ed. by P. Petiau and M. Porneuf, J. de Phys. 43, Colloque C3-471. Dawson, S., J. S. Hagelin and L. Hall, 1981 Phys. Rev. D23, 2666. (e Witt, B. S., 1967 Phys. Rev. 162, 1195, 1239. Derman, E. and D. R. T. Jones, 1977 Phys. Rev. 162, 1195, 1239. Derman, E., 1979 Phys. Rev. 162, 1195, 1239. Derman, E., 1979 Phys. Rev. D19, 319. Faddeev, L. D. and V. N. Popov, 1967 Phys. Letters 25B, 29. Faddeev, L. D. and A. A. Slavnov, 1980 Gauge Fields; Introduction to Quantum Theory (Benjamin, Massachusetts and London). Permis, E., 1934 Z. Phys. 88, 161. Ferniberg, G. and S. Weinberg, 1961 Phys. Rev. 74, 1430. Feynman, R. P., 1948 Phys. Rev. 76, 769. Feynman, R. P., 1948 Phys. Rev. 76, 769. Feynman, R. P., 1963 Acta Phys. Polon. 26, 697. Fleischer, J. and F. Jergerlehner, 1981 Phys. Rev. D23, 2001. Fleischer, H., M. Gell-Mann and H. Leutwyler, 1973	Davier. M., 1982		0.1
July 1982, ed. by P. Petiau and M. Porneuf, J. de Phys. 43, Colloque C3-471. 1.2 Dawson, S., J. S. Hagelin and L. Hall, 1981 Phys. Rev. D23, 2666. 7.2 de Witt, B. S., 1967 Phys. Rev. 162, 1195, 1239. 1.3, 2.2 Derman, E. and D. R. T. Jones, 1977 Phys. Rev. 162, 1195, 1239. 5.1 Derman, E. and D. R. T. Jones, 1977 Phys. Letters 70B, 449. 5.1 Derman, E., 1979 Phys. Rev. D19, 319. 5.1 Faddeev, L. D. and V. N. Popov, 1967 Phys. Letters 25B, 29. 1.3, 2.2 Faddeev, L. D. and A. A. Slavnov, 1980 Gauge Fields; Introduction to Quantum Theory (Benjamin, Massachusetts and London). 2.2 Feinberg, G. and S. Weinberg, 1961 Phys. Rev. 1934 5.1 Premi, E., 1934 5.1 Z. Phys. 88, 161. 1.1 Feynman, R. P., 1948 1.1 Phys. Rev. 74, 1430. 1.1 Feynman, R. P., 1948 1.1 Feynman, R. P., 1948 1.1 Feynman, R. P., 1963 1.1 Feynman, R. P., 1963 1.1 Feynman, R. P., 1963 1.3, 2.2 Fleischer, J. and F. Jergerlehner, 1981 7.1 Phys. Rev. D2, 2001. 7.1	Proceedings of the 21st International Conference on High Energy Physics, Part	is	
Dawson, S., J. S. Hagelin and L. Hall, 1981 Phys. Rev. D23, 2666. 7.2 de Witt, B. S., 1967 Phys. Rev. 162, 1195, 1239. 1.3, 2.2 Derman, E. and D. R. T. Jones, 1977 Phys. Letters 70B, 449. 5.1 Derman, E., 1979 5.1 Phys. Rev. D19, 319. 5.1 Faddeev, L. D. and V. N. Popov, 1967 Phys. Letters 25B, 29. 1.3, 2.2 Faddeev, L. D. and A. A. Slavnov, 1980 Gauge Fields; Introduction to Quantum Theory (Benjamin, Massachusetts and London). 2.2 Feinberg, G. and S. Weinberg, 1961 Phys. Rev. Letters 6, 381. 5.1 Fermi, E., 1934 2. Phys. Rev. 74, 1430. 1.1 Feynman, R. P., 1948 Phys. Rev. 76, 769. 1.1 Feynman, R. P., 1949 Phys. Rev. 109, 193. 1.1 Feynman, R. P., 1943 1.1 Feynman, R. P. and M. Gell-Mann, 1958 1.1 Feynman, R. P. 1963 1.3 Acta Phys. Polon. 26, 697. 1.3, 2.2 Fleischer, J. and F. Jergerlehner, 1981 1.3, 2.2 Phys. Rev. D23, 2001. 7.1 Fritzsch, H., M. Gell-Mann and H. Leutwyler, 1973 7.1	July 1982, ed. by P. Petiau and M. Porneuf, I. de Phys. 43. Colloque C3-471.	-	1.2
Phys. Rev. D23, 2666. 7.2 de Witt, B. S., 1967 1.3, 2.2 Derman, E. and D. R. T. Jones, 1977 1.3, 2.2 Derman, E. and D. R. T. Jones, 1977 5.1 Phys. Letters 70B , 449. 5.1 Derman, E., 1979 5.1 Phys. Rev. D19 , 319. 5.1 Faddeev, L. D. and V. N. Popov, 1967 1.3, 2.2 Paddeev, L. D. and A. A. Slavnov, 1980 6auge Fields; Introduction to Quantum Theory (Benjamin, Massachusetts and London). 2.2 Feinberg, G. and S. Weinberg, 1961 2.2 Fermi, E., 1934 5.1 Z. Phys. Rev. Letters 6 , 381. 5.1 Fermi, E., 1934 1.1 Feynman, R. P., 1948 1.1 Phys. Rev. 76 , 769. 1.3 Feynman, R. P., 1948 1.1 Feynman, R. P., 1949 1.1 Feynman, R. P., 1949 1.1 Feynman, R. P. and M. Gell-Mann, 1958 1.1 Feynman, R. P. 1963 1.3 Acta Phys. Polon. 26 , 697. 1.3, 2.5 Fleischer, J. and F. Jergerlehner, 1981 7.1 Phys. Rev. D23 , 2001. 7.1 Fritzsch, H., M. Gell-Mann and H. Leutwyler, 1973	Dawson, S., I. S. Hagelin and L. Hall, 1981		
de Witt, B. S., 1967 Phys. Rev. 162, 1195, 1239. 1.3, 2.2 Derman, E. and D. R. T. Jones, 1977 Phys. Letters 70B, 449. 5.1 Derman, E., 1979 Phys. Letters 70B, 449. 5.1 Derman, E., 1979 Phys. Rev. D19, 319. 5.1 Faddeev, L. D. and V. N. Popov, 1967 1.3, 2.2 Faddeev, L. D. and A. A. Slavnov, 1980 Gauge Fields; Introduction to Quantum Theory (Benjamin, Massachusetts and London). 2.2 Feinberg, G. and S. Weinberg, 1961 Phys. Rev. Letters 6, 381. 5.1 Fermi, E., 1934 1.1 2.2 Z. Phys. 88, 161. 5.1 Feynman, R. P., 1948 1.1 Phys. Rev. 74, 1430. 1.1 Feynman, R. P., 1949 1.1 Phys. Rev. 76, 769. 1.1 Feynman, R. P., 1949 1.1 Feynman, R. P., 1943 1.1 Feynman, R. P., 1943 1.1 Feynman, R. P., 1963 1.3 Acta Phys. Polon. 26, 697. 1.3, 2.2 Fleischer, J. and F. Jergerlehner, 1981 1.3, 2.5 Phys. Rev. D23, 2001. 7.1 Fritzsch, H., M. Gell-Mann and H. Leutwyler, 1973 1.3	Phys. Rev. D23 , 2666.		7.2
Phys. Rev. 162, 1195, 1239. 1.3, 2.2 Derman, E. and D. R. T. Jones, 1977 1.3, 2.2 Derman, E., 1979 5.1 Phys. Rev. D19, 319. 5.1 Faddeev, L. D. and V. N. Popov, 1967 1.3, 2.2 Prist, Rev. D19, 319. 5.1 Faddeev, L. D. and V. N. Popov, 1967 1.3, 2.2 Faddeev, L. D. and A. A. Slavnov, 1980 6auge Fields; Introduction to Quantum Theory (Benjamin, Massachusetts and London). 2.2 Feinberg, G. and S. Weinberg, 1961 2.2 Phys. Rev. Letters 6, 381. 5.1 Fermi, E., 1934 5.1 Z. Phys. 88, 161. 1.1 Feynman, R. P., 1948 1.1 Phys. Rev. 74, 1430. 1.1 Feynman, R. P., 1949 1.1 Phys. Rev. 76, 769. 1.1 Feynman, R. P., 1949 1.1 Feynman, R. P. and M. Gell-Mann, 1958 1.1 Feynman, R. P., 1963 1.3, 2.2 Fleischer, J. and F. Jergerlehner, 1981 1.3, 2.2 Fleischer, J. and F. Jergerlehner, 1981 7.1 Phys. Rev. D23, 2001. 7.1 Fritzsch, H., M. Gell-Mann and H. Leutwyler, 1973 7.1	de Witt, B. S., 1967		
Derman, E. and D. R. T. Jones, 1977 Phys. Letters 70B, 449. 5.1 Derman, E., 1979 Phys. Rev. D19, 319. 5.1 Faddeev, L. D. and V. N. Popov, 1967 Phys. Letters 25B, 29. 1.3, 2.2 Faddeev, L. D. and A. A. Slavnov, 1980 Gauge Fields; Introduction to Quantum Theory (Benjamin, Massachusetts and London). 2.2 Feinberg, G. and S. Weinberg, 1961 Phys. Rev. Letters 6, 381. 5.1 Fermi, E., 1934 1.1 Z. Phys. 88, 161. 1.1 Feynman, R. P., 1948 1.1 Phys. Rev. 74, 1430. 1.1 Feynman, R. P., 1949 1.1 Phys. Rev. 76, 769. 1.1 Feynman, R. P., 1949 1.1 Feynman, R. P., 1963 1.3 Acta Phys. Polon. 26, 697. 1.3, 2.2 Fleischer, J. and F. Jergerlehner, 1981 7.1 Phys. Rev. D23, 2001. 7.1 Fritzsch, H., M. Gell-Mann and H. Leutwyler, 1973 7.1	Phys. Rev. 162 , 1195, 1239.	. 3.	2.2
Phys. Letters 70B, 449. 5.1 Derman, E., 1979 Phys. Rev. D19, 319. Faddeev, L. D. and V. N. Popov, 1967 5.1 Faddeev, L. D. and V. N. Popov, 1967 1.3, 2.2 Faddeev, L. D. and A. A. Slavnov, 1980 6auge Fields; Introduction to Quantum Theory (Benjamin, Massachusetts and London). 2.2 Feinberg, G. and S. Weinberg, 1961 2.2 Fermi, E., 1934 5.1 Z. Phys. 88, 161. 5.1 Feynman, R. P., 1948 1.1 Feynman, R. P., 1949 1.1 Feynman, R. P., 1949 1.1 Feynman, R. P., 1949 1.1 Feynman, R. P., 1963 1.3 Acta Phys. Polon. 26, 697. 1.3, 2.2 Fleischer, J. and F. Jergerlehner, 1981 7.1 Phys. Rev. D23, 2001. 7.1 Fritzsch, H., M. Gell-Mann and H. Leutwyler, 1973 7.1	Derman, E. and D. R. T. Jones, 1977	- •	
Derman, E., 1979 Phys. Rev. D19, 319. 5.1 Faddeev, L. D. and V. N. Popov, 1967 Phys. Letters 25B, 29. 1.3, 2.2 Faddeev, L. D. and A. A. Slavnov, 1980 Gauge Fields; Introduction to Quantum Theory (Benjamin, Massachusetts and London). 2.2 Feinberg, G. and S. Weinberg, 1961 Phys. Rev. Letters 6, 381. 5.1 Fermi, E., 1934 5.1 Z. Phys. 88, 161. 1.1 Feynman, R. P., 1948 1.1 Feynman, R. P., 1949 1.1 Feynman, R. P., 1963 1.1 Feynman, R. P., 1963 1.1 Feynman, R. P., 1963 1.3, 2.2 Fleischer, J. and F. Jergerlehner, 1981 7.1 Phys. Rev. D23, 2001. 7.1 Fritzsch, H., M. Gell-Mann and H. Leutwyler, 1973 7.1	Phys. Letters 70B , 449.		5.1
Phys. Rev. D19, 319. 5.1 Faddeev, L. D. and V. N. Popov, 1967 1.3, 2.2 Faddeev, L. D. and A. A. Slavnov, 1980 6auge Fields; Introduction to Quantum Theory (Benjamin, Massachusetts and London). Feinberg, G. and S. Weinberg, 1961 2.2 Fermi, E., 1934 5.1 Z. Phys. 88, 161. 5.1 Feynman, R. P., 1948 1.1 Feynman, R. P., 1948 1.1 Feynman, R. P., 1949 1.1 Feynman, R. P. and M. Gell-Mann, 1958 1.1 Feynman, R. P. 1963 1.1 Feynman, R. P., 1963 1.3 Acta Phys. Polon. 26, 697. 1.3, 2.2 Fleischer, J. and F. Jergerlehner, 1981 7.1 Phys. Rev. D23, 2001. 7.1 Fritzsch, H., M. Gell-Mann and H. Leutwyler, 1973 7.1	Derman, E., 1979		
Faddeev, L. D. and V. N. Popov, 1967 1.3, 2.2 Paddeev, L. D. and A. A. Slavnov, 1980 6auge Fields; Introduction to Quantum Theory (Benjamin, Massachusetts and London). 2.2 Feinberg, G. and S. Weinberg, 1961 Phys. Rev. Letters 6, 381. 5.1 Fermi, E., 1934 5.1 Z. Phys. 88, 161. 1.1 Feynman, R. P., 1948 1.1 Phys. Rev. 74, 1430. 1.1 Feynman, R. P., 1949 1.1 Feynman, R. P. and M. Gell-Mann, 1958 1.1 Feynman, R. P. 1963 1.1 Feynman, R. P., 1963 1.3 Acta Phys. Polon. 26, 697. 1.3, 2.2 Fleischer, J. and F. Jergerlehner, 1981 7.1 Phys. Rev. D23, 2001. 7.1 Fritzsch, H., M. Gell-Mann and H. Leutwyler, 1973 7.1	Phys. Rev. D19, 319.		5.1
Phys. Letters 25B, 29. 1.3, 2.2 Faddeev, L. D. and A. A. Slavnov, 1980 Gauge Fields; Introduction to Quantum Theory (Benjamin, Massachusetts and London). Peinberg, G. and S. Weinberg, 1961 2.2 Fermir, E., 1934 5.1 Z. Phys. Rev. Letters 6, 381. 5.1 Fermi, E., 1934 5.1 Feynman, R. P., 1948 1.1 Phys. Rev. 74, 1430. 1.1 Feynman, R. P., 1949 1.1 Phys. Rev. 76, 769. 1.1 Feynman, R. P. and M. Gell-Mann, 1958 1.1 Phys. Rev. 109, 193. 1.1 Feynman, R. P., 1963 1.3, 2.2 Acta Phys. Polon. 26, 697. 1.3, 2.2 Fleischer, J. and F. Jergerlehner, 1981 7.1 Phys. Rev. D23, 2001. 7.1 Fritzsch, H., M. Gell-Mann and H. Leutwyler, 1973 7.1	Faddeev, L. D. and V. N. Popov, 1967		
 Faddeev, L. D. and A. A. Slavnov, 1980 Gauge Fields; Introduction to Quantum Theory (Benjamin, Massachusetts and London). Feinberg, G. and S. Weinberg, 1961 Phys. Rev. Letters 6, 381. Fermi, E., 1934 Z. Phys. 88, 161. Feynman, R. P., 1948 Phys. Rev. 74, 1430. Feynman, R. P., 1949 Phys. Rev. 76, 769. Feynman, R. P. and M. Gell-Mann, 1958 Phys. Rev. 109, 193. Feynman, R. P., 1963 Acta Phys. Polon. 26, 697. Fleischer, J. and F. Jergerlehner, 1981 Phys. Rev. D23, 2001. Fritzsch, H., M. Gell-Mann and H. Leutwyler, 1973 	Phys. Letters 25B . 29.	. 3.	2.2
Gauge Fields; Introduction to Quantum Theory (Benjamin, Massachusetts and London). 2.2 Feinberg, G. and S. Weinberg, 1961 2.1 Phys. Rev. Letters 6, 381. 5.1 Fermi, E., 1934 5.1 Z. Phys. 88, 161. 1.1 Feynman, R. P., 1948 1.1 Feynman, R. P., 1949 1.1 Feynman, R. P. and M. Gell-Mann, 1958 1.1 Feynman, R. P., 1963 1.3 Acta Phys. Polon. 26, 697. 1.3, 2.2 Fleischer, J. and F. Jergerlehner, 1981 1.3 Phys. Rev. D23, 2001. 7.1 Fritzsch, H., M. Gell-Mann and H. Leutwyler, 1973 7.1	Faddeev, L. D. and A. A. Slavnov, 1980		
London). 2.2 Feinberg, G. and S. Weinberg, 1961 94 Phys. Rev. Letters 6, 381. 5.1 Fermi, E., 1934 1.1 Z. Phys. 88, 161. 1.1 Feynman, R. P., 1948 1.1 Feynman, R. P., 1949 1.1 Feynman, R. P. and M. Gell-Mann, 1958 1.1 Feynman, R. P. and M. Gell-Mann, 1958 1.1 Feynman, R. P., 1963 1.3 Acta Phys. Polon. 26, 697. 1.3, 2.2 Fleischer, J. and F. Jergerlehner, 1981 1.3, 2.2 Phys. Rev. D23, 2001. 7.1 Fritzsch, H., M. Gell-Mann and H. Leutwyler, 1973 7.1	Gauge Fields; Introduction to Quantum Theory (Benjamin, Massachusetts an	d	
Feinberg, G. and S. Weinberg, 1961 5.1 Phys. Rev. Letters 6, 381. 5.1 Fermi, E., 1934 1.1 Z. Phys. 88, 161. 1.1 Feynman, R. P., 1948 1.1 Feynman, R. P., 1949 1.1 Feynman, R. P. and M. Gell-Mann, 1958 1.1 Feynman, R. P. and M. Gell-Mann, 1958 1.1 Feynman, R. P., 1963 1.3 Acta Phys. Polon. 26, 697. 1.3, 2.2 Fleischer, J. and F. Jergerlehner, 1981 1.3, 2.2 Phys. Rev. D23, 2001. 7.1 Fritzsch, H., M. Gell-Mann and H. Leutwyler, 1973 7.1	London).		2.2
Phys. Rev. Letters 6, 381. 5.1 Fermi, E., 1934 1.1 Z. Phys. 88, 161. 1.1 Feynman, R. P., 1948 1.1 Feynman, R. P., 1949 1.1 Feynman, R. P. and M. Gell-Mann, 1958 1.1 Feynman, R. P. and M. Gell-Mann, 1958 1.1 Feynman, R. P., 1963 1.3 Acta Phys. Polon. 26, 697. 1.3, 2.2 Fleischer, J. and F. Jergerlehner, 1981 1.3, 2.2 Phys. Rev. D23, 2001. 7.1 Fritzsch, H., M. Gell-Mann and H. Leutwyler, 1973 7.1	Feinberg, G. and S. Weinberg, 1961		
Fermi, E., 1934 1.1 Z. Phys. 88, 161. 1.1 Feynman, R. P., 1948 1.1 Phys. Rev. 74, 1430. 1.1 Feynman, R. P., 1949 1.1 Feynman, R. P., 1949 1.1 Feynman, R. P., 1949 1.1 Feynman, R. P. and M. Gell-Mann, 1958 1.1 Feynman, R. P. and M. Gell-Mann, 1958 1.1 Feynman, R. P., 1963 1.1 Feynman, R. P., 1963 1.3 Acta Phys. Polon. 26, 697. 1.3, 2.2 Fleischer, J. and F. Jergerlehner, 1981 1.3 Phys. Rev. D23, 2001. 7.1 Fritzsch, H., M. Gell-Mann and H. Leutwyler, 1973 7.1	Phys. Rev. Letters 6, 381.		5.1
Z. Phys. 88, 161. 1.1 Feynman, R. P., 1948 1.1 Phys. Rev. 74, 1430. 1.1 Feynman, R. P., 1949 1.1 Feynman, R. P., 1949 1.1 Feynman, R. P., 1949 1.1 Feynman, R. P. and M. Gell-Mann, 1958 1.1 Feynman, R. P. and M. Gell-Mann, 1958 1.1 Feynman, R. P., 1963 1.1 Feynman, R. P., 1963 1.3 Acta Phys. Polon. 26, 697. 1.3, 2.2 Fleischer, J. and F. Jergerlehner, 1981 1.3 Phys. Rev. D23, 2001. 7.1 Fritzsch, H., M. Gell-Mann and H. Leutwyler, 1973 7.1	Fermi, E., 1934		
Feynman, R. P., 1948 1.1 Phys. Rev. 74, 1430. 1.1 Feynman, R. P., 1949 1.1 Phys. Rev. 76, 769. 1.1 Feynman, R. P. and M. Gell-Mann, 1958 1.1 Feynman, R. P. and M. Gell-Mann, 1958 1.1 Feynman, R. P., 1963 1.1 Feynman, R. P., 1963 1.3 Acta Phys. Polon. 26, 697. 1.3, 2.2 Fleischer, J. and F. Jergerlehner, 1981 1.3 Phys. Rev. D23, 2001. 7.1 Fritzsch, H., M. Gell-Mann and H. Leutwyler, 1973 7.1	Z. Phys. 88, 161.		1.1
Phys. Rev. 74, 1430. 1.1 Feynman, R. P., 1949 1.1 Phys. Rev. 76, 769. 1.1 Feynman, R. P. and M. Gell-Mann, 1958 1.1 Phys. Rev. 109, 193. 1.1 Feynman, R. P., 1963 1.1 Acta Phys. Polon. 26, 697. 1.3, 2.2 Fleischer, J. and F. Jergerlehner, 1981 1.1 Phys. Rev. D23, 2001. 7.1 Fritzsch, H., M. Gell-Mann and H. Leutwyler, 1973 1.3	Feynman, R. P., 1948		
Feynman, R. P., 1949 1.1 Phys. Rev. 76, 769. 1.1 Feynman, R. P. and M. Gell-Mann, 1958 1.1 Phys. Rev. 109, 193. 1.1 Feynman, R. P., 1963 1.3 Acta Phys. Polon. 26, 697. 1.3, 2.2 Fleischer, J. and F. Jergerlehner, 1981 7.1 Phys. Rev. D23, 2001. 7.1 Fritzsch, H., M. Gell-Mann and H. Leutwyler, 1973 1.3	Phys. Rev. 74, 1430.		1.1
Phys. Rev. 76, 769. 1.1 Feynman, R. P. and M. Gell-Mann, 1958 1.1 Phys. Rev. 109, 193. 1.1 Feynman, R. P., 1963 1.3 Acta Phys. Polon. 26, 697. 1.3, 2.2 Fleischer, J. and F. Jergerlehner, 1981 7.1 Phys. Rev. D23, 2001. 7.1 Fritzsch, H., M. Gell-Mann and H. Leutwyler, 1973 1.3	Feynman, R. P., 1949		
Feynman, R. P. and M. Gell-Mann, 1958 1.1 Phys. Rev. 109, 193. 1.1 Feynman, R. P., 1963 1.3, 2.2 Acta Phys. Polon. 26, 697. 1.3, 2.2 Fleischer, J. and F. Jergerlehner, 1981 7.1 Phys. Rev. D23, 2001. 7.1 Fritzsch, H., M. Gell-Mann and H. Leutwyler, 1973 7.1	Phys. Rev. 76, 769.		1.1
Phys. Rev. 109, 193. 1.1 Feynman, R. P., 1963 1.3, 2.2 Acta Phys. Polon. 26, 697. 1.3, 2.2 Fleischer, J. and F. Jergerlehner, 1981 7.1 Phys. Rev. D23, 2001. 7.1 Fritzsch, H., M. Gell-Mann and H. Leutwyler, 1973 7.1	Feynman, R. P. and M. Gell-Mann, 1958		
Feynman, R. P., 1963 1.3, 2.2 Acta Phys. Polon. 26, 697. 1.3, 2.2 Fleischer, J. and F. Jergerlehner, 1981 7.1 Phys. Rev. D23, 2001. 7.1 Fritzsch, H., M. Gell-Mann and H. Leutwyler, 1973 7.1	Phys. Rev. 109, 193.		1.1
Acta Phys. Polon. 26, 697. 1.3, 2.2 Fleischer, J. and F. Jergerlehner, 1981 7.1 Phys. Rev. D23, 2001. 7.1 Fritzsch, H., M. Gell-Mann and H. Leutwyler, 1973 7.1	Feynman, R. P., 1963		
Fleischer, J. and F. Jergerlehner, 1981 Phys. Rev. D23 , 2001.7.1Fritzsch, H., M. Gell-Mann and H. Leutwyler, 1973	Acta Phys. Polon. 26, 697.	.3,	2.2
Phys. Rev. D23, 2001. 7.1 Fritzsch, H., M. Gell-Mann and H. Leutwyler, 1973	Fleischer, J. and F. Jergerlehner, 1981		
Fritzsch, H., M. Gell-Mann and H. Leutwyler, 1973	Phys. Rev. D23 , 2001.		7.1
	Fritzsch, H., M. Gell-Mann and H. Leutwyler, 1973		

References

Phys. Letters 47B, 365.	1.1
Fujikawa, K., B. W. Lee and A. I. Sanda, 1972	2 5
Flys. Rev. D0, 2923. Fukuda H. V. Miyamoto and S. Tomonaga 1040	2.0
Prog Theor Phys 4 47 121	1.1
Gamba, A., R. E. Marshak and S. Okubo, 1959	10.2
Proc. Natl. Acad. Sci. 45, 881.	1.2
Georgi, H. and S. L. Glashow, 1974	
Phys. Rev. Letters 32, 438.	1.1
Glashow, S. L., 1961	
Nucl. Phys. 22, 579.	1.1
Glashow, S. L., J. Iliopoulos and L. Maiani, 1970	
Phys. Rev. D2 , 1285.	1.2
Goldstone, J., 1961	
Nuovo Cim. 19, 154.	2.3
Green, M. and M. Veltman, 1980	
Nucl. Phys. B169, 137; B175, 547(E).	5.1, 6.2, 7.1
Hara, Y. 1964	
Phys. Rev. 134, B701.	1.2
Harsert, F. J. et al., 1973	
Phys. Letters 46B , 121.	5.1
Hepp, K. 1966	0.5
Comm. Math. Phys. 2, 301.	2.5
Higgs, P. W., 1964a	
Phys. Letters 12, 132.	1.1, 2.3
Higgs, P. W., 1964b Dhyp. Days Letters 19 509	1109
Higgs D W 1066	1.1, 2.3
Phys $R_{av} = 145$ 1156	1123
High Z. 1978	1.1, 2.0
Sorvushiron Kenkyu (Kyoto) 58, 207 (in Japanese).	1.2
Hioki, Z., 1982a	
Prog. Theor. Phys. 67, 1165.	5, 1, 5, 3, 6, 1, 6, 2
Hioki, Z., 1982b	,,,
Prog. Theor. Phys. 65, 2134.	5.1, 6.1, 7.1
Hung, P. Q. and J. J. Sakurai, 1979	
Phys. Letters 88B, 91.	1.2
Inoue, K., A. Kakuto, H. Komatsu and S. Takeshita, 1980	
Prog. Theor. Phys. 64, 1008.	1.3, 7.1
Jarlskog, C., 1970	
Letters Nuovo Cim., 4, 377.	5.1
Jogleker, S. D. and B. W. Lee, 1976	
Ann. of Phys. 97 , 160.	2.5
Jonker, M. et al. (CHARM Collaboration), 1980	
r nys. Letters 93D , 203. Kallen C. 1052	5.1
Holy Phys Acto 25 417	3.9
Katavama V K Matumoto S Tanaka and F Vamada 1062	5.2
Prog Theor Phys 28 675	1.2
Kazama, Y. and Y. P. Yao, 1980a, b. c	1.2
Phys. Rev. D21 , 116, 1138; D22 , 514.	7.2
Kazama, Y. and Y. P. Yao, 1982	112
Phys. Rev. D25 , 1605.	7.2

Kinoshita, T. and A. Sirlin, 1959	
Phys. Rev. 113, 1652.	5.3, 6.1
Kinoshita, T., 1962	
J. Math. Phys. 3, 650.	5.3
Kobayashi, M. and T. Maskawa, 1973	
Prog. Theor. Phys. 49, 652.	1.2
Komatsu, H., 1978	
Prog. Theor. Phys. 59, 2013.	1.2
Konuma, M. and T. Oka, 1978	
Prog. Theor. Phys. 60, 1842.	1.2
Konuma, M. and T. Maskawa, 1981	
Proceedings of the 4th Kyoto Summer Institute on Grand Unified Theory	ries and
Related Topics, ed. by M. Konuma and T. Maskawa (World Scientific Pr	ublishing
Co., Singapore).	1.1
Kugo, I., 19/7	0.5
Frog. Theor. Flys. 57, 595.	2.5
Rugo, I. and I. Ojima, 1978	1 2 2 2 2 4
Fliys. Letters 73D , 459.	1.3, 2.2, 2.4
Rugo, 1. and I. Ojima, 1979 Drog Theor Phys Suppl No. 66 1	1222
Kugo T and S Hohara 1081	1.3, 2.2
Prog. Theor. Phys. 66, 1044	3.4
I log. 1 Heor. 1 Hys. 00, 1044.	0.4
Phys Reports 72C 185	1 1
$I_{\text{autrup}} = 1067$	1.1
Mat Eves Medd Dan Vid Selsk 35 29	1.3
Lee B W and I Zinn-Justin 1972 a. h. c.	1.0
Phys. Rev. D5 , 3121, 3137, 3155.	1.3.2.5
Lee. B. W., 1974	
Phys. Rev. D9, 933.	2.5
Lee, B. W., 1976	
Methods in Field Theory, Les Houches 1974, ed. by R. Balian and J. Zi	nn-Justin
(North-Holland), 79.	2.5
Lee, T. D. and M. Nauenberg, 1964	
Phys. Rev. 133, B1549.	5.3
Lee, T. D., M. Rosenbluth and C. N. Yang, 1949	
Phys. Rev. 75, 905.	1.1
Lehmann, H., 1954	
Nuovo Cim. 11, 342.	3.2
Lehmann, H., K. Symanzik and W. Zimmermann, 1955	
Nuovo Cim. 1, 205.	3.2
Lemoine, M. and M. Veltman, 1980	
Nucl. Phys. B164 , 445.	7.1
Lewin, L., 1958	
Dilogarithms and Associated Functions (MacDonald and Co. Ltd., London	n). C, F
Llewellyn Smith, C. H. and J. F. Wheater, 1981	
Phys. Letters 105B, 486.	7.1
Maki, Z., M. Nakagawa and S. Sakata, 1962	
Prog. 1 heor. Phys. 28, 870.	1.2
Maki, L., 1964 a, b	1.0
rrog. 1 neor. rnys. 31, 331, 333.	1.2
Marciano, w. J. and A. Sirlin, 19/5	E 0
NUCI. FRYS. D88, 86.	5.4

Marciano W I and H Pagels 1078	
Phys. Reports 36C, 137.	1.1
Marciano, W. J., 1979	
Phys. Rev. D20 , 274.	7.2
Marciano, W. J. and A. Sirlin, 1980 Phys. Rev. D22 , 2695.	7.1
Marciano, W. J. and A. Sirlin, 1981	
Phys. Rev. Letters 46, 163.	6.2
Mo, L. W., 1982	
Neutrino Physics and Astrophysics, Proceedings of the International Confere Neutrino '80, Erice, ed. by E. Fiorini (Plenum Press, New York), 191.	nce on 5.1
Nakanishi, N., 1958	5.0
Prog. Theor. Phys. 19, 159.	5.3
Nakanishi, N., 1966 Dree Theor Dive 25, 1111	1 2
Frog. Theor. rhys. 55 , 1111.	1.5
Nakanishi, N., 1979 Drag Theor Dive 69, 1995	0.0
Prog. Theor. Phys. 62, 1385.	2.2
Nambu, I. and G. Jona-Lasinio, 1901 Dhua Day 199 245	
Fliys. Rev. 122, 343.	2.3
Pascnos, E. A. and M. Wirdel, 1962 Nucl Phys B104 180	71
Passarino G and M Veltman 1079	1.1
Nucl Phys B160 151	5.1.7.1
Phillippe R. 1982	012, 112
Phys. Rev. D26 , 1588.	7.1
Prescott, C. Y. et al., 1978	••
Phys. Letters 77B, 347.	1.2, 5.1
Prescott, C. Y. et al., 1979	·
Phys. Letters 84B, 524.	5.1
Reines, F., H. S. Gurr and H. W. Sobel, 1976	
Phys. Rev. Letters 37, 315.	1.2, 5.1
Ross, D. A., 1973	
Nucl. Phys. B51, 116.	5.1, 7.1
Ross, D. A. and J. C. Taylor, 1973	
Nucl. Phys. B51, 125.	3.3
Sakakibara, S., 1981	
Phys. Rev. D24 , 1149.	7.1
Sakata, S., H. Umezawa and S. Kamefuchi, 1951 Phys. Rev. 84, 154	2.5
Sakata, S., H. Umezawa and S. Kamefuchi, 1952	2.0
Prog. Theor. Phys. 7, 377.	2.5
• Salam, A., 1968	
<i>Elementary Particle Theory</i> , ed. by N. Svartholm (Almqvist and Wiksell, holm), 367.	Stock- 1.1
Salomonson, P. and Y. Ueda, 1975 Phys. Rev. D11, 2606	5171
Schwinger I 1948	5.1, 7.1
Phys. Rev. 74, 1439.	1.1
Schwinger, J., 1949 a, b	
Phys. Rev. 75, 651; 76, 790.	1.1
Sehgal, L. M., 1977	
Phys. Letters 71B, 99.	1.2
Sirlin, A., 1980	
Phys. Rev. D22 , 971.	6.1, 7.1

References

Sirlin, A. and W. J. Marciano, 1981	
Nucl. Phys. B189, 442.	6.1, 7.1
Slavnov, A. A., 1972	
Theor. and Math. Phys. 10, 99.	2.5
Sterman, G. and S. Weinberg, 1977	
Phys. Rev. Letters 39, 1436.	5.3
Sudarshan, E. C. G. and R. E. Marshak 1958	
Proceedings of the International Conference on Mesons and Rec	ently Discovered
Particles, Padova-Venezia, 1957 (Italian Physical Society), p. V-	14.
T Hoott, G., 1971a	
² Hys. Letters 37B , 195.	5.1
t flooit, G., 1971D Nual Dhua B92 172	111005
1 Hooft G 1071c	1.1, 1.3, 2.0
Nucl Phys B35 167	25
't Hooft, G. and M. Veltman, 1972a	2.0
Nucl. Phys. B44 , 189.	А
't Hooft, G. and M. Veltman, 1972b	
Nucl. Phys. B50, 318.	2.5
't Hooft, G. 1973	
Nucl. Phys. B61, 455.	2.5
't Hooft, G. and M. Veltman, 1979	
Nucl. Phys. B153, 365.	C, F
Takahashi, Y., 1957 Nuovo Cim. 6, 370.	2.4.3.4
Tati, T. and S. Tomonaga, 1948	2, 0. 1
Prog. Theor. Phys. 3, 391.	1.1
Taylor, J. C., 1971 Nucl. Phys. B33 , 436.	2.5
Tsai, Y. S., 1960 Phys. Rev. 120 , 269.	5.3
Tsai, Y. S., 1965	
Phys. Rev. 137B , 730.	5.3
Umezawa, H. and S. Kamefuchi, 1951 Prog. Theor. Phys. 6 543	2.0
Van Nieuwenhuizen P 1971	5.2
Nucl. Phys. B28 , 429.	5.3
Veltman, M., 1980	
Phys. Letters 91B, 95.	6.1, 7.1
Ward, J. C., 1950	
Phys. Rev. 78, 182.	2.4, 3.4
Weinberg, S., 1964 Phys. Rev. 135, 1040	2 /
Meinherg S 1967	5.4
Phys. Rev. Letters 19, 1264.	1.1
Weinberg, S., 1973	
Phys. Rev. D8, 3497.	2.5
Wheater, J. F. and C. H. Llewellyn Smith, 1982	
Nucl. Phys. B208 , 27.	6.1, 7.1
Yang, C. N. and K. L. Mills, 1954 Phys. Pay. 06, 101	1 1
1 myo. Acv. 30, 191. Vulcawa H 1025	1.1
Proc. PhysMath. Soc. Japan (3) 17, 48.	1.1
Zimmermann, W., 1969	
Comm. Math. Phys. 15, 208.	2.5